## MOUSE SPLENOCYTE PROLIFERATION AND ITS MODULATION BY SEX STEROIDS

HELEN GODFREY

A thesis submitted for the degree of Doctor of Philosophy

The University of Aston in Birmingham

**JUNE 1988** 

This copy of the thesis has been supplied on condition that anyone who consults it is understood to recognise that its copyright rests with its author and that no quotation from the thesis and no information derived from it may be published without the author's prior, written consent.

### THE UNIVERSITY OF ASTON IN BIRMINGHAM

### MOUSE SPLENOCYTE PROLIFERATION AND ITS MODULATION BY SEX STEROIDS

by

#### Helen Godfrey

#### Submitted for the degree of Doctor of Philosophy

#### SUMMARY

Concanavalin A, a T cell mitogen enhanced DNA synthesis in murine splenocytes. Amongst the early signals prior to this event was an increase in cytosolic calcium derived from both intra- and extracellular sources. The requirements for extracellular calcium persisted for four hours after the lectin administration which itself was needed for six hours. Putative calcium channel antagonists and calmodulin inhibitors blocked the increase in DNA synthesis. The calcium signal was mimicked by application of the ionophore, A23187, although no increase in DNA synthesis occurred. An activator of protein kinase C, 12-0-tetradecanoylphorbol 13-acetate, had little effect in isolation but the combined application of these two agents greatly enhanced DNA synthesis. The natural mediators of these events are presumed to be inositol trisphosphate and diacylglycerol derived from phosphatidylinositol bisphosphate hydrolysis. Lectin application and protein kinase C activation both increased intracellular pH possibly as a result of Na<sup>+</sup>/H<sup>+</sup> exchange since amiloride an inhibitor of this antiporter inhibited lectin induced DNA synthesis. The calcium and hydrogen ionic changes occur within minutes of lectin application; the protracted requirement for this mitogen suggests further signalling mechanisms occur to elicit maximum DNA synthesis in these cells.

Gonadectomy caused an increase in thymic and splenic weight. Splenocytes derived from castrated mice showed no change in mitogen response whereas those from ovariectomised mice demonstrated a reduced lectin sensitivity. Testosterone,  $5\alpha$ 

dihydrotestosterone, a and ß oestradiol all inhibited lectin induced DNA synthesis but

only at pharmacological concentrations. Testosterone glucuronide and cholesterol were without effect. Studies with mouse serum fractions of differing steroidal status were unable to confirm the presence or absence of serum factors which might mediate the effects of steroid on lymphoid cells, all fractions tested inhibited lymphocyte transformation.

Both interleukin-2 and lipopolysaccharide induced splenocyte mitogenesis was also impaired by high steroid concentrations <u>in vitro</u>, suggesting that steroids mediate their effect by a non-specific, non-receptor-mediated event.

Keywords:

Splenocytes Sex steroids Proliferation Signal transduction Calcium

#### ACKNOWLEDGEMENTS

I would like to thank everyone who has helped me during the course of this study. I extend thanks to those in the Department of Pharmaceutical Sciences for their help and guidance. I am grateful to Helen Hussey for her technical assistance and Tracy Knight for her expertise in the typing of this manuscript.

I would particularly like to thank my colleagues, Frances Khan, Ansar Malik, Rod Haynes and Meenu Wadhwa for their help, support, companionship and inspiring discussions.

I am grateful to all my friends and family for their constant encouragement, empathy and tolerance especially Sarah, Louise, Viki, Celia and Graham.

My special thanks to Alan Perris for his guidance, interest and long-suffering involvement throughout my education in this Department.

## CONTENTS

			Page
Summary	y		2
Acknowledgements			3
Contents			4
List of Diagrams			5
List of F	igures		6
List of Ta	ables		10
CHAPTE	ER ONE:	INTRODUCTION	12
1.1	Preface		12
1.2	The Cell	l Cycle	13
1.3	Extracel	lular Growth Substances	17
1.4	Recepto	rs	33
1.5	Signal T	ransmission	47
	1.5.1	Cyclic nucleotides	50
	1.5.2	Calcium and Phosphoinositide Metabolism	55
	1.5.3	Magnesium	71
	1.5.4	Monovalent lons	75
	1.3.3	Ornithine Decarboxylase and Polyamines	78
16	1.5.0 Normlan	Cell Cycle Dependent Genes	82
1.0	The Con	tic Growth	84
1./	The Con	lirol of Spienic Lymphocyte Proliferation	91
CHAPTE	ER TWO:	MATERIALS AND METHODS	106
2.1	Experim	ental Animals	106
2.2	Lympho	cyte Proliferation	107
	2.2.1	Preparation of Splenocyte Cultures	107
	2.2.2	Estimation of DNA Synthesis	108
	2.2.3	Proliferative Assay: <sup>3</sup> H-Thymidine Incorporation	109
	2.2.4	Interleukin 2 Assay/Proliferative Model	109
	2.2.5	Preparation of Con A Blast Assay	110
2.3	Measure	ment of Intracellular Ca <sup>2+</sup> levels	110
	2.3.1	Measurement of Intracellular $Ca^{2+}$ by the	
24	<b>C</b> 1.	fluorescent Indicator Fura-2	114
2.4	Calcium	Uptake Studies	118
2.5	Measure 2.5.1	ment of Intracellular pH (pHi)	120
	2.5.1	Fluesseent of Intracellular pH by a	
	252	Massurement of Intracellular III	121
	2.3.2	Fluorescent Indicator DCECE (2) 711	
		(carboxyethyl) 5(6) carboxyl fluorencia)	
2.6	Statistica	l Analysis	124
CHADTE	DTIDE		
CHAPTE	K THKE	E: EAPERIMENTAL RESULTS	126
CHAPTE	R FOUR	DISCUSSION	192
APPENDIX		220	
			220
REFERENCES			221

# LIST OF DIAGRAMS

		Page
Diagram 1	A model of the cell cycle	16
Diagram 2	A diagrammatic representation of the generation of putative second messengers and some of their possible interactions during stimulus-mitosis coupling	81
Diagram 3	A diagrammatic representation of the proposed mechanism for regulation of cell proliferation by proto-oncogenes	89
Diagram 4	Measurement of cytosolic calcium levels in fura-2 loaded murine splenocytes	116
Diagram 5	The effect of a weak base (5m $MNH_4Cl$ ) on the pHi of quene 1 loaded murine splenocytes	123

## LIST OF FIGURES

		Page
Figure 1	Dose response curve of the effect of Concanavalin A on thymidine incorporation in isolated TO murine splenocytes	127
Figure 2	Dose response curve of the effect of Concanavalin A on thymidine incorporation in isolated $C_{57}BL/6$ murine splenocytes	128
Figure 3	Effect of Verapamil on Con A induced murine splenocyte proliferation	129
Figure 4	Effect of Nifedipine on Con A (5 $\mu$ g ml <sup>-1</sup> ) induced murine splenocyte proliferation	130
Figure 5	Effect of Diltiazem on Con A (5 $\mu$ g ml <sup>-1</sup> ) induced murine splenocyte proliferation	131
Figure 6	Effect of Con A (5 $\mu$ g ml <sup>-1</sup> ) and ionophore A23187 (10 <sup>-7</sup> M) on calcium uptake in isolated murine splenocytes	133
Figure 7	The effect of Con A (5 $\mu$ g ml <sup>-1</sup> ) on the [Ca <sup>2+</sup> ]i of fura-2 loaded murine splenocyted incubated in Ca <sup>2+</sup> free medium followed by subsequent addition of Ca <sup>2+</sup> ions (0.6 mM) to the extracellular environment	137
Figure 8	Effect of calmodulin inhibitors on Con A (5 $\mu$ g ml <sup>-1</sup> ) induced murine splenocyte proliferation	139
Figure 9	Effect of Dibutyryl cyclic AMP on Con A (5 $\mu$ g ml <sup>-1</sup> ) induced murine splenocyte proliferation	141
Figure 10	Effect of Forskolin on Con A (5 $\mu$ g ml <sup>-1</sup> ) induced murine splenocyte proliferation	143
Figure 11	Effect of sequential additions of EGTA and $CaCl_2$ during the 48 hour incubation on Con A (5 µg ml <sup>-1</sup> ) induced murine splenocyte proliferation	145
Figure 12	Dose-response curve of the effect of $\alpha$ methyl- mannoside on Con A (5 µg ml <sup>-1</sup> ) induced murine splenocyte proliferation	146
Figure 13	Effect of sequential additions of $\alpha$ methyl-mannoside (50 mM) during the incubation period on Con A (5 µg ml <sup>-1</sup> ) induced murine splenocyte proliferation	146

		Page
Figure 14	Dose response curves of the effects of A23187 and TPA alone and in combination on thymidine incorporation in isolated murine splenocytes	148
Figure 15	Effects of TPA with and withoput A23187 (10 <sup>-8</sup> M) on thymidine incorporation in isolated murine splenocytes	149
Figure 16	Effects of TPA with A23187 ( $10^{-9}$ M) and without ionophore on thymidine incorporation in isolated murine splenocytes	150
Figure 17	Effects of $4\alpha$ , 12,14 phorbol didecanoate with and without A23187 ( $10^{-7}$ M) on thymidine incorporation in isolated murine splenocytes	153
Figure 18	Effects of the co-mitogens A23187 ( $10^{-7}$ M) and TPA ( $10^{-9}$ M) with varying concentrations of Con A	154
Figure 19	Comparison of the effects of Con A (5 $\mu$ g ml <sup>-1</sup> ) on the [Ca <sup>2+</sup> ]i of fura-2 loaded murine splenocytes preincubated in normal medium or medium treated with TPA (10 <sup>-7</sup> M) for 10 minutes	157
Figure 20	Effects of preincubation with phorbol ester (10 minutes) on the Con A (5 $\mu$ g ml <sup>-1</sup> ) induced murine splenocyte proliferation	159
Figure 21	Effect of Amiloride on Con A (5 $\mu$ g ml <sup>-1</sup> ) induced murine splenocyte proliferation	160
Figure 22	Comparison of the effects of Con A (5 $\mu$ g ml <sup>-1</sup> ) and TPA (10 <sup>-9</sup> M) addition upon the pHi levels of BCECF loaded murine splenocytes	162
Figure 23	Effects of Testosterone, $5\alpha$ Dihydrotestosterone ( $5\alpha$ DHT) and Testosterone Glucuronide on Con A ( $5 \ \mu g \ ml^{-1}$ ) induced murine splenocyte proliferation	165
Figure 24	Effects of $\alpha$ and $\beta$ Oestradiol on Con A (5 µg ml <sup>-1</sup> ) induced murine splenocyte proliferation	166
Figure 25	Effect of Cholesterol on Con A (5 $\mu$ g ml <sup>-1</sup> ) induced murine splenocyte proliferation	167
Figure 26	Effect of sequential additions of Testosterone $(10^{-4}M)$ during the incubation period on Con A (5 µg ml <sup>-1</sup> ) induced murine splenocyte proliferation	169

		Page
Figure 27	Comparison of thymus and spleen weights of sham operated (SH) gonadectomised (GX) and control (C) female TO mice at 4, 6 and 9 weeks after surgery	170
Figure 28	Comparison of Thymus and spleen weights of sham operated (SH), gonadectomised (GX) and control (C) male TO mice after 4, 6 and 9 weeks after surgery	172
Figure 29	Comparison of dose-response curves of the effect of Con A on thymidine incorporation in splenocytes isolated from post-pubertal sham operated (SH) gonadectomised (GX) and control (C) male TO mice four weeks after surgery	173
Figure 30	Comparison of dose-response curves of the effect of Con A on thymidine incorporation in splenocytes isolated from post-pubertal sham operated (SH), gonadectomised (GX) and control (C) female TO mice four weeks after surgery	174
Figure 31	Comparison of the effects of control (C), gonadectomised (GX) and sham operated (SH) male and female mouse serum on normal splenocytes (isolated from male mice) incubated with Con A $(2 \ \mu g \ m l^{-1} \ or \ 5 \ \mu g \ m l^{-1})$	176
Figure 32	Comparison of the effects of varying concentrations of foetal calf and mouse serum on Con A (5 $\mu$ g ml <sup>-1</sup> ) induced murine splenocyte proliferation and the effect of mouse serum on thymidine incorporation in splenocytes incubated with Con A (5 $\mu$ g ml <sup>-1</sup> ) and 10% foetal calf serum (FCS)	178
Figure 33	Dose-response curve of the effect of recombinant human Interleukin 2 on thymidine incorporation in 'Con A blasts'.	180
Figure 34	Effect of Nifedipine on the IL2 (100 U ml <sup>-1</sup> ) induced 'Con A blast' proliferation	181
Figure 35	Comparison of the effects of IL2 (100 U ml <sup>-1</sup> ) and physiological saline addition upon the $[Ca^{2+}]i$ of fura-2 loaded Con A blasts	182
Figure 36	Effect of TFP on the IL2 (100 U ml <sup>-1</sup> ) induced 'Con A blast' proliferation	183
Figure 37	Effect of Amiloride on the IL2 (100 U ml <sup>-1</sup> ) induced proliferation of 'Con A blasts'	184

		Page
Figure 38	Effect of sex steroids on the IL2 (100 U ml <sup>-1</sup> ) induced proliferation of 'Con A blasts'	186
Figure 39	Comparison of the effect of Interleukin 1 $\alpha$ with and without Con A (0.35 µg ml <sup>-1</sup> ) on thymidine incorporation in isolated murine splenocytes	187
Figure 40	Comparison of the effect of Interleukin 1ß with and without Con A (0.35 $\mu$ g ml <sup>-1</sup> ) on thymidine incorporation in isolated murine splenocytes	188
Figure 41	Dose response curve of the effect of Lipopoly- saccharide (LPS) on thymidine incorporation in isolated murine splenocytes	189
Figure 42	Effect of sex steroids on the LPS (25 $\mu$ g ml <sup>-1</sup> ) induced proliferation of murine splenocytes.	190

# LIST OF TABLES

		Page
Table 1	Effect of A23187 10 <sup>-7</sup> M on the intracellular calcium concentration of murine splenocytes	134
Table 2	The effect of Verapamil on the Con A induced increase in intracellular calcium concentration $[Ca^{2+}]i$ of murine splenocytes	135
Table 3	The effect of Diltiazem on the Con A induced increase in $[Ca^{2+}]i$ of murine splenocytes	135
Table 4	The effect of extracellular $Ca^{2+}$ ions on the Con A (5 µg ml <sup>-1</sup> ) induced increase in $[Ca^{2+}]i$ in murine splenocytes	136
Table 5	The effect of preincubation with 8-Br cAMP on the Con A induced increase in intracellular calcium concentration of murine splenocytes	140
Table 6	The effect of preincubation with dibutyryl cAMP on the Con A induced increase in intracellular calcium concentration of murine splenocytes	142
Table 7	The effect of simultaneous administration of calcium ionophore A23187 and phorbol esters on intracellular calcium concentrations of murine splenocytes	151
Table 8	The effect of preincubating (3 mins) murine splenocytes with varying concentrations of TPA on the Con A induced increase in intracellular calcium concentration	155
Table 9	The effect of preincubating (3 mins) murine	156
	splenocytes with $4\alpha 12,13$ phorbol didecanoate on the Con A induced increase in intracellular calcium concentration	
Table 10	The effect of preincubating (10 mins) murine	156
	splenocytes with TPA $(10^{-7}M)$ and $4\alpha 12,13$ phorbol didecanoate $(10^{-7}M)$ on the Con A induced increase in intracellular calcium concentration	
Table 11	The effect of DFMO (5 mM) on the Con A induced increase in cytosolic $Ca^{2+}$ concentration of murine splenocytes	163

		Page
Table 12	The influence of DFMO on Con A induced DNA synthesis in murine splenocytes	163
Table 13	The effect of testosterone $(10^{-4}M)$ on the Con A induced increase in cytosolic calcium in murine splenocytes	168

# CHAPTER ONE INTRODUCTION

### 1.1 Preface

Ideally cell growth occurs only when required either for development or for tissue maintenance. Although the growth in size of individual cells is a small component in the growth of tissues and organs, an increase in the number of cells is of much greater significance in both normal and neoplastic growth. Since the increase in cell number is achieved by mitotic division of cells, an understanding of normal growth and its aberrations must be based on a study of the mechanisms that regulate cell division.

The control of normal growth in eukaryotic cells is achieved by a complex signalling system beginning with the action of a variety of extracellular effectors which, having bound to specific receptors in or on the cell, are able to initiate changes in intracellular processes which ultimately lead to DNA synthesis and mitosis (Rozengurt & Collins 1983; Berridge, 1984a). The molecular mechanisms by which these extracellular signals are transduced across the plasma membrane to provoke a complex cellular response which eventually leads to mitosis are still not fully understood.

The proliferation rate of animal cells is largely determined by the relative times that they spend in the cell cycle as opposed to quiescence. An initiation event is required to propel a cell towards mitosis; in its absence cells enter a resting state in which they remain until an effector provides the necessary signal starting the chain of events leading to DNA synthesis and division once again (Pardee, 1974).

### 1.2 <u>The Cell Cycle</u>

The cell cycle is the period between the division of one cell and the subsequent division of a daughter cell. However this strict definition is normally broadened to include the cellular and biochemical aspects that regulate cell proliferation (Baserga, 1981; Ronot & Adolphe, 1986).

DNA synthesis (the S phase) occurs during a discrete period during interphase and is an essential prelude to mitosis (the M phase). Before and after the S phase, cells engage in growth and metabolic activity but not in chromosome replication. Cells preparing for DNA replication are in the  $G_1$  phase whilst cells preparing for mitosis are considered to be in the  $G_2$  period. The four consecutive phases  $G_1$ , S,  $G_2$  and M first described by Howard and Pelc (1953) are collectively known as the cell cycle.

Subdivision of the intermitotic period into  $G_1$ , S and  $G_2$  has generated many attempts to elucidate the process controlling the progression of cells from one mitosis to the next. The duration of S and  $G_2$  do not vary greatly between populations since both phases are predetermined and specifically related to division (Smith and Martin, 1973; Ronot & Adolphe, 1986). However the greatest variation in duration of the interphase is found in the  $G_1$  phase. When a cell cycle is long most of the prolongation is in the  $G_1$  phase and when a cell cycle is short as it is in egg cells and in certain rapidly proliferating cell lines, there is no measurable  $G_1$  phase at all (Mazia, 1974; Liskay and Prescott, 1978; Baserga, 1985).  $G_1$  has therefore attracted particular interest as the phase in which proliferation is regulated. Hidden in the  $G_1$  phase are the key processes that make it possible for the cell to enter the S phase and thereby commit itself to division in the future. Restraints or 'blocks' prevent uncontrollable proliferation in non-neoplastic cells and normally occur in the  $G_1$  phase, (Pardee, 1974; Pardee et al, 1978; Baserga, 1981; Whitfield, 1982; Murray, 1987a) but may also occur in the  $G_2$  and  $G_0$  phases of the cell cycle (Gelfant, 1981). Neoplastic growth occurs because cancerous cells circumvent these restriction points and grow autonomously (Walker, 1986).

The growth status of cell populations has been categorised in three ways, thus there are proliferating or 'cycling' cells, differentiated cells and quiescent cells. Proliferating cells continuously divide going from one mitosis to the next, through the four consecutive phases of the cell cycle.

A further group of cells may temporarily leave the cell cycle and become quiescent because they are unable to pass through a critical control or restriction point in  $G_1$  due to failure of key processes. They remain in a quiescent or  $G_0$ phase until environmental conditions stimulate their re-entry into the cell cycle at  $G_1$  (Fantes, 1986).

The G<sub>0</sub> phase may encompass cells with different degrees of readiness to resume cell division, eg. serum starved or nutrient deprived cells may pass the restriction point close to the  $G_1/S$  boundary fairly rapidly upon serum or nutrient addition whilst in other cells such as peripheral blood lymphocytes and hepatocytes the recruitment is more complex and protracted (MacManus et al, 1972; Klaus & Hawrylowicz, 1984). Recent work by Zetterberg and colleagues observing 3T3 cells under the microscope so that age related effects could be precisely analysed have determined that cells are only sensitive to serum deprivation for between 3 and 4 hours after mitosis. This period is referred to as  $G_1$  pm (post-mitotic). This infers that cells leave the cycle and enter quiescence from a discrete period of the  $G_1$  phase. The remainder of the  $G_1$  phase is termed  $G_1$  ps (pre S phase). Commitment to mitosis is thought to occur at a control point at the end of  $G_1$  pm (Larsson et al, 1985; Zetterberg & Larsson, 1985; Fantes, 1986) and is determined by appropriate growth factors in the serum.

Interestingly it appears that DNA synthesis is not initiated at an invariant time after the completion of the commitment process in  $G_1$  pm. The time spent in  $G_1$  ps varies considerably from cell to cell, thus the variation in cycle duration attributed to the  $G_1$  phase is now considered to be in this latter part of  $G_1$ ,  $G_1$  ps. The timing of DNA synthesis and subsequent mitosis is thought therefore to be adjusted during the  $G_1$  ps interval.  $G_1$  ps variability is consistent with a transition-probability event (Smith and Martin, 1973; Brooks, 1980) but could equally reflect variability in cell size (Killander & Zetterberg, 1965; Shields et al, 1978). Thus although the cells are committed to mitosis in  $G_1$  pm the time at which the S phase and subsequently division occurs may depend on the cells reaching a critical size in  $G_1$  ps (Zetterberg & Larrson, 1985).

Moreover the possibility exists that there are separate commitment and division time regulatory points in mammalian cells comparable to those demonstrated recently in the fusion yeast *Schizosaccharomyces pombe*. In addition to a 'start' control point in  $G_1$  there is a second regulatory point in the cell cycle between  $G_2$  and mitosis (Hayles and Nurse, 1986). At this latter control position at the  $G_2/M$  boundary the cell monitors cell size and nutrient availability. Under favourable conditions the cell becomes committeed to immediate entry into



Diagram 1: A model of the cell cycle (see text for details).

mitosis, while if they are unfavourable it enters a quiescent state. Interestingly the same gene product cdc 2 (cell division cycle) is required for both these regulatory points in *S.pombe* (Murray, 1987a). Human homologues of cdc 2 gene may act at the  $G_1$  restriction point in mammalian cells and also at a  $G_2/M$  transition point. A mitotic inducer, maturation promotion factor (MPF) is postulated to initiate mitosis in vertebrate cells at a mitotic control point at  $G_2/M$  comparable to that found in the fusion yeast, and may therefore be an activated form of the cdc 2 gene homologue protein (Murray, 1987a; Murray, 1987b; Lee & Nurse, 1987).

Thus, committment to enter the cell cycle occurs at a restriction point at the end of  $G_1$  pm. Cells may be induced to pass through this control point by the appropriate chemical factors or other stimuli occurring at the critical time. Once committed to mitosis cells then pass through the  $G_1$  ps phase. The timing of the subsequent S phase and mitosis may be adjusted by unknown critical factors during this interval. A second control point initiating mitosis occurs late in  $G_2$  in which maturation promotion factor (MPF) is thought to be involved. If such critical factors are missing during  $G_1$  or  $G_2$  or even during the S phase the cells may become arrested and enter a quiescent phase ( $G_{10}$ ,  $G_{20}$  or  $S_0$ ). They will remain there until the appropriate stimulus provokes them to re-enter the cell cycle (see Diagram 1).

### 1.3 Extracellular Growth Substances

Normal cells proliferate as an adaptive response to a sequence of external signals. A mitogen is a factor which acts as a primary signal to induce a cell to undergo DNA synthesis and proliferation.

Among the various agents involved in growth regulatory processes are certain classical protein or polypeptide endocrine hormones.

Perhaps the best example of this group of classical hormones which regulate growth is growth hormone (GH), which promotes longitudinal bone growth by inducing chondrocytes in the epiphyseal growth plate to divide. In addition to skeletal growth, GH also promotes cell proliferation in a variety of cells such as skeletal muscle, liver, kidney, heart, adipose tissue and intestinal epithelium (Goldspink and Goldberg, 1975; Isaksson et al, 1985). Many studies have shown that the effect of growth hormone on different growth processes, particularly skeletal growth is indirect, mediated by GH dependent plasma factors, the somatomedins. However there is evidence that GH may have direct effects on longitudinal bone growth *in vivo* (Isaksson, 1982; Isaksson et al, 1985) and on cultured chondrocytes *in vitro* (Madsen et al, 1983).

Thyroid hormones have also long been implicated in growth. Thus, thyroidectomy in young rats leads to a significant reduction in skeletal growth together with morphological and histological abnormalities of skin, brain and other organs. Upon administration of thyroid hormone, normal growth and development resumes (Gibson et al, 1977). There are indications that thyroid hormone synergises with growth hormone and that both are essential for normal skeletal growth and development (Tonna, 1973).

Another hormonal vector controlling mitotic activity and body growth may be parathyroid hormone via its ability to modulate plasma calcium concentrations (Whitfield et al, 1969; Perris et al, 1970; Rixon & Whitfield, 1972). In the absence of the parathyroid gland there is thymic atrophy and bone marrow hypoplasia and the heightened mitotic activity which normally follows haemorrhage and immunological challenge is reduced (Perris et al, 1971; Edwards et al, 1981). A functional parathyroid gland is also essential for the normal

regenerative response of liver cells after partial hepatectomy (Rixon & Whitfield, 1976).

Growth *in vivo* is dependent on the anabolic hormone insulin. In addition to its well characterised effects on metabolism insulin has for a long time also been implicated in the stimulation of growth and proliferation of cells *in vitro* (Gey & Thalhimer, 1924). Since then insulin has been shown to stimulate a variety of cells to grow (Straus, 1981). Insulin alone, under certain culture conditions, is able to stimulate DNA synthesis (Temin, 1967; Teng et al, 1976; Coppock et al, 1980; and Koontz and Iwahashi, 1981). Typically however its stimulatory effect on growth is enhanced by and in some cases requires the presence of other hormonal growth factors. Insulin is generally believed to belong to the class of factors which are required for progression of "competent" cells through  $G_1$  and S (Scher et al, 1979; Stiles et al, 1979; Callahan et al, 1985).

The synergistic effects of insulin with other growth factors is remarkably well illustrated in 3T3 cells. Quiescent cells can be stimulated to reinitiate DNA synthesis by specific combinations of mitogens which act synergistically when added to serum free cultures. Maximal DNA synthesis occurs when a growth factor which raises cAMP levels is administered in combination with another mitogen which activates protein kinase C and enhances ion fluxes. Some substances such as platelet derived growth factor (PDGF) have the ability to activate both pathways and achieve maximal stimulation alone. Interestingly, insulin may act synergistically with a factor from either group to maximally promote DNA synthesis, and thus plays a central role in the generation of early signals leading to proliferation in 3T3 cells (Rozengurt & Collino, 1983; Rozengurt, 1985; Rozengurt & Mendoza, 1985).

Other peptide hormones which promote growth and maturation in their respective target tissues are adrenocorticotrophic hormone, thyroid stimulating

hormone, follicle stimulating hormone, luteinizing hormone and erythropoietin (Tata, 1976; Nandi et al, 1980). Placental lactogen, relaxin, thrombin, transferrin and vasopressin also exhibit growth promoting activity on occasions (James & Bradshaw, 1984).

Certain steroids are also extremely important in the normal growth process. Thus, pubertal growth in girls is driven by low levels of oestrogen (Cutler et al, 1986). Oestradiol and progesterone powerfully stimulate cell division in mammalian uterine, vaginal and mammary tissue. Oestrogen will also stimulate proliferation of other tissues such as cultured decidual cells (Peleg & Linder, 1980) and rat hepatocytes, (Pietras & Szego, 1979). Additionally, oestrous-cycle controlled cell proliferation is apparent in the adrenal cortex of female rats (Pappritz et al, 1977).

Oestrogens are thought to exert some of their mitogenic effects via specific growth factors termed oestromedins. Thus, it is proposed that oestrogens may induce certain target tissues (uterus, kidney and pituitary) to synthesize or secrete polypeptide growth factors that enter the circulation and promote the growth of distant target tissue tumours (Sirbasku, 1978; Sirbasku & Moo, 1982; Sirbasku et al, 1982). More recently it appears that in addition to their endocrine actions, oestromedins may exert potent autocrine activity toward normal or neoplastic cells, or both, from the tissue of origin (Sirbasku et al, 1985). In contrast to their growth promoting role, oestrogens exert an inhibitory effect on erythropoiesis (Dukes et al, 1961). The immune system is also modulated by oestradiol and other steroids. Steroid administration clearly produces thymic atrophy whilst removal of steroids by ovariectomy, castration or adrenalectomy increases the size of the gland (Dougherty, 1952; Kinoshita et al, 1974; Castro, 1974; Sobhon & Jirasttham, 1974; Fitzpatrick et al, 1985; Greenstein et al, 1986).

Androgens stimulate growth of the male reproductive tract; testosterone is the predominant androgen produced and is synthesised in the Leydig cells of the testes. Testosterone may be converted to an even more powerful androgen 5a dihydrotestosterone within some target cells (Mainwaring, 1977; Mooradian et al, 1987) other tissues aromatize testosterone to oestradiol. Androgens control the growth of diverse tissues such as kidney (Bardin et al, 1978) skin fibroblasts (Keenan et al, 1975), skeletal muscle and fibroblast cells in culture (Powers et al, 1975). Androgenic steroids also exhibit a stimulatory effect on erythropoiesis (Fried et al, 1964; Gurney & Fried, 1965; Fried & Morley, 1985). The adolescent growth spurt is under androgen control and this is superimposed on the effects of growth hormone and thyroxine (Gupta, 1977; Cutler et al, 1986). There is evidence to suggest that growth hormone is necessary for the growth promoting action of testosterone, thus, in the hypophysectomized rat, testosterone alone does not increase the growth of any tissues other than accessory sexual glands (Scow & Hagan, 1965). Interestingly, testosterone influences the production and release of growth hormone in intact animals (Birge et al, 1967; Macleod et al, 1969; Jansson et al, 1984; Jansson et al, 1985).

In addition to these classical growth hormones there exists a further group of growth regulatory agents, the polypeptide growth factors (PGFs). Numerous recent reviews are a testimony to their potential importance in normal and neoplastic growth (Rozengurt & Collins, 1983; James & Bradshaw, 1984; Heldin & Westermark, 1984; Paul, 1985; Burgess, 1986; Goustin et al, 1986; & Weinstein, 1987). PGFs are distinct from the more classical endocrine hormones in that little is known about their role in the animal. Many growth factors have been isolated from serum and other body fluids by assaying the growth promoting activity in cells cultured in medium containing little or no serum. Earlier studies illustrated that not only do animal cells in culture require essential nutrients and maintenance at optimal pH and temperature mimicing conditions *in vivo* but also the presence of serum in the culture medium if proliferation is to occur (Holley & Kiernan, 1968; Holley, 1975). Arrest of growth in  $G_1$  causing cells to enter the quiescent phase  $G_{10}$  occurs when cells are deprived of serum or when the growth promoting activity of the serum present is exhausted. On addition of fresh serum cells re-enter the cell cycle (Pardee et al, 1978; Rozenburt, 1980). However, not until nutrient media and purified growth factors were available, was the study of the mitogenic actions of individual growth factors under defined culture conditions possible (Bottenstein et al, 1979; Shipley & Ham, 1981).

PGFs have since been shown to represent a large family of regulatory agents with many subsets characterised by structural similarities suggesting common ancestries. No single definition can be used to describe the extensive collection of PGFs which have so far been characterised. However, all are polypeptides and each PGF has an individual specific high affinity receptor located on the plasma membrane of target cells. Receptor ligand binding results in molecular responses that generally increase anabolism of the cell and modulation in expression of specific genes. The transmission of the regulatory signal is mediated by one or more second messengers but no universal mechanism has yet been determined. Lastly, PGFs and their receptors are extensively internalized by receptor-mediated endocytosis which ultimately leads to degradation of the ligand and of receptor recyling (Rozengurt & Collins, 1983; James & Bradshaw, 1984 Paul, 1987). PGFs are unlike the more classical endocrine hormones, since the majority of them are thought to act in a paracrine mode. However, there are notable exceptions, such as insulin like growth factors I and II (IGFs) which are endocrine (Froesch et al, 1985) and transforming growth factors (TGFs) which are autocrine (Anzano et al, 1985; Sporn & Roberts, 1985; Burgess, 1986).

Growth factors found in serum are presumed to be largely derived from platelets released during the clotting process (Holley & Kreman, 1974; Vogel et al, 1978; Childs et al, 1982; Oka & Orth, 1983). In addition PGFs are present in many tissues, adult and embryonic, and are thought to be released by most cells in culture (Shields, 1978). Specific combinations of growth factors are often required by normal cells for maximal stimulation of DNA synthesis (Barnes & Sato, 1980; Walthall & Ham, 1981; Rozengurt & Collins, 1983), it is proposed therefore that most cells will have membrane receptors for more than one growth factor (Wrann et al, 1980; Bowen-Pope et al, 1983).

Growth factors are thought to operate at different points of the cell cycle. According to the model of growth control initially proposed by Pledger and co-workers (1977) for density arrested Balb/C 3T3 cells, PGFs could be subdivided into two groups, the 'progression' and 'competence' factors. "Competence" factors prime the cells and enable them to respond to "progression" factors which stimulate the cells to traverse G1 and enter S phase, thus competence and progression represent two functionally distinct subphases of G0/G1 (Pledger et al, 1978; Stiles et al, 1979; Scher et al, 1982; Wharton et al, 1982; Leof et al, 1982). It is unclear how these two phases relate to the  $G_1$  pm and  $G_1$  ps subphases of the cell cycle (Zetterberg & Larsson, 1985), possibly "competence" factors enable cells to pass the control point at the end of G1 pm and become committed to mitosis, whilst "progression" factors allow these cells to traverse through G<sub>1</sub> ps into the S phase of the cell cycle. Competence factors such as PDGF have a restricted target specifically and act predominantly in the region in which they are produced (Alexander & Currie, 1984; Olashaw & Pledger, 1985), whilst factors in the progressive category include the systemically active insulin

like growth factors and epidermal growth factor (EGF) which stimulates a wide variety of cell types, both epithelial and mesenchymal (O'Keefe & Pledger, 1983; Alexander & Currie, 1984; Goustin et al, 1986).

The multiplicity of growth factors found in various tissues, the different cell type specificity of PGFs and the requirement for more than one PGF generally for maximal stimulation of specific cell types allows for the fine tuning of relative proliferation rates necessary for the co-ordinated growth of cells to form tissues during development and for adult tissue maintenance.

Included in the category of competence factors is platelet derived growth factor (PDGF) one of the major mitogens found in serum. Early studies showed that it is released from platelets into serum during blood clotting and that it is a potent mitogen for cultured fibroblasts, smooth muscle cells and glial cells (Balk, 1971; Ross et al, 1974; Scher et al, 1974; Antoniades et al, 1975; Westermarke & Wasteson, 1976; Scher et al, 1979). In the intact animal, PDGF may play a role in wound repair and in maintenance of the vascular system (Balk, 1971; Ross et al, 1974; Deuel & Huang, 1984; Lipton, 1985). More recently PDGF has been used as an experimental probe in the study of cell cycle control in fibroblasts. Thus, PDGF confers proliferative "competence" on density arrested Balb/C 3T3 mouse cells, ie. they are potentially able to leave  $G_0$  and enter the cell cycle (Pledger et al, 1977; Pledger et al, 1978; Stiles et al, 1979). It is thought that PDGF promotes the rapid synthesis of a family of cytoplasmic proteins within quiescent G<sub>0</sub> cells (Pledger et al, 1981). These proteins, gene products of the competence gene family which include c-myc and c-fos, confer "competence" and allow the cells to respond to "progressive" factors and traverse from G1 into the S phase of the cell cycle (Cochran et al, 1983; Cochran et al, 1984; Callahan et al, 1985; Lipton, 1985; Ollashaw & Pledger, 1985). Other growth factors which

belong to the PDGF family are the osteosarcoma derived growth factor, the glioma derived growth factor and the transforming protein of Simian sarcoma virus (Scher et al, 1979; Westermark et al, 1983; James & Bradshaw, 1984; Deuel and Huang, 1984; Heldin et al, 1986).

Pituitary and brain fibroblast growth factor (FGF) has also been shown to be a "competence" factor for Balb/c and Swiss 3T3 cell lines (Jimenez de Asua, 1977; Stiles et al, 1979; Jimenez de Asua, 1980). FGF is a potent mitogen which is considered to have a role to play in blastema cell proliferation *in vivo* during regeneration of lost appendages in the newt (Gospodarowicz & Mescher, 1981). Additionally FGF demonstrates angiogenic activity (Gospodarowicz et al, 1979) and induces cartilage regeneration *in vivo* (Jentzsch et al, 1980). FGF has both short and long-term effects on the morphology and growth pattern of many cells *in vitro*. It increases the rate of cellular proliferation, stabilises their phenotypic expression, stimulates cell migration and extends the culture lifespan (Gospodarowicz, 1984; Gospodarowicz, 1985).

Many other types of growth factors with broad target tissue specificities have been described which are included in the "progressive" category according to the model proposed by Pledger and colleagues (1977). These include epidermal growth factor (EGF), insulin and insulin-like-growth factors (sometomedins). Thus, once density-arrested Balb/c 3T3 cells are rendered competent by transient exposure to PDGF; EGF and either insulin or insulin-like growth factor I (IGF I) enable the cells to progress through  $G_1$  to the S phase. Moreover it appears that EGF is only required in addition to insulin or IGF I to induce the cells to leave  $G_0$ and progress for the first 6 hours of the 12 hour  $G_0/G_1$  lag and then IGF I alone is sufficient for progression through the last 6 hours of the lag period (Leof et al, 1982). Human fibroblasts, unlike 3T3 cells, only appear to require EGF for progression through  $G_0/G_1$  *in vitro*, probably due to autocrine secretion of IGF I (Clemmons et al, 1981; O'Keefe & Pledger, 1983). EGF has been isolated from murine submaxillary glands and visceral tissues (Cohen, 1959, 1962). It is mitogenic for a variety of cultured cells of ectodermal, mesodermal and endodermal origin (Carpenter & Cohen, 1979; Schlessinger et al, 1983). EGF promotes the proliferation of the basal cell layer of various epithelia of ectodermal origin *in vivo* (Cohen & Taylor, 1974; Cohen, 1984). The mitotic effect of EGF is strongly potentiated by insulin (Shipley et al, 1984), and EGF appears to be an important potentiator of one of the class of transforming growth factors (TGFß) (Roberts et al, 1981) illustrating once again that PGFs often act in combination to induce maximal DNA synthesis.

Other members of the EGF family are the group of transforming growth factors (TGFs) (Roberts et al, 1981; Marquardt et al, 1983; Brown & Blakeley, 1984; Roberts & Sporn, 1985). These polypeptides have the ability to confer the transformed phenotype on normal cells. Two types of TGFs have been well characterised, but only one of these, TGF $\alpha$  is in fact related to epidermal growth factor and binds to the EGF receptor (Todara et al, 1980; Marquardt et al, 1983; Keski-Oja et al, 1987). TGF $\alpha$  was originally observed in the conditioned medium of murine cells transformed by sarcoma virus as EGF competing activity that was associated with anchorage-independent growth of fibroblastic rat kidney NRK cells (Delarco & Todaro, 1978). Although TGF $\alpha$  is produced by different malignant cells including those transformed by viruses, it does not appear to be produced by normal adult cells, nevertheless it may have a benign physiological activity in developing tissues (Keski-Oja et al, 1987). By binding to authentic EGF receptors TGF $\alpha$  may permanently stimulate cell proliferation, thus transformed cells are able to synthesise, release and respond to their own growth

factors (Heldin & Westermark, 1984). TGFB is produced by a variety of normal and malignant cells, it is neither structurally nor functionally related to EGF and mediates its effects by distinct receptors (Roberts et al, 1983; Roberts & Sporn, 1985). Many of the mitogenic effects of TGFß are probably an indirect result of the activation of certain growth factor genes in the target cell (Keski-Oja et al. 1987). TGFB appears to have a bifunctional nature since depending on the cell types, growth conditions and the presence of other growth factors in culture it may exhibit growth stimulatory or growth inhibitory effects (Roberts et al, 1985; Nakamura et al, 1985; Roberts & Sporn, 1985). The apparently universal presence of TGFB in non-neoplastic tissues suggests that in addition to its possible action in mediating cell transformation, this factor might have an important role in controlling normal cell growth and function. Thus, it is thought that TGFB may function as a wound healing agent (Sporn et al, 1983; Assoian & Sporn, 1986). Evidently the possibility exists that TGFB in combination with other growth factors or inhibitors has the potential to play a key role in the control of normal and malignant cell proliferation (Keski-Oja et al, 1987).

In addition to the aforementioned "progression" factor, EGF, other growth factors which share the ability of stimulating competent cells to progress to the S phase of the cell cycle and ultimately mitosis include insulin and the insulin like growth factors (Stiles et al, 1979; O'Keefe & Pledger, 1983). IGFs were originally included in a group of factors termed sometomedins, a family of circulating peptides produced mainly in the liver which act as intermediaries in the stimulation of growth by growth hormone (Phillips & Vassilopoulou-Sellin, 1980; Hall & Sara, 1983). It is now believed that somatomedin C is chemically identical to IGF I (Spencer et al, 1983) and multiplication stimulating activity (MSA) has been shown to be the rat homologue of human IGF II (Marquardt et al, 1981). IGFs are capable of mimicing all of the biological effects of insulin on metabolism

and growth (Zapf et al, 1978) and are able to stimulate *in vitro* DNA synthesis and cell division not only in rat cartilage (Salmon & Duvall, 1970) but also in numerous other cell types, very often in combination with other PGFs (Van Wyk, 1984; Froesch et al, 1985; Moses & Pilistine, 1985; Zapf & Froesch, 1986; Flier et al, 1986). Recently it has been illustrated that IGF I stimulates growth in normal rats *in vivo* (Hizuka et al, 1986). Serum levels of IGF I are strongly dependent on growth hormone, reduced in states of hormone deficiency and increased in states of growth hormone excess (acromegaly) (Zapf et al, 1981). There is also evidence that IGF I levels may be modulated by sex steroids, thus in puberty an increase in IGF I is correlated with an increase in plasma oestradiol and testosterone. In young subjects low doses of these sex steroids enhance IGF I secretion, whilst oestrogen at higher doses and/or older subjects has an inhibitory effect on IGF I (Caufriez & Copinschi, 1986). The role of IGF II *in vivo* is unclear (Zapf et al, 1985) although it may play an important role in rat foetal growth and development (Moses & Pilistine, 1986).

Nerve growth factor (NGF) belongs to the insulin family of PGFs (Bradshaw, 1983) and although NGF was the first substance to bear the label "growth factor" (Levi-Montalcini, 1952) its role as a mitogen is controversial. NGF is essential for the survival and differentiation of a number of neural crest derivatives, including sympathetic and sensory neurones, it is essential for the maintenance of their specific functions and plays a role in regeneration (Levi-Montalcini, 1968; Thoenen & Barde, 1980; Calissano et al, 1984; Levi Montalcini, 1987). Early studies indicated that NGF might also have a mitogenic effect on these neurones (Levi Montalcini, 1960), but subsequent work implied that NGF promotes cell survival or differentiation rather than proliferation (Hendry, 1977; Thoenen et al, 1985). However recently it has been shown that NGF does indeed induce a mitotic response in cultured chromaffin cells from

young rats. These cells divide in the presence of NGF and become neuronal in the continued presence of NGF (Lillien & Claude, 1985). Thus, it is possible that NGF may have a potential mitotic effect on other neural crest derivatives at early stages of development *in vivo*.

Relatively well characterised growth factors have also been described which regulate the proliferation or differentation of haematopoietic progenitor cells, the haematopoetic growth factors (HGFs); these include colony stimulating factors (CSFs) (Burgess & Metcalf, 1980; Metcalf, 1985) and erythropoietin (Miyake et al, 1977). There are four well defined CSFs which differ in their effects within the hierarchial framework of haematopoietic progenitor cells and in the different cell lineages they affect (Yunis et al, 1983; Burgess, 1985; Arai et al, 1986). Multi-CSF also known as haematopoietic cell growth factor (HCGF) and interleukin 3 (Dexter et al, 1985; Ihle, 1985), has a very broad haematopoietic target specificity, GM-CSF stimulates granulocyte and macrophage progenitor cells whilst G-CSF and M-CSF restrict their effects to the granulocyte or macrophage cell lineages, respectively (Nicola & Metcalf, 1985). Characterisation of haematopoietic growth factors (HGFs) has been made from culture studies in which unfractionated bone marrow cells, thymocytes or spleen cells have been stimulated with growth factors produced by cell lines or tissues in vitro. Although haemopoiesis in vivo occurs within a stromal cell environment, indications are that haemopoiesis in the absence of stromal cells may be provoked in vitro if the appropriate growth factors are present in the culture medium (Metcalf, 1981). So far only two HGFs have been detected in animals, erythropoietin (Goldwasser, 1985) and GM-CSF (Metcalf, 1971; Metcalf, 1985). Erythropoietin controls the production of red blood cells (Goldwasser et al, 1985) and the correlation of erythropoietin levels and subsequent erythropoiesis is good. Indeed in vivo injection of this growth factor increases erythropoiesis in a dose dependent fashion

(Sherwood & Goldwasser, 1979).

Another group of regulatory proteins "the interleukins" have been described which regulate many aspects of the immune response. Many factors have been described, some of which either stimulate or inhibit lymphocyte cell division (Gillis & Mizel, 1981; Oppenheim & Gery, 1982; Duff & Durum, 1983; Schreier, 1984; Robb, 1984; Hemler et al, 1984; Nowell et al, 1985; Lachman , 1985; Smith, 1985; Smith & Rennick, 1986; Arai et al, 1986; Klaus 1986). The roles of the two most studied, interleukin 1 and interleukin 2, may be summarised, thus; interleukin 1 (IL1) a macrophage derived factor and the T cell derived T cell growth factor interleukin 2 (IL2) are known to take part in T cell activation and proliferation. Thus, T lymphocytes can be activated to move from the  $G_0$  into the  $G_1$  phase of the cell cycle by the synergistic action of IL1 and antigen (or lectin). The expression of interleukin 2 receptors on activated blasts allows them to

undergo proliferative expansion under the influence of IL2 secreted by themselves or by another activated T cell subpopulation (Wagner et al, 1980; Klaus & Hawrylowicz, 1984; Robb, 1984; Pfizenmaier et al, 1984).

Thus, there is a good deal of evidence which suggests that proliferation of normal cells is under the control of exogenous growth factors and classical hormones, many of which, although by no means all have been described in this section. Nevertheless, there is an opposing view which holds that, in the natural state all well nourished cells engage in proliferation and that *in vivo* this mitotic exuberance is suppressed by some form of negative feedback inhibitor. Historically, these inhibitors have been named "chalones", (Bullough & Laurence, 1968; Bullough, 1975; Allen & Smith, 1979). Because of the inherent difficulties in measuring cell inhibition in a specific manner and problems in the purification of the various chalones, the biological evidence for the existence of "chalones" has yet to be unequivocably established (Patt & Houck, 1980). Recently the isolation and purification of TGFß and a few other endogenous inhibitors of cell proliferation from several sources, including normal tissues, has provided some experimental basis for the hypothesis of negative regulation of cell growth (McMahon et al, 1982; Assosian et al, 1983; Roberts et al, 1985; Sporn & Roberts, 1985; Wang & Hsu, 1986). BSC 1 growth inhibitor was originally isolated from medium conditioned by BSC 1 cells (derived from African green monkey kidney cells) (Holley et al, 1980). It is now considered to be identical with TGFß (Holley et al, 1985). TGFß has been found in many normal tissues, and has been isolated from serum-free conditioned medium of normal and transformed cells. TGFß binds to specific cell-surface receptors in responsive cells and is established as a negative growth regulator that may function in the autocrine pathway (Sporn & Roberts, 1985).

Growth inhibitory activity has also been shown for a family of proteins, the "interferons" usually identified by their ability to protect cells against virus infections (Stewart, 1979). Cell growth inhibitory effects of interferons have been demonstrated both in cell culture and in tumour bearing animals (Taylor-Papadimitrou et al, 1982). Many cell types have shown varying degrees of sensitivity to the growth inhibiting properties of interferons. Nevertheless the proliferative activity of some normal cells including fibroblasts, epithelial cells and lymphocytes has been inhibited (Balkwill et al, 1978; Gurari-Rotman et al, 1978) in addition to inhibition of tumorigenic cells such as the Daudi line of Burkitts lymphoma-derived B lymphoblastoid cells (Broder et al, 1982). Furthermore, it appears that no single stage in the cell cycle is uniquely sensitive to interferon action (Clemens & McNurlan, 1985; Taylor-Papadimitriou et al, 1985).

Various soluble factors produced by leukocytes have been described that inhibit bone marrow colony formation *in vitro*, they include various types of interferons, prostaglandins, tumour necrosis factor and natural killer cell-derived

haematopoietic colony-inhibiting activity (Degliantoni et al, 1985). Recently the existence of an autocrine growth inhibitor has been postulated which inhibits the IL2 driven T cell proliferative response in a cloned T cell line D10 (Horowitz et al, 1986).

Rat platelets contain two inhibitors of growth for adult rat hepatocytes. These have been termed platelet derived growth inhibitors, PDGI $\alpha$  and PDGI $\beta$ (Nakmaura et al, 1986). The latter has very similar properties to TGF $\beta$  whereas PDGI $\alpha$  is distinct from TGF $\beta$  and other cell growth inhibitors, such as "liver chalone" isolated from adult rat liver (Iype & McMahon, 1984).

A growth regulatory factor which reversibly inhibits DNA synthesis and proliferation of fibroblasts has also been isolated. This factor termed FGRs (fibroblast growth regulator), is found in medium conditioned by density-inhibited mouse 3T3 cells (Hsu & Wang, 1986; Blat et al, 1986).

An extremely potent growth inhibitor has been recovered and partially purified from the culture medium of density-inhibited U79 cells (tumorigenic chinese hamster cells). Medium conditioned by exposure to these cells was able to reversibly inhibit growth and DNA synthesis of proliferating cultures not only of the same cell line but also of the Balb/c 3T3 A31 murine cell line (Koga et al, 1986).

Most of the examples of growth inhibition by regulatory factors have been illustrated in tissue culture systems and it remains to be determined whether knowledge about growth inhibitors can be applied to the intact animal. A possible role for growth inhibitors *in vivo* may be the prevention of overgrowth of tissues during wound healing. Thus there is limited evidence to suggest that cell proliferation may be controlled by the balance between growth promoting and inhibiting factors.

### 1.4 <u>Receptors</u>

Control of growth is governed by a signalling system that translates external information into a limited variety of internal signals termed 'second messengers'. The majority of hormones and growth factors signal their target cells usually through the formation of a recognition complex at the exterior of the plasma membrane, although the steroids involved in growth are thought to have their initial recognition interaction in the cytoplasm or nucleus of the cell (Clark & Peck, 1977).

For non-penetrating ligands the receipt and transduction of extracellular signals is a cell surface phenomenon dependent upon the existence of membrane associated receptors. The definition of a receptor is still evolving, but in its limited sense can be thought of as the initial recognition site on the cell surface with which a ligand interacts or binds (Kohn & Shifrin, 1982). The criteria that make a binding site a receptor are tissue specificity, stereospecificity, saturability, reversibility and high affinity (Rozengurt & Collins, 1983; Laduron, 1984). Most of the receptors or recognition molecules are large protein species which appear to be in a dynamic rather than a static state, both in terms of the turnover of these membrane localised entities (eg. de novo synthesis, internalization and degradation) and in terms of the mobility of certain receptors within the plane of the plasma membrane (Hollenberg, 1982; Helmreich & Elison, 1984). The cell surface receptor proteins span the membrane; the ligand binding domain protrudes from the external side of the cell surface and bears sugar residues which are thought to protect against proteolysis. Stability is also achieved by disulphide bridges between cysteine residues in the extracellular portion. Groups of charged basic amino acids just outside the membrane domain on the cytoplasmic face impose a steep energy barrier against movement of the receptor out of the plane of the membrane (Geisow, 1986). With the exception of insulin and IGF1 receptors,

PGF receptors are composed of single polypeptide chains. The molecular weight and subunit structures of the PGF receptors (except those for insulin and IGF1) do not show any obvious relationships despite their common involvement in growth regulation. This may indicate independent evolution of each ligand/receptor pair, despite the clear ancestral relationship of some of the hormones themselves. Alternatively, the techniques of analysis may have obscured similarities in their structures (James & Bradshaw, 1984).

Theoretically, hormone and PGF action can be regulated at many stages before, during and after their interaction with the cell surface receptor. Clearly receptor regulation can play an important role in modulating the cells response to an agonist. Desensitization is the tendency of biological responses to decrease over time despite the continuous presence of hormone or growth factor. Mitogenic desensitization can be achieved by two means; thus homologous or heterologous ligands can reduce either the number of receptors or alter the affinity of the receptor for its ligand (Rozengurt & Collins, 1983; Sibley & Lefkowitz, 1985). There is increasing evidence that receptor phosphorylation plays a key role in the regulation of receptor function; thus a major mechanism for heterologous desensitization of adenylate cyclase appears to be feedback regulation of the receptors via phosphorylation by the cAMP-dependent protein kinase (protein kinase A). In avian erythrocytes, following stimulation of adenylate cyclase by any of several hormones, intracellular cAMP levels rise, protein kinase A is activated and the ß adrenergic receptors become phosphorylated exclusively on serine residues (Stadel et al, 1983, 1986; Sibley et al, 1984a; Nambi et al, 1985). Protein kinase C is also able to phosphorylate the ß adrenergic receptor, thus treatment of intact erythrocytes with phorbol esters directly activates protein kinase C leading to receptor phosphorylation and desensitization (Sibley et al, 1984b; Kelleher et al, 1984). The inference is that protein kinase C activated by

phosphatidylinositol (PI) hydrolysis promoted by various ligand/receptor interactions may also play a role in some cases of heterologous phosphorylation and desensitization of the  $\beta$  adrenergic receptor (Sibley et al, 1987). Other receptors coupled to PI turnover also exhibit agonist-induced desensitization. For example treatment with phorbol esters leads to the desensitization of the

 $\alpha_1$ -adrenergic receptor (Leeb-Lundberg et al, 1985). A classical negative feedback loop may exist such that diacylglycerol, one of the second messengers generated by PI hydrolysis, activates protein kinase C. This kinase subsequently

causes the response to wane by phosphorylating and thus, desensitizing the  $\alpha_1$  adrenergic receptor (Sibley et al, 1987).

Certain of the plasma membrane receptors for polypeptide growth factors are themselves tyrosine kinases, these include EGF (Ushiro & Cohen, 1980; Carpenter, 1983) PDGF (Ek et al, 1982; Nishimura et al, 1982), insulin (Kasuga et al, 1982; Roth & Cassell, 1983; Kasuga et al, 1983) and IGF1 (Jacobs et al, 1983; Rubin et al, 1983). The receptor proteins span the membrane, the external portion binds the growth factor and a catalytic domain inside the cell accomplishes phosphorylation. Thus, the binding of the growth factor and tyrosine phosphorylation are activities of a single receptor molecule (Hunter, 1982; Sefton & Hunter, 1984; Hunter & Cooper, 1985). Receptor function of these PGFs may be regulated both by autophosphorylation of tyrosine residues and by phosphorylation catalyzed by other kinases.

Insulin receptors are integral membrane glycoproteins composed of  $2\alpha$ and 2ß subunits linked covalently by disulphide bonds. The ß subunit has a cytoplasmic domain that is homologous with the EGF receptor and the *src* family of protein-tyrosine kinases (Hunter, 1982; Hampe et al, 1984; Sefton & Hunter,

1984; Ullrich et al, 1985). When insulin binds to the  $\alpha$  subunits, the kinase activity of the ß subunits is enhanced. In cell free conditions, insulin-stimulated phosphorylation of the insulin receptor occurs exclusively on several tyrosine residues (Van Obberghen et al, 1983; Czech, 1985; Kahn, 1985; Pike & Krebs, 1986; White & Kahn, 1986; Goren et al, 1987).

The receptor autophosphorylation bestows kinase activity upon the cytoplasmic domain and several exogenous substrates may be phosphorylated. Once activated in this way the receptor kinase becomes insulin-independent (Klein et al, 1986; Yu & Czech, 1986). In intact cells receptors undergo phosphorylation on serine and threonine residues in addition to tyrosine residues implying that other protein kinases may also be involved in insulin receptor regulation. Indeed it has been demonstrated that activation of protein kinase A and protein kinase C in intact cells by other agonists leads to phosphorylation of the insulin receptor, decreased insulin binding and reduced protein-tyrosine kinase activity (Takayama et al, 1984; Haring et al, 1986). Thus insulin receptors appear to be under positive control, through autophosphorylation of the ß subunit leading to enhanced tyrosine kinase activity toward exogenous substrates and negative control though phosphorylation of serine and threonine residues (Sibley et al, 1987).

Activation of the EGF receptor also leads to enhanced protein phosphorylations on tyrosine residues on the receptor molecule itself (autophosphorylation) and on intracellular proteins (Sefton & Hunter, 1984; Hunter & Cooper, 1985; Pike & Krebs, 1986). In intact cells, treatment with EGF leads to the EGF receptor being phosphorylated on serine and threonine residues in addition to tyrosine residues and it appears likely that these phosphorylations are involved in the regulation of receptor function (King & Copper, 1986; Chinkers & Garbers, 1986). Protein kinase C is able to phosphorylate threonine residues on the EGF receptor (Hunter et al, 1984; Davis
& Czech, 1985). Protein kinase C phosphorylation of the EGF receptor inhibits receptor tyrosine kinase activity (Cocher et al, 1984; Downward et al, 1985), it decreases receptor affinity for EGF (Davis & Czech, 1986a) and appears to promote internalization of the receptor possibly by phosphorylating threonine 654 (Beguinot et al, 1985; Fearn & King, 1985). Not only does EGF stimulation promote rapid autophosphorylation of tyrosine residues, it also induces phosphorylation of threonine 654 leading to receptor internalization and tyrosine kinase inhibition (Chinkers & Garbers, 1986). This pathway thus provides a means of homologous and heterologous desensitization. Heterologous ligands that activate protein kinase C may exert negative regulation on the EGF receptor. Indeed, many heterologous ligands upon binding with their own specific receptors on 3T3 cells inhibit EGF binding indirectly by diminishing the apparent affinity of the EGF receptor population for EGF. The term "transmodulation" has been ascribed to this novel mechanism whereby structurally unrelated growth promoting agents such as phorbol esters, vasopressin, FDGF and PDGF are able to alter the affinity of the plasma membrane receptors for EGF (Rozengurt et al, 1981, 1982; Rozengurt & Collins, 1983) It is tempting to speculate that these transmodulating agents decrease the number or affinity of EGF receptors via a common mechanism; the activation of protein kinase C. Indeed, PDGF and vasopressin indirectly activate protein kinase C by inducing the formation of its endogenous activator diacylglycerol following PI hydrolysis, in 3T3 cells (Brown et al, 1984). However, recent evidence suggests that phorbol esters and PDGF modulate EGF binding via different mechanisms (Olashaw et al, 1986). PDGF causes a more protracted loss of EGF binding which requires preliminary protein synthesis (Olashaw & Pledger, 1987). A remarkable correlation exists between the agents that reduce EGF binding in 3T3 cells and those that act synergistically with EGF to promote proliferation of these cells (Brown et al, 1984; Collins &

Rozengurt, 1984; Davis et al, 1985). However, despite these correlations a causal relationship between heterologous receptor modulation and mitogenesis is not yet established (Olashaw et al, 1986).

In general then, it may be said that phosphorylation of tyrosine confers enhanced activity of receptors whereas serine or threonine phosphorylation diminishes receptor function. It should be pointed out that growth factor stimulated tyrosine kinase activity of the receptor might catalyse the phosphorylation and activation of serine and threonine protein kinases which could result in a cascade of phosphorylations some of which could be significant in proliferative control (Pike, 1983).

Fundamental to an understanding of how tyrosine kinase regulates cell proliferation is the identification of the primary substrate(s) of the growth factor receptor associated tyrosine kinase activity. A likely candidate appears to be a protein p42 which is phosphorylated on tyrosine residues by several mitogenic agents (Phorbol ester, PDGF, EGF and IGF1), although the precise function of this protein awaits identification (Bishop et al, 1983; Cooper et al, 1984; Foukes & Rosner, 1985; Hunter et al, 1985; Sefton, 1985).

Receptor phosphorylation may also play a role in the regulation of receptor distribution; an emerging paradigm is that receptor phosphorylation promotes receptor internalization. Receptor-mediated endocytosis is a mechanism by which cells internalize many macromolecules including certain growth factors such as EGF, PDGF, Insulin, Interleukin 1 and 2. (Hopkins, 1985; Goldstein et al, 1985; Hopkins 1986a; Hopkins 1986b; Dunn et al, 1986; Duprez & Dautry-Varsat, 1986; Mizel et al, 1987; Sibley et al, 1987).

The process is initiated following ligand binding when these receptors slide laterally and cluster in specialized regions of the membrane, "coated pits". These invaginations of the plasma membrane have a highly ordered lattice,

constructed in the protein clathrin, which may be involved in trapping the ligand receptor complex. These clathrin coated pits invaginate into the cell within minutes and form coated endocytic vesicles. After shedding their clathrin coats several vesicles fuse to form endosomes. The ligand may then be carried to lysosomes for degradation whilst the receptors may cycle back to the cell surface. Alternatively both receptor and ligand may be degraded or both be recycled (Goldstein et al, 1985).

Phosphorylation of receptors is proposed as a potential mechanism for signalling entry possibly by promoting receptor binding to clathrin. Protein kinase C activity has been demonstrated to be associated with the phosphorylation of transferrin (Klausner et al, 1984) insulin (Hachiya et al, 1987) and EGF receptors (Fearn & King, 1985; Beguinot et al, 1985; Chinkers and Garbers, 1986) and their internalisation. Protein kinase C activation may thus represent a general signalling mechanism that modulates the internalization of the membrane receptors.

Receptor mediated endocytosis functions in the rapid control of receptor number and in the removal of the growth factor or hormone from the circulation (Carpenter & Cohen, 1979; Terris et al, 1979). This form of receptor modulation, "down regulation" also plays a key role in reducing a target cell's sensitivity. Thus it is apparent that exposure to physiological concentrations of ligand induces a rapid and sometimes prolonged removal of receptors from target cell surfaces (Beguinot et al, 1984; Stoscheck & Carpenter, 1984). Down regulation can easily be demonstrated in tissue culture (Soll et al, 1974) and in chronic conditions *in vivo* where high levels of hormone are continuously present. In diabetic conditions, obese hyperinsulinaemic individuals have their insulin receptor populations reduced by more than 80% below normal, this process of 'down regulation' is thought to be the most likely cause of the disease (Archer et al, 1975; Soll et al, 1975). In less abnormal physiological conditions this form of receptor modulation is thought to occur in systems such as the reproductive system where there are pulsatile surges of hormone release and cyclical changes in receptor number (Dockray & Hopkins, 1982; Mooradian et al, 1987).

Heterologous ligands may also affect the number of receptors, for example treatment with phorbol esters reduces the number of EGF receptors in Hela cells (Lee & Weinstein, 1978, 1979). PDGF is thought to promote a transient down regulation of the EGF receptor in murine 3T3 cells (Wrann et al, 1980) in contrast to reports from other laboratories which suggest that PDGF mediates its effects on EGF binding in these cells by altering the affinity of the EGF receptor (Rozengurt & Collins, 1983).

Cellular desensitization to mitogenic ligands has thus the potential to regulate the proliferative response since it is crucial in determining the ultimate level of response. The interaction of the processes involving mitogenic desensitization and cell committment to DNA synthesis may provide a delicate mechanism to regulate growth control (Rozengurt & Collins, 1983). The extent of the proliferative response may also be effected by an 'up regulation' of hormone receptors in some instances (Hollenberg, 1982). Thus IL2 'up regulates' both high and low affinity IL2 receptors on T lymphoblasts (previously stimulated with lectin) (Diamantstein et al, 1986). In addition, an increase in the affinity of a receptor for its ligand may be promoted by heterologous ligands. Thus, a ten fold increase in the affinity of the IGF II receptor for its ligand is induced by insulin in isolated adipocytes (Oppenheimer et al, 1983) and rat hepatoma cells (Massague et al, 1982) despite the fact that IGF II receptor has essentially no affinity for insulin (Czech et al, 1983). Insulin is also believed to induce a redistribution of IGF II receptors from a microsomal compartment (which under basal conditions, contains most of the receptors) to the plasma membrane (Corvera & Czech, 1985).

Oestradiol has been shown to increase the number of EGF receptors 3 fold in rat uterine membranes within 12 hours of administration, probably by *de novo* synthesis of the EGF receptor. This raises the possibility that events coupled to the EGF receptor may play a role in oestrogen stimulated growth (Mukku & Stancel, 1985). Alternatively or in addition oestradiol may enhance the effect of EGF on growth.

Some PGF receptors such as those for insulin, (Krupp & Livingston, 1978) EGF (King & Cuatrecasas, 1982), NGF (Landreth & Shooter, 1980) and Interleukin-2 (Diamantstein et al, 1986) are reported to show two classes of affinity, as judged by the binding of radioiodinated ligand to various receptor preparations. In the case of NGF, it has been shown that the phaeochromocytoma PC12 cell possesses receptors with two different affinities, NGF binds first to receptors of low affinity and this binding induces a conversion of a proportion of the receptors to a higher affinity state. The change in affinity may be due to conformational changes in the receptor, interaction of the occupied receptor with other receptors or interaction with effector proteins in the plasma membrane. This change in receptor affinity may be a prerequisite for the internalization of the bound NGF or a sorting mechanism to select for only that NGF which is to be translocated to the nucleus (Landreth & Shooter, 1980; Hosang & Shooter, 1987). Thus, NGF-NGF receptor interaction is compatible with the mobile receptor hypothesis, receptors in the plane of the membrane change the apparent affinity with which they bind hormone when they interact with other effector molecules (possibly receptors) in the membrane (Jacobs & Cuatrecasas, 1976).

Classically, polypeptide hormones and growth factors bind to their specific receptors at the cell surface and initiate a chain of intracellular events mediated by second messengers which are considered in more detail in the next section. There is evidence however that certain polypeptide hormones and growth

factors also interact with the nuclei of their target cells (Burwen & Jones, 1987). Nuclear binding sites satisfy the criteria for hormone receptors, and have been demonstrated in target cells for EGF (Marti et al, 1987), NGF (Rakowicz-Szulczynska et al, 1986) and PDGF (Rakowicz-Szulczynska & Koprowski, 1986) by antireceptor antibodies raised against plasma membrane receptors. The origin of nuclear receptors is not known, they may be present before the arrival of the ligand to the cell and be distinct from their plasma membrane counterparts. Alternatively, they may be delivered to the nucleus together with their ligands in vesicles formed during receptor-mediated endocytosis (Burwen & Jones, 1987). A physiological role for nuclear receptors remains to be elucidated but may represent a mechanism for the phosphorylation of nuclear proteins by tyrsoine kinase activity present in the receptor molecule. Phosphorylation of specific nuclear proteins may result in a cascade of nuclear events involved in the initiation of mitosis (Burwen & Jones, 1987).

Although association of polypeptide hormones and growth factors with nuclei of target cells represents a novel alternative mechanism of effecting a biological response, steroid hormones are classically considered to interact with the genome following receptor binding. The mechanism of action of steroid hormones has generally been considered to incorporate the following stages. The steroid enters the target cell by simple diffusion and combines with a high affinity cytoplasmic receptor. The steroid-receptor complex undergoes tansformation (activation) and is transferred to the nucleus where it binds to selective sites in the chromatin. The hormone-receptor interaction with the genome leads to a modulation of RNA and protein synthesis. The synthesis of specifically induced proteins is finally responsible for the physiological response to the steroid (Jensen et al, 1968, 1982; Feldman et al, 1972; Edelman, 1975; Baxter & Harris, 1975).

More than 90% of steroid hormones that circulate in the blood are bound to plasma proteins; however it is the unbound fraction that is endowed with biological activity (Mooradian et al, 1987). The lipophilic nature of the hormone facilitates easy permeation of the plasma membrane by simple diffusion. However, the mechanisms of translocation of steroids across the membrane, the processes on and in the membrane and inside the cell regulating the influx are still subjects of conjecture. Thus, there is increasing evidence indicating that steroid hormones interact with components in the plasma membrane resulting in the transport of the hormones into the cell (Rao, 1981; Rao et al, 1977; Pietras & Szego, 1984). Once inside the cell, the hormone binds to a high affinity receptor, specific to the steroid and not to the response (Baulieu, 1984). The hormone acts to induce receptor transformation to the active chemical form that is capable of penetrating the nuclei (Gorski et al, 1968; Jensen & DeSombre, 1972). Although the precise mechanism(s) of transformation are unknown the possibilities include association or dissociation of subunits, conformational changes, addition or removal of cytosol factors, limited site-specific proteolysis and phosphorylation or dephosphorylation (Grody et al, 1982; Sherman & Stevens, 1984; Walters, 1985). The location of non-transformed receptor is particularly controversial (Schrader, 1984; Walters, 1985; King, 1986; Wolff et al, 1987; Jensen, 1987). Recent observations using monoclonal antibodies suggest that the unoccupied oestrogen receptor does not reside in the cytosol in intact cells as formerly thought (Jensen et al, 1968, 1972) but is exclusively found in the nucleus (King & Greene, 1984; Welshons et al, 1984). Furthermore similar observations have been made with the progesterone receptor (Perrot-Applanat et al, 1985), indicating that there is no translocation of the receptor from cytoplasm to nucleus under the influence of the hormone. Rather than being a compartmental shift from the cytosol to the nucleus, the change in receptor (receptor transformation), induced

upon hormone binding might be more subtle. Thus, steroid receptors may be permanently associated with the genes they regulate, and have only to rotate or undergo a conformational change to induce their effects (Shrader, 1984).

Nevertheless, steroid-receptor complexes are nuclear regulatory elements wherever they originate in the cell. The mechanism by which hormone receptor complexes increase specific transcription is still largely unknown (Baulieu, 1984). The activated or transformed receptor has a greater affinity for chromosomal sites in the cell nucleus; receptor association has been reported with almost all nuclear components; (Yamamoto & Alberts, 1976; Anderson, 1984). The selectivity of steroid action on transcription might reflect receptor binding to specific DNA sequences within or near regulated genes called steroid response elements (REs). Interaction of steroid receptors with REs may enhance transcription of specific proteins by potentiating the activity of particular transcription factors at nearby promoters, leading to enhancedmRNA synthesis, altered protein synthesis and ultimately regulation of cell function (Yamamoto, 1985). Alternatively, the selectivity of hormone-receptor action could reflect site specific receptor binding to a non-DNA 'acceptor' molecule. The avian progesterone receptor is claimed to interact with a particular class of non-histone proteins. This acceptor molecule may mediate the effect of the steroid-receptor complex and cause gene derepression perhaps by preventing production of repressor molecules (Thrall & Spelsberg, 1980; Yamamoto, 1985).

The numbers of steroid receptors in a given cell may vary in an analogous fashion to plasma membrane receptors under physiological circumstances. Thus, variations of progesterone receptor number are apparent in the guinea pig uterus during the oestrous cycle; the synthesis of progesterone receptors is induced by oestradiol at proestrous whilst progesterone causes 'down regulation' of its own receptor during the luteal phase (Baulieu et al, 1984).

Furthermore, it has been demonstrated that heterologous ligands such as EGF and insulin are able to phosphorylate tyrosine residues on hen oviduct progesterone receptor. The significance of phosphorylation of progesterone receptors, regarding their cellular functions has not yet been delineated although, ligand affinity, receptor transformation and/or affinity for nuclear sites could all potentially be modulated (Ghosh-Dastidar et al, 1983; Woo et al, 1986). Phosphorylation on tyrosine of the purified oestradiol receptor by an endogenous tyrosine kinase *in vitro* confers hormone binding capacity to this receptor (Migliaccio et al, 1984). More recently it appears that the oestradiol receptor is phosphorylated on tyrosine *in vivo*(Migliaccio et al, 1986). These findings illustrate that like PGF receptors, steroid receptors also interact with a protein tyrosine kinase. The fact that certain PGF receptors exhibit tyrosine kinase activity themselves whilst steroid receptors depend on endogenous tyrosine kinases for phosphorylation does not detract from the functional similarities between these receptors (Migliaccio et al, 1986).

Recent cloning and sequencing of oestradiol receptors (Walter et al, 1985; Green et al, 1986; Krust et al, 1986), glucocorticoid receptors (Hollenberg et al, 1985; Weinberger et al, 1985a, 1985b) has shown that both contain a large degree of homology with the v-erb-A gene product of the avian erythroblastosis virus (AEV). The AEV genome contains two cell-derived genes termed v-erb-A and v-erb-B. AEV causes erythroleukaemia in chickens; the transformation is due to the product of v-erb-B and is potentiated by v-erb-A gene product (Frykberg et al, 1983). 'v-erb-B' encodes a protein that is remarkably similar to the cytoplasmic portion (which incorporates the tyrosine kinase domain) of the EGF receptor (Downward et al, 1984). These homologies imply that the interaction between v-erb-A and v-erb-B gene products might be analogous to that observed between oestradiol receptor and the endogenous 'oestradiol receptor' tyrosine kinase (Migliaccio et al, 1984). It follows from this hypothesis that there is likely to be homology between oestradiol receptor kinase and v-erb-B gene product. Because of the homology between v-erb-A gene product and the oestradiol and glucocorticoid receptors it was suggested that the gene product of c-erb-A (the cellular counterpart of v-erb-A) was likely to be a receptor for a steroid-related ligand (Krust et al, 1986). Surprisingly, however it has been recently demonstrated that the product of c-erb-A gene specifically binds thyroid hormone and may therefore be the thyroid hormone receptor (Sap et al, 1986; Weinberger et al, 1986). It thus appears that steroid and thyroid hormones which are neither structurally nor biosynthetically related have receptors which have evolved from a common ancestor gene. This implies that there is a large superfamily of genes whose products are transcriptional regulatory proteins. This superfamily of receptor genes contains potential oncogenes encoding 'enhancer' factors (erb-A gene products) which can bind to response elements in the genome and induce ultimately the synthesis of specific proteins leading to the cellular response. 'Altered' enhancer factors, by interfering with the transcriptional regulation of crucial target genes might be important in certain types of carcinogenesis (Sap et al, 1986; Weinberger et al, 1986; Green & Chambon, 1986). Steroids are known to stimulate the growth of certain tumours, this may reflect an ability to enhance the effect of an oncogene, thus steroid receptors from neoplastic tissues may differ as v-erb-A and c-erb-A do (Parker, 1987).

Thus, there is an increasing body of evidence which does not fit with the current conceptual framework that a sharp demarcation exists between the mechanism of action of polypeptide hormones/PGFs and steroid hormones. This is further amplified and complicated by the recognition that the classical receptor mediated pathway of hormone action is not sufficient to account for all the known effects of steroids. Thus, Baulieu and co-workers (1978) demonstrated that

several steroid hormones are able to promote the maturation of *Xenopus laevis* oocytes by reinitiating meiosis although these cells do not contain classical steroid cytosolic receptors. Recently, a plasma membrane steroid receptor has been identified in *Xenopus laevis* (Sadler & Maller, 1982, 1984; Sadler et al, 1985). Steroid binding sites have been demonstrated on plasma membranes in several tissues (Harrison et al, 1976; Koch et al, 1977; Pietras & Szego, 1979, 1980; Towle & Sze, 1983; Haukkamaa, 1987). Thus, there is some evidence to indicate that steroids may mediate some of their effects via plasma membrane receptors rather than their classical mode of action which involves binding to intracellular receptors (Duval et al, 1983).

## 1.5 Signal Transmission

The importance of the internalization of certain PGFs involved in growth by receptor mediated endocytosis in signalling mitosis is still largely unknown. In contrast, the role of internalization in regulating the sensitivity of the cell to the mitogen is firmly established (see section 1.4). Many of the hormones/PGFs which have been shown to internalize have both short term and longer term trophic effects on their target cells. Thus, EGF binding elicits a series of rapid events including tyrosine kinase activation of its receptor and increases in ion flux (Sawyer & Cohen, 1981; Carpenter, 1984). These effects occur within seconds and precede internalization; DNA synthesis begins many hours later and could feasibly depend in part upon internalization of the ligand/receptor complex as well as on the generation of second messengers. After internalization the receptor and hormone are subsequently degraded, probably by lysosomal proteases. The role of cellular degradation in the effect of EGF stimulated DNA synthesis in 3T3 cells has been investigated by the use of pharmacological agents which inhibit peptide degradation. The EGF stimulated DNA synthesis is not affected by inhibition of its degradation suggesting that intracellular processing of EGF is not necessary for its mitogenic effects (Greenberg & Rozengurt, 1982). In other studies to determine the role of internalization in EGF-stimulated mitosis, receptor specific antibodies have been developed which will bind to the receptor but are unable to stimulate DNA synthesis unless they are cross-linked by another reagent (Shechter et al, 1979; Schreiber et al, 1983). These observations have led to the hypothesis that microclustering of receptors (a prelude to internalization) may play a role in the early stages of the mitotic signal (Schlessinger et al, 1983). However, since the initial internalization of EGF is not sufficient for EGF induced mitogenesis, the major role of internalization is thought by some to be the adjustment of the cells sensitivity to EGF (Aharonov et al, 1978). No consensus of opinion has yet been attained regarding internalization and mitosis (Hopkins, 1985). Perhaps both the rapid generation of Second messengers and internalization are necessary preludes in the stimulation of DNA synthesis by certain PGFs and hormones.

Another mechanism whereby growth promoting factors may initiate metabolic activities leading to DNA synthesis without the necessity of a second messenger has been proposed. A growth factor or hormone, on binding to its specific receptor may cause a conformational change allowing the complex to diffuse laterally in the plane of the membrane, the "mobile receptor hypothesis" (Jacobs & Cuatrecasas, 1976). This ligand/receptor complex is able to interact with membrane localised effector systems for which the complex has high affinity and change their functional properties (Cuatrecasas, 1974; Rozengurt, 1976). Since changes in the organisation of microtubules can alter protein-protein interactions (Berlin et al, 1974), the effect on DNA synthesis of agents known to dis *r*upt the microtubule network has been investigated (Friedkin et al, 1979; Friedkin & Rozengurt, 1981). These agents were shown to enhance the initiation of DNA synthesis to a variety of mitogens although the mechanism remains to be elucidated (Friedkin et al, 1979; Wang & Rozengurt, 1983). Alternatively, signal transduction might depend on the ability of the cytomatrix to transmit forces from the plasma membrane to the nucleus and components such as actin fibres, and microtubules might be involved (Packard, 1986).

Since these 'direct' effects of PGFs in the promotion of DNA synthesis and mitosis remain nebulous a great deal of energy is still directed towards gaining an understanding of the mechanism(s) of action of growth promoting factors and hormones. A central problem being to elucidate how after binding to specific receptors on the plasma membrane such mitogens are able to induce metabolic responses in the cell leading to proliferation since, after all, internalization of PGFs/hormones is far from universal. There is evidence suggesting the opening of ion permeability pathways through the plasma membrane, mobilization of ions from intracellular stores, alterations in the concentration of cyclic nucleotides, and phosphoinositide metabolism may play a role in signal transduction (Rozengurt, 1980, 1985; Rozengurt & Collins, 1983; James & Bradshaw, 1984; Paul, 1985; Downes & Michell, 1985; Berridge, 1986).

It is unlikely that the mechanism of action for all PGFs and classical hormones will be identical since it has already been acknowledged that the mitogenic responses to PGFs may be subdivided into 'progression' or 'competence'. Cells rendered 'competent' to replicate their DNA by PGFs in the 'competent' category may be stimulated to enter the S phase and subsequently mitosis by 'progression' factors. Additional evidence that growth factors do not operate through a universal mechanism is that some agents are able to initiate mitogenesis in a relatively short time whereas others require long periods of exposure to accomplish the same effect (James & Bradshaw, 1984). A remarkable feature of many mitogenic agents added to quiescent cells in serum-free medium is that they demonstrate synergy when administered in specific

combinations. These synergistic effects may be explained by their ability to trigger the generation of separate internal signals which act in a complimentary manner. This synergy can be further enhanced by the depolymerization of microtubules which is thought to act at a later stage (Rozengurt, 1980, 1985; Rozengurt & Collins, 1983). The regulation of cell proliferation by low concentrations of interacting growth factors may illustrate a flexible mechanism for the fine regulation of growth of individual cells in the intact animal (Rozengurt, 1980).

Thus, it becomes apparent that there are likely to be several intracellular signals which can couple the stimulus to the proliferative response, for a variety of growth factors and hormones in different cells, under different circumstances. In what follows, the early signals induced by PGFs/hormones leading to DNA synthesis and mitosis will be described, together with the possible interactions of these signal transduction pathways.

## 1.5.1 Cyclic nucleotides

cAMP has been implicated as an intermediate in the action of many peptide hormones since its discovery. Ligand-receptor binding induces membrane bound adenylate cyclase to catalyse the formation of cyclic AMP (Rall & Sutherland, 1962). Interposed between the receptor and effector molecule (adenylate cyclase) are two transducing or coupling proteins the G proteins (Gilman, 1984; Bourne et al, 1987). Each of these proteins binds  $Mg^{2+}$  and guanine nucleotides. The Gs protein is responsible for mediating the effects of stimulatory hormone receptors, the other, Gi protein mediates the effects of inhibitory hormone receptors (Codina et al, 1984; Jakobs et al, 1984; Birnbaumer et al, 1985). In fact adenylate cyclase systems to which receptors couple are three component systems formed of a catalytic unit C (which forms cyclic AMP plus Mg PPi from the substrate Mg ATP) and the Gs and Gi proteins (Birnbaumer, 1985). The catalytic unit serves as a monitor of the activity states of Gs and Gi. Addition of a stimulatory hormone upon binding to its receptor activates the Gs protein which subsequently results in increased C activity and the formation of cAMP. Phosphodiesterase located in the cytosol is the enzyme responsible for the degradation of cAMP (Wells & Hardman, 1977). The increased concentration of cAMP acts as a second messenger activating its protein kinase which potentially results in further phosphorylation reactions leading to the biological response.

The role of cAMP in the regulation of cell proliferation is highly controversial (Pastan, 1975; Friedman et al, 1976; Rozengurt et al, 1981; Ralph, 1983). Original studies regarding the role of cyclic AMP in cell proliferation demonstrated that the presence of the cyclic nucleotide in the culture medium of fibroblasts inhibited cell doubling, cultures of growing cells contained lower intracellular cAMP concentrations than cultures of non-growing cells and furthermore, adding serum to growth arrested cells promoted a decline in intracellular cAMP levels (Otten, 1971; Froehlick & Rachmeler, 1972; Rudland et al, 1974). Thus it was generally believed that cAMP was an inhibitory regulator of cell proliferation (Kram, 1973). However it now seems apparent that cAMP is stimulatory during the entire proliferative process in certain cell types and that temporal increases in cAMP have positive regulatory effects in other cell types even though the continued presence of cAMP might inhibit subsequent events necessary for division (Haddox & Greenfield, 1982).

In 3T3 cells and other fibroblast cells, increased levels of cAMP were generally believed to reduce the rate of growth and inhibit the stimulation of DNA synthesis promoted by serum addition to quiescent cells (Willingham et al, 1972; Burger et al, 1972; Bombik & Berger, 1973). An objection to these studies however was that the effects were elicited by high concentrations of cAMP

analogues and could be regarded as non-specific (Friedman et al, 1976).

Further support for a possible 'negative' control of cell proliferation by cAMP comes from observations that cAMP at high intracellular concentrations usually stops growth in the  $G_1$  phase of the cell cycle (Ralph, 1983). Moreover, the addition of serum to quiescent 3T3 fibroblasts has been shown to rapidly decrease the cellular cAMP content and promote entry into DNA synthesis (Pastan et al, 1975). Thus, there is considerable support for cAMP as a negative or inhibitory regulator controlling the normal mammalian cell cycle (Pardee et al, 1978). More recently the idea that cAMP inhibits growth has been challenged on the basis that cAMP analogues or cAMP elevating agents such as cholera toxin or prostaglandin E1 act synergistically with specific mitogens to stimulate DNA synthesis in quiescent 3T3 cells (Rozengurt et al, 1981; Rozengurt, 1982a.b; Rozengurt, 1985). In addition, cAMP stimulates DNA synthesis in a variety of other cells, such as adult rat hepatocytes (Armato et al, 1981), cultured guinea pig skin (Carney, 1982), cartilage (Kawashima et al, 1980) and MDCK kidney epithelial cells (Taub et al, 1979). T51B rat liver cells deprived of Ca<sup>2+</sup> also exhibit a pre-DNA synthesis surge of cAMP following Ca<sup>2+</sup> addition (Boynton & Whitfield, 1981). It is believed that this cAMP surge and a subsequent burst of cAMP-PK activity are required for events in the late  $G_1$  phase of the cell cycle (Boynton et al, 1985). Furthermore many other cells including fibroblasts and lymphocytes generate a transient cAMP surge as they near the end of the G1 phase (Boynton & Whitfield, 1983). This cAMP surge is believed to result from the activation of the c-ras proto-oncogene. There is evidence that c-ras protein mimics the action of adenylate cyclase's GTP binding  $G_S$  subunit. The implication is that cells may trigger this replication-related burst of adenylate cyclase activity by producing the highly conserved GTP-binding c-ras proteins. An increase in

cAMP and therefore cAMP-PK activity activates a cdc (cell division cycle) master gene(s) whose products lead to chromosome replication (Whitfield et al, 1985; Whitfield et al, 1987). cAMP may also have a regulatory role in  $G_2$  since a transient cAMP surge precedes both mitosis and meiosis in many kinds of cells (Boynton & Whitfield, 1983; Whitfield et al, 1987). This  $G_2$  surge like the early  $G_1$  surge must subside before the cell is able to progress to mitosis (Boynton & Whitfield, 1983). An interesting hypothesis is that the  $G_2$  cAMP surge might trigger the production of maturation promotion factor (MPF) (Whitfield et al, 1987).

The inconsistent reports of cyclic AMP metabolism in relation to the growth of cultured cells referred to earlier can perhaps be accounted for now. Thus, it is apparent that increased intracellular cAMP concentrations are stimulatory at one stage of the cell cycle and inhibitory at another; therefore, the administration of cAMP analogues to synchronized cells in different phases of the cell cycle may contribute to the contrasting observations.

The role of cGMP in the regulation of cell proliferation is probably even more controversial than cAMP; this is because there is little evidence of any receptor-mediated modulation of cGMP. Moreover, there is no real evidence that changes in cGMP have physiological effects within the cell (Hunt & Martin, 1980; Houslay, 1985) A negative regulatory role is implicated in some cases; addition of serum to 3T3 fibroblasts induces a rapid fall in cGMP concentration and promotes cell growth (MacManus et al, 1978). In contrast, cGMP has a stimulatory effect on DNA synthesis in cultured guinea pig skin (Carney, 1982) and primary neonatal hepatocytes (Armato et al, 1981).

The role of cGMP in proliferation of lymphocytes also remains hotly contested (Kaever & Resch, 1985). There are reports implicating cGMP as the

second messenger for B-cell activation; mitogenic agents were found to raise the concentration of intracellular cGMP but not cAMP (Coffey et al, 1977) and exogenous cGMP, its derivatives or cGMP elevating agonists are able to promote a mitogenic response in several cells (Seifert & Rudland, 1974; Hadden et al, 1979). On the contrary, other workers found that mitogenesis elicited small increases of cAMP rather than cGMP (Wedner et al, 1975; Parker, 1976). Moreover, other workers failed to correlate changes in either cyclic nucleotide concentration with the mitogen stimulated DNA synthesis (Watson, 1976). Haddox and co-workers proposed the 'Yin-Yang' hypothesis, which suggested that cell proliferation may be controlled by the relative concentrations of cAMP and cGMP within the cell. A transient increase in cGMP and an associated fall in cAMP was thought to trigger the initiation of cell division (Goldberg et al, 1975). More recently the role of cGMP has been implicated in mediating the action of atrial natriuretic factor in certain tissues (Leitman, 1985). cGMP is also considered to play a role in visual transduction in rod photoreceptors. The cGMP concentration transiently decreases following illumination, due to the stimulation of cGMP phosphodiesterase by rhodopsin. Transducin, a member of the family of G proteins mediates this signal-coupling (Kwok-Keung Fung, 1985). Perhaps in a similar fashion the role of cGMP in cell proliferation will be resurrected. This indeed seems likely since the tumor promoting phorbol esters increase cGMP concentration in a variety of cell types probably as a result of protein kinase C activation which ultimately phosphorylates and activates guanylate cyclase (Zwiller et al, 1985).

The role of cyclic nucleotides in the initiation of cellular proliferation is therefore difficult to interpret and made more complex by the observation that the cations,  $Ca^{2+}$  and  $Mg^{2+}$  are intimately connected with the cyclic nucleotides in several ways. Thus,  $Mg^{2+}$  is an absolute requirement for adenylate cyclase

activity and indeed may regulate its activity (Maguire, 1984).  $Ca^{2+}$  has many interactions with the cAMP system causing a rise or fall in cAMP concentrations in different cells by activating phosphodiesterase and regulating adenylate cyclase activity (Rasmussen & Barrett, 1984). Moreover,  $Ca^{2+}$  ions stimulate guanylate cyclase activity and exogenous cGMP will promote an immediate cAMP surge in thymic lymphocytes (Whitfield, 1979). Thus, the precise roles of the cyclic nucleotides may not be delineated at present and remain controversial, it is possible that cells may use precise, temporally distinct fluctuations in the local intracellular concentrations of the molecules or in their ratio, as signals serving to regulate cell cycle progression. Other cellular components such as cations and protein kinase C may initiate or modulate such fluctuations which ultimately control the cellular response.

## 1.5.2 Calcium and Phosphoinositide Metabolism

Calcium ions play an essential role in many key biological processes such as maintenance of cellular organisation, nerve conduction, bone formation, coupling of stimuli to their mechanical, secretory or metabolic responses (Rasmussen, 1970; Rasmussen & Goodman, 1977; Rasmussen & Barrett, 1984; Campbell, 1986; Rasmussen et al, 1986). There is also substantial evidence that calcium is implicated in stimulus-mitosis coupling (Rebhun, 1977; Swierenga et al, 1981; Durham & Walton, 1982; Whitfield et al, 1979, 1985, 1987). The role of calcium as a second messenger for regulation of numerous physiological processes linking stimuli from extracellular origins to intracellular environments is possible due to the steep 10,000 fold calcium gradient which is known to exist across the plasma membrane (Rasmussen, 1970; Rasmussen & Barrett, 1984). Therefore when a small quantity of calcium enters cytoplasm from the external medium or from internal membrane bounded stores there can be a transient change of several orders of magnitude in  $Ca^{2+}$  concentration, these intracellular  $Ca^{2+}$  transients regulate a variety of biological processes. This steep gradient can exist because under resting conditions the plasma membrane is relatively impermeable to  $Ca^{2+}$  and extrusion of  $Ca^{2+}$  from the cell to counteract the inevitable slight passive influx of ions is achieved by two mechanisms associated with the plasma membrane, an ATP dependent  $Ca^{2+}$  pump ( $Ca^{2+}$  ATPase) (Schatzman & Vincenzi, 1969; Huggins & England, 1985; Schatzman, 1985; Pederson & Carafoli, 1987) and by Na<sup>+</sup>/Ca<sup>2+</sup> exchange (Dipolo & Beange, 1983, 1986). The relative importance of the  $Ca^{2+}$  pump and Na<sup>+</sup>/Ca<sup>2+</sup> exchange mechanisms appear to vary between tissues (Cavero & Spedding, 1983; Huggins & England, 1985). In addition  $Ca^{2+}$  may be sequestered into intracellular organelles such as mitochondria, (Fiskum & Lehninger, 1980) and endoplasmic reticulum (Rasmussen & Barrett, 1984) or sarcoplasmic reticulum in some tissues (Feher & Briggs, 1982).  $Ca^{2+}$  ions may also be immobilised by forming complexes with acidic residues of phospholipids or proteins (Gill, 1982).

A rise in intracellular free  $Ca^{2+}$  may be accomplished in several ways, by influx of extracellular  $Ca^{2+}$  ions, mobilisation of  $Ca^{2+}$  from intracellular stores or by the inhibition of calcium extrusion.  $Ca^{2+}$  entry into the cell normally occurs through specific  $Ca^{2+}$  channels in the plasma membrane, the conductance of which is controlled by gates that are opened by critical changes in cellular polarisation (Voltage operated channel: VOC) or by receptor activation (Receptor operated channel: ROC) (Bolton, 1979). In smooth muscle a third channel, generally referred to as a  $Ca^{2+}$  leak into the cell is thought to be responsible for allowing  $Ca^{2+}$  into the resting cell, the resting influx is largely through these and is not susceptible to blockade by  $Ca^{2+}$  entry blockers (Bolton, 1985).  $Na^+/Ca^{2+}$ exchange may also enhance  $Ca^{2+}$  entry when intracellular  $Na^+$  is high (DiPolo & Beagué, 1986). The intracellular message generated by external stimuli is

transferred to the response mechanism by calcium receptor proteins that bind Ca2+ with high affinity. There are two classes of  $Ca^{2+}$  receptor proteins: (1) true Ca<sup>2+</sup> receptor proteins such as calmodulin, troponin C, parvalbumin and myosin light chain, which have no intrinsic enzymatic activity; on binding to  $Ca^{2+}$  they undergo a conformational change that alters their association with other proteins, response elements (REs), thereby changing the activity of these REs; and (2)  $Ca^{2+}$  regulated enzymes which bind  $Ca^{2+}$  directly and have no  $Ca^{2+}$  binding subunit, eg. protein kinase C (Rasmussen & Barrett, 1984). A further group of tissue specific calcium binding proteins, distinct from calmodulin which respond to the calcium signal, are the calcimedins and related proteins (Owens et al, 1984; Smith & Dedman, 1986 Geisow & Walker, 1986; Heizman & Berchtold, 1987). Nevertheless, it is well known that calmodulin (CaM) mediates many of the Ca<sup>2+</sup> regulated events in the eukaryotic cell and is the most universally distributed of the true Ca<sup>2+</sup> receptor proteins (Wang & Waisman, 1979; Means & Dedman, 1980; Klee et al, 1980; Cheung, 1982; Klee & Newton, 1985). It is felt by some that for CaM to activate an enzyme at least three of the four  $Ca^{2+}$  binding sites on CaM must be occupied resulting in the formation of Ca<sub>3</sub> CaMRE and Ca<sub>4</sub> CaMRE, the high activity states of the response element (RE) (Cox, 1984; Rasmussen & Barratt, 1984). However, other workers have detected three different conformers of calmodulin the  $Ca^{2+}$  free protein,  $Ca^{2+}_{2}$  calmodulin and the fully liganded calmodulin (Ca<sup>2+</sup><sub>4</sub> calmodulin) (Manalan & Klee, 1984; Klee & Newton, 1985). The different conformers may allow calmodulin to translate quantitative changes in  $Ca^{2+}$  concentration into qualitatively different physiological responses if different enzymes have the power to recognise the various conformers (Klee et al, 1980). Calmodulin has been shown to interact with an ever increasing number of enzymes and proteins and is thought to mediate directly or indirectly the Ca<sup>2+</sup>

regulation of such activities as glycogen and cyclic nucleotide metabolism, protein phosphorylations, microtubule assembly,  $Ca^{2+}$  flux and cell division (Chafouleas & Means, 1982; Veigl et al, 1984; Klee & Newton, 1985).

When a hormone or other extracellular messenger interacts with its specific receptor and raises the level of intracellular calcium ions, the flow of information through the CaM branch of the Ca<sup>2+</sup> messenger system is initiated. The increase in free cytosolic calcium concentration leads to the association of  $Ca^{2+}$  with binding sites on CaM. The binding of Ca<sup>2+</sup> to these sites leads to activation of a variety of REs, the physiological response of the cell results from these activations (Rasmussen & Barrett, 1984).

There is increasing evidence that phosphoinositide metabolism plays a vital role in the  $Ca^{2+}$  messenger system. Michell and colleagues in the mid-1970's found that mobilisation of  $Ca^{2+}$  in the cytoplasm of cells stimulated by hormones, neurotransmitters and other agents acting via cell surface receptors was nearly always associated with an increased turnover of inositol phospholipids (Michell, 1975; Michell et al, 1977). The link between the phosphoinositide response and the mobilisation of calcium from intracellular stores is attributed to the second messenger inositol trisphosphate (InsP<sub>3</sub>) (Berridge, 1984b, 1987; Berridge & Irvine, 1984).

This hypothesis is supported by the finding that  $InsP_3$  induces the release of  $Ca^{2+}$  ions from the endoplasmic reticulum (ER) of permeabilised pancreatic cells (Streb et al, 1983) and hepatocytes (Burgess et al, 1984; Joseph et al, 1984). Inositol bisphosphate, inositol phosphate and inositol 1,2 cyclic phosphate were unable to release  $Ca^{2+}$  from the ER of permeabilised pancreatic cells, suggesting that the release mechanism is specific for the trisphosphate (Streb et al, 1983). This has been substantiated by testing a range of inositol phosphates for their ability to mobilise  $Ca^{2+}$  from intracellular stores in other cells (Berridge & Irvine,

1984). The available evidence suggests that  $Ca^{2+}$  is mobilised from the ER rather than the mitochondria (Prentki et al, 1984; Biden et al, 1984) whilst InsP<sub>3</sub> induces calcium release from sarcoplasmic reticulum of skeletal muscle (Volpe et al, 1985). It is considered that InsP<sub>3</sub> acts by binding to a specific intracellular receptor on the ER to stimulate release of  $Ca^{2+}$  ions (Burgess et al, 1984; Hirata et al, 1985; Spat et al, 1986). Occupancy of this receptor releases  $Ca^{2+}$  into the cytosol via an unknown mechanism, which may however, involve a G protein (Dawson, 1985). Despite the evidence that points to a role for inositol lipid breakdown in receptor controlled  $Ca^{2+}$  mobilisation there is an opposing view that the inositol lipid breakdown may be a response to the  $Ca^{2+}$  mobilised in certain cells by receptor activation rather than a cause of the  $Ca^{2+}$  mobilisation (Cockcroft et al, 1981, Cockcroft, 1984).

The perturbation of membrane phospholipids following ligand receptor binding is thought to be the transducing mechanism that initiates a signal cascade, resulting not only in the mobilisation of calcium, but also the activation of protein kinase C, the release of arachidonic acid and the stimulation of guanylate cyclase leading to cGMP formation (Berridge, 1984b, 1986, 1987; Berridge et al, 1985; Downes & Michell, 1985; Downes, 1985).

The initial event following agonist-receptor binding is the hydrolysis of phosphatidylinositol 4,5, bisphosphate to inositol trisphosphate (InsP<sub>3</sub>) and diacylglycerol (DG) by phospholipase C, both of which have roles as intracellular messengers. These two second messengers are produced very rapidly, act at low concentrations and are removed by specific mechanisms once the external signal is removed (Berridge, 1984b, 1987).

As in the adenylate cyclase and rhodopsin/cGMP phosphodiesterase systems there is increasing evidence that a G protein may function in signal

transduction by coupling the surface receptor to phospholipase C (Wallace & Fain, 1985; Verghese et al, 1985; Cockcroft & Gomperts, 1985; Bradford & Rubin, 1986; Uhing et al, 1986; Baldassare & Fisher, 1986; Litosch & Fain, 1986). Evidence suggests that this G protein, Gp whilst related to other members of the ubiquitous family of G proteins (Dohlman et al, 1987; Bourne et al, 1987) is distinct from those involved in receptor regulation of adenylate cyclase (Merritt et al, 1986). When a hormone which mediates its effects by PI hydrolysis interacts with its receptor, the exchange of bound GDP to GTP on the  $\alpha$  subunit of the G protein is catalysed. This leads to dissociation of the heterotrimer to  $\alpha$  and  $\beta\gamma$ subunits. The released  $\alpha$  subunit activates phospholipase C and this enzyme catalyses the hydrolysis of PIP<sub>2</sub>. The  $\alpha$ -subunit possesses GTPase activity and so hydrolyses the bound GTP, thus terminating the action of phospholipase C (Cockcroft, 1987) The p21 protein encoded by the ras oncogene is also a typical G protein and is able to activate adenylate cyclase (Sadler et al, 1986). It may also have effects on phosphoinositide metabolism since ras-transformed cells produce inositol phosphates more efficiently than do normal cells in response to carbamylcholine (Chiarugi et al, 1985) and transformation by ras of two fibroblast cell lines (NIH 3T3 and NRK cells) leads to enhanced levels of DG and inositol phosphates (Fleischman et al, 1986; Wakelam et al, 1986).

The hydrolysis of PIP<sub>2</sub> represents a bifurcation in the signal pathway because both products function as second messengers. Diacylglycerol (DG) stimulates protein kinase C (PKC) (Nishizuka, 1984, 1986; Takai et al, 1985) and decreases the apparent  $K_m$  a and b for the calcium-mediated activation of protein kinase C (Kishimoto et al, 1980). A prior elevation in cytosolic Ca<sup>2+</sup> ions may enhance the activation of the enzyme by diacylglycerol illustrating the co-operative nature of PKC activation by both Ca<sup>2+</sup> and DG (Wolf et al, 1985; Dougherty & Niedel, 1986). During the activation of PKC, a process which also requires phosphatidylserine (Takai et al, 1979), the enzyme seems to move from the cytosol onto the membrane. Thus, there is evidence that IL2 and IL3 upon binding to their receptors promote a rapid and transient redistribution of PKC from the cytosol to the membrane (Farrar et al, 1985; Farrar & Anderson, 1985). Since IL3 does not appear to induce phosphoinositide turnover, it seems that certain PGFs might be able to activate PKC without generating DG (Berridge, 1987). Independent activation of PKC may be explained by the discovery of multiple forms of this enzyme (Coussens et al, 1986; Woodgett & Hunter, 1987). Although IL2 and IL3 share a common feature of transmembrane signalling with phorbol esters, ie. translocation of PKC, other growth factors do not appear to. In fibroblasts treated with phorbol ester, translocation of PKC was apparent whilst PDGF, FGF and Bombesin did not alter intracellular PKC localisation (Halsey et al, 1987).

The DG/PKC pathway is likely to play a significant role in cell proliferation in certain instances since the tumour promoting phorbol esters act by mimicking DG (Castagna et al, 1982) indeed PKC is believed to be the cellular receptor for phorbol esters (Fiyika et al, 1984; Parker et al, 1984). Activated PKC has been shown to phosphorylate many different proteins when studied *in vitro*, the identification of substrate proteins in the intact cell has been more problematical although vinculin (Werth & Pastan, 1984) is phosphorylated and the Na<sup>+</sup>/H<sup>+</sup> exchanger is believed to be activated by PKC. Evidence for this is that not only do many growth factors promote an increase in pH when stimulating cells to divide but that phorbol esters have also been shown to induce an increase in pH presumed to be under the control of DG/PKC pathway (Moolenaar et al, 1983; Takai et al, 1985).

The two second messengers generated following hydrolysis of  $PIP_2$  activate two separate ionic events both of which have been implicated in the control of cell proliferation (Michell, 1979, 1982ab; Berridge, 1984a; Downes & Michell, 1985). Further evidence for the role of inositol lipid turnover in the generation of signals that control cell proliferation is that PDGF a growth factor that stimulates  $PIP_2$  hydrolysis is encoded by the proto-oncogene c-sis (Doolittle et al, 1983; Robbins et al, 1983; Waterfield et al, 1983). and two 'tyrosine kinase' oncogenes src and ras stimulate polyphoshoinositide synthesis probably by encoding inositol lipid kinases (Sugimoto et al, 1984; Macara, 1984). The possible role of oncogenes in encoding proteins implicated in the inositol lipid signalling system will be discussed in more detail in Section 1.6.

The concept that the two limbs of the phosphoinositide metabolism pathway act synergistically with each other to regulate cell processes including cell proliferation is based on the findings that the effects triggered by agonists could be induced by a combination of calcium ionophore and phorbol ester. The calcium ionophore mimicks the effect of InsP<sub>3</sub> in raising intracellular calcium whereas the phorbol ester activates PKC. Many different cellular processes including DNA synthesis in lymphocytes can be activated by a combination of these two agents (Mastro & Smith, 1983; Truneh et al, 1985; Guy et al, 1985; Klaus et al, 1986).

An emerging paradigm is that growth factors and other agents belonging to the "competence" category eg. (PDGF, thrombin, phytohaemagglutinin) are capable of stimulating rapid inositol lipid breakdown. Another group of agents such as vasopressin and bradykinin stimulate  $PIP_2$  hydrolysis and synergise with insulin and/or EGF in stimulating fibroblast proliferation and therefore seem likely to act as competence factors (Macara, 1984). These 'competence' stimuli produce a rapid activation of c-fos proto oncogene and accumulation of the nuclear protein that it encodes (Greenberg & Ziff, 1984; Kruijer et al, 1984; Muller et al, 1984). This is followed by similar activation and expression of the c-myc proto-oncogene (Kelly et al, 1983). Thus, PIP<sub>2</sub> hydrolysis might constitute a 'competence' signal that would allow cells to divide on exposure to an appropriate progression signal such as insulin and/or EGF. However, a cautious approach in accepting this hypothesis is necessary since enhanced PIP<sub>2</sub> hydrolysis is not restricted to the critical  $G_0/G_1$  point where competence factors act. Thus ,stimulated inositol lipid turnover is sustained in dividing cells (Halenda & Feinstein, 1984) and rapid inositol lipid turnover has been demonstrated during the  $G_1$  and S phases of the cell cycle (Rampini & Dubois, 1981) the period during which the 'progression' factors are thought to act.

A link between the turnover of phosphatidylinositol and mobilisation of intracellular calcium in signal transduction is fairly well established, but it is also believed that the hydrolysis of PIP<sub>2</sub> may have a role to play in the entry of calcium across the plasma membrane (Michell et al, 1977; Michell, 1982a). In several types of cells, stimulation by receptors which trigger an initial, rapid increase in the intracellular free Ca<sup>2+</sup> concentration also give rise to a second, more prolonged elevated rise in Ca<sup>2+</sup> concentrations which appears to be dependent on the concentration of extracellular Ca<sup>2+</sup> (Putney, 1978; Joseph et al, 1985; Hesketh et al, 1985). Several possible molecular mechanisms underlying the control of receptor operated channels have been proposed. One idea advanced to explain how hydrolysis of these phospholipids might lead to a Ca<sup>2+</sup> influx is that phosphatidic acid (PA) formed from DG may function as a calcium ionophore (Putney et al, 1980; Salmon & Honeyman, 1980; Mitchell, 1982a). Other workers consider that the Ca<sup>2+</sup> influx which accompanies the phosphatidylinositol

turnover is not due to the ionophoretic action of PA but to Na<sup>+</sup>/Ca<sup>2+</sup> exchange. Thus, treatment of isolated canine cardiac sarcolemmal vesicles with phospholipase D (which converts phospholipids to phosphatidic acid) increases Na<sup>+</sup>/Ca<sup>2+</sup> exchange activity by up to 400% (Philipson & Nishimoto, 1984). The effect was specific for  $Na^+/Ca^{2+}$  exchange and the enhanced  $Ca^{2+}$  uptake was Na<sup>+</sup> gradient dependent. Electrogenic Na<sup>+</sup>/Ca<sup>2+</sup> exchange can transport Ca<sup>2+</sup> in both directions across the plasma membrane (Aickin et al, 1984). Whether stimulation of Na<sup>+</sup>/Ca<sup>2+</sup> exchange by PA will result in a net influx or efflux of  $Ca^{2+}$  will be a function of the ionic gradient and membrane potential. PA stimulated Na<sup>+</sup>/Ca<sup>2+</sup> exchange might also be important in returning elevated cytosolic  $Ca^{2+}$  to its normal level following the phosphoinositide response (Philipson & Nishimoto, 1984). A more recent model for receptor-regulated entry of calcium proposes that the initial emptying of the intracellular  $Ca^{2+}$  pool by InsP<sub>3</sub> leads to an opening of a pathway from the extracellular milieu to the intracellular pool, this fills with  $Ca^{2+}$  from outside the cell and is subsequently released into the cytosol by the continued presence of InsP3 (Putney, 1986).

Even more recently a number of novel inositol phosphates have been shown to be produced upon hormonal stimulation of cells, some of which may in the future be designated second messengers (Palmer et al, 1986; Michell, 1986). Indeed one of these,  $Ins(1,3,4,5)P_4$  is thought to play an important function in modulating intracellular  $Ca^{2+}$  levels. Phosphorylation of 1,4,5IP<sub>3</sub> a step that may be stimulated by the increased cytosolic  $Ca^{2+}$  concentration (following mobilisation of internal stores) (Michell, 1986b; Biden & Wollheim, 1986; Zilberman et al, 1987) gives  $InsP_4$ .  $InsP_4$  is believed to regulate transfer of  $Ca^{2+}$ from the extracellular space to the intracellular ER pool. For as long as 1,4,5,IP<sub>3</sub> (and/or 1,3,4IP<sub>3</sub>) is present the ER  $Ca^{2+}$  gate is open and  $Ca^{2+}$  can enter the cytosol from the extracellular milieu via the ER (Irvine & Moor, 1986; Nahorski & Batty, 1986; Houslay, 1987; Taylor, 1987).

Despite the recent interest linking hydrolysis of membrane phosphoinositides with calcium gating and calcium mobilisation from intracellular stores following ligand receptor binding, in some instances agonists may be capable of opening calcium channels in the plasma membrane without any change in phosphoinositide metabolism. But no matter how an increase in  $[Ca^{2+}]i$  is achieved, termination of the cell's response is the reduction in the cytosolic  $Ca^{2+}$ concentration which may be accomplished by  $Na^+/Ca^{2+}$  exchange and/or by energy dependent pumps which extrude  $Ca^{2+}$  from the cell or store it in subcellular organelles. Accumulated  $Ca^{2+}$  is stored largely in the mitochondrial matrix (Fiskum & Lehninger, 1982; Akerman & Nicholls, 1983). It is slowly lost from this pool into the cytosol and then into the extracellular environment without flooding the cytosol, so that reactivation of the cell does not occur (Rasmussen & Barrett, 1984). The role of mitochondria in regulating the cytosolic calcium concentration is questioned by some workers who feel that the major role of the  $Ca^{2+}$  transport system in mitochondria is to relay changes in cytoplasmic  $Ca^{2+}$ concentrations into the mitochondrial matrix. Ca<sup>2+</sup> ions stimulate intramitochondrial oxidative metabolism, thus actively promoting the synthesis of ATP. The increase in ATP supply meets the enhanced demand of energy requiring processes in the stimulated cell. They consider that  $Ca^{2+}$  ions should be considered to act as a second messenger in 'stimulus-response-metabolism' coupling (Denton & McCormack, 1980, 1985; McCormack & Denton, 1986).

The evidence presented here illustrates that  $Ca^{2+}$  is a potential major second messenger for regulation of numerous physiological processes. Direct evidence implicates  $Ca^{2+}$  in the regulation of cell proliferation and transformation. The evidence is clearest with regard to egg cells; following activation a rise in intracellular Ca<sup>2+</sup> concentration has been directly observed using the photoprotein aequorin (Cuthbertson et al, 1987). Lymphocytes in culture can be activated into 'capping' and ultimately into DNA synthesis by a variety of agents; extracellular  $Ca^{2+}$  has been shown to be essential for activation and that a small rise in cytosolic  $Ca^{2+}$  ions ([ $Ca^{2+}$ ]i) occurs in response to mitogenic doses of plant lectins (Tsien et al, 1982; Lichtman et al, 1983; Hesketh et al, 1983). Serum, PDGF & EGF immediately raise the [Ca<sup>2+</sup>]i in human fibroblasts (Moolenaar et al, 1984; Mix et al, 1984). The elevation of  $Ca^{2+}$  concentration in the external medium may under certain conditions trigger resting 3T3 cells to divide (Dulbecco & Elkington, 1975). Furthermore an increase in [Ca<sup>2+</sup>]i following serum stimulation has also been observed in swiss 3T3 cells and is thought to involve the phosphoinositide signal pathway (Lopez-Rivas & Rozengurt, 1983; McNeil et al, 1985). Many individual growth factors and other mitogens have also been demonstrated to cause rapid transient increases in [Ca<sup>2+</sup>]i in swiss 3T3 cells (Morris et al, 1984; Ives & Daniel, 1987).

There is also evidence that Ca<sup>2+</sup> controls cell replication *in vivo*; cellular proliferation in rat liver following partial hepatectomy is prevented by hypocalcaemia induced by prior parathyroidectomy (Rixon & Whitfield, 1976). DNA snythesis may be restored by appropriately timed calcium chloride or parathyroid hormone administration post-hepatectomy. The isoproterenol-induced activation of the rat parotid gland cells is similarly inhibited by prior removal of the parathyroid glands to elicit hypocalcaemia and may be reversed by appropriate hormone administration (Tsang et al, 1981).

The evidence suggests therefore that in normal cells an adequate supply of calcium is necessary for cell proliferation; the  $Ca^{2+}$  sensitive event may be at the late  $G_1/S$  boundary (Whitfield et al, 1981, 1985) or early in the  $G_1$  phase of the

cycle (Paul & Ristow, 1979). In resting lymphocytes it is believed that the transition of cells from the resting state  $G_0$  into the cell cycle is regulated by an increase in the [Ca<sup>2+</sup>]i above a critical point, cells remain in the cell cycle unless  $[Ca^{2+}]$  i falls below a lower limit in which case the cells enter G<sub>0</sub> quiescent phase. If the  $[Ca^{2+}]i$  rises above an upper limit of about 1µM cells are inhibited and if prolonged the high concentration of  $[Ca^{2+}]i$  may be cytotoxic (Metcalfe et al. 1980; Hesketh et al, 1982). Since it is now well established that calmodulin mediates many of the Ca<sup>2+</sup> regulated events in the eukayotic cell (Wang & Waisman, 1979; Means & Dedman, 1980; Vergl et al, 1984) it is not surprising that calmodulin may also play a role in cell cycle progression. Indeed when calmodulin levels were determined in synchronised CHO-K1 cells throughout the cell cycle, it was found that during late G1 and/or early S phases, calmodulin levels begin to increase and reach the maximum level at the beginning of the M phase (Chafouleas & Means, 1982; Sasaki & Hidaka, 1982). Although increased total calmodulin levels have been observed in these and other cultured cells prior to or just after the onset of DNA synthesis (Criss & Kakiuchi, 1982; Bourgeade et al, 1983; Boynton & Whitfield, 1983) the levels remain unchanged in certain NRK cells and parotid cells (Conner et al, 1983; Durkin et al, 1983; Gonzales et al, 1984). One hypothesis is that  $Ca^{2+}$  calmodulin must reach a critical concentration to enable progression from  $G_1$  to S to occur. Specific functions that  $Ca^{2+}$  calmodulin performs which may be important at the  $G_1/S$  transition are regulation of the cytoskeleton, protein phosphorylations in general and histone phosphorylation in particular (Chafouleas & Means, 1982). One of the proteins that may be activated by calcicalmodulin is the cytoplasmic stimulatory factor which can induce DNA synthesis in isolated nuclei; this protein may represent the

molecular trigger for the  $G_0/G_1 \longrightarrow S$  transition (Gutowski & Cohen, 1983).

In addition to acting in concert with the DG/PKC limb of the PIP2 hydrolysis pathway,  $Ca^{2+}$  or  $Ca^{2+}$ . calmodulin often functions in concert with cAMP to couple stimulus to response (Rasmussen & Goodman, 1977; Rasmussen & Barrett, 1984). Likewise in stimulus-mitosis coupling, calcium does not always act alone, the co-trigger of the late G1 events in some cells is a transient cAMP surge which precedes the initiation of DNA synthesis (Whitfield, 1982; Whitfield et al, 1985). A model has been proposed to account for their interaction in promoting DNA synthesis. A rise in intracellular calcium activates calmodulin which in turn stimulates phospholipase A2, this releases arachidonate from various phospholipids. Arachidonate is converted to a prostaglandin which stimulates adenylate cyclase causing a transient increase in cAMP concentration. The cAMP binds to a type II cAMP dependent protein kinase which is translocated to the nucleus and pohosphorylates an unidentified protein or proteins leading to the initiation of DNA synthesis (Gillies, 1982). Interaction between the two second messengers may occur in controlling the phosphorylation state of proteins, thus a cAMP dependent protein kinase might control the phosphorylation of one protein in an enzyme cascade whilst a  $Ca^{2+}$  dependent protein kinase controls the phosphorylation of a subsequent enzyme in the cascade (Rasmussen & Barrett, In addition, in some circumstances both  $Ca^{2+}$  and cAMP dependent 1984). protein kinases may catalyse the phosphorylation state of the same substrate protein at different phosphorylation sites (Le Peuch et al, 1979, 1982).

Evidence has been presented which demonstrates the ability of some second messengers to act in combination. It appears that the hydrolysis of membrane phospholipids may represent a fundamental transducing mechanism that initiates a signal cascade comprising a variety of putative second messengers

which act co-operatively. Thus, hormones and growth factors which bind to receptors that hydrolyse phosphoinositides have the potential to produce  $Ca^{2+}$  signals, to activate protein kinase C which may lead to activation of a neutral Na<sup>+</sup>/H<sup>+</sup> exchange carrier (Vicentini & Villereal, 1986). Both an increase in pH and in cytosolic Na<sup>+</sup> concentration have been implicated as signals which may induce proliferation (Cassell et al, 1983; Moolenaar et al, 1983). In some cells arachidonic acid may also be released (Irvine et al, 1982; Rittenhouse, 1982) which may be converted to prostaglandins, thromboxanes and leukotrienes (Berridge, 1984, 1987).

Receptors involved in PIP<sub>2</sub> hydrolysis also appear to be linked to the generation of cGMP (Michell, 1975; Berridge, 1981). It has been suggested that guanylate cyclase may be regulated by either arachidonic acid or one of its metabolites (Peach, 1981; Takai et al, 1982; Gerzer et al, 1983). Thus, the generation of several of these putative second messengers appears to be inexorably linked. Interaction of some or all of these second messengers may contribute to the final response, the different signals may also provide the versatility necessary to introduce subtle variations in the control mechanisms. These second messengers also interact at other stages in the signal pathway. Thus, although it is evident that PIP<sub>2</sub> hydrolysis leads to mobilisation of Ca<sup>2+</sup> ions, it is also recognised that PIP<sub>2</sub> hydrolysis <u>depends</u> on  $Ca^{2+}$  ions. Phospholipase C (PLC) the enzyme responsible for the hydrolysis of PIP2 and subsequent production of the two second messengers DG and InsP3 requires at least the normal resting intracellular Ca<sup>2+</sup> concentration (Smith et al, 1986). Indeed at lower concentrations of [Ca<sup>2+</sup>]i the activity of the enzyme is reduced or even abolished (Moscat et al, 1986; Renard et al, 1987). Moreover since Ca<sup>2+</sup> ionophores

stimulate the phosphatidylinositol cycle in several types of cells (Laychock, 1983; Charest et al, 1985; Murayama & Ui, 1985) it appears that  $Ca^{2+}$  ionophores induce increases in  $[Ca^{2+}]i$  which are capable of activating phospholipase C without acting through a specific receptor coupling system (Moscat et al, 1986; Cockcroft, 1987). Similarly, high  $Ca^{2+}$  concentrations (1mM) are able to activate PLC in polymorphonuclear leukocyte membrane preparations in the absence of another stimulus (Smith et al, 1985).

Calcium ions are also required for the activation of protein kinase C (PKC) in addition to phospholipids and diacylglycerol (Takai, 1979; Schwantke et al, 1985). It appears that when a hormone or growth factor activates a cell via the phosphoinositide signal pathway the initial transient rise in  $[Ca^{2+}]i$  is important not only in activating the CaM branch but also in determining the amount of PKC that becomes membrane bound and activated (Rasmussen & Barrett, 1984). The emerging hypothesis is that when the calcium messenger system couples stimulus to sustained cellular response, the flow of information occurs via two distinct branches which have distinct temporal roles. The CaM branch is involved in initiating the response and the PKC branch in sustaining the response. The magnitude of the sustained response appears to depend on the amount of PKC associated with the membrane and the rate of  $Ca^{2+}$  cycling across the membrane (Rasmussen et al, 1986). cAMP is also capable of modulating the  $Ca^{2+}$ messenger system, a rise in cAMP can lead to either a negative or a positive change in the set point depending on the particular tissue (Rasmussen et al, 1986). In a similar vein, PKC activation also has effects on cAMP formation; in intact platelets activation of PKC with phorbol ester leads to inhibition of prostaglandin and forskolin stimulated cAMP formation (Williams et al, 1987), whereas in certain other cells (rat pinealocyte and phaeochromocytoma cells) activation of PKC enhances agonist induced cAMP formation (Sugden et al, 1985;

Hollingsworth et al, 1986).

## 1.5.3 Magnesium

Although calcium ions have been implicated in the control of cell division and the cell cycle for a long time it is only relatively recently that magnesium has also been recognised as a possible regulator of cell proliferation (Rubin, 1975; Sanui & Rubin, 1982a; Walker & Duffus, 1983; Walker, 1986). Magnesium is thought to be ideally suited to play a primary role in the co-ordinate control of metabolism and cell proliferation since it is required as a cofactor by more intracellular enzymes than any other inorganic cation.

In mammalian cells total intracellular magnesium is maintained at approximately 10 mM over a broad range of external Mg<sup>2+</sup> concentrations suggesting an efficient homeostatic mechanism (Rink et al, 1982). Little is known about cellular magnesium homeostasis although a specific magnesium transport system has been described (Maguire & Grubbs, 1983). In addition an inwardly directed Mg<sup>2+</sup> pump has been implicated but not characterised in mammalian cell membranes (Sanui & Rubin, 1982b). Studies with chicken erythrocytes have indicated that the regulation of intracellular free  $Mg^{2+}$  is achieved by a gating process at the cell surface which is active at increased concentrations of intracellular free  $Mg^{2+}$  allowing exchange of 1  $Mg^{2+}$  for 2 Na<sup>+</sup> until the physiological level of free  $Mg^{2+}$  is reached. This efflux is independent of calcium (Gunther et al, 1984; Gunther & Vormann, 1986). The measurement of free  $Mg^{2+}$ , performed by several different methods gives a value of approximately 1 mM (Rink et al, 1982), which is in the range of, or below, the concentration necessary for maximum activity of many magnesium dependent enzymes (Sanui & Rubin, 1982b). Thus, there is the possibility that small changes in intracellular Mg<sup>2+</sup> levels may have far reaching effects on cellular metabolism and proliferation. Evidence which supports the hypothesis that Mg<sup>2+</sup> may act as a

second messenger is that  $Mg^{2+}$  transport and therefore internal  $Mg^{2+}$ concentrations are subject to hormonal control. ACTH, adrenaline and noradrenaline cause an elevation in the accumulation of Mg<sup>2+</sup> by adipocyte plasma membrane vesicles (Elliot & Rizack, 1974). Insulin specifically induces Mg<sup>2+</sup> uptake in fibroblasts and adipocytes (Cech et al, 1980). Enhanced magnesium uptake has also been demonstrated in rat pancreatic islet cells by stimulators of B-cell function (Henquin et al, 1983, 1986). A hormone sensitive Mg<sup>2+</sup> transport system has been demonstrated in murine S49 lymphocytes wherein Mg<sup>2+</sup> influx is inhibited by activation of ß adrenergic receptors. The hypothesis is that free Mg<sup>2+</sup> has an important role in the regulation of hormonal responsiveness; following hormone-receptor interaction a subcytoplasmic pool becomes depleted of Mg<sup>2+</sup> such that a second hormonal stimulation would elicit a decreased response both from a lack of  $Mg^{2+}$  to activate adenylate cyclase and from a lack of Mg<sup>2+</sup> to form the initial high affinity agonist receptor complex (mediated by the G protein), (Maguire, 1984). A similar hormone sensitive Mg<sup>2+</sup> transport system has been demonstrated in other cell lines and in response to other hormones (eg.PGE<sub>1</sub>) and adenosine, and it may be that hormonal modulation of  $Mg^{2+}$ transport leading to desensitisation is a general property of all receptor cyclase systems regardless of the hormone involved (Erdos & Maguire, 1983).

There is evidence that  $Mg^{2+}$  can modulate hormone stimulated responses in two distinct ways; firstly by interfering with the Ca<sup>2+</sup> homeostatic system and thus mediating its effect by ultimately altering the intracellular Ca<sup>2+</sup> concentration and secondly by influencing ligand-receptor interaction. Magnesium influences intracellular calcium homeostasis in several ways, it can regulate both calcium exit and entry processes and influence intracellular distribution. Thus, magnesium stimulates Ca<sup>2+</sup> influx into sarcoplasmic reticulum and inhibits calcium dependent efflux (Stephenson, 1981). Magnesium may also retard calcium efflux into the
extracellular space (Altura & Altura, 1981) or when the  $Mg^{2+}$  concentrations are lower, promote efflux (Turlapaty & Altura, 1978). Thus, in this manner,  $Mg^{2+}$ may act as a physiological regulator of cytosolic Ca<sup>2+</sup> levels within the cell so that Ca<sup>2+</sup> ultimately assumes the role of second messenger.

Extracellular magnesium ions can induce an increase in the number of functional surface receptors available to an agonist (Pearlmutter & Soloff, 1979) and thus it is feasible that magnesium may help to stimulate the cellular response by causing the hormone to bind to more receptors. Physiological levels of Mg<sup>2+</sup> induce a high affinity state of the ß receptor in many cellular systems including S49 cells (Maguire & Erdos, 1978). The role of  $Mg^{2+}$  in stimulus response coupling in systems utilising adenylate cyclase/cAMP formation is very important. A  $Mg^{2+}$  site on the G protein is responsible for altering the ability of the GTP regulatory unit to modulate both agonist affinity for the hormone receptor and the interaction of the catalytic subunit with the GTP regulatory unit (Cech & Maguire, 1982; Higashijima et al, 1987). A further  $Mg^{2+}$  site on the catalytic subunit is responsible for Mg<sup>2+</sup> activation of catalytic activity (Maguire, 1984). In most cells the free  $Mg^{2+}$  concentration is insufficient to promote this activity, however the mechanism of hormonal activation of adenylate cyclase involves a marked increase in enzyme affinity for free  $Mg^{2+}$  and allows a higher rate of cAMP synthesis (Cech et al, 1980; Iyengar & Birnbaumer, 1982).

It thus appears that free  $Mg^{2+}$  has the ability to perform important roles in the regulation of cell function following ligand receptor interaction (Grubbs & Maguire, 1987). Indeed there are those who consider that the intracellular regulation of magnesium ions may represent a plausible mechanism for the control of events leading to the initiation of DNA synthesis and cell division (Rubin, 1975, 1977; Rubin et al, 1979; Sanui & Rubin, 1982b). Rubin (1975, 1977) proposed that this cation can regulate the 'co-ordinate' response reactions which

follow growth stimulation in animal cells. The hypothesis is that small changes in intracellular  $Mg^{2+}$  concentration alter the kinetics of many enzymes which in turn determine the rate of cell proliferation. Evidence of an important role for Mg<sup>2+</sup> in cell proliferation is illustrated by the dependence of various cells on the availability of extracellular magnesium (Wyatt, 1961; Kamine & Rubin, 1976; McKeehan & Ham, 1978; Rubin & Chu, 1978; Abboud et al, 1985). A number of studies have correlated intracellular magnesium levels with rates of cell proliferation. Thus, it appears that elevated intracellular magnesium concentrations are needed to maintain rapid rates of normal cell division (Cameron et al, 1980). Rubin (1975) has demonstrated that intracellular magnesium rather than calcium is closely linked to growth of fibroblasts and believes that magnesium may be the major factor in governing proliferation rate in these cells. More recently studies with lymphocytes show that cells in different phases of growth contain varying amounts of cell magnesium such that low concentrations of Mg<sup>2+</sup> correspond with low proliferative activity (Hosseini & Elin, 1985). It is believed by Rubin and co-workers that the effects of calcium on cell proliferation may be indirect, mediated by altering Mg<sup>2+</sup> availability within the cell (Rubin, 1975; Rubin & Doide, 1976). An alternative view is that the observed effects of magnesium on cell proliferation may be due to alterations in calcium homeostasis (Levine & Coburn, 1984). Regulation of intracellular Ca<sup>2+</sup> by the outwardly directed plasma membrane pump, mitochondrial and ER sequestration are all subject to Mg<sup>2+</sup> alterations (Nicholls, 1978; Black et al, 1980; Pershadsingh et al, 1980).

Although the relative importance of  $Ca^{2+}$  and  $Mg^{2+}$  in the regulation of cell proliferation is the subject of much controversy (Berridge, 1975; Rubin, 1976; Sanui & Rubin, 1982b), there is evidence that in some cells at least, both of these cations are important. Calcium is believed to be the trigger for late  $G_1$  progression with more of a permissive role for magnesium throughout the cell cycle

(Whitfield, 1982).

Human diploid 'fibroblast' like cells derived from fetal lung have an absolute requrement for both  $Ca^{2+}$  and  $Mg^{2+}$  even when the other is present in saturating amounts. Initiation of DNA synthesis and mitosis is mediated by a  $Ca^{2+}$  dependent process followed by a magnesium dependent process (McKeehan & Ham, 1978). NGF stimulated neurite outgrowth in phaeochromocytoma PC12 cells is also regulated by both  $Ca^{2+}$  and  $Mg^{2+}$  (Koika, 1983). Both cations are also believed to have regulatory roles in lectin stimulated DNA synthesis in human lymphocytes (Abboud et al, 1985).

### 1.5.4 Monovalent Ions

A significant literature has developed linking fluxes not only of divalent cations, but also of monovalent ions with initiation or maintenance of cells in a state of active replication. Changes in ion pumping activity may be an important regulatory device in linking surface and intracellular events, since the asymmetric distribution of K+and Na+ profoundly affects the transport of non-electrolytes, membrane potential and protein synthesis (Rozengurt, 1980; Lubin, 1982). Marked alterations in fluxes of monovalent cations are among the earliest changes that occur in mitogen stimulated lymphocytes (Quastel & Kaplan, 1970; Prasad et al, 1987), EGF and NGF induced proliferation in PC12 cells (Boonstra et al, 1983) and mitogen stimulated quiescent 3T3 cells (Rozengurt et al, 1983; Rozengurt & Burns, 1984). A number of reports have demonstrated correlations between the activity of Na-K-ATPase and cell growth (Rozengurt et al, 1979, 1980; Rozengurt & Mendoza, 1980; Varon & Shaper, 1983; Moore, 1983; Rozengurt, 1985; Takagi et al, 1986). Na<sup>+</sup> efflux via the ouabain-sensitive Na<sup>+</sup>/K<sup>+</sup> pump leads to K<sup>+</sup> accumulation, this increase in K<sup>+</sup> may activate cell metabolism (Lubin, 1982) and cell K+ concentrations could therefore be important

in growth control (Lopez-Rivas et al, 1982; Takagi et al, 1986).

The possibility that K<sup>+</sup>ions are important in growth control is also implicated in another way since voltage gated K<sup>+</sup> channels are thought to play a role in T lymphocyte mitogenesis. Voltage-dependent potassium channels have been found to be the predominant ion channels in human T lymphocytes and phytohaemagluttinin (PHA) (at concentrations that induce proliferation) was found to alter K<sup>+</sup> channel gating within 1 minute of addition to the bathing solution. [<sup>3</sup>H]-thymidine incorporation by T lymphocytes following PHA stimulation was inhibited by classical K<sup>+</sup> blockers (De Coursey et al, 1984). However, the K<sup>+</sup> conductance in human lymphocytes may be in part due to the presence of Ca<sup>2+</sup>activated potassium channels (Matteson & Deutch, 1984).

Increased Na<sup>+</sup> entry is one of the earliest responses to mitogenic agents in many quiescent cells including human fibroblasts (Owen & Villereal, 1983), hamster fibroblasts (Poussegar et al, 1982), rat liver cells (Koch & Leffert, 1979), neuroblastoma (Moolenaar et al, 1981), lymphocytes (Felber & Brand, 1983), BSC-1 epithelial cells (Rothenburg et al, 1982) rat phaeochromocytoma cells (Boonstra et al, 1983) and glial cells (Benos & Sapirstein, 1983).

Mitogen activated Na<sup>+</sup> influxes are amiloride sensitive and are mediated via the ubiquitous Na<sup>+</sup>/H<sup>+</sup> antiporter (Schuldiner & Rozengurt, 1982; Paris & Pousségur, 1983; Leffert & Koch, 1986; Frelin et al, 1986). It is believed that in the presence of growth factors the Na<sup>+</sup>/H<sup>+</sup> antiporter operates at more alkaline pHi values, due to increased affinity of the pHi-sensor in the allosteric H<sup>+</sup> binding site (Paris & Pousségur, 1984; Grinstein et al, 1985). Recent studies have shown that synthetic diacylglycerol and phorbol ester (activators of PKC) both activate Na<sup>+</sup>/H<sup>+</sup> exchange and it is now believed that PKC is a transducer of growth factor-mediated activation of the Na<sup>+</sup>/H<sup>+</sup> antiporter (Pousségur, 1985; Rozengurt, 1985). However since the activity of the Na<sup>+</sup>/H<sup>+</sup> antiport is also enhanced by

some mitogens that do not activate PKC it is likely that other mechanisms may be involved (Vara et al, 1985), possibly a phosphorylation step activated by  $Ca^{2+}$ . calmodulin (Villereal et al, 1985). The fact that an increased influx of Na<sup>+</sup> is an early event following ligand-receptor binding does not necessarily imply that the increase in [Na<sup>+</sup>]i is the mitogenic signal since the interactions of ion concentrations, exchanges and pumps are complicated, but certainly the increase in Na<sup>+</sup> permeability in mitogen stimulated cells sets in motion an array of ion distributions and metabolic events associated with proliferation (Leffert & Koch, 1985; Moolenaar, 1986). The stimulation of Na<sup>+</sup> influx is believed to enhance the activity of the Na<sup>+</sup>/K<sup>+</sup> pump (Lopez-Rivas et al, 1982) and thus an increase in K<sup>+</sup> concentration may be the important signal in the initiation of cell proliferation. Concurrent with an influx of Na<sup>+</sup> ions is an efflux of protons via the Na<sup>+</sup>/H<sup>+</sup> antiporter leading to a rise in intracellular pH (pHi) which can then serve as an intracellular messenger (Frelin et al, 1986). An interest in intracellular pH as a possible ionic signal for cell activation arose from studies with sea urchin eggs; a rapid rise in pHi occurred after fertilisation (Johnson et al, 1976). Increases in cytoplasmic pH following mitogenic stimulation have also been demonstrated in other cells (Moolenaar et al, 1983; Burns & Rozengurt, 1984; Grinstein & Goetz, 1985; Villereal et al, 1985).

Many mitogens cause both rapid chances in intracellular pH and  $Ca^{2+}$ , Hesketh and colleagues (1985) suggest a highly conserved sequence of ionic events take place in the mitogenic stimulation of eukaryotic cells. In both thymocytes and fibroblasts a pHi response invariably follows a  $Ca^{2+}$  signal, it is believed that the two signals are uncoupled; thus the pHi increase is not a direct consequence of  $Ca^{2+}$  signal, but rather the two signals are generated independently. Although phosphoinositode metabolism can account for these two ionic responses in many instances there are cases where other mechanisms must be responsible, for example EGF generates both signals in 3T3 fibroblasts without causing PIP<sub>2</sub> hydrolysis (Besterman et al, 1986). In some circumstances increased [Ca<sup>2+</sup>]i is sufficient to activate the Na<sup>+</sup>/H<sup>+</sup> exchanger (Owen & Villereal, 1982), the Ca<sup>2+</sup> induced activation of the antiporter in lymphocytes is thought to be mediated by changes in cell volume (Grinstein, 1987). In 3T3 fibroblasts an increased intracellular Ca<sup>2+</sup> is neither necessary nor sufficient for activation of the Na<sup>+</sup>/H<sup>+</sup> exchanger (Ives & Daniel, 1987). The precise roles of Ca<sup>2+</sup> and PKC in activation of the Na<sup>+</sup>/H<sup>+</sup> antiporter await further clarification.

### 1.5.5 Ornithine Decarboxylase and Polyamines

Polyamines and new polyamine biosynthesis have also been associated for some time with the traverse of cells through the cell cycle. Ornithine decarboxylase (ODC) is the first enzyme in the biosynthetic pathway of polyamines and is rate limiting (Pegg, 1986). Cell growth is always accompanied by high ODC activity, for reviews see (Janne et al, 1978; McCann, 1980; Heby & Anderson, 1980; Pegg & Williams-Ashman, 1981; Pegg & McCann, 1982). Polyamines are low molecular weight substances and because of their flexibility and cationic nature interact and support the synthesis of nucleic acids, proteins and phospholipids (Algranati & Goldenberg, 1977). ODC not only acts as an important regulatory enzyme in the generation of polyamines but is also believed to regulate ribosomal RNA synthesis and thus, protein synthesis (Russell et al, 1976; Russell, 1981). The induction of polyamine synthesis in lymphocyte activation, requires gene expression and occurs several hours after mitogen stimulation with peak ODC activity occurring at eighteen hours (Fidelus et al, 1984).

More recent evidence suggests that ODC may also take part in growth signal transduction. Thus, treatment of T lymphocytes, with mitogenic

monoclonal antibodies or lectins induces a rapid activation of ODC within minutes (Scott et al, 1985; Mustelin et al, 1986). This early activation is independent of de novo protein synthesis but requires an intact cytoskeleton and energy metabolism (Scott et al, 1985). This early induction of pre-existing ODC molecules appears to be specifically linked to the initiation of T lymphocyte proliferation and since non hydrolysable GTP analogues demonstrate rapid ODC activation a G protein is thought to be an important link in the transduction of the mitogenic signal (Mustelin et al, 1986). Recent work has demonstrated that ODC is covalently linked to the cytoplasmic surface of the cell membrane by phosphatidylinositol. An endogenous phospholipase C enzyme is thought to cleave the phosphotidylinositol bearing ODC into diacylglycerol and ODC linked to inositol-1-monophosphate, once cleaved, the activity of ODC rises several fold (Mustelin et al, 1987). The rapid activation of ODC in T cells also requires an elevation in the free cytosolic calcium (Scott et al, 1984). This suggests that ODC activation follows a  $Ca^{2+}$  signal in inducing T lymphocytes to divide. The exact function of the rapid ODC activation as opposed to the long term cycloheximide-sensitive ODC activation is unknown. The fact that ODC is attached to the inositol group of PI adds a new dimension to inositol phospholipid turnover and suggests the possibility that attachment and liberation of ODC from phosphatidylinositol may be an important mechanism of growth regulation (Mustelin et al, 1987).

Another intriguing facet in phosphoinositide turnover is that in A431 cells polyamines stimulate the phosphorylation of phosphatidylinositol to phosphatidylinositol-4-phosphate and might therefore have a role to play in replenishing the pool of polyphosphoinositides to be used in signal transduction (Vogel & Hoppe, 1986). Rapid ODC activation following mitogenic stimulation might serve to initiate polyamine synthesis, the polyamines then phosphorylate

phosphatidylinositol to phosphatidylinositol-4-phosphate, a further phosphorylation produces  $PIP_2$  the hydrolysis of which leads to the generation of InsP<sub>3</sub> and DG, both of which function as second messengers to initiate the signal cascade.

Another model for signal transduction which involves the polyamines suggests that these polycations serve as intracellular messengers to increase free cytosolic  $Ca^{2+}$  by stimulating  $Ca^{2+}$  influx and mobilising  $Ca^{2+}$  from intracellular sites via a cation exchange reaction. A transient increase in ODC activity and a subsequent rise in polyamine levels are early events following receptor activation by testosterone (Koenig et al, 1983a) isoproteronol (Koenig et al, 1983b) thyroid hormone (Koenig et al, 1984a) and insulin (Koenig et al, 1984b). Thus polyamine synthesis has been found to be essential for hormone induced  $Ca^{2+}$ fluxes and the mediation of several  $Ca^{2+}$ -dependent membrane transport processes including endocytosis, hexose transport and amino acid transport. ODC and polyamines may also have an important part to play in the transduction and transmission of voltage-mediated signals and in stimulus-secretion coupling in nerve cells (Iqbal & Koenig, 1985).

Unravelling the network of second messengers involved in the control of cell proliferation is a daunting task. The number and variety of putative intracellular messengers under consideration is large arguing against the likelihood of a universal second messenger. The interaction of the proposed signal pathways presents a complex puzzle, making it difficult to consider second messengers in isolation (see Diagram 2) (Bauer et al, 1987). Moreover it is by no means clear how so relatively few second messengers are able to couple so many external signals to diverse cellular responses (Lichstein & Rodbard, 1987). Nevertheless many early events have been characterised which are assumed to couple extracellular signals and mitosis. The mechanisms by which these second



messengers cause alterations in gene expression still await much clarification and the expression of the so called cell cycle dependent genes inducible by different mitogens is an intensive field of study.

### 1.5.6 Cell Cycle Dependent Genes

The search for genes (and gene products) that regulate cell division aims at elucidating genes that are preferentially expressed in a specific phase of the cell cycle and to understand their significance in the mitogenic response. A number of genes have already been identified whose expression is thought to be cell cycle-dependent although this does not necessarily imply that they <u>regulate</u> cell cycle progression.

Transcription of the nuclear proto-oncogenes, c-fos and c-myc are among the earliest events following activation of quiescent cells (Taylor, 1986; Sassone-Corsi & Verma, 1987, March 1987) and they are thought to act as molecular switches, which may trigger the transition of cells from a quiescent to proliferating state (Curran et al, 1985). These two proto-oncogenes are members of the 'competence' gene family induced by PDGF in Balb/c-3T3 cells (Callahan et al, 1985). The induction of c-fos and/or c-myc by growth factors has been demonstrated in a variety of cells, including 3T3 cells (Greenburg & Ziff, 1984; Curran et al, 1985), A431 cells (Curran et al, 1985; McCaffrey et al, 1987), thymocytes (Moore et al, 1986), T-lymphocytes (Kelly, 1983), B lymphocytes (Kelly et al, 1984) and phaeochromocytoma cells PC12 (Morgan & Curran, 1986). Because the transcripts from c-myc increase approximately 20 fold shortly after cells are stimulated to proliferate and decline before the onset of DNA synthesis (Kelly et al, 1983, 1984) it has been suggested that the expression of c-myc may be specific to the  $G_1$  phase of the cell cycle (Kelly et al, 1984). However other work indicates that the increase in c-myc mRNA following the

stimulation of quiescent cells is the result of an activational event which renders the cells competent to enter the cell cycle. The level of c-myc oncogene mRNA then remains invariant throughout the remainder of the cell cycle (Thompson et al, 1985). This is in agreement with other reports which suggest that c-myc protein synthesis is dissociated from the cell cycle in human and avian cells (Hann et al, 1985) and in *Xenopus* oocytes (Godeau et al, 1986).

Two other genes whose expression has been associated with cellular proliferation (Johnson et al, 1982; Sittman et al, 1983; Gibson et al, 1986) are thymidine kinase (TK) and histone 2b (H2b) and they do demonstrate consistent variations in the cell cycle. The mRNA levels of these two genes increase during the S phase in continuously growing cells and decrease when cell proliferation stops in density-arrested cultures (Thompson et al, 1985).

Although much attention has been focussed on the expression of two proto-oncogenes c-myc and c-fos, another proto-oncogene Harvey c-ras has also been labelled cell-cycle dependent (Goyette et al, 1984; Campisi et al, 1984). NIH 3T3 cells induced to divide by adding serum to the culture medium are unable to enter the S phase of the cell cycle after microinjection of anti-ras antibody implying that the protein product of the ras proto-oncogene is required for the initiation of the S phase in NIH 3T3 cells (Mulcahy et al, 1985).

The cellular oncogene p53 is also thought to be expressed in a cell cycle dependent manner; p53 is barely detectable in quiescent cells but is markedly increased in 3T3 cells stimulated by serum (Reich & Levine, 1984). The protein product levels of p53 increase dramatically in mouse lymphocytes stimulated with concanavalin A (Milner & Milner, 1981).

The inducibility of the four proto-oncogenes *c-fos*, *c myc*, Harvey *c-ras* and p53 by different mitogens in various cell types from different species implies that these genes are involved in cell cycle progression (Gibson et al, 1986).

### 1.6 <u>Neoplastic Growth</u>

Cells within the body are normally under strict control to ensure that proliferation only occurs when required for development or for body maintenance. According to the prevailing environmental conditions a normal cell can either be growing and proceeding through the cell cycle or be quiescent. The decision to proceed through another cycle or enter a quiescent phase  $G_0$ , takes place at the restriction point 'R' in the cycle (see Section 1.2). Growth control can be considered as a switching mechanism dependent upon external factors that act at this point 'R'. Normal cells can be transformed into malignant cells by a variety of chemical carcinogens, RNA or DNA viruses and radiation. Growth control in transformed cells is relaxed, these cells can grow under conditions that would restrict the growth of normal cells (Pardee et al, 1982).

The characteristics of transformed malignant cells are many and varied. A number of these features may be characteristic of rapidly proliferating normal cell populations and some transformed malignant cells may not exhibit all of the characteristics (Ruddon, 1981). These changes include: cytological changes (eg, increased nuclear: cytoplasmic ratio, formation of clusters and cords of cells), alterations in growth characteristics (immortality, loss of 'contact inhibition', loss of 'anchorage dependence', loss of 'restriction point' control) and a lower serum requirement. Malignant transformation of cells such as fibroblasts abolishes their requirements for specific serum growth factors such as PDGF (Currie, 1981). Indeed the loss of requirement for exogenous growth factors appears to be the paramount feature defining the malignant phenotype (Alexander & Currie, 1984). The simplest explanation for autonomous proliferation in the absence of exogenous growth factor stimulation is the autocrine secretion of transforming growth factors (TGFs) (see Section 1.3). Transformed cells cultured in serum

free medium produce and release growth factors with the ability to confer the transformed phenotype on normal fibroblasts (Keski-Oja et al, 1987). Upon removal of TGF normal cells resume normal growth patterns and the permanence of the transformed phenotype in malignant cells was therefore attributed to their capacity to produce TGF. However, since TGFs have been extracted from normal tissues (Roberts & Sporn, 1985) and authentic growth factors such as EGF and PDGF can in some circumstances confer upon normal cells many of the features associated with a transformed phenotype (Kaplan & Ozanne, 1983), the malignant phenotype does not appear to depend solely upon the synthesis of special growth factors by cancer cells. A hypothesis to explain malignant growth is that neither the growth factor nor the receptor for this factor is different from those in normal cells but it is their simultaneous presence which may distinguish malignant from normal. Thus, an essential characteristic of malignancy may be the production of a TGF and its receptor (Alexander & Currie, 1984).

Uncontrolled activation or expression of any regulatory component in the signal pathway that conveys information from the cell surface to the site of DNA replication in the nucleus has the potential to promote uncontrolled cell proliferation (see section 1.5). Thus, an alteration in cyclic nucleotide levels has been implicated in transformed malignant cells although it isn't clear whether changes in cAMP levels, or the cAMP response system are responsible for the appearance of the transformed phenotype. A number of transformed cell types have lower cAMP levels than their non transformed counterparts (Hunt & Martin, 1979; Elgebaly et al, 1982).

Likewise, an alteration in the concentration of  $Mg^{2+}$  may be responsible for many of the phenotypic traits of transformed cells according to some. A spontaneously transformed clone of Balb/c3T3 cells could be induced to resume

the characteristics of normal cells when the extracellular concentration of  $Mg^{2+}$  was sharply reduced (Rubin, 1981). It is possible that transformation is brought about at least in part by a defect in regulating the distribution of  $Mg^{2+}$  in cells, such that transformed cells have higher  $Mg^{2+}$  levels than normal cells which correlates with their higher rate of DNA synthesis (Rubin, 1982; Walker, 1986). The loss of intracellular  $Mg^{2+}$ , in low concentations of extracellular  $Mg^{2+}$  restores density-dependent inhibition of DNA synthesis, a characteristic feature of non-transformed cells (Rubin, 1982).

In contrast, others suggest that neoplasticity may be associated with a gradual loss of sensitivity to calcium ions. In some cells there is a good correlation between loss of  $Ca^{2+}$  control and progression towards tumorigenicity (Boynton & Whitfield, 1978). There are also several observations that growth of normal cells is controlled by extracellular calcium levels whilst that of tumour cells is more independant of calcium (Hickie et al, 1983). Furthermore the concentration of the calcium regulatory protein calmodulin has been reported to be higher in transformed cells than in their normal counterparts (Chafouleas et al, 1981). For example, transformed chick embryo fibroblasts have a 2-4 fold increase in calmodulin levels compared to normal chick embryo fibroblasts (Van Eldik et al, 1982). However other workers do not support the contention that transformed cells obligatorily contain elevated levels of calmodulin, although a role for calmodulin-mediated regulation may be reflected by the inability of transformed cells to down regulate this protein (Veigl et al, 1984b). Nevertheless, calmodulin remains attractive as a mediator of changes induced by transformation, since transformation results from pertubation of major cascade regulation systems many of which may be controlled by calci-calmodulin (Veigl et al, 1984a).

Calmodulin levels may increase in conjunction with the *de novo* appearance of a tumour specific calcium binding protein oncomodulin.

Oncomodulin is related to the parvalbumins and unlike calmodulin has only two binding sites per molecule, a high affinity calcium/magnesium site and a calcium specific site (Williams et al, 1987). Oncomodulin has been extracted from oncogenic and foetal tissue (MacManus et al, 1985), its function is unknown although it has been shown to mimic calmodulin in its ability to activate several enzymes (MacManus & Whitfield, 1983). The appearance of oncomodulin may be responsible for the ability of tumour cells to bypass the restriction point at the  $G_1/S$  boundary and to grow in medium deficient in calcium (Boynton et al, 1982).

Protein kinase C may also be involved in neoplastic growth since potent tumour promoters such as phorbol esters are able to directly activate this enzyme both *in vitro* and *in vivo* (Castagna et al, 1982). The evidence available suggests that many of the pleiotropic actions of tumour promoters are mediated through this protein kinase. An uncontrollable production of an active form of protein kinase C, whether the product of a cellular or a viral gene has the potential to promote carcinogenesis in the same way that tumour promoting phorbol esters do (Nishizuka, 1984b). Recent work shows that TPA increases cellular diacylglycerol (the endogenous activator of protein kinase C) through a mechanism not involving phosphoinositide hydrolysis in Swiss mouse 3T3 fibroblasts. There is a close correlation between the effect of phorbol ester on DG formation and mitotic activity in 3T3 cells, although it is not known how the enhanced DG formation contributes to the activation of protein kinase C by TPA (Takuwa et al, 1987).

Cancer genes have been discovered in the chromosomes of tumour cells; these so called oncogenes represent the driving force behind the uncontrollable growth of many cancer cells and they become activated in the conversion of a normal cell into a cancer cell. Oncogenes isolated from tumours are highly homologous to benign genes in normal cells, the proto-oncogenes (Weinberg,

1983). The proto-oncogenes were discovered initially because of their great resemblance to certain retroviral genes responsible for the malignant transformation of their host cells. These oncogenic retroviral genes are called viral oncogenes (v-onc). Retroviruses carry a single gene which is responsible for the induction of cancer and they are structurally altered versions of the proto-oncogenes (c-onc) from which they were initially derived (Weinberg, 1983). Expression of viral oncogenes in cells is thought to cause cancer by mimicking the actions of the corresponding proto-oncogenes in an abnormal fashion (Alitalo et al, 1987). Since chemical carcinogens in contrast to retroviruses cannot introduce new genetic information into target cells they must distort the function of proto-oncogenes. There are three principal ways leading to oncogenic activation of proto-oncogenes, point mutations, chromosomal rearrangements and gene amplication (Spandidos, 1986; Nishimura & Sekiya, 1987; Alitalo et al, 1987).

Growth factors have been shown to increase the transcription of certain proto-oncogenes, eg. (c-myc, c-fos) (see section 1.5.6), the products of which in turn regulate the transcription of other genes necessary for stimulation of cell proliferation. The emerging paradigm is that a set of proto-oncogenes may encode for the various components of the growth factor-receptor-response pathway, and indicate points at which alterations may arise inducing cancer (Berridge, 1984a; Macara, 1985; Goustin et al, 1986; Alitalo et al, 1987; Nishimura & Sekiya, 1987). Thus, each of the sequential steps following binding of growth factor or other mitogen to their receptors may represent a site where oncogenes might mediate their effects (see diagram 3). The viral oncogene of Simian sarcoma retrovirus, v-sis codes for a protein that is almost identical to PDGF suggesting that cells carrying a genetic alteration of the c-sis gene may produce excessive amounts of PDGF leading to transformation through autocrine stimulation



Diagram 3: A diagrammatic representation of the proposed mechanism for regulation of cell proliferation by proto-oncogenes.

Abbreviations are: PDGF, platelet-derived growth factor; EGF, epidermal growth factor; G, G-protein; PLC, phospholipase C; PI, phosphatidylinositol; PIP<sub>2</sub>, phosphatidylinositol-4,5-bisphosphate; InsP<sub>3</sub>, inositol-1,4,5-triphosphate; DG, diacylglycerol; PKC, protein kinase C; Ca<sup>2+</sup> CAM, calci calmodulin; identified oncogenes are shown in brackets; (-), negative regulation; ? unknown routes to gene expression. See text for further details.

(Doolittle et al, 1983; Waterfield et al, 1983).

Certain genetic defects may result in overproduction of receptors for growth factors, indeed recent studies have shown that *erb*-B oncogene of avian erythroblastosis virus encodes a protein that is remarkedly similar to the internal domain and transmembrane part of the EGF receptor. This truncated receptor may be capable of initiating the signal pathway independently of EGF (Downward et al, 1984).

The p21 protein encoded by the c-H-*ras* gene has been demonstrated to have GTP-binding activity and GTPase activity and is related to the G proteins that regulate adenylate cyclase activity (Nishimura & Sekiya, 1987). The viral p21 Ki-*ras* protein appears to stimulate adenylate cyclase activity in rat kidney cells and it is possible that this ability results in oncogenesis (Franks et al, 1987).

The existence of a functional relationship between growth factor receptors and oncogene products is further supported by the findings that certain oncogene products have transmembrane and tyrosine kinase domains like certain PGF receptors (eg. PDGF, EGF and insulin receptors). Thus the products of a family of oncogenes *src*, *yes*, *abl*, *fps*, *fes*, *ros* and *fgr* possess tyrosine kinase activity and are associated with the cell membrane (Hunter, 1984). Since their discovery, a number of cellular substrates for viral tyrosine kinases have been identified although their role in cell proliferation remains hypothetical (Marcara, 1985). An alternative mechanism of action for these oncogenic products is that they control PI metabolism. A phosphoinositide kinase activity is associated with two retroviral gene products, pp60 v-*src* (Sugimoto et al, 1984) and pp68 v-*ros* (Macara et al, 1984) and transformation of chick embryo fibroblasts by these retroviruses is associated with increased phosphoinositide turnover *in vivo*. Regarding v-*ros*, the presence of the oncogene product results in increases in the second messengers resulting from phosphoinositide turnover, IP<sub>3</sub> and DG (Macara et al, 1985). The evidence for an intimate link between proto-oncogenes and PI turnover is increasing (Whitman et al, 1986) and models have been proposed demonstrating these links, one of which incorporates the idea of 'competence and progression' factors (see Section 1.3) (Berridge, 1984a; Macara, 1985).

The conversion of a normal cell to a malignant one is thought by some to be a process involving transformation by at least two oncogenes (Lacal et al, 1986). Oncogenes can be divided into two groups, those functioning in the nuclei and those functioning in the cytoplasm (including the cell membrane) (Weinberg, 1985). Activation of an oncogene from each group may be necessary for carcinogenesis (Nishimura & Sekiya, 1987).

In conclusion then, in normal cells proliferation is under the strict control of a set of proto-oncogenes. Certain environmental factors (eg. chemical carcinogens, radiation) may cause an abnormal expression of these genes (ie. point mutations, chromosone alterations and gene amplification), the resulting interference with the normal flow of information along the signal transduction pathway manifests itself in uncontrolled proliferation - cancer. In addition, viral oncogenesis may result from infection with a retrovirus, activation of the v-onc leads to neoplastic growth. Other viruses may cause abnormal growth due to the insertion of viral DNA within the host cells genome leading to mutation and abnormal gene expression.

## 1.7 The Control of Splenic Lymphocyte Proliferation

The purpose of this study is to investigate the mechanism(s) of T lymphocyte proliferation. It will attempt to elucidate the role of putative second messengers and any interaction amongst them following stimulation with mitogens. The ability of sex steroids to modulate lymphocyte mitogenesis will

also be examined. The experimental system employed has been that of cultured lymphocytes isolated from the mouse spleen. The main antigen recognising cells of the immune system are the T and B lymphocytes which operate in conjunction with accessory cells derived from the myeloid lineage such as macrophages. In very broad general terms T lymphocytes mediate cellular immunity and B lymphocytes become antibody forming cells responsible for humoral immunity. The spleen is a secondary lymphoid organ containing a mixture of T and B lymphocytes in approximately equal proportions (Sprent, 1977). The two components of splenic parenchyma are red pulp, consisting of splenic cords lined with macrophages and venous sinusoids and white pulp containing lymphoid follicles. T- and B-cell areas are segregated in these follicles such that T lymphocytes enclose a follicular arteriole whilst B lymphocytes surround them. The lymphoid tissue of the spleen has several functions; it traps blood borne antigen in the macrophage lined cords and this trapped antigen is then transported to the lymphoid interior by phagocytic migration, T cells exported from the thymus collect here and the lymphoid tissue provides both cellular and humoral immune responses to trapped antigens.

Immunity to those infectious organisms which have developed the capacity for living and multiplying within the cells of the host, eg. viral infections, is the responsibility of T lymphocytes. T cells may be long lived and persist for many years without dividing, however in response to an antigen which binds to a specific receptor on the T lymphocyte (the T cell receptor) the lymphocyte enlarges and divides rapidly. The T cell responds to an antigen by clonally dividing and differentiating into one of several kinds of T cell specific to the antigen. Thus, cytotoxic T lymphocytes bind to antigen displayed on the surfaces of the infected cell and kill the cell. Suppressor T lymphocytes act to inhibit the immune response to an antigen some time after the response has been set in motion.

Helper T lymphocytes bind to antigen on the surface of a B cell that has already bound to the antigen. The helper T cell then releases lymphokines which enable B lymphocytes to divide and differentiate (Swain & Dutton, 1983; Reinherz & Schlossman, 1983). Each functional lymphocyte class bears a distribution of T cell specific surface antigens characteristic of that class; thus mouse cells bear various Ly (Lyt) antigens (Plaut, 1987).

The recognition of antigen by T cell receptors involves the phenomenon of major histocompatibility (MHC) restriction. Thus, helper cells proliferate in response to antigen only when the latter is presented on the surface of macrophages in conjunction with Class II products of the major histocompatibility complex (Rosenthal & Shevach, 1973). Cytotoxic T cells will likewise only lyse targets bearing allogeneic Class I MHC antigens (Zinkernagel & Doherty, 1977). It is felt that the MHC cell surface glycoproteins act as a guide to ensure that the correct T cell can interact with the correct antigen or target (Klein & Nagy, 1982). The T cell receptor is believed to be a molecular complex comprising at least five

polypeptide chains, two of these chains are highly variable disulphide linked  $T_1 \alpha$ and  $\beta$  subunits which together mediate antigen recognition. They are non-covalently associated with three CD3 subunits which are believed to mediate signal transduction when the T cells are activated by receptor binding (Mener et al, 1986; Reinherz, 1987). The mechanism by which resting T cells are activated in an antigen-specific response may be summarised thus; the T cell receptor recognises the antigen/MHC encoded protein complex on the antigen presenting macrophage, this bonding stimulates the release of IL1 by the macrophage and that in turn activates the T cell, which expresses IL2 receptors, the activated blasts undergo proliferative expansion under the influence of IL2 produced by another activated T cell subpopulation (Zanders, 1985; MacDonald & Nabholz, 1986; Meuer et al, 1986). Preceding activation, T cells are in  $G_0$ , the quiescent phase of the cell cycle, the first activation signal is provided by the interaction between the antigen specific T receptor and antigen associated with MHC proteins, IL1 also plays a role in this event. As a result of this signal, IL2 receptors appear. Transition of T cells from  $G_1$  and progression into the mitotic cycle then occurs as a result of the second signal, the production of IL2 and its binding to high affinity receptors (Klaus & Hawrylowicz, 1984; Isakov et al, 1987).

The proliferation of lymphocytes in response to plant lectins is often taken as a model for the proliferative aspect of immune responses (Paetkau et al, 1976). The advantage of using plant lectins in a model system to examine signal transduction and intracellular events in T-lymphocyte activation is that they stimulate a very large proportion of T lymphocytes to divide; hence they are described as "polyclonal activators". Concanavalin A (Con A) is the best characterised of the plant lectins and was isolated many years ago (Sumner, 1919). Con A binds specifically to α-D-glucopyranosides and α-D-mannopyranosides and to polysaccharides or glycoproteins containing such residues (Sumner & Howell, 1936). These properties of Con A were known for a long time before it was first reported to be mitogenic (Powell & Leon, 1970). The binding of Con A to sugar molecules on cell surface receptors is an irreversible phenomenon (Powell & Leon, 1970) and although binding occurs on both T and B lymphocytes, Con A preferentially stimulates T cells to divide (Hickel et al, 1985). Con A has been chosen in this study in preference to phytohaemagglutinin (PHA) another T cell mitogen since it is more potent than PHA and causes less agglutination and cell damage (Stobo et al, 1972).

Although the interaction of ligands with T-cell surface receptors and the subsequent division of T cells has been widely studied, the biochemical events

occurring between these two extreme points still await much clarification. Thus, it is unknown whether the signal transduction pathways set in motion by various exogenous stimuli (antigens, lectins or growth factors) act via distinct cell surface receptors or whether T-cell subsets, each with their own genetically programmed functions have differing signal requirements to activate these functions (Isakov et al, 1986). Several signal transduction pathways have been suggested to play a part in T cell activation (for reviews see Zanders, 1985; Gelfand et al, 1987; Linch et al, 1987). In what follows the early signals induced by antigen, lectin and interleukin leading to T cell activation will be described.

Calcium ions have long been implicated in T lymphocyte activation. Initial evidence came from the demonstration that  $Ca^{2+}$  ions were necessary in the external medium for T cells to proliferate (Whitney & Sutherland, 1972). Further support came from the observation that an increase in  ${}^{45}Ca^{2+}$  uptake can be triggered by T cell mitogens (Freedman et al, 1975; Whitney & Sutherland, 1983) and that in certain instances the calcium ionophore A23187 is mitogenic (Luckasen et al, 1974). Following the advent of a series of fluorescent dyes (Tsien et al, 1982; Grynkiewicz et al, 1985), increases in cytosolic calcium [Ca<sup>2+</sup>]i have been monitored following the interaction of T lymphocytes with lectins (Tsien et al, 1982; Hesketh et al, 1985; Gelfand et al, 1984); specific antigen (Nisbet-Brown et al, 1985), or monoclonal antibodies that react with certain T cell surface proteins (O'Flynn et al, 1984; Imboden & Stobo, 1985; Oettgen et al, 1985). IL2 secretion but not IL2 receptor expression appears to require an increase in [Ca<sup>2+</sup>]i (Mills et al, 1985a; Gelfand et al, 1986a). However, the role of  $[Ca^{2+}]i$  in the interaction of IL2 with its high affinity receptor, the second signal in T cell activation is controversial. Changes in [Ca<sup>2+</sup>]i following IL2 binding have not been detected in human or murine T lymphocytes or IL2 dependent cell lines by some workers (Mills et al, 1985b; Le Grue, 1987) whereas others have detected a rise in [Ca<sup>2+</sup>]i

in rat thymocytes and IL2 dependent cell lines stimulated with IL2 (Gearing et al, 1985a; Rossio et al, 1986). The other interleukin involved in T cell activation IL1, does not appear to use an increase in  $[Ca^{2+}]i$  in its transmembrane signalling (Abraham et al, 1987).

Evidence favouring a role for  $Mg^{2+}$  ions in T lymphocyte proliferation is sparse, although lectin induced DNA synthesis in T cells is inhibited by about 30% in  $Mg^{2+}$  free media, suggesting that  $Mg^{2+}$  ions in addition to  $Ca^{2+}$  ions are important in T cell proliferation (Abboud et al, 1985). Magnesium is also thought to mediate the response of rat thymic lymphocytes to high concentrations of various mitogens including adrenaline, isoprenaline, dopamine, glucagon and acetylcholine (Perris & Morgan, 1976; Morgan et al, 1984). Further evidence for a role for magnesium in T cell proliferation is the diminished immune response observed in  $Mg^{2+}$  deficiency (Gaudin-Harding, 1981). Con A stimulation of DNA synthesis in thymocytes from Mg deficient rats is reduced by 40% (Gunther & Averdunk, 1979), however the diminished immune response in  $Mg^{2+}$ deficiency is not caused by a defective Con A induced  $Ca^{2+}$  signal (Vormann & Gunther, 1987) inferring that other biochemical mechanisms must be responsible for the reduction of humoral and cellular immunity and in addition, that  $Mg^{2+}$ plays an important part in Con A stimulated proliferation.

Monovalent ion fluxes have also been linked with T lymphocyte proliferation. Thus, elevated rates of Na<sup>+</sup> influx have been reported in lectin stimulated T cells (Averdunk, 1976). Potassium channels are thought to open more rapidly after lectin stimulation and because K<sup>+</sup> channel blockers (eg. quinine) inhibit mitogenesis in activated T cells, K<sup>+</sup> ions may have a role to play in signal transduction (Matteson & Deutsch, 1984). However since several investigations have implied that these K<sup>+</sup> channels may be calcium dependent, it may be that Ca<sup>2+</sup> mobilisation is the important primary event, the opening of monovalent cation channels and membrane potential changes then results in secondary processes ultimately leading to cell division (Zanders, 1985).

Cyclic nucleotides are established as mediators of many rapid activation processes in different cells and have been implicated in the initiation of lymphocyte activation. However, as indicated previously (see Section 1.5.1) there are considerable contradictions in the data so that it is by no means clear whether cAMP or cGMP signals are stimulatory or inhibitory or whether they are indeed generated at all by T cell mitogens. More recent work presents evidence that cGMP does not play a role in the mitogenic activation of human peripheral lymphocytes and rabbit thymocytes stimulated with Con A whilst cAMP may be involved as an inhibitory signal (Kaever & Resch, 1985).

Many of the aforementioned putative second messengers which are thought to be implicated in T lymphocyte proliferation may be linked to the hydrolysis of membrane phospholipids following ligand/receptor interaction (see Section 1.5.2). The role of phosphatidylinositol breakdown in T lymphocyte proliferation has been under intense examination. Thus, stimulation of T cells with antibodies to the CD3/T-cell receptor complex causes turnover of phosphatidylinositol to form InsP3 and DG, this signal transduction pathway which leads to calcium mobilisation and the activation of PKC can lead to T cell activation when other required signals are provided by accessory cells and Interleukin 1 (Imboden & Stobo, 1985; Imboden et al, 1985; Cockcroft et al, 1985; Ledbetter et al, 1987). Several investigators have used T lymphocytes treated with mitogenic lectins as models for cellular proliferation, in each case formation of InsP3 and DG from PIP2 hydrolysis was observed (Fisher & Mueller, 1968; Allan & Michell, 1977; Hui & Harmony, 1980; Moore et al, 1984; Metcalfe et al, 1985). Diacylglycerol activates PKC by increasing its affinity for Ca<sup>2+</sup>. Following activation, PKC translocates from the cytosol to the membrane

and has the potential to catalyse the phosphorylation of numerous protein substrates at serine and threonine residues by using ATP as a phosphate donor. One of the substrates for PKC may be the Na<sup>+</sup>/H<sup>+</sup> exchanger, the activation of which results in an increase in intracellular pH (pHi), an event that accompanies and seems to be causally related to cell proliferation (see Section 1.5.4). Early measurements of pHi using fluorescent probes failed to detect a significant alkalinization in lectin-stimulated T cells (Rink et al, 1982; Rogers et al, 1983). However more recent measurements have detected an increase in pHi in Con A-stimulated thymocytes (Hesketh et al, 1985). In mouse splenic lymphocytes a cellular alkalinization has also been detected following mitogenic stimulation with Con A (Gerson et al, 1982). Further evidence for the role of PIP<sub>2</sub> hydrolysis and the resulting generation of a dual signal pathway in T cell activation comes from studies using tumour promoters and calcium ionophores which mimic the effects of the physiological signals InsP3 and DG respectively (Isakov et al, 1986; Gelfand et al, 1987; Isakov et al, 1987). In addition, the enzyme responsible for PIP<sub>2</sub> hydrolysis, phospholipase C has been isolated in the plasma membrane fractions from murine splenocytes (Kamisaka et al, 1986) inferring that PIP2 hydrolysis is indeed a possibility in T cell activation. Attempts to correlate IL2 binding to its high affinity receptor with phosphatidylinositol turnover have however given ambiguous results. Thus, although an IL2 induced cytoplasmic alkalinization through stimulation of the Na<sup>+</sup>/H<sup>+</sup> antiport has been demonstrated (Mills et al, 1985a), no changes could be detected in labelled inositol phosphates (Mills et al, 1986) or in the intracellular distribution of PKC (Gelfand et al, 1987). In contrast other workers believe that the IL2 signal does involve the activation of inositol phospholipid turnover (Bonvini et al, 1986; Farrar et al, 1986), Ca<sup>2+</sup> mobilisation (Rossio et al, 1986) and PKC translocation (Farrar & Anderson,

1985) which is inferred to be the result of DG generation/activation. Hence much confusion persists as to the nature of the secondary mediators in IL2 signalling. The mechanism of IL1 action is even more nebulous but does not appear to be associated with PIP<sub>2</sub> hydrolysis (Abraham et al, 1987).

The early induction of ornithine decarboxylase activity is also thought to be linked to the initiation of human T lymphocyte proliferation. The T cell mitogen, Con A and monoclonal antibody OKT3 cause a rapid activation of ornithine decarboxylase (ODC) (Scott et al, 1985). When T cells are selectively depleted of guanine nucleotides by treatment with mycophenolic acid, the early mitogen-induced activation of ODC is completely inhibited suggesting that a G-protein(s) is involved in the transduction of the proliferation signal in human T cells (Mustelin, 1987). The intracellular target system for this G protein may include phosphoinositide breakdown and ODC activation (Mustelin et al, 1987) (see Section 1.5.5).

Thus it is apparent that controversies exist in the delineation of intracellular signals involved in T lymphocyte proliferation. Therefore an important feature of the current study has been to attempt to clarify certain facets of stimulus-mitosis coupling in murine T lymphocytes, aiming towards a better understanding of this complex phenomenon. Another aim of the present work has been to gain an insight into the mechanism(s) of sex steroid modulation of the immune system.

There is a great deal of evidence to suggest that the endocrine system has the capacity to directly influence the development and function of the immune system (for a recent review see Kiess & Belohrodsky, 1986). Of the numerous hormones known to regulate the immune response, gonadal steroids have long been implicated (Dougherty, 1952). Observations from these and more recent studies demonstrate that gonadectomy increases the size of the thymus

(Glucksmann & Cherry, 1968; Castro, 1974) whilst sex steroid administration causes involution (Slijivic & Warr, 1973). Most studies suggest that the thymus is an important organ during development but that once sexual maturity is reached, involution occurs (Simpson et al, 1975; Tosi et al, 1982). Endogenous oestrogens or androgens may be causal factors in this naturally occurring process although some workers have shown that ageing of grafted thymi in young mice is independent of the systemic hormonal conditions of the host (Metcalf, 1966). More recent studies suggest that sex steroids are indeed implicated in thymic involution, since the thymus of 12-15 month old rats which is grossly atrophied can be markedly restored in size, 30 days following orchidectomy (Fitzpatrick et al, 1985). Moreover the regeneration of the thymus in old rats can be inhibited by testosterone administration (Greenstein et al, 1986) and oestradiol (Fitzpatrick & Greenstein, 1987). In contrast the age-related involution of the human thymus is not considered by some, to be related to endocrinological changes taking place at puberty (Steinmann et al, 1986). Intriguingly there are still others who believe that age related thymic involution is a myth (Simmons, 1987). Thus there is some controversy over the roles of sex steroids in thymic involution, and it is likely that it is a more complex phenomenon than originally imagined. Indeed there are many possible mechanisms which may lead to thymic involution; cells may die intrathymically, fail to arrive from the bone marrow, may be exported at a faster rate, develop longer cell cycles or simply cease to divide (Clarke & MacLennan, 1986).

Despite these ambiguities surrounding the effects of sex steroids on thymic involution, there is a large amount of clinical and experimental evidence supporting the contention that gonadal steroids interact with the immune system at both physiological and pharmacological concentrations. This is based on several observations; thus sexual dimorphism exists in the immune response, the immune

response is altered during pregnancy and the organs responsible for the immune response contain specific receptors for gonadal steroids (Grossman, 1985).

Many studies have suggested that the female of various mammalian species out performs the male when measured in terms of immune responsiveness (Eidinger & Garrett, 1972). The mechanisms underlying this sex difference are imperfectly understood. Two possibilities include a direct control exerted by X or Y chromosomal genes and/or an indirect modulatory influence of sex steroids. Indeed the majority of studies have shown that immune responses can be altered by sex hormones (Ahmed et al, 1985). Immunoglobulin production is greater in females than in males, particularly IgM (Myers & Petersen, 1985). A possible mechanism for this stimulation of antibody production proposes that oestradiol inhibits suppressor T cell activity. Since suppressor T cells prevent B cells from producing antibody it follows that inhibition of suppressor T cell function will enhance B-cell maturation and increase antibody production (Paavonen et al, 1981). Another report suggests however that the increase in antibody titres is the result of a direct action by oestradiol on B cells to specifically increase the synthesis of the IgM class of antibodies (Myers & Petersen, 1985). The supposed sex difference in cell-mediated immunity is surrounded by controversy. In some studies females are reported to have a greater capability for cellular immunity than males, (Eidinger & Garrett, 1972). Thus, females reject tumours and homografts with greater efficiency than males (Graff et al, 1969) and cells derived from females perform better in mixed lymphocyte cultures than cells derived from males (Lieber et al, 1969). Other reports suggest that females have decresed cellular immune responses as compared to males due in part to the presence of oestradiol (Huber et al, 1982; Myers et al, 1986), which prolongs graft survival time (Waltman et al, 1971), reduces delayed type hypersensitivity responses (Kalland & Fosberg, 1978) and decreases natural killer activity (Kalland, 1980).

A differential sex susceptibility to various autoimmune diseases has long been recognised. Although the aetiology of these disorders is not well understood, a large body of evidence indicates that the progression of autoimmune diseases may be influenced by steroid hormones (Lahita, 1985). For example, 90% of patients suffering systemic lupus erythematosus are female (DuBois & Tuffanelli, 1964). A similar sex related expression of disease is known to occur in animal models of autoimmune diseases (Roubinian et al, 1977; Ansar Ahmed et al, 1985). Even though there are ambiguities in the reported sexual dimorphism of the immune response many studies reveal that pharmacological and physiological levels of oestrogens do modulate immune functions. Thus, immunological effects associated with administration of pharmacological doses of natural and synthetic oestrogen include thymic atrophy (Dougherty, 1952), myelotoxicity (Boorman et al, 1980), stimulation of the reticulo-endothelial system (Nicol et al, 1964) suppression of cell-mediated immunity (Myers et al, 1986) and depression of natural killer cell activity (Kalland, 1980). Physiological levels of oestrogen provide some degree of immune regulation as clearly evidenced by the association of oestrogens with the pathogenesis of autoimmunity (Roubinian et al, 1978). There is also a correlation between increased serum oestradiol levels during pregnancy and depression of cell-mediated immunity and lymphopenia (Hamilton & Hellstrom, 1977; Mathur et al, 1979).

Testosterone has been implicated in the weaker immune responses of males compared to females (Cohn, 1979a). However the precise effects of androgens on lymphomyeloid function remain poorly understood due in part to conflicting observations. For example, in healthy animals androgen treatment decreases antibody responses to dinitrophenol (Oettgen et al, 1966) but has no effect on immune responses to sarcoma cells (Batchelor, 1968), on graft rejection or on graft versus host responses (Fujii et al, 1975). Similarly, administration of

testosterone suppresses (Fujii et al, 1975) enhances (Healy & Neeman, 1967) or has no influence (Frey-Wettstein & Craddock, 1970) on antibody responses to sheep red blood cells. Testosterone also has inconsistent effects on autoimmune diseases, ie.testosterone inhibits development of autoallergic thyroiditis (Kappas et al, 1963) and retards murine lupus in the NZB/W (Shear et al, 1981) but not the NZB strain (Steinberg et al, 1981). Effects of androgens on bone marrow such as erythropoiesis, granulopoiesis and stem cell division are generally accepted (Mooradian et al, 1987). The relationship between androgens and the immune response has been studied in healthy intact mice which demonstrate high or low androgen responsiveness according to a seminal vesicle bioassay. In these mice genetically controlled high sensitivity to androgen correlates with a low immune performance. High androgen responder (HAR) strains demonstrate lower polyclonal, proliferative responses to lipopolysaccharide suggesting a fundamental B cell weakness, alternatively T cell suppression could play a role in the low immune response. The proliferative responses of HAR males to Con A indicate that the mitogenic capacity of their T cells is not reduced (Cohn, 1979a & b, 1986). Orchidectomy induces thymic hypertrophy and increases in the mass of peripheral lymph nodes and spleen (Castro, 1974) whilst administration of androgens leads to thymic atrophy (Dougherty, 1952) and inhibition of growth and development in the bursa of Fabricius in chickens (Norton & Wira, 1977; Verheul et al, 1986). In many early studies, high doses of steroids were used such that confusion exists between the effects of steroid replacement in gonadectomized animals and the pharmacological effects on castrated or intact animals. In a more recent study a dose of dihydrotestosterone, which maximally suppresses the thymic hypertrophy which follows castration does not alter thymic weight in intact animals. This suggests a potential physiological role rather than a toxic effect (Pearce et al, 1981). Similarly, physiological concentrations of

testosterone inhibit the regeneration of atrophied thymus in orchidectomised mice (Fitzpatrick et al, 1985).

Both oestrogens and androgens have been reported to be both immunoinhibitory and immunostimulatory. However, despite all the confusions alluded to earlier, most agree sex steroids act mainly to suppress the cell-mediated immune system. Oestrogens and androgens may suppress either the same or different T lymphocyte subpopulations. Oestrogens can also act as stimulators of the humoral immune system, this may be a direct action on the B lymphocytes or regulated through the cell-mediated immune system by stimulation of helper and inhibition of suppressor T lymphocytes. Androgens may also stimulate the humoral immune response since they can be converted to oestrogens in vivo. The ratio of oestrogen to androgen may determine whether the circulating hormones will be immunostimulatory or immunoinhibitory (Grossman, 1985). Not only can the sex steroids regulate the immune system, but it is apparent that the circulating levels of these steroids may be modulated by the immune system. A hypothetical scheme to account for the interaction of the gonadal steroids on T lymphocyte function via a bidirectional hypothalamic-pituitary-gonadal-thymic axis (HPG-thymic axis) has been proposed (Grossman, 1985).

Evidence has been presented which suggests that the function of the cellular immune system is depressed by sex steroids. The mechanism of action of sex steroids on immune responses *in vitro* has been poorly investigated. The objective of the present study is to determine the mode of action by which sex steroids modulate the cell mediated immune system, specifically through an effect on lectin induced T lymphocyte proliferation (an experimental model considered to represent the proliferative aspect of the immune response). Several reports demonstrate that sex steroids do indeed have effects on *in vitro* human lymphocyte transformation, but this steroid induced inhibition of proliferation occurs at

unphysiological concentrations (Schiff et al, 1975; Mendelsohn et al, 1977; Niefield et al, 1977; Wyle & Kent, 1977). In a further study using mouse thymus and spleen cell populations, sex steroid hormones were shown to inhibit lymphoblast transformation but only at pharmocological concentrations. Since sex steroid receptors are believed to be located in the epithelial reticulum of the lymphoid organ rather than in lymphocytes themselves, (Grossman et al, 1979a,b) it has been proposed that the inhibitory effect of high concentrations of sex steroids *in vitro* is a non-specific, non-receptor-mediated event (Homo et al, 1980). An objective of this study is to discern the mechanism(s) by which sex steroids inhibit lymphocyte transformation and to elucidate whether it has any relevance to the modulation of the immune system *in vivo* by gonadal hormones. Additionally the effects of orchidectomy and ovariectomy on *in vitro* T cell function will be investigated since gonadectomy, in addition to causing hypertrophy of lymphoid organs (Dougherty, 1952; Castro, 1974) also results in improved *in vivo* immune function (Graff et al, 1969; Castro & Hamilton, 1972).

#### CHAPTER TWO

# MATERIALS AND METHODS

The experimental techniques employed in this study have been chosen to investigate the effect and mechanism of action of several mitogenic stimuli with a view to gaining a better understanding of T cell proliferation. In addition, attempts have been made to elucidate the nature of the sex steroid modulation of the proliferative aspect of the cellular immune response. The interrelationships between mitogens, sex steroids, ionic environment and mitotic activity in the mouse splenic T lymphocyte have been specifically studied. Thus, methods for the estimation of DNA synthesis, determination of intracellular calcium ion concentration and intracellular pH using fluorescent dyes and measurement of  ${}^{45}Ca^{2+}$  fluxes are described in this section.

### 2.1 Experimental Animals

Postpubertal male albino mice of the TO strain (Bantin and Kingman Ltd) weighing approximately 25g have been used unless otherwise specified to avoid unnecessary fluctuations resulting from age-related changes in lymphoid function (Walters & Claman, 1975) and oestrous cycle endocrine periodicity (Smith et al, 1975). The animals were maintained under constant laboratory conditions, which included a standard Heygate's rat and mouse breeding diet (Heygate and Sons Ltd) and tap water *ad libitum*.

The animals were sacrificed by cervical dislocation after being lightly anaethetised with diethyl ether. This method is preferential to the use of barbiturates since they have an antimitotic action (Baserga & Weiss, 1967) whereas ether has no short term effects on total red and white blood cell levels or plasma corticosteroid concentration (Besch & Chou, 1971).

## 2.2 Lymphocyte Proliferation

The small lymphocyte can be forced out of quiescence into an active state by a variety of stimuli both *in vivo* and *in vitro* with consequent synthesis of RNA, protein, and DNA and eventual cell division.

Concanavalin A (Con A) a potent T cell mitogen is capable of binding to the lymphocyte membrane and stimulating a sizeable proportion of T lymphocytes to divide (Larsson & Coutinho, 1979). Hence, it offers a simple *in vitro* system for studying the mechanism of induction of cell cycling in normally quiescent lymphocytes and as such may help elucidate apparently similar events which occur in the cellular immune response.

### 2.2.1 Preparation of Splenocyte Cultures

The cell suspensions were prepared using a modification of the techniques described by Ford (1978). Spleens were aseptically removed and rinsed in two changes of culture medium to remove superficial debris. Splenic lymphocytes were released into sterile culture medium by gently teasing the spleen avoiding breaking it up into fragments. Large aggregates of cells were subsequently removed by sedimentation and the cell suspension thus remaining centrifuged at 200g for 10 minutes. The cell pellet was resuspended in fresh culture medium and the cell viability estimated by trypan blue exclusion (see Appendix A) was consistently greater than 95%. The cell concentration was adjusted to  $5 \times 10^6$  cells ml<sup>-1</sup> and 0.5 ml aliquots were dispensed into sterile plastic culture tubes (Sterilin Ltd) to which the additions of Con A (Sigma Chemical Company Ltd) and/or other drugs were performed as desired in small aliquots from stock solutions made up in sterile 0.9% saline or 0.9% saline-ethanol/dimethyl sulphoxide. Cultures were set up in triplicate in sterile tissue culture microtitre plates (Flow Laboratories Ltd) with each well receiving 100 µl of cell suspension (5 x 10<sup>5</sup> cells). Cultures were maintained for 48 hours unless stated otherwise at 37°C in a

humidified 5% CO2 atmosphere.

The culture medium consisted of RPMI 1640 medium (Sigma Chemical Company Ltd) supplemented with penicillin (5000 IU/ml) and streptomycin 5000  $\mu$ g ml<sup>-1</sup> (Flow Laboratories Ltd) and 10% foetal calf serum (FCS) (Flow Laboratories Ltd). No reduction in the viability of the total cell population was observed over the 48 hour incubation period (as assessed by Trypan Blue exclusion).

### 2.2.2 Estimation of DNA Synthesis

The measurement of the rate of DNA synthesis is the basis for the majority of assay systems used in the quantitation of the lymphocyte response to mitogens. The degree to which radioactivity from <sup>3</sup>H-thymidine is actually incorporated into the DNA of lectin stimulated lymphocytes is taken as a measure of the proliferation, and thus of the immuno-competence of the cells (Oppenheim, 1969). Despite the popular use of this technique to estimate DNA synthesis there are potential pitfalls. The amount of exogenous thymidine in the culture system is critical and needs to be sufficiently great to maintain adequate intracellular pools for up to 4 hours whilst not causing radiation damage and reduction of the rate of DNA synthesis (Waite & Hirschorn, 1978).

Two pathways exist in lymphocytes for the utilisation of thymidine, the "salvage" pathway utilises exogenous or preformed thymidine while the *de novo* pathway provides thymidine by synthesis from formate and serine. Thus, the possibility must be appreciated that the incorporation of exogenous labelled thymidine may vary with different conditions and may be different in various lymphoid cell types (Wolberg, 1971). Indeed, tritiated thymidine incorporation into DNA is not thought to reflect DNA synthesis in the rat thymocyte since the production of *de novo* thymidine is considered to dilute out the isotopic pool and specific activity within the cell (Youdale & MacManus, 1975). Another factor outlined in a recent report suggests that the incorporation of radioactivity in human lymphocytes cannot be used as a quantitative
indicator of cell proliferation since several components of blood catabolise nucleosides, including thymidine, rendering it unincorporable (Bodycote & Wolfe, 1986). Nevertheless, tritiated thymidine incorporation into DNA is considered to be a useful assessment of the rate of DNA synthesis in many lymphocytes when measurement of the incorporation of the label is made using short pulses of relatively high concentrations of thymidine (Ling & Kay, 1975).

Since DNA synthesis is both temporally and physiologically far removed from the initial event of activation the quantitative nature of the response may be obscured. For instance, in cultures incubated for more than 48 hours cell death and simultaneous cell division complicate the analysis of the response (Wilson & Thompson, 1968).

## 2.2.3 Proliferative Assay: <sup>3</sup>H Thymidine Incorporation

Proliferative responses were assayed by incorporation of 50  $\mu$ l of <sup>3</sup>H methyl thymidine (1  $\mu$  Ci/well) (Specific activity 5 Ci/mmol, Amersham International plc) for the last four hours of incubation. Cultures were harvested using a Titertek Cell Harvester (water wash model) (Flow Laboratories Ltd) and <sup>3</sup>H-thymidine incorporation measured by liquid scintillation counting (Packard tri-carb counter, model no 2660) using a toluene based scintillant made up thus, 750 ml toluene, 250 ml 2-methoxyethanol, 0.1g POPOP (1,4, Di 2-(5-phenyloxazolyl-benzene) and 3.0g PPO (2,5, Diphenyloxazole) (BDH Chemicals Ltd).

#### 2.2.4 Interleukin 2 Assay/Proliferative Model

Stimulation of splenocytes with mitogenic plant lectins induces a subpopulation of lymphocytes to become responsive to IL2. Hence, preparations of lectin activated cells can be used to assay the lymphokine and an experimental model subsequently developed to investigate IL2 induced proliferation, in particular, signal

109

transduction and the effect of sex steroids. Mouse splenocytes can be easily activated using Con A to provide lymphocyte preparations which respond to IL2, the major disadvantage being that these cells may also respond to other stimulators (eg. IL-1, and residual plant lectin) (Gearing et al, 1985b).

## 2.2.5 Preparation of Con A Blast Assay

Splenocyte cultures were prepared (see Section 2.2.1) at a cell density of 2 x  $10^{6}$  cells ml<sup>-1</sup> in culture medium containing 5 x  $10^{-5}$ M 2-mercaptoethanol and 2.5 µg ml<sup>-1</sup> Con A and incubated for 72 hours. The blasts were washed 3 times and resuspended at 1 x  $10^{6}$  cells ml<sup>-1</sup> in culture medium containing 5 x  $10^{-5}$  M 2-mercaptoethanol and 25 mM  $\alpha$  methyl mannoside (Sigma Chemical Company Ltd) which inhibits the mitogenic effect of any residual Con A. 50 µl aliquots of doubling dilutions of recombinant Human IL2 (a gift from Dr AJH Gearing) prepared in RPMI 1640 medium were plated out in triplicate in sterile tissue culture multi-well trays (Flow Laboratories Ltd), 50 µl of cell suspension was added to each well and cultures incubated for 18 hours. 50 µl (1 µ Ci) of <sup>3</sup>H-Thymidine (Specific Activity 5 Ci/mmol Amersham International plc) was then added to individual wells. Cells were harvested 4-6 hrs later, and incorporated radioactivity determined as before (Section 2.2.3).

# 2.3 <u>Measurement of Intracellular Ca<sup>2+</sup> Levels</u>

There is substantial evidence consistent with a role for cytoplasmic  $Ca^{2+}$  as a key regulator in the proliferative response of lymphocytes and many other cells (see Section 1.5.2). However, much of the evidence for calcium's role as intracellular regulator comes indirectly, from experiments involving manipulation of extracellular calcium, measurement of  $Ca^{2+}$  fluxes and studies of  $Ca^{2+}$  on isolated enzymes and systems. Direct evidence requires that the initial stimulus should cause an increase in intracellular  $Ca^{2+}$  which is necessary for the proliferative response. Measurement of

cytoplasmic free Ca<sup>2+</sup> could provide evidence for intracellular Ca<sup>2+</sup> as the mediator of the cell's response to external stimuli and differentiate this role from other passive roles for Ca<sup>2+</sup> in the cell. In addition, it would enable the source of Ca<sup>2+</sup> for stimulus-mitosis coupling to be delineated, ie. intracellular and/or extracellular. Moreover it would allow  $^{45}$ Ca<sup>2+</sup> fluxes to be accurately interpreted. Thus, techniques for monitoring free Ca<sup>2+</sup> levels in intact cells are crucial in determining the role of Ca<sup>2+</sup> in stimulus-mitosis coupling.

An ideal intracellular free  $Ca^{2+}$  indicator should be sensitive within a range of 10nM —> 10 µM and be selective for  $Ca^{2+}$  in relation to other ions. The response of the indicator to free  $Ca^{2+}$  should be linear so that quantitation is simple. Additionally, the change in signal should be great for a small fraction of  $Ca^{2+}$  bound. Ideally, the response of the indicator should be quicker than the changes in  $[Ca^{2+}]i$ being measured. Incorporation of the indicator into intact cells must be adequate to monitor free  $Ca^{2+}$  concentrations without perturbing intracellular  $Ca^{2+}$  homeostasis or disrupting the cell. Once incorporated the indicator must not leak out nor be taken up by intracellular organelles. The ideal indicator must obviously not be toxic to the cells in question.

Indicators which are currently used on the basis of the above criteria may be categorised into distinct groups.

#### (a) <u>Chemiluminescent Indicators</u>

 $Ca^{2+}$  activated photoproteins such as aequorin and obelin contain a covalently bound chromophore which emits light when  $Ca^{2+}$  binds to the protein. They have found wide applicability in measuring free  $Ca^{2+}$  changes in large cells but have proved inappropriate for small cells including lymphocytes because of the difficulty of incorporation without perturbation of cell function (Snowdowne & Borle, 1984; Campbell et al, 1985). (b)  $Ca^{2+}$  Selective microelectrodes

These microelectrodes are based on the principle of separating the cytosol from a constant reference solution with a membrane which is selectively permeable for  $Ca^{2+}$  ions. The activity of  $Ca^{2+}$  ions is measured by a potentiometric signal. The major disadvantage of these microelectrodes is that they sense the chemical environment only at their tip and may therefore miss a localised change in free  $[Ca^{2+}]i$  which may be functionally very important (Tsien & Rink, 1980). Moreover their use requires large robust cells which are not damaged by penetration of the microelectrode.

#### (c) <u>Metallochromic indicators</u>

Indicators such as Arzenazo III and Antipyralazo III change colour upon binding  $Ca^{2+}$ , but since Arzenazo III is susceptible to interference from  $Mg^{2+}$  and causes significant  $Ca^{2+}$  buffering and Antipyralazo III is not sensitive enough to detect resting  $[Ca^{2+}]$  i they fail to meet the requirements of an ideal indicator (Dormer et al, 1985).

## (d) <u>Nuclear Magnetic Resonance Indicators</u>

These indicators, 4-fluoro-BAPTA and 5-fluoro-BAPTA, show a shift in resonance upon binding to  $Ca^{2+}$ . They have been used to investigate free  $Ca^{2+}$  levels in lymphocytes but since the spectrum accumulates over a long period (~ 20 mins) it would be unsuitable for the measurement of  $Ca^{2+}$  transients (Metcalfe et al, 1985).

#### (e) <u>Fluorescent Dyes</u>

Quin 2 belongs to a new family of fluorescent  $Ca^{2+}$  indicators and has been a popular choice for the measurement of cytosolic free  $Ca^{2+}$  concentration ([ $Ca^{2+}$ ]i). The monitoring of free  $Ca^{2+}$  levels by quin 2 particularly in small mammalian cells is

112

partly due to the relative ease of incorporation and partly because of its straightforward calibration (Tsien, 1980; Tsien et al, 1982). Quin 2 is loaded into intact cells by incubating them with a membrane-permeant ester derivative. Cytosolic esterases split off the ester groups and leave the membrane-impermeant quin 2 trapped in the cytosol. Increases in quin 2 fluorescence then signal increased [Ca<sup>2+</sup>]i. Although quin 2 has disclosed a lot of important biological information, including increases in [Ca<sup>2+</sup>]i in lymphocytes following lectin stimulation (Tsien et al, 1982), it has always had severe and acknowledged limitations (Tsien et al, 1982, 1984). Indeed earlier studies performed with rat thymocytes showed that high loading concentrations of quin 2/AM were necessary to produce an adequate signal, these higher concentrations of quin 2 appeared to be toxic, to buffer intracellular Ca<sup>2+</sup> transients and furthermore to leak out of the cells in substantial amounts. Recently, a new generation of  $Ca^{2+}$  indicators with greatly improved fluorescent properties have been developed (Grynkiewicz et al, 1985) and the preferred dye for the majority of applications is now considered to be fura-2. Fura-2 has some major advantages over quin 2 including a sixfold higher extinction coefficient and five fold greater quantum efficiency, such that a given intensity of fluorescence requires approximately thirty times less fura-2 than quin 2, a decrease in intracellular dye loadings can be achieved and buffering of cytosolic [Ca<sup>2+</sup>]i transients can be avoided. Fura-2 bleaches much less rapidly than quin 2 and is therefore the dye of choice for fluorescence microscopy (Tsien et al, 1985). Fura-2 shifts its excitation peak from about 340 nM to 360 nM upon binding  $Ca^{2+}$ , both free and bound species have fairly high quantum efficiencies and pairs of excitation wavelengths may be used for ratio operation. The ratio of the fluorescences at two suitably chosen wavelengths can signal calcium whilst cancelling out most, if not all of the possible variability due to instrument efficiency and content of effective dye. Another advantage of fura-2 over quin 2 is its better selectivity for Ca<sup>2+</sup> over divalent cations such as  $Mg^{2+}$ ,  $Mn^{2+}$  and  $Zn^{2+}$ .

The main disadvantage of fura-2 compared to quin 2 is the need for dual wavelength instrumentation to take full advantage of the wavelength shifts. Nevertheless, fura-2 is still considered a better choice for monitoring cytosolic calcium even if the ratio method is prohibited due to lack of the sophisticated instrumentation required. Since fura-2 seems eminently suitable for measuring transient changes in  $[Ca^{2+}]i$  in lymphocytes (Tsien et al, 1985) and has many advantages over quin 2 it was selected for use in this study with mouse splenocytes.

# 2.3.1 Measurement of Intracellular Ca<sup>2+</sup> by the Fluorescent Indicator Fura-2

Since fura-2 is membrane impermeant it has to be incorporated into cells as an acetoxymethyl ester fura-2/AM. Once inside the cell, it is hydrolysed by cytoplasmic esterases which effectively trap the free acid, fura-2. Fura-2/AM supplied as a freeze-dried oil (Calbiochem) was dissolved in dry dimethylsulphoxide (DMSO) and desiccated at -20°C in small aliquots prior to use.

Fura-2 has a dissociation constant (Kd) of 224 nm at 37°C and shows optimal fluorescence at an emission wavelength of 510 nm using an excitation wavelength set at 340 nm. Fura-2 does not appear to be effected by pH changes within the physiological range and neither does it bind to membranes (Grynkiewicz et al, 1985) and thus is an effective method of monitoring changes in  $[Ca^{2+}]i$ .

#### Loading Cells with Fura-2

Mouse splenocytes were isolated as described in Section 2.2.1 and adjusted to  $5 \times 10^7$  cells ml<sup>-1</sup> in RPMI 1640 medium supplemented with 10% FCS. Fura-2/AM was added to the cell suspension at a final concentration of 5  $\mu$ M (shown to give optimal fluorescence signals without showing signs of cytotoxicity) and this was then incubated in stoppered culture tubes (Sterilin Ltd) in a roller drum assembly to allow full and even uptake of the dye for 10 minutes at 37°C. The suspension was then diluted 10 fold with culture medium and incubated for a further 20-30 minutes. The cells were then washed, resuspended and stored in RPMI medium prior to use.

#### Fluorescence Measurements

Freshly loaded cells were spun for 30 seconds in a Beckman microfuge and resuspended in 3 mls of physiological saline at  $2.5 \times 10^6$  cells per ml and loaded into a 1cm square quartz cuvette. This procedure minimises any fura-2 carry over. Physiological saline is preferred to the usual RMPI 1640 medium since substances such as phenol red may filter excitation and emission wavelengths and/or contribute their own fluorescence. Physiological saline comprised 145 mM NaCl, 5mM KCl, 1mM Na<sub>2</sub>HPO<sub>4</sub>, 5mM Glucose and 10 mM HEPES adjusted to pH 7.4 at 37°C with NaOH. Mg<sup>2+</sup> (1.0mM) and Ca<sup>2+</sup> (0.6 mM) were added from stock solutions of their respective sulphate and chloride as required.

For fluorescence measurements a Perkin-Elmer LS-5 spectrofluorimeter was used with excitation wavelength set at 340 nm (2.5 nm slit width) and emission measured at 510 nm (5 nm slit width). Cells were continuously stirred and maintained at 37°C whilst the fluorescence signal was plotted in arbitrary units. Additions of small aliquots (< 30  $\mu$ l) of concentrated reagents were made to the cell suspension (giving a final volume of < 3.030 mls) and any change in fluorescence monitored. Maximum fluorescence values ( $F_{max}$ ) were obtained by adding 40  $\mu$ l of a 10% solution of Triton X-100 R (reduced) (Sigma Chemical Company Ltd) thus permeabilising the cells to calcium and ensuring that essentially all fura-2 was complexed with calcium. Minimum fluorescence values ( $F_{min}$ ) were then achieved by the addition of 50  $\mu$ l of 0.5M EGTA in the presence of 25  $\mu$ l of 10% NaOH to achieve a pH value > 8.3. A typical trace is presented in Diagram 4.



Diagram 4: Measurement of cytosolic calcium levels in fura-2 loaded murine splenocytes.

The fluorescence values correspond to resting calcium levels of 143  $\cdot$  nM (R) and stimulated levels of 193 nM (S) which represents an increase of 1.35 (S/R). (See Section 2.3.1 for details).

## [Ca<sup>2+</sup>]i Concentrations

Intracellular free calcium concentrations were calculated according to methods described by Grynkiewicz et al (1985). Thus:

$$[Ca^{2+}]i = \frac{F - F_{min}}{F_{max} - F} \times 224 \text{ nM}$$

where F is the fluorescence reading.  $F_{max}$  is the F for saturated fura-2 whilst  $F_{min}$  is the F for free fura-2. This method assumes that there is minimal quenching of fura-2 by heavy metal ions within the cells and little leakage of fura-2 from the cells during the experiment. Fura-2 leakage did not appear to be a problem during the course of experiments since they were invariably short-lived (~ 5 mins). However since batches of cells had to be prepared for four sequential experiments at one time it was necessary to have the cell suspensions incubating at 37°C in the spectrofluorimeter for about 10-20 minutes prior to use. Fura-2 leakage did appear to contribute to a gradual increase in the calculated resting value during the series of four experiments. For example on one occasion the resting value appeared to gradually increase from 124 nM in the first experiment to 218 nM in the last even though the cells originally came from the same freshly loaded stock. Under the circumstances it seems that qualitative changes in [Ca<sup>2+</sup>]i could accurately be demonstrated whilst quantitation of the results might be misleading, thus increases in  $[Ca^{2+}]i$  have been expressed such that the resting or basal  $[Ca^{2+}]i$  is equal to unity and an increase in  $[Ca^{2+}]i$  is represented by a figure greater than 1 (see Results section, Chapter 3). Since there are some uncertainties in the interpretation of fura-2 data and it is unknown whether changes in [Ca<sup>2+</sup>]i observed by mitogen application to fura-2 loaded cells are comparable to those which occur in cells unperturbed by the presence of the dye it seemed appropriate to combine these studies with an indirect technique which could quantify transient transmembrane fluxes of radiolabelled  $Ca^{2+}$  during mitogenic stimulation.

#### 2.4 <u>Calcium Uptake Studies</u>

Many calcium uptake studies have used procedures involving extensive centrifugation, washing and resuspension of cells to remove unincorporated isotope from the extracellular fluid. These procedures are not fast enough to prevent a trans-plasmamembrane readjustment of cell calcium and therefore tracer exchange. Furthermore, such procedures are likely to result in metabolic alterations when pelleted cells are resuspended. Not surprisingly, the reported values of  ${}^{45}Ca^{2+}$  uptake in mitogen-stimulated cells using such techniques are conflicting (Deutsch & Price, 1982). These problems may be overcome by using a modification of the centrifugation technique developed by Freedman, Raff and Gomperts (1975). In this technique, the cell associated  ${}^{45}Ca^{2+}$  can be distinguished from the unincorporated extracellular tracer by centrifugation through a chemically inert and water impermeable column of oil. In the original technique cells were lysed in formic acid at the bottom of the microfuge tube, to release  ${}^{45}Ca^{2+}$  to be counted. However, preliminary studies showed that elimination of this layer produced more consistent results. Thus, polypropylene microcentrifuge tubes (Beckman PRO-22) were prepared by addition of 175 µl MS 550 silicone oil (BDH Chemicals Ltd) and subsequently spun to ensure an air bubble-free column.

Mouse splenocyte cultures were prepared as described (see Section 2.2.1) and cell density adjusted to  $10^8$  ml<sup>-1</sup>. Cultures were preincubated at 37°C for at least thirty minutes before isotope addition to allow recovery from the preparative procedures. Cell suspensions were subsequently added to prewarmed tubes containing an equal volume of medium comprising isotope (final concentration 5 µCi ml<sup>-1</sup>, specific activity, 31.61 m Ci mg<sup>-1</sup>, Du Pont de Nemours, Germany) and other agents of interest (resulting in a cell suspension of 5 x 10<sup>7</sup> cells ml<sup>-1</sup>). Triplicate 200 µl aliquots were removed at appropriate intervals thereafter and placed on top of the column and immediately centrifuged for 11/2 minutes in a Beckman microfuge. The

incubation time for each tube was taken from the time of addition of cells to isotope to the beginning of the centrifuge run, since the microfuge achieves maximum force within five seconds.

The contents of the tube were then frozen by immersion in liquid nitrogen and the bottom (~ 1 cm) containing the cell pellet was cut off and placed in scintillation vials containing NE-260 scintillation fluid (Nuclear Enterprises Ltd). The inherent properties of the oil present potential errors to this technique but previous studies revealed that silicone oil (MS 550) did indeed have the required characteristics. Thus, the oil is chemically inert, highly water impermeable and allows maximal cell passage with minimal supernatant carry over. Carry over of isotope from the aqueous phase (presumably trapped within the interstitial spaces between the cells) was 0.13% or less of the total supernatant activity. Such amounts were trivial compared to the cell-associated <sup>45</sup>Ca<sup>2+</sup> and were therefore ignored. Controls (blanks) were run during the course of each experiment by centrifuging 200 µl of the cell free isotope (5 µi Ci ml<sup>-1</sup>) through the oil to monitor any changes in the physical properties of the oil, the subsequent counts were always found to be negligible. Radioactive calcium was measured using liquid scintillation spectroscopy (after 24 hours in contact with the scintillant to allow maximum dispersal) in a Packard tri-carb counter, model no 2660. An internal standard was used to correct for decay and the efficiency of the B-detector corrected using an on-line handling system.

Cell-associated  ${}^{45}Ca^{2+}$  comprises both transmembrane influx of the ion and non-specific adsorption to the plasma membrane. Although some workers have chilled cells to eliminate 'active' transmembrane  $Ca^{2+}$  transport and used cell-associated  ${}^{45}Ca^{2+}$  values at 4°C to represent non-specific membrane binding (Freedman et al, 1975) preliminary investigations suggested that this assumption was invalid. Thus, membrane-bound  ${}^{45}Ca^{2+}$  at 4°C estimated by this technique was invariably greater than or equal to cell associated  ${}^{45}Ca^{2+}$  for control cells incubated with isotope for 30 minutes at 37°C (results not shown). The act of chilling cells possibly alters plasma membrane properties with the effect of increasing non-specific adsorption of  $^{45}Ca^{2+}$ . Therefore it seemed more appropriate to consider membrane bound  $^{45}Ca^{2+}$ , to be cell associated calcium at zero time, ie. immediately after mixing cells with isotope and rapid centrifugation. The Ca<sup>2+</sup> taken up into the cells was then calculated from the following equation, all values were precorrected for counting efficiency and background activity.



where total dpm represents counts of a 200  $\mu$ l aliquot of isotope-laden cell suspension without spinning through the oil layer. Since the basal medium contains 0.6mM Ca<sup>2+</sup> (equivalent to 24  $\mu$ g ml<sup>-1</sup>) total dpm represents counts in a 200  $\mu$ l aliquot and is therefore equivalent to 4.8  $\mu$ g (4800ng).

## 2.5 Measurement of Intracellular pH (pHi)

An increase in pHi following a rise in cytoplasmic free  $Ca^{2+}$  ( $[Ca^{2+}]i$ ) may occur during the stimulation of cell growth in a variety of cells. Thus, serum, growth factors, mitogenic lectins and other agents that induce an increase in  $[Ca^{2+}]i$  also activate Na<sup>+</sup>/H<sup>+</sup> exchange. Since the change in  $[Ca^{2+}]i$  generally precedes the activation of the antiport, it has been suggested that there may be a common sequence of changes in  $[Ca^{2+}]i$  and pHi when different types of cells enter the cell cycle (Hesketh et al, 1985) (see section 1.5.4). It therefore seemed appropriate to combine the studies on intracellular  $Ca^{2+}$  measurement with determinations of pHi. The pHi of lymphocytes from different sources and of cultured lymphoblasts has been determined by a variety of techniques, including weak acid and base partition (Gerson & Kiefer, 1982), <sup>31</sup>P and <sup>19</sup>F nuclear magnetic resonance (NMR) (Rink et al, 1982) and direct titration of freeze-thawed cells (Rink et al, 1982). Since, these techniques are unable to resolve the time course of any changes in pHi which may occur within seconds of stimulation (Rogers et al, 1983) the use of fluorescent indicators to measure pHi changes was considered more promising.

## 2.5.1 Measurement of Intracellular pH by a Fluorescent Indicator, Quene 1

Quene 1, a fluorescent intracellular pH indicator is related to quin 2 and is designed to respond rapidly to changes in normal pHi values and to have negligible affinity for  $Ca^{2+}$  at normal levels of  $[Ca^{2+}]i$  (Rogers et al, 1983). Quene 1 has excitation and emission maxima at 390 and 530 nm respectively, and shows a 30-fold increase in fluorescence between pH 5 and 9. The fluorescence is insensitive to  $Ca^{2+}$ and Mg<sup>2+</sup> at free concentrations up to 10<sup>-4</sup>M and to the proportions of Na<sup>+</sup> and K<sup>+</sup> at total concentrations of Na<sup>+</sup> and K<sup>+</sup> from 100 to 200 mM. The indicator is loaded into cells using the tetraacetoxymethyl ester derivative which is hydrolysed in the cells to give the membrane impermeant tetracarboxylate anion. Once incorporated quene 1 does not perturb glycolysis or the ATP level in resting cells at concentrations up to 0.8 mM (Rogers et al, 1983). Thus, the properties of quene 1 are well suited to its use as an intracellular indicator of pH.

#### Loading Cells with Quene 1

Mouse splenocytes were prepared as described in Section 2.2.1 and adjusted to  $1 \ge 10^7$  cells ml<sup>-1</sup> in RPMI 1640 medium supplemented with 10% FCS. The cells were incubated for 30 minutes at 37°C with 1 µM quene 1 acetoxymethyl ester (shown to give optimal fluorescence signals without showing signs of cytotoxicity) (Amersham International plc) from a stock solution in DMSO (final concentration < 1% (v/v)) before washing by centrifugation and resuspension in fresh culture medium at a cell density of 5 x 10<sup>6</sup> cells ml<sup>-1</sup>, for a further 30 minute incubation period prior to use.

#### Fluorescence Measurements

Cell aliquots (1.5 ml) were spun for thirty seconds in a Beckman microfuge and resuspended in 3 mls of physiological saline ( $2.5 \times 10^6$  cells ml<sup>-1</sup>) and loaded into a 1cm square quartz cuvette. Cells were allowed to equilibrate prior to fluorescence measurements which were performed in a Perkin-Elmer LS-5 spectrofluorimeter (excitation 390nm, emission 530 nm). Cells were continuously stirred and maintained at 37°C whilst the fluorescence signal was plotted in arbitrary units. The agents of interest were added from stock solutions and any change in fluorescence monitored. The fluorescence of lymphocytes loaded with quene 1 may be calibrated by the addition of 0.05% Triton X-100R in 0.5 mM EGTA and 0.5 mM EDTA to lyse the cells and release quene 1 into the extracellular medium. Serial additions of 1 M Tris and 0.5M HCL solutions calibrate the fluorescence of the released quene 1 as a function of pH (Rogers et al, 1983) Although these steps were performed and changes in fluorescence recorded (see Diagram 5) no correlation has been made to pHi values.

Preliminary studies showed that quene 1, whilst recording changes in fluorescence (and therefore pHi) to a weak base (5 mM NH<sub>4</sub> Cl) (Diagram 5) did not monitor any change following stimulation with Con A (n = 3) or TPA (12-0-tetradecanoyl phorbol 13-acetate) (Sigmal Chemical Company Ltd) (n = 3) during the thirty minute period following addition.

The absence of any effect on pHi of these two agents is in agreement with other studies where quene 1 failed to monitor pHi changes in lectin stimulated thymocytes (Rogers et al, 1983). However, since lectin stimulation of mouse splenocytes does lead to an increase in pHi when measured with weak acid DMO (5,5-dimethyloxazolidine-2,4-dione) (Gerson et al, 1982) further studies were undertaken with another fluorescent indicator to eliminate the possibility that quene 1 might be insensitive to pHi changes in mouse splenocytes.



Diagram 5: The effect of a weak base (5 mM NH<sub>4</sub> Cl) on the pHi of quene 1 loaded murine splenocytes.

The fluorescence intensity of quene 1 loaded murine splenocytes in physiological saline prepared as described in Section 2.5.1. Arrows indicate additions of (i) NH<sub>4</sub>Cl 5 mM (15  $\mu$ l of 1M NH<sub>4</sub> Cl) or Con A (5  $\mu$ g ml<sup>-1</sup>) or TPA (10<sup>-9</sup>M) (ii) 0.05% Triton X-100R with 0.5 mM EGTA and 0.5 mM EDTA (15  $\mu$ l of 10% Triton X-100R + 15  $\mu$ l 100 mM EGTA + 15  $\mu$ l 100 mM EDTA) (iii) 2.5 mM Tris (7.5  $\mu$ l of 1M Tris) and (iv) sequential additions of 1.25 mM HCl (7.5  $\mu$ l aliquots of 0.5 M HCl).

#### 2.5.2 <u>Measurement of Intracellular pH by the Fluorescent Indicator BCECF (2',</u> <u>7'bis (Carboxyethyl) - 5 (6)-Carboxyl fluorescein)</u>

BCECF a fluoresent pH indicator originally used by Rink, Tsien and Pozzan (1982) is superior to the carboxyfluoresceins from which it is derived. It allows measurement of a nearly linear relationship of fluorescence emission vs pH in the physiologically important pH range of 6.5 to 7.5 and does not leak out of cells as rapidly as the carboxyfluoresceins. BCECF has excitation and emission maxima at 500nm and 530nm respectively. The indicator whilst having a high sensitivity to small changes in pHi does not appear to be toxic nor affect  $[Ca^{2+}]i$  (Rink et al, 1982).

BCECF may be loaded into cells in the form of the membrane-permeant, substantially non-fluorescent ester derivative BCECF/AM. Once BCECF/AM penetrates the cell membrane it is cleaved by cytoplasmic esterases to yield free fluorescent BCECF. This indicator system has been used successfully to monitor intracellular pH changes in mammalian lymphocytes (Hesketh et al, 1985; Metcalfe et al, 1985; Grinstein & Cohen, 1987), fibroblasts (Moolenaar et al, 1983) and gastric cells (Paradiso et al, 1984). It therefore seemed desirable to investigate its ability to detect reputed pHi changes in stimulated mouse splenocytes (Gerson et al, 1982).

#### Loading Cells with BCECF

Mouse splenocytes were isolated as described in Section 2.2.1 and adjusted to  $3 \times 10^7$  cells ml<sup>-1</sup> in RPMI 1640 medium supplemented with 10% FCS. Cells were loaded with 1 µM BCECF (a gift from Dr Mike Thompson) from a stock solution in DMSO (final concentration <0.05% v/v) shown to give optimal fluorescence signals without adversely affecting cell viability (determined by Trypan Blue exclusion). Following incubation for thirty minutes at 37°C the cells were centrifuged and resuspended in fresh culture medium at a cell density of  $6 \times 10^6$  cells ml<sup>-1</sup>.

#### Fluorescence Measurements

Aliquots (1.5 ml) of freshly loaded cells were spun for thirty seconds in a Beckman microfuge and resuspended in 3 mls of physiological saline (3 x  $10^6$  cells m1<sup>-1</sup>) and loaded into a 1cm square quartz cuvette. Following equilibration, fluorescence measurements were recorded in a Perkin Elmer S-5 spectrofluorimeter (excitation 500 nm, slit width 2.5 nm, emission 530 nm, slit width, 5.0 nm). Cells were continuously mixed with a magnetic stirrer and maintained at 37°C whilst the fluorescence signal was plotted in arbitrary units. Small aliquots (< 30 µl) of concentrated reagents were added (final volume < 3.030 ml) and any change in fluorescence monitored. The simplest calibration can be achieved by releasing the dye into the medium after lysing the cells and then recording the fluorescent signal at known pHs (Rink et al, 1982). However since the dye behaves differently in the physiological saline than inside the cells (excitation peak ~ 5 nm red shifted in intracellular dye) an appropriate correction on the calibration scale has to be made for each batch of cells by the method of Thomas et al (1979), using nigericin. Since the main purpose of using BCECF was to detect changes in pHi, this elaborate calibration method was not performed. Preliminary studies revealed that BCECF was able to record pHi changes following stimulation of mouse splenocytes (see Results section, Chapter 3).

## 2.6 <u>Statistical Analysis</u>

Groups of data have been compared using Anovar one way analysis of variance. The least significant difference (LSD) has been determined by substituting the value of 't' at a given probability level. Thus, in each of the figures (Chapter 3) asterisk(s) denote the degree of significance \* P < 0.05, \*\* P < 0.01 and \*\*\* P < 0.001. Values have been expressed as the means  $\pm$  the standard error of the mean (SEM) and n represents the number of separate experiments in each group.

# CHAPTER THREE EXPERIMENTAL RESULTS

Preliminary studies demonstrated that Concanavalin A (Con A) induces an increase in DNA synthesis in murine splenocytes which reaches a maximum by 48 to 72 hours (results not shown). The dose-response curve of Con A stimulation for spleen cells incubated for 48 hours (Figure 1) shows that mitogen-induced stimulation increased with Con A concentrations up to 5  $\mu$ g ml<sup>-1</sup> and then decreased with increasing concentrations of Con A. The stimulation indices proved to be rather variable as witnessed by the large SEM's (standard error of mean) although the trends were similar for individual experiments in which cells are derived from a single male TO mouse. With a view to achieving greater consistency in the Con A-induced lymphocyte transformation model, the effect of Con A on murine splenocytes derived from highly inbred male C<sub>57</sub> BL/6 mice was examined (Figure 2). Having established that the variability could not be reduced by employing a different murine strain all subsequent experiments were performed using splenocytes from male TO mice stimulated with the optimal concentration of Con A.

Among the earliest responses common to a variety of mitogen stimulated cells is an increase in cytosolic Ca<sup>2+</sup> concentration ([Ca<sup>2+</sup>]i) (see Section 1.5.2). Calcium influx may thus mediate stimulus-mitosis coupling of the cells in this study. If calcium movement into murine splenocytes is involved in signal transduction then Con A induced DNA synthesis should be compromised by the addition of calcium antagonists to the culture medium. Three specific calcium antagonists thought to have different sites of action upon the so-called long or L-calcium channel were employed in this study (see Discussion, Chapter 4 for further details). The effects of the drugs verapamil, nifedipine and diltiazem on Con A (5  $\mu$ g ml<sup>-1</sup>) induced splenocyte proliferation are illustrated in Figures 3, 4 and 5 respectively. To assess the effects of drugs on Con A induced stimulation the level of [<sup>3</sup>H]-thymidine incorporation with



Figure 1: Dose-response curve of the effect of Concanavalin A on thymidine incorporation in isolated TO murine splenocytes

Results expressed as stimulation index (SI) represent the mean ( $\pm$  SEM) of triplicate determinations in nine experiments

SI = \_\_\_\_\_

Incorporation in the unstimulated control (cpm/well)

n = the number of separate experiments for each treatment values significantly different from those of the other cultures and indicated by asterisks



Figure 2: Dose-response curve of the effect of Concanavalin A on thymidine incorporation in isolated  $C_{57}$  BL/6 murine splenocytes.

Values significantly different from those of the other cultures are indicated by asterisks. Other details as in Figure 1.



Figure 3: Effect of Verapamil on Con A induced murine splenocyte proliferation.

The level of thymidine incorporation in cultures stimulated by 5  $\mu$ g ml<sup>-1</sup> Con A alone is taken as 100% and the incorporation in cultures incubated with 5  $\mu$ g ml<sup>-1</sup> Con A plus various concentrations of verapamil expressed as a percentage thereof. Values significantly different from corresponding cultures without the drug are indicated by asterisks.



Figure 4: Effect of Nifedipine on Con A (5  $\mu$ g ml<sup>-1</sup>) induced murine splenocyte proliferation.

Values significantly different from corresponding cultures without the drug are indicated by asterisks. Other details as in Figure 3.





Values significantly different from corresponding cultures without the drug are indicated by asterisks. Other details as in Figure 3.

Con A alone was taken as 100% and incorporation in the presence of drugs expressed as a percentage thereof. All three agents suppressed proliferation in a dose-related manner, a significant inhibition of  $[^{3}H]$ -thymidine uptake was found at 5 x 10<sup>-5</sup> M verapamil, 10<sup>-6</sup> M nifedipine and 10<sup>-4</sup> M diltiazem. To exclude the possibility that suppression of T cell proliferation is due to cytotoxicity, cell viability was examined in control cultures and those containing the drugs at the above concentrations. 90-95% of splenocytes were viable at the end of the incubation period as assessed by trypan blue exclusion, under all culture conditions. Thus,  $Ca^{2+}$  influx appears to be a pre-requisite for the Con A induced mitotic response in splenocytes. To verify this hypothesis the radioisotope of calcium,  ${}^{45}Ca^{2+}$ , was used to monitor calcium influx in splenocytes at rest and during their recruitment from quiescence by Con A. Preliminary studies revealed that the time course of  ${}^{45}Ca^{2+}$  uptake in Con A treated splenocytes was not significantly different from control cells (Figure 6) implying that either Con A induced mitotic stimulation does not involve an enhanced Ca<sup>2+</sup> influx or that the technique employed (see Section 2.4 for details) fails to monitor an enhanced  $Ca^{2+}$  influx. Since cells treated with the calcium ionophore A23187 (10<sup>-7</sup>M) also failed to demonstrate any increase in  ${}^{45}Ca^{2+}$  uptake over control values during a thirty minute period it appears likely that this microcentrifugation technique is unsuitable for monitoring Ca<sup>2+</sup> fluxes in murine splenocytes.

Since  ${}^{45}Ca^{2+}$  tracer studies were unable to define the role of  $Ca^{2+}$  fluxes in Con A induced proliferation, it seemed appropriate to monitor intracellular calcium levels following lectin stimulation. For this purpose fura-2 (see Section 2.3.1) was employed. The calcium ionophore A23187 was used to demonstrate the effectiveness of the procedure. An increase in cytosolic  $Ca^{2+}$  concentration is represented by a figure greater than one, since resting levels are considered equal to unity. A23187  $10^{-7}M$  caused a rapid (within one minute) increase in fluorescence (Table 1) whereas the addition of an equal volume of physiological saline did not influence the  $[Ca^{2+}]i$ over a ten minute period (results not shown).



Figure 6: Effect of Con A (5  $\mu$ g ml<sup>-1</sup>) and ionophore A23187 (10<sup>-7</sup>M) on calcium uptake in isolated murine splenocytes

Each value represents the mean ( $\pm$  SEM) of triplicate determinations in three experiments.

Resting [Ca <sup>2+</sup> ]i levels (R) (nM)	Stimulated [Ca <sup>2+</sup> ]i levels (nM) (S)	Increase (S/R)	Mean increase ± SEM
189	560	2.96	2.23 ± 0.37
145	251	1.73	
137	273	1.99	

Table 1 - Effect of A23187 10<sup>-7</sup>M on the intracellular calcium concentration of murine splenocytes

Medium cation concentration was 0.6 mM Ca<sup>2+</sup>, 1.0 mM Mg<sup>2+</sup> in all fura-2 studies unless otherwise stated.

Splenocytes retained their ability to respond to calcium ionophore even after two hours incubation at 37°C although some fura-2 leakage was apparent (see Section 2.3.1); cell viability was also maintained at greater than 90% throughout this period. When intracellular calcium was monitored following Con A treatment a rapid increase in  $[Ca^{2+}]i$  was observed reaching a maximum by 11/2 minutes (see Diagram 4 for a typical recording). The rise in  $[Ca^{2+}]i$  in response to Con A in murine splenocytes demonstrates an increase of 1.45 ( $\pm$  0.16 SEM n = 39) over resting levels. The  $[Ca^{2+}]i$  prior to stimulation, ie. the resting level is 186 nM ( $\pm$  10 SEM n = 39). The  $Ca^{2+}$  signal persists for at least one hour following interaction of Con A with its receptor (results not shown) and possibly longer. Attempts to monitor  $[Ca^{2+}]i$ following Con A stimulation during the whole of the 48 hour incubation period were unsuccessful; fura-2 loading performed at intervals after Con A application gave variable results such that no trends were distinguishable.

The Ca<sup>2+</sup> signal generated by Con A treatment could result from a stimulation of Ca<sup>2+</sup> influx, an inhibition of active Ca<sup>2+</sup> efflux across the plasma membrane and/or mobilisation of Ca<sup>2+</sup> from an intracellular store(s) (see Section 1.5.2). To elucidate the nature of the Ca<sup>2+</sup> signal the effect of the calcium channel blockers was investigated. Verapamil (5 x 10<sup>-5</sup>M) at a concentration known to suppress Con A induced proliferation (Figure 3) significantly diminished the lectin

# stimulated $[Ca^{2+}]i$ increase (Table 2).

Table 2: The effect of Verapamil on the Con A induced increase in intracellular calcium concentration  $[Ca^{2+}]i$  of murine splenocytes

	Treatment	
	Con A (5 $\mu$ g ml <sup>-1</sup> )	Con A (5 $\mu$ g ml <sup>-1</sup> ) Verapamil (5 x 10 <sup>-5</sup> M)
Increase in	A Constant of States	
[Ca <sup>2+</sup> ]i above	1.59	1.35
basal levels	1.50	1.40
(considered	1.25	1.18
equal to unity)	1.27	1.16

Whilst investigating the effects of various agents on the Con A induced Ca<sup>2+</sup> signal, experiments involving the potential antagonists were conducted alongside control 'runs' of Con A alone wherever possible and the subsequent results analysed by the 'paired T test'. Verapamil (5 x 10<sup>-5</sup>M) has a significant (P < 0.05) effect on the Con A induced increase in [Ca<sup>2+</sup>]i in murine splenocytes. In contrast, the concentration of diltiazem (10<sup>-4</sup> M) found to inhibit proliferation (Figure 5) was without effect on the increase in [Ca<sup>2+</sup>]i promoted by Con A (Table 3).

	induced mercase mile	I or murme
splenocytes		

Table 3: The effect of Diltiazem on the Con A induced increase in  $[C_{0}2^{+1}]$  of music

	Treatment		
	Con A (5 μg ml <sup>-1</sup> )	Con A (5 $\mu$ g ml <sup>-1</sup> ) + Diltiazem (5 x 10 <sup>-4</sup> M)	
Increase in	1.40	1 34	
$[Ca^{2+}]i$	1.48	1.54	
over basal levels	1.40	1.37	
(considered equal to unity)			

Attempts to monitor the effect of nifedipine, on the Con A stimulated rise in  $[Ca^{2+}]i$ were unsuccessful due to its ability to interfere with the fluorescence signal, possibly by filtering out excitation and emission energies.

The equivocal nature of these results may reflect the differing abilities of the calcium antagonists to block a calcium influx component of the <u>early</u>  $Ca^{2+}$  signal. The effects of the  $Ca^{2+}$  channel blockers on the Con A induced  $Ca^{2+}$ signal were only monitored for approximately five minutes. However, it must be borne in mind that in a 48 hour incubation period they may cause other alterations in the intracellular calcium concentrations. Additional interpretations include the possibility that this early  $Ca^{2+}$  signal may be irrelevant in the signal transduction pathway following Con A stimulation; or that some feature of the early  $Ca^{2+}$  signal may be important or finally, that these calcium antagonists exert effects on proliferation unrelated to calcium.

The Ca<sup>2+</sup> signal in response to Con A was partially diminished when Ca<sup>2+</sup> was absent from the extracellular medium (achieved by washing twice and resuspending in Ca<sup>2+</sup> free medium), when Ca<sup>2+</sup> was subsequently added to the medium a further increase in  $[Ca^{2+}]i$  was monitored (Figure 7, Table 4) suggesting that the Con A induced early Ca<sup>2+</sup> signal depends on both a mobilisation of intracellular Ca<sup>2+</sup> and an influx of Ca<sup>2+</sup> ions across the plasma membrane.

Resting $[Ca^{2+}]i$ levels (nM) in $Ca^{2+}$ free media (R)	Con A (5 µg ml <sup>-1</sup> ) addition: stimulated [Ca <sup>2+</sup> ]i levels (nM) (S)	Increase in [Ca <sup>2+</sup> ]i (S/R)	Ca <sup>2+</sup> addition (0.6 mM): stimulated [Ca <sup>2+</sup> ]i levels (nM) (S')	Increase in [Ca <sup>2+</sup> ]i (S'/R)
200 155 137 164 136	230 192 190 208 178	1.15 1.24 1.39 1.27 1.31	297 232 266 247 216	1.49 1.50 1.94 1.51 1.59
Mean + SEM 158.4±11.69	199.6±8.98	1.27±0.04	251±14.04	1.61±0.09

<u>Table 4: The effect of extracellular  $Ca^{2+}$  ions on the Con A (5 µg ml<sup>-1</sup>) induced</u> increase in  $[Ca^{2+}]i$  in murine splenocytes



Figure 7: The effect of Con A (5  $\mu$ g ml<sup>-1</sup>) on the [Ca<sup>2+</sup>]i of fura-2 loaded murine splenocytes incubated in Ca<sup>2+</sup> free medium followed by subsequent addition of Ca<sup>2+</sup> ions (0.6 mM) to the extracellular environment.

The fluorescence values correspond to resting calcium levels of 155 nM in  $Ca^{2+}$  free medium which rise to 192 nM following Con A stimulation, a further increase to a  $[Ca^{2+}]i$  of 232 nM occurs upon addition of  $Ca^{2+}$  ions to the medium.

Removal of  $Ca^{2+}$  from the medium for short time periods did not appear to alter the resting  $[Ca^{2+}]i$  levels from cells incubated in medium with the cation present (compare resting levels in Tables 1 and 4). The fluorescence signal of fura-2 in splenocytes incubated for up to 10 minutes in  $Ca^{2+}$ -free physiological saline was not increased over 5-10 minutes after addition of external calcium (results not shown). This suggests that addition of  $Ca^{2+}$  ions to cells in  $Ca^{2+}$  free saline has no effect *per se* on the  $[Ca^{2+}]i$  in murine splenocytes.

Although the viability of cells appeared to be unaffected during brief periods in  $Ca^{2+}$ -free media (results not shown)  $Ca^{2+}$  depletion may have had other effects on the cells in addition to the effects on the  $Ca^{2+}$  signal. Nevertheless these studies suggest that the early  $Ca^{2+}$  signal generated by Con A stimulation in murine splenocytes is dependent on  $Ca^{2+}$  in the medium and mobilisation of  $Ca^{2+}$ from an intracellular pool.

A justifiable supposition is that the increase in intracellular calcium ion concentration probably activates the calcium regulatory protein, calmodulin (see Section 1.5.2). Consequently, the effects of two calmodulin antagonists, trifluoperazine (TFP) and M&B,13,573 (May & Baker Ltd) on the Con A induced mitotic response were investigated. As predicted both agents suppressed proliferation in a dose-dependent fashion, with approximately 50% inhibition being achieved at  $10^{-7}$ M in both cases (Figure 8); at these concentrations cell viability was unaffected (results not shown) confirming that cytotoxicity was not responsible for the diminished [<sup>3</sup>H]-thymidine incorporation. Higher concentrations of TFP ~ 50  $\mu$ M are thought to exhibit non-specific effects on protein kinase C and reduce ATP levels (see Discussion, Chapter 4 for further details).

Thus, whilst these preliminary studies demonstrate that a considerably lower TFP concentration inhibits splenocyte proliferation it is evident that the effects of calmodulin inhibitors must be interpreted cautiously. Nevertheless, the inhibition of Con A induced proliferation by calmodulin antagonists corroborates the evidence gained from the proliferative studies employing  $Ca^{2+}$  channel blockers.





Values significantly different from those of cultures without drugs are indicated by asterisks.

Thus,  $Ca^{2+}$  appears to be involved in the signal transduction pathway following Con A stimulation. However, to establish whether the early  $Ca^{2+}$  signal is obligatory (rather than a passive accompaniment to some other unknown signal) for subsequent DNA synthesis to occur it is essential to find an agent which not only blocks the early  $Ca^{2+}$  signal demonstrable in fura-2 studies but which also inhibits [<sup>3</sup>H]-thymidine incorporation following 48 hours incubation with Con A. Agents that elevate intracellular cAMP or membrane-permeant cyclic AMP analogues such as 8-bromo cyclic AMP (8-Br cAMP) and dibutyryl cyclic AMP also block the early  $Ca^{2+}$  signal and inhibit subsequent DNA synthesis in response to Con A in murine thymocytes (Hesketh et al, 1985; Metcalfe et al, 1985). Neither of these membrane permeant cyclic AMP analogues impaired the Con A induced early  $Ca^{2+}$  signal at any concentration tested (Tables 5 & 6) following a 3 minute preincubation period and only a rather unphysiological concentration of dibutyryl cAMP (10<sup>-4</sup>M) caused a significant reduction in DNA synthesis (Figure 9).

Increase in [Ca <sup>2+</sup> ]i above basal levels (considered equal to unity)	Con A (5 µg ml <sup>-1</sup> ) 1.43 1.46 1.61	) Treatment Con A (5 µg ml <sup>-1</sup> ) + Concentration (M) 8-Br cAMP		
		10 <sup>-5</sup> 10 <sup>-6</sup> 10 <sup>-7</sup>	1.55 1.40 1.57	

Table 5: The effect of preincubation with 8-Br cAMP, on the Con A induced increase in intracellular calcium concentration of murine splenocytes





Values significantly different from those of cultures with Con A alone are indicated by an asterisk.

Increase in	Con A (5 µg ml <sup>-1</sup> )	Treatment Con A (5 $\mu$ g ml <sup>-1</sup> ) + dibutyryl cAMP Concentration (M)	
	1.61	10-5	1.57
[Ca <sup>2+</sup> ]i above	1.46	10-6	1.39
basal levels (considered equal to unity)	1.61	10-7	1.58

Table 6: The effect of preincubation with dibutyryl cAMP on the Con A induced increase in intracellular calcium concentration of murine splenocytes

Since these membrane permeant cyclic AMP analogues were unable to block either the early  $Ca^{2+}$  signal or DNA synthesis except at high concentrations it seemed appropriate to investigate the effects of forskolin an agent which elevates intracellular cAMP, on Con A induced DNA synthesis. It is apparent that an increase in endogenous cAMP does indeed inhibit [<sup>3</sup>H]-thymidine incorporation in a dose-related manner (Figure 10) unless forskolin has effects on DNA synthesis unrelated to cAMP production. Forskolin was initially dissolved in DMSO but the final concentrations of DMSO had no effect on [<sup>3</sup>H]-thymidine incorporation. (Results not shown). Since fura-2 studies were not undertaken to investigate the effects of forskolin on the early Ca<sup>2+</sup> signal it has not yet been established how forskolin mediates its inhibitory effects. Forskolin alone had no effect on basal [<sup>3</sup>H]-thymidine incorporation in murine splenocytes (results not shown), however, this does not imply that cAMP is not involved in stimulus-mitosis coupling in murine splenocytes. Thus, the timing of a cAMP signal may be critical in recruiting quiescent cells to enter the cell cycle (see Section 1.5.1).

Attempts to directly correlate the early  $Ca^{2+}$  signal with the subsequent DNA synthesis measured 48 hours after Con A stimulation have proved difficult. Thus, no agent has been found which abolishes the early  $Ca^{2+}$  signal in addition to inhibiting [<sup>3</sup>H]-thymidine incorporation 48 hours later. Nevertheless, the studies with verapamil demonstrate that the early  $Ca^{2+}$  signal does indeed have some relevance for





Values significantly different from those of cultures without forskolin are indicated by an asterisk.

the initiation of DNA synthesis. An attempt was made to establish the chronology of the Ca<sup>2+</sup> requirement during Con A stimulation. If extracellular Ca<sup>2+</sup> was chelated throughout the incubation period then DNA synthesis was obliterated (Figure 11). But if  $Ca^{2+}$  presence was allowed to persist for the first four hours no inhibition occurred. If  $Ca^{2+}$  ions were restored to  $Ca^{2+}$  free medium at four hours following Con A stimulation, DNA synthesis was still severely reduced (Figure 11); these results indicate that the Ca<sup>2+</sup> requirement comes early in the incubation period. To determine the crucial period of time which elapses between Con A application and commitment to division (the 'activation' period) experiments were conducted where the mitogenic effect of Con A was systematically inhibited by the addition of a methyl mannoside to the culture medium. At a concentration previously shown to maximally inhibit Con A induced DNA synthesis (Figure 12), a methyl mannoside (50 mM) suppressed <sup>[3</sup>H]-thymidine incorporation in murine splenocytes when applied during the first 6 hours of incubation after which time it had no effect (Figure 13). Thus, receptor occupancy is necessary for the first 4-6 hours of Con A stimulation whilst extracellular calcium is only required for the first 4 hours of the 48 hours incubation period implying that in addition to the rise in  $[Ca^{2+}]$  i following stimulation - the putative mitotic signal, Con A occupancy also serves some additional purpose(s).

Many mitogens that use  $Ca^{2+}$  as a second messenger induce hydrolysis of membrane phosphoinositides resulting in a transducing mechanism that initiates a signal cascade comprising mobilisation of  $Ca^{2+}$  and the activation of protein kinase C (see Section 1.5.2). Con A is known to stimulate the breakdown of phosphatidylinositol (4,5)-bisphosphate (PtdInsP<sub>2</sub>) to inositol (1,4,5)-trisphosphate (InsP<sub>3</sub>) in murine thymocytes (Moore et al, 1984; Metcalfe et al, 1985). The two limbs of this signal pathway can be mimicked by the calcium ionophore A23187 and the phorbol ester TPA (12-0-tetradecanoyl phorbol 13-acetate) which is assumed to activate specifically protein kinase C (Metcalfe et al, 1985) (see Section 1.5.2).


Figure 11: Effect of sequential additions of EGTA and CaCl<sub>2</sub> during the 48 hour incubation on Con A (5  $\mu$ g ml<sup>-1</sup>) induced murine splenocyte proliferation.

Values significantly different from those of cultures with Con A and basal medium are indicated by asterisks. Basal medium contains 0.6 mM  $Ca^{2+}$  sequential additions of CaCl<sub>2</sub> (0.6 mM) are made to cultures in medium rendered Ca<sup>2+</sup> free by the addition of EGTA (0.7 mM).







INCUBATION TIME BEFORE ADDITION OF a METHYL MANNOSIDE (Hours)

Figure 13: Effect of sequential additions of  $\alpha$  methyl-mannoside (50 mM) during the incubation period on Con A (5 µg ml<sup>-1</sup>) induced murine splenocyte proliferation.

Values significantly different from corresponding cultures without the drug are indicated by asterisks.

Therefore proliferative studies were undertaken to determine if this transducing mechanism is involved in Con A stimulated murine splenocyte proliferation. These studies demonstrated that A23187 alone did not stimulate DNA synthesis at any concentration tested (Figure 14) whilst TPA alone caused a very modest (in comparison to Con A) increase in [<sup>3</sup>H]-thymidine incorporation (Figure 14). However, when varying concentrations of A23187 in combination with TPA (10<sup>-9</sup>M) were investigated for their mitotic potential, a synergistic interaction emerged with A23187 (10<sup>-7</sup>M) and TPA (10<sup>-9</sup>M) (Figure 14) showing a stimulation index equivalent to about 50% of the Con A (5 µg ml<sup>-1</sup>) stimulation index under identical conditions (results not shown). To elucidate whether the  $Ca^{2+}$  signal and protein kinase C activation were able to compensate for one another, lower concentrations of A23187, (10<sup>-8</sup>M and 10<sup>-9</sup>M) in conjunction with varying concentrations of TPA were tested for their proliferative actions. Neither concentration of ionophore produced a significant increase in [<sup>3</sup>H]-thymidine incorporation with any concentration of TPA (Figures 15 and 16) suggesting that the magnitude of the  $Ca^{2+}$  signal generated by ionophore and protein kinase C activation by TPA are critical in stimulating DNA synthesis.

The opportunistic comitogens, TPA  $(10^{-9}M)$  and A23187  $(10^{-7}M)$ presumably bypass the physiological pathway acting through PtdInsP<sub>3</sub> breakdown, by increasing  $[Ca^{2+}]i$  and activating PKC directly. Indeed, TPA does seem to be exerting its affects by enhancing PKC activity rather than by a non-specific membrane perturbation since the synergism could not be repeated using 4  $\alpha$ -phorbol-12,13didecanoate, a phorbol ester which has no PKC enhancing activity (Castagna et al, 1982) (Figure 17). These studies indicate that the phosphoinositide transducing mechanism may be important in murine splenocyte proliferation and that PKC activation is necessary in addition to a Ca<sup>2+</sup> signal and it is predicted that they may be compulsory secondary responses (or signals) for mitogenic stimulation by Con A.



Figure 14: Dose response curves of the effects of A23187 and TPA alone and in combination on thymidine incorporation in isolated murine splenocytes.

Values significantly different from those of the other cultures are indicated by asterisks.



Figure 15: Effects of TPA with and without A23187 ( $10^{-8}$  M) on thymidine incorporation in isolated murine splenocytes.



Figure 16: Effects of TPA with A23187  $(10^{-9}M)$  and without ionophore on thymidine incorporation in isolated murine splenocytes

Fura-2 studies substantiate the hypothesis that a  $Ca^{2+}$  signal alone is insufficient to trigger mitosis since A23187 10<sup>-7</sup>M caused an early  $Ca^{2+}$  signal of greater magnitude than Con A (Table 1) although it was not mitogenic by itself (Figure 14). The early  $Ca^{2+}$  signal was unaffected by the simultaneous addition of TPA  $(10^{-9}M)$  (Table 1 & 7) although the two acted synergistically to increase DNA

synthesis (Figure 14). Since A23187  $(10^{-7}M)$  and 4  $\alpha$ , 12, 13-phorbol didecanoate  $(10^{-9}M)$  together produced a Ca<sup>2+</sup> signal of the same proportion (Table 7) but do not act as co-mitogens (Figure 17) it appears that in addition to the early Ca<sup>2+</sup> signal there must be another signal presumably PKC activation. Neither phorbol ester had any effect on resting [Ca<sup>2+</sup>]i levels at these concentrations (results not shown).

	A23187 (10 <sup>-7</sup> M) + TPA (10 <sup>-9</sup> M)	Treatment A23187 ( $10^{-7}M$ + 4 $\alpha$ , 12, 13 phorbol didecanoate ( $10^{-9}M$ )
--	---	--

2.46

3.48

2.45

1.64

2.29

1.76

 $[Ca^{2+}]i$ 

levels

above basal

(considered equal to unity)

Table /: The effect of simultaneous administration of calcium iononhore A 22	107 1
phorbol esters on intracellular calcium concentration of calcium tonophore A25	18/ and
present esters on intracentular calcium concentrations of murine splenocytes	

Preliminary studies revealed that when the  $Ca^{2+}$  signal and the putative PKC activation generated by Con A were mimicked by appropriate combinations of A23187 and TPA the subsequent DNA synthesis was reduced by approximately 50% compared with the responses to Con A (results not shown). This implies that the lectin promoted a further unidentified signal in addition to the Ca<sup>2+</sup> signal and any PKC activation. Further studies demonstrated surprisingly, that no concentration of Con A tested, further enhanced DNA synthesis stimulated by the co-mitogens TPA  $(10^{-9}M)$  and A23187  $(10^{-7}M$  (Figure 18). Moreover the stimulation indices of the optimal concentration of Con A (5 µg ml<sup>-1</sup>) and the combination of A23187 and TPA are virtually identical (Figure 18). These results are in stark contrast to preliminary studies which suggested that Con A (5 µg ml<sup>-1</sup>) produced a much greater signal, and possibly relate to fluctuations between different batches of TPA since in later studies the concentration of TPA required to synergise with A23187 was rather variable (results not shown). Unfortunately, these ambiguous results are unable to confirm or deny the necessity of a third signal in addition to the Ca<sup>2+</sup> mobilisation and PKC activation.

There are indications in the literature that TPA at higher concentrations can interfere with receptor dependent increases in cytosolic Ca<sup>2+</sup> (Naccache et al, 1985; Staddon & Hansford, 1986; Muldoon et al, 1987). Indeed, if murine splenocytes were preincubated with varying concentrations of TPA for 3 minutes the Ca<sup>2+</sup> signal measured by fura-2 fluorescence in response to Con A was diminished by doses of  $10^{-8}$ M TPA and above (Table 8). Cells preincubated with the co-mitogenic concentration of TPA ( $10^{-9}$ M) did not demonstrate a diminution in the Ca<sup>2+</sup> signal (Table 8).



Figure 17: Effects of  $4\alpha$ , 12,13 phorbol didecanoate with and without A23187 (10<sup>-7</sup>M) on thymidine incorporation in isolated murine splenocytes

Values significantly different from other cultures are indicated by asterisks.





Values significantly different from other cultures are indicated by an asterisk.

Table 8: The effect of preincubating (3 mins) murine splenocytes with varying concentrations of TPA on the Con A induced increase in intracellular calcium concentration

	Treatr	nent		
	Con A (5 µg ml <sup>-1</sup> )	Concentration (M)	Con A (5 µg ml <sup>-1</sup> ) + TPA	
	1.61	10-9	1.58	
	1.94 1.62 1.61	10-8	1.25 1.63 1.47	
Increase in [Ca <sup>2+</sup> ]i above basal levels (considered equal to unity)	1.35 1.46 1.43 1.82 1.76 1.61	10-7	1.19 1.15 1.23 1.25 1.25 1.25 1.22	

TPA (10<sup>-7</sup>M) has a highly significant (P < 0.01) effect on the Con A induced increase in  $[Ca^{2+}]i$ .

 $4 \alpha 12,13$  phorbol-didecanoate which lacks tumorigenicity and the ability to activate PKC had no effect on the Con A induced increase in  $[Ca^{2+}]i$  at a similar concentration (Table 9), implying that the inhibition of the Ca<sup>2+</sup> signal by TPA is mediated by protein kinase C. Table 9: The effect of preincubating (3 mins) murine splenocytes with 4  $\alpha$  12.13 phorbol didecanoate on the Con A induced increase in intracellular calcium concentration

	Treatment	
	Con A (5 µg ml <sup>-1</sup> )	Con A (5 $\mu$ g ml <sup>-1</sup> ) + 4 $\alpha$ 12,13 phorbol didecanoate (10 <sup>-7</sup> M)
Increase in	1.46	1.65
[Ca <sup>2+</sup> ]i above	1.53	1.69
basal levels	1.59	1.49
(considered equal	1.76	1.75
to unity)	1.62	1.67
	1.61	1.47

Even when the preincubation time was increased to ten minutes this inactive phorbol ester had no effect on the Con A induced intracellular  $Ca^{2+}$  increase (Table 10) whilst TPA (10<sup>-7</sup>M) had a marked effect, although the Ca<sup>2+</sup> signal was never completely obliterated (Table 10), (Figure 19).

Table 10: The effect of preincubating (10 minutes) murine splenocytes with TPA (10<sup>-7</sup>M) and 4  $\alpha$  12,13 phorbol didecanoate (10<sup>-7</sup>M) on the Con A induced increase in intracellular calcium concentration

	Con A (5 $\mu$ g ml <sup>-1</sup> )	Treatment	Con A (5 $\mu$ g ml <sup>-1</sup> ) + phorbol ester (10 <sup>-7</sup> M)
Increase in [Ca <sup>2+</sup> ]i above basal levels (considered equal to unity)	1.42 1.33 1.22 1.29 1.63	ТРА	1.10 1.11 1.10 1.07 1.25
	1.42 1.29 1.65	4α12,13 phorbol didecanoate	1.40 1.27 1.65

TPA (10<sup>-7</sup>M) has a highly significant (P < 0.01) effect on the Con A induced increase in  $[Ca^{2+}]i$ 



Figure 19: Comparison of the effects of Con A (5  $\mu$ g ml<sup>-1</sup>) on the [Ca<sup>2+</sup>]i of fura-2 loaded murine splenocytes preincubated in normal medium or medium treated with TPA (10<sup>-7</sup>M) for 10 minutes.

The fluorescence values correspond to resting calcium levels of 124 nM(1) and 134 nM(2) rising to 137 nM and 190 nM respectively following Con A stimulation.

To elucidate the effects of the diminished  $Ca^{2+}$  signal on subsequent DNA synthesis, murine splenocytes were preincubated with TPA (10<sup>-7</sup>M) and other concentrations of TPA for 10 minutes prior to administration of Con A 5 µg ml<sup>-1</sup>. After 48 hours incubation DNA synthesis was increased 2 fold when compared to control cells (Con A alone) (Figure 20) following TPA treatment. Under identical conditions, the inactive phorbol ester, 4 $\alpha$  12,13 phorbol didecanoate had no effect on the Con A induced stimulation (Figure 20). This is further confirmation of the hypothesis that the Ca<sup>2+</sup> signal alone is insufficient to commit cells to division.

One of the consequences of PKC activation is the stimulation of Na<sup>+</sup>/H<sup>+</sup> exchange which leads to an increase in intracellular pH. To define a role for PKC activation in signal transduction following the interaction of lectin with its receptor, the effect of amiloride an agent used to inhibit the Na<sup>+</sup>/H<sup>+</sup> antiport (Besterman et al, 1985) on Con A stimulated DNA synthesis was investigated. Amiloride significantly inhibited [<sup>3</sup>H]-thymidine incorporation into murine splenocytes at concentrations of 10<sup>-6</sup>M and higher (Figure 21). These results suggest that the Na<sup>+</sup>/H<sup>+</sup> antiport is indeed involved in T-lymphocyte proliferation possibly as a result of PKC activation. Na<sup>+</sup>/H<sup>+</sup> exchange is in principle detectable as a cytoplasmic alkalinization, thus investigations were conducted using the pHi sensitive fluorescent probe Quene 1 (Rogers et al, 1983). No pHi increases were detected following Con A (5 µg ml<sup>-1</sup>) stimulation or as a result of application of the 'co-mitogen' TPA (10<sup>-9</sup>M) a direct activator of protein kinase C (results not shown). Although this fluorescent probe detected pHi increases following treatment with NH4Cl 5mM (see Diagram 5) no significant cytoplasmic alkanization was monitored following administration of either ligand, thus either this model system is insensitive to such pHi changes that do occur or these mitogens do not signal T cells to divide via Na<sup>+</sup>/H<sup>+</sup> exchange. It seemed appropriate therefore to repeat the experiments utilising a further pHi sensitive fluorescent probe BCECF, (bis (carboxyethyl)-carboxyfluorescein. In these studies Con A (5  $\mu$ g ml<sup>-1</sup>) induced a pHi increase which develops more slowly than the Ca<sup>2+</sup>



Figure 20: Effects of preincubation with phorbol ester (10 minutes) on the Con A (5  $\mu$ g ml<sup>-1</sup>) induced murine splenocyte proliferation

Values significantly different from those of corresponding cultures without phorbol esters are indicated by an asterisk.





Values significantly different from corresponding cultures without amiloride are indicated by an asterisk.

signal and only reaches a maximum after ~ 10 minutes (compare Diagram 4 and Figure 22). The co-mitogenic concentration of TPA ( $10^{-9}$ M) also generated a pHi increase (Figure 22) whilst physiological saline over the same period had no effect (results not shown). When cells from the same mouse were loaded with either quene 1 or BCECF, a pHi increase following Con A (5 µg ml<sup>-1</sup>) was demonstrable only in BCECF loaded splenocytes suggesting that quene 1 is an unsatisfactory fluorescent probe in this particular system. This is in agreement with earlier quene 1 studies in thymocytes (Rogers et al, 1983) and perhaps this indicator should now be considered obsolete.

The  $[Ca^{2+}]i$  and pHi responses induced by Con A stimulated T lymphocytes are presumably generated by the breakdown of PtdInsP<sub>2</sub> releasing InsP<sub>3</sub> and diacylglycerol. Protein kinase C activation is then probably responsible for the increased Na<sup>+</sup>/H<sup>+</sup> exchange leading to cytoplasmic alkalinization (see Section 1.5.2).

Many putative second messengers have been linked to T-lymphocyte activation (see Section 1.7) of these, a Ca<sup>2+</sup> signal and pHi increase are implicated in Con A stimulation of murine splenocytes. Another, second messenger, ornithine decarboxylase is thought to be involved not only in T lymphocyte proliferation but in the generation of Ca<sup>2+</sup> signals in other cell types (see Section 1.5.5). Investigations were therefore performed to determine the effects of an irreversible inhibitor of ornithine decarboxylase, DFMO ( $\alpha$ (Difluoromethyl) ornithine hydrochloride monohydrate) (a gift from Dr Tom Brown, May & Baker Ltd) on the Con A induced Ca<sup>2+</sup> signal in murine splenocytes. DFMO (5 mM), (a concentration known to irreversibly inhibit ornithine decarboxylase (Mamont et al, 1978)) was without significant effect on the Con A induced Ca<sup>2+</sup> signal (Table 11).





Fluorescence recordings demonstrate qualitative changes in the pHi of murine splenocytes stimulated with both mitogens. No attempt has been made to calibrate these values (see Chapter 2 for details).

Table 11: The effect of DFMO (5 mM) on the Con A induced increase in cytosolic  $Ca^{2+}$  concentration of murine splenocytes

	Treatment	
Ster Star	Con A (5 µg ml <sup>-1</sup> )	Con A (5 μg ml <sup>-1</sup> ) DFMO (5 mM)
Increase in [Ca <sup>2+</sup> ]i	1.27	1.30
above basal levels	1.27	1.10
(considered equal to unity)	1.31	1.20

DFMO has no significant effect on the Con A induced increase in [Ca<sup>2+</sup>]i

Furthermore, no concentration of DFMO tested inhibited [<sup>3</sup>H]-thymidine

incorporation into Con A stimulated murine splenocytes (Table 12).

Table 12: The influence of DFMO on Con A induced DNA synthesis in murinesplenocytesn = 4

Tre	atment
DFMO Concentration (M)	% of Con A (5 µg ml <sup>-1</sup> ) alone induced [ <sup>3</sup> H]- thymidine incorporation
10-6	82.83 ± 7.25
10-4	$111.55 \pm 23.68$ $123.28 \pm 21.66$
10-3	$85.65 \pm 16.19$
$5 \times 10^{-3}$	$75.32 \pm 12.69$
10-2	$96.23 \pm 16.76$

Since normal cellular growth requires polyamines and the only route to putrescine is via the enzyme ornithine decarboxylase (Pegg & McCann, 1982) it must be assumed that the inhibitor DFMO was lacking in activity and therefore the fura-2 studies should be viewed with scepticism. Having outlined as far as possible the mechanism of Con A induced murine splenocyte proliferation, the effects of known physiological modulators of immunological proliferation were investigated. An objective of this study was to discern the mechanism(s) by which sex steroids inhibit lymphocyte transformation and to determine whether it has any relevance to the modulation of the immune system *in vivo* by gonadal steroids (see Section 1.7). Thus, the ability of testosterone and certain metabolites to inhibit the stimulation induced by 5  $\mu$ g ml<sup>-1</sup> Con A in splenocytes was investigated. Testosterone is indeed an inhibitor of DNA synthesis but only at a pharmacological concentration of 10<sup>-4</sup>M (Figure 23). 5  $\alpha$  dihydrotestosterone (5  $\alpha$  DHT) an active metabolite more potent than testosterone in certain than testosterone itself (Figure 23); whereas testosterone glucuronide, a conjugated

metabolite was without effect (Figure 23).  $\alpha$ -oestradiol and  $\beta$ -oestradiol were equally effective in blocking proliferation but only at pharmacological doses of  $10^{-4}$ M (Figure 24). The effects of cholesterol, a C27 compound from which steroids are derived, failed to have an inhibitory effect on Con A stimulation at any concentration tested (Figure 25). These results clearly show that sex steroids are able to inhibit mitogen induced blast transformation albeit at high concentrations. To exclude the possibility that inhibition of DNA synthesis was due to cytotoxicity cell viability was examined in control cultures and those containing steroids at the inhibitory concentrations. 90-95% of splenocytes were viable at the end of the incubation period as assessed by trypan blue exclusion, under all culture conditions. The steroids were initially dissolved in ethanol, but the final concentrations of ethanol had no effect on [<sup>3</sup>H]-thymidine incorporation (results not shown). In order to define the mechanism of the immuno-suppressive action of sex steroids, the effect of testosterone on the Con A induced increase in [Ca<sup>2+</sup>]i measured by the Ca<sup>2+</sup> sensitive fluorescent dye fura-2 was investigated. Testosterone (10<sup>-4</sup>M) had no effect on the Ca<sup>2+</sup> signal (Table 13)



Figure 23: Effects of Testosterone,  $5\alpha$ Dihydrotesterone ( $5\alpha$ DHT) and Testosterone Glucuronide on Con A ( $5 \mu g m l^{-1}$ ) induced murine splenocyte proliferation

Values significantly different from corresponding cultures without steroids are indicated by asterisks



Figure 24: Effects of  $\alpha$  and  $\beta$  Oestradiol on Con A (5 µg ml<sup>-1</sup>) induced murine splenocyte proliferation.

Values significantly different from corresponding cultures without steroids are indicated by asterisks.



Figure 25: Effect of Cholesterol on Con A (5  $\mu$ g ml<sup>-1</sup>) induced murine splenocyte proliferation.

implying that its inhibitory action must be mediated elsewhere.

A CONTRACTOR	Treatment	
	Con A (5 µg ml <sup>-1</sup> )	Con A (5 $\mu$ g ml <sup>-1</sup> ) + Testosterone 10 <sup>-4</sup> M
Increase in	1.41	1.30
[Ca <sup>2+</sup> ]i above	1.29	1.30
basal levels	1.51	1.47
(considered equal	1.72	1.64
to unity)	1.46	1.33
	1.59	1.61

Table 13: The effect of testosterone (10<sup>-4</sup>M) on the Con A induced increase in cytosolic calcium in murine splenocytes

Investigations were undertaken to elucidate the time during the 48 hour culture period at which testosterone  $(10^{-4}M)$  exerts its abrogation of the Con A induced stimulation of murine splenocytes. Surprisingly, testosterone  $(10^{-4}M)$  caused a significant inhibition of DNA synthesis even when administered after 44 hours of incubation with Con A (5 µg ml<sup>-1</sup>) (Figure 26). In view of the high concentrations of sex steroids necessary to inhibit lymphocyte stimulation *in vitro*, it is probable that other factors contribute to the observations of sex steroid inhibited lymphocyte function in the intact animal (see Section 1.7).

Studies were therefore undertaken to investigate the effects of gonadectomy on the lymphoid tissue of mice. The spleens and thymus glands of post-pubertal TO mice were excised at four, six and nine weeks following surgery. Surgery *per se* had little effect since controls never differed significantly from sham-operated animals. In female mice no significant increases were found in the spleen index (spleen to body weight ratio) at any post-operative time (Figure 27), whilst the thymic index of ovariectomised mice was larger than control and sham-operated mice at 9 weeks after surgery (Figure 27).



Figure 26: Effect of sequential additions of Testosterone  $(10^{-4}M)$  during the incubation period on Con A (5 µg ml<sup>-1</sup>) induced murine splenocyte proliferation.

Values significantly different from those of cultures incubated with Con A alone are indicated by asterisks.



WEEKS AFTER SURGERY



Figure 27: Comparison of thymus and spleen weights of sham operated (SH), gonadectomised (GX) and control (C) female TO mice at 4, 6 and 9 weeks after surgery.

Results are expressed as spleen index (Spleen mg/g body weight) and thymus index (thymus weight mg/g body weight). Values significantly different from those of organ indices of control animals are indicated by asterisks.

In contrast, post-pubertal orchidectomy resulted in a significant increase in the spleen index at 4 weeks post-operation which had disappeared by six weeks (Figure 28). The thymic index of orchidectomised mice was significantly greater than control and sham-operated mice on all occasions following surgery (Figure 28). Thus, although the sex steroids, oestradiol and testosterone had similar effects on proliferation *in vitro*, their effects on lymphoid tissue *in vivo* are apparently quite different.

Investigations were conducted to investigate the hypothesis that gonadectomy may potentiate T cell function in mice. Thus, murine splenocytes derived from control, sham-operated and gonadectomised mice, four weeks after surgery were incubated for 48 hours with varying concentrations of the T cell mitogen Con A. Large variations were apparent in the spleen cell mitogen response for mice of both sexes. The spleen cells of castrate mice responded in a similar way as controls to lectin stimulation, 5  $\mu$ g ml<sup>-1</sup> being the concentration inducing maximum [<sup>3</sup>H]-thymidine incorporation in both cases (Figure 29). In contrast, the spleen cells of ovariectomised mice did not respond as well as spleen cells derived from sham-operated and control mice to Con A 5 µg ml<sup>-1</sup> (Figure 30). Furthermore, spleen cells from control and sham-operated mice displayed increased [<sup>3</sup>H]-thymidine incorporation at a lower concentration of Con A (1  $\mu$ g ml<sup>-1</sup>) than spleen cells from ovariectomised mice (Figure 30). Briefly then, both oestradiol and testosterone (at high concentrations) inhibited T cell proliferation in vitro; orchidectomy led to significant increases in lymphoid tissue mass particularly that of the thymus, whereas ovariectomy had relatively little effect on the thymic index and no effect at all on the spleen to body weight ratio post-surgery. Orchidectomy did not appear to potentiate T cell function since splenocytes derived from orchidectomised animals responded to mitotic stimulation by Con A in an identical fashion as cells from control animals. In contrast, these preliminary studies demonstrate that ovariectomy impaired the spleen cell mitogen response implying that oestrogen potentiates T cell function in vivo. These findings contradict reports in the literature which suggest that both sex steroids depress the cell mediated immune response (see Section 1.7).



Figure 28: Comparison of Thymus and Spleen weights of Sham operated (SH), gonadectomised (GX) and control (C) male TO mice at 4, 6 and 9 weeks after surgery.

Values significantly different from those of organ indices of control animals are indicated by asterisks. Other details as in Figure 27.



Figure 29: Comparison of dose-response curves of the effect of Con A on thymidine incorporation in splenocytes isolated from post-pubertal sham operated (SH) gonadectomized (GX) and control (C) male TO mice four weeks after surgery.

Values significantly different from those of other cultures within the same treatment group (SH, GX or C) are indicated by asterisks.



Figure 30: Comparison of dose-response curves of the effect of Con A on thymidine incorporation in splenocytes isolated from post-pubertal sham operated (SH), gonadectomised (GX) and control (C) female TO mice four weeks after surgery.

Values significantly different from those of other cultures within the same treatment group (SH, GX or C) are indicated by asterisks.

There are indications that gonadal steroids may act indirectly in the modulation of the immune system in rats thus, oestradiol is thought to depress the cell mediated immune response through the generation of thymic serum factors (Grossman et al, 1982; Grossman & Roselle, 1983; Grossman, 1985; Myers et al, 1986). To determine whether serum factors were involved in the gonadal steroid modulation of the immune system in mice, splenocytes were incubated in vitro for 48 hours in the presence of Con A (2.5  $\mu$ g ml<sup>-1</sup> or 5  $\mu$ g ml<sup>-1</sup>), tissue culture media and 20% specific mouse serum fractions and pulse labelled with tritated thymidine. Mouse serum fractions were prepared from control, gonadectomised and sham-operated animals. Remarkably, splenocytes incubated with mouse sera from all sources failed to respond to the mitogenic concentration of Con A in the accustomed way. Thus, splenocytes derived from the same animal cultured in 20% mouse serum demonstrated a stimulation index of <2 in response to Con A 5  $\mu$ g ml<sup>-1</sup> compared with a mean stimulation index of 67.59 ( $\pm$  38.20 SEM n = 4) when incubated with the same concentration of lectin but with 10% foetal calf serum in the medium (Figure 31). When splenocytes derived from male animals were incubated with the different male serum fractions no significant differences between the serum groups were observed upon addition of Con A, all the stimulation indices were approximately 1 (ie. no stimulation at all). In contrast female control sera significantly enhanced DNA synthesis in response to Con A (2  $\mu$ g ml<sup>-1</sup>) in splenocytes derived from female animals (Figure 31), this adds further support to earlier indications that oestrogen potentiates T cell function.

To ascertain whether these surprising findings indicate that mouse serum fails to support lymphocyte proliferation or whether the concentration of 20% mouse serum was too high and potentially toxic, further investigations were conducted. To rule out the possibility that high concentrations of serum might be cytotoxic, splenocytes (derived from male animals) were incubated with Con A (5  $\mu$ g ml<sup>-1</sup>) and varying concentrations of foetal calf serum. The combination of 10% foetal calf serum with Con A (5  $\mu$ g ml<sup>-1</sup>) stimulated the greatest [<sup>3</sup>H]-thymidine incorporation (Figure

175



Figure 31: Comparison of the effects of control (C), gonadectomized (GX) and sham operated (SH) male and female mouse serum on normal splenocytes (isolated from male mice) incubated with Con A (2  $\mu$ g ml<sup>-1</sup> or 5  $\mu$ g ml<sup>-1</sup>)

Sera were prepared from the various groups of mice by collecting the blood by cardiac puncture, allowing it to clot at room temperature. It was centrifuged, decanted and stored at  $-20^{\circ}$ C. Before use, the sera was thawed and added to medium at a concentration of 20% and sterilised by filtration with 0.45 micron filter. Values significantly different from those of corresponding cultures incubated with the same concentration of Con A are indicated by an asterisk.

32), however splenocytes incubated with a similar concentration of Con A but with 30% foetal calf serum still exhibited a much higher stimulation index than with 20% mouse serum (Figures 31 and 31). Furthermore, no concentration of normal mouse serum tested in combination with Con A ( $5 \mu g m l^{-1}$ ) was able to support the enhanced DNA synthesis in murine splenocytes, normally achieved at this lectin concentration (Figure 32). Intriguingly not only does mouse serum not support splenocyte proliferation but it actually inhibits in a dose-dependent manner DNA synthesis in murine splenocytes stimulated with Con A ( $5 \mu g m l^{-1}$ ) in the presence of 10% foetal calf serum (Figure 32). Cell viability was examined in control cultures and those containing mouse serum at the above concentrations. 90-95% of splenocytes were viable at the end of the incubation period as assessed by trypan blue exclusion, under all culture conditions. Unfortunately, these studies with mouse serum fractions although interesting, are unable to confirm or deny the presence of thymic serum factors which might mediate the effects of gonadal steroids on the immune system.

Although the proliferation of lymphocytes in response to plant lectins is often taken as a model for the proliferative aspect of immune responses (see Section 1.7), it seemed appropriate to investigate the mechanism of action of a more physiological growth factor, Interleukin 2. The mechanism by which quiescent T cells are activated incorporates the following components, the T cell receptor recognises the antigen MHC encoded protein complex on the antigen presenting macrophage, this bonding stimulates the release of IL1 by the macrophage and that in turn activates the T cell which expresses IL2 receptors. The activated blasts undergo proliferative expansion under the influence of IL2 produced by another activated T cell sub-population (see Section 1.7). Stimulation of murine splenocytes with mitogenic plant lectins induces a subpopulation of lymphocytes to become responsive to IL2. Hence, preparations of lectin activated cells can then be used to investigate the mechanism of IL2 induced proliferation (Gearing et al, 1985). Thus, murine splenocytes which had been activated with Con A (Con A blasts) (see Section 2.2.5 for details) were incubated with varying concentrations of recombinant human IL2.



Figure 32: Comparison of the effects of varying concentrations of foetal calf and mouse serum on Con A (5  $\mu$ g ml<sup>-1</sup>) induced murine splenocyte proliferation, and the effect of mouse serum on thymidine incorporation in splenocytes incubated with Con A (5  $\mu$ g ml<sup>-1</sup>) and 10% foetal calf serum (FCS).

Normal mouse serum was prepared from male mice as described in Figure 30. Values significantly different from those of cultures within the same treatment group are indicated by asterisks.

This lymphokine stimulated DNA synthesis in a dose dependent manner with  $[{}^{3}H]$ -thymidine incorporation into the Con A blasts reaching a plateau at a concentration of 100 U ml<sup>-1</sup> (Figure 33). In the absence of IL2, incorporation of tritiated thymidine into Con A blasts was negligible. Nifedipine a calcium channel blocker suppressed IL2 (100 U ml<sup>-1</sup>) induced proliferation at a concentration of  $10^{-5}M$  (Figure 34). Cell viability in control cultures and those containing the drug at this concentration remained at 90-95%. This implies that calcium ions must enter the cytosol from the medium before proliferation can occur, unless this calcium antagonist has effects on proliferation unrelated to calcium.

Use of the calcium sensitive fluorescent probe fura-2 indicated that Con A blasts had an intracellular calcium concentration of 208 nM ( $\pm$  26 SEM n = 6) compared with quiescent lymphocytes which had basal levels of 186 nM ( $\pm$  10 SEM n = 39). Following addition of 1L2 (100 U ml<sup>-1</sup>) the [Ca<sup>2+</sup>]i appeared to rise (see Figure 35 for a typical recording), however addition of the same volume of physiological saline produced an identical effect (Figure 32). These findings suggest that Con A blasts are susceptible to pronounced and rapid fura-2 leakage and it is therefore impossible to determine whether or not IL2 (100 U ml<sup>-1</sup>) generates a Ca<sup>2+</sup> signal.

The effect of a calmodulin antagonist TFP on IL2 induced proliferation of Con A blasts was investigated since there are indications that IL2 might signal cells to divide by generating a  $Ca^{2+}$  signal (see Section 1.7). TFP (10<sup>-5</sup>M) significantly blocked proliferation of murine Con A blasts (Figure 36), this concentration of TFP did not affect cell viability.

The role of PKC activation in IL2 stimulation is controversial (section 1.7) it seemed appropriate therefore to investigate the consequences of treatment with amiloride, an agent used to inhibit the Na<sup>+</sup>/H<sup>+</sup> antiport (believed to be activated by PKC). Amiloride (10<sup>-5</sup>M) significantly inhibited [<sup>3</sup>H]-thymidine incorporation into Con A blasts stimulated with IL2 (Figure 37).



Figure 33: Dose-response curve of the effect of recombinant Human Interleukin 2 on thymidine incorporation in 'Con A blasts'.

Values significantly different from those of other cultures are indicated by asterisks.


Figure 34: Effect of Nifedipine on the IL2 (100 U ml<sup>-1</sup>) induced 'Con A blast' proliferation

The level of thymidine incorporation in cultures stimulated by IL2 (100 U ml<sup>-1</sup>) alone is taken as 100% and the incorporation in cultures incubated with IL2 (100 U ml<sup>-1</sup>) plus various concentrations of nifedipine expressed as a percentage thereof.

Values significantly different from those of corresponding cultures without the drug are indicated by asterisks.



Figure 35: Comparison of the effects of IL2 (100 U ml<sup>-1</sup>) and physiological saline addition upon the  $[Ca^{2+}]i$  of fura-2 loaded Con A blasts.

The fluorescence values correspond to resting calcium levels of 172 nM in the upper trace and 224 nM in the lower trace.





Values significantly different from those of corresponding cultures without the drug are indicated by asterisks.



Figure 37: Effect of Amiloride on the IL2 (100 U ml<sup>-1</sup>) induced proliferation of 'Con A blasts'.

Values significantly different from those of corresponding cultures incubated without the drug are indicated by asterisk(s).

An objective of this study has been to define the mechanism(s) of sex steroid modulation of the immune system. Thus, the effects of sex steroids on the IL2 induced proliferation of Con A blasts were investigated. The hypothesis being that more physiological concentrations of sex steroids might inhibit proliferation induced by this T cell growth factor as opposed to the pharmacological concentrations necessary to block lectin stimulated growth. However, oestradiol and testoesterone still only inhibited DNA synthesis at high concentrations (Figure 38). Nevertheless, this inhibition was not due to cytotoxicity since cell viability was between 90-95% in control cultures and cultures treated with steroids.

Interleukin 1 can stimulate the proliferation of thymocytes in the presence of low amounts of mitogenic lectin (Praetkau et al, 1976). The proliferation is believed to be due to IL2 released by IL1 stimulated T-cells. There are at least two members of the IL1 family, IL1 $\alpha$  and IL1 $\beta$  (March et al, 1985), and both of these lymphokines were investigated for their mitotic potential in murine splenocytes alone and in the presence of Con A (0.35 µg ml<sup>-1</sup>) for 72 hours. All concentrations of IL1 $\alpha$  tested promoted a small increase in DNA synthesis (Figure 39). Administration of varying doses of this lymphokine with Con A (0.35 µg ml<sup>-1</sup>) had no additive effect on the [<sup>3</sup>H]-thymidine incorporation stimulated by lectin alone (Figure 39). IL1 $\beta$  induced a significant, albeit small increment in DNA synthesis at a concentration of 10<sup>-9</sup>g ml<sup>-1</sup> (Figure 40). No concentration of IL1 $\beta$  significantly enhanced the tritiated thymidine incorporation provoked by Con A (0.35 µg ml<sup>-1</sup>) alone (Figure 40).

Investigations designed to elucidate the nature of sex steroid modulation of immune function have thus far concentrated on the effects of gonadal steroids on T lymphocyte activity. It seemed appropriate therefore to investigate the actions of sex steroids on B cell function. Lipopolysaccharide, a bacterial endotoxin which specifically stimulates B cells to divide (Greaves and Janossy, 1972), increases DNA synthesis in murine splenocytes with maximum stimulation occurring at a dose of 25  $\mu$ g ml<sup>-1</sup> (Figure 41). Preliminary studies demonstrated that oestradiol inhibits B cell

185



Figure 38: Effect of sex steroids on the IL2 (100 U ml<sup>-1</sup>) induced proliferation of 'Con A blasts'.

Values significantly different from those of corresponding cultures incubated without steroids are indicated by asterisks.





In this experiment splenocytes were cultured for 72 hours, <sup>3</sup>H-thymidine being present during the last 4 hours of incubation.



Figure 40: Comparison of the effect of Interleukin 1 ß with and without Con A (0.35  $\mu$ g ml<sup>-1</sup>) on thymidine incorporation in isolated murine splenocytes.

Values significantly different from those of other cultures within the same treatment group are indicated by an asterisk. Other details as in Figure 39.



Figure 41: Dose response curve of the effect of Lipopolysaccharide (LPS) on thymidine incorporation in isolated murine splenocytes.

Values significantly different from those of other cultures are indicated by an asterisk.





The level of thymidine incorporation in cultures stimulated by LPS alone is taken as 100% and the incorporation in cultures incubated with LPS plus various concentrations of steroid expressed as a percentage thereof. Values significantly different from those of cultures incubated without steroids are indicated by asterisks.

proliferation at a pharmacological concentration of  $10^{-4}$ M (Figure 42) whilst testosterone ( $10^{-5}$ M) significantly inhibits DNA synthesis in murine splenocytes stimulated with LPS ( $25 \ \mu g \ ml^{-1}$ ) (Figure 42).

The generation and possible interaction of the putative intracellular signals involved in T lymphocyte proliferation stimulated by the plant lectin, Con A and the lymphokine, Interleukin 2 will be discussed in the following section in the light of these experimental findings. The possible mechanism(s) of sex steroid modulation of lymphoid cell function will also be examined with regard to the evidence presented here.

## CHAPTER FOUR DISCUSSION

Cell-mediated immunity provides the main defence against intracellular organisms. The mechanism by which resting T cells are activated in an antigen specific response is believed to incorporate the following steps; T cell receptors interact with antigen in association with Class II products of the major histocompatibility complex on the surface of the antigen-presenting macrophages. The combination of antigen and macrophage-derived interleukin 1 (IL1) activate the T cells which express interleukin 2 (IL2) receptors and subsequently undergo proliferative expansion, under the influence of IL2 secreted by another activated T cell sub-population. The two arms of the cell-mediated response, generation of cytotoxic cells and the release of soluble factors (the lymphokines) are mediated by different T cell subsets, the cytotoxic T cells and helper T cells respectively (see Section 1.7). Preceding activation, T cells are resting in G<sub>0</sub>, after interaction with antigen and IL1, intracellular biochemical responses occur that promote entry into the G1 phase rendering them responsive to an additional signal(s) (IL2) which enables cells to traverse from  $G_1$  into the S phase (DNA synthesis) and progress into mitosis. Despite extensive study, the mechanism(s) by which membrane signals are transduced into the nucleus and thence regulate the cascade of activation and proliferation steps, awaits delineation. Lectin stimulation of T lymphocytes, often used as a model of the proliferative aspect of the immune response has been investigated here in an attempt to gain a better understanding of this complex phenomenon.

In accordance with other studies, Con A, a T cell mitogen enhanced DNA synthesis in murine splenocytes (Figure 1). Mitogen-induced stimulation of T-lymphocytes increased with Con A concentrations up to 5  $\mu$ g ml<sup>-1</sup> and then

decreased with increasing concentrations of Con A. Two hypotheses have been suggested to account for this inhibitory effect of supraoptimal lectin concentration. Thus, rapid capping and internalization of mitogen receptors occurs such that receptors are removed from the cell surface prematurely before commitment to the S phase has occurred (Pozzan et al, 1981). Alternatively or in addition, supra-optimal concentrations are thought to bind to proteins (other than mitogen receptors) on the membrane, thereby blocking essential surface-modulating activity controlled by the cytoskeleton (McClain & Edelman, 1976).

T lymphocyte proliferation induced by the optimal mitogenic concentration of Con A, 5 µg ml<sup>-1</sup> was significantly inhibited by three specific calcium antagonists, verapamil (5 x 10<sup>-5</sup>M), nifedipine (10<sup>-6</sup>M) and diltiazem (10<sup>-4</sup>M), (Figures 3, 4 and 5 respectively), suggesting that a  $Ca^{2+}$  influx is obligatory in stimulus-mitosis coupling. Calcium antagonists are widely believed to block Ca<sup>2+</sup> entry into cells by interacting with voltage-operated  $Ca^{2+}$  channels (VOC's) (Spedding, 1987). The concept that VOC's are more sensitive to calcium channel blockers than ROC's (receptor-operated channels) is supported by the finding that contractions of vascular smooth muscle induced by calcium or potassium depolarization are potently inhibited by calcium antagonists whereas contractions induced by hormones such as noradrenaline are refractory to these drugs (Bolton, 1979; Cauvin et al, 1983). In contrast, several other studies indicate that in some tissues the ROC is more sensitive to calcium entry blockers than the VOC. Moreover, it appears that the ROC may vary from tissue to tissue in its sensitivity to calcium channel blockers (Van Breeman and Siegel, 1980). Thus, the ROC's of the cerebral circulation appear to be more susceptible to the action of calcium antagonists than those of the mesenteric arteries (Towart, 1981; Cauvin et al, 1983). Thus, the widely held belief that VOC's are more sensitive than ROC's to calcium antagonists does not always apply.

There are believed to be three classes of  $Ca^{2+}$  channels, a small transient current is carried by the T (for transient) channel; a larger current is carried by the N (neuronal) channel (only present in certain neurones) and a third class of channel the L (long) channel, a slowly inactivating channel. Of the three classes of Ca<sup>2+</sup> channel only the L channel is thought to be susceptible to the classical calcium antagonists (Spedding, 1987). Because these organic  $Ca^{2+}$  channel blockers exert their actions at low concentrations and exhibit stereospecificity it appears likely that they are recognised by specific structures of the  $Ca^{2+}$  channel. The diversity of the molecular structures of Ca<sup>2+</sup> channel blockers is consistent with differing modes and sites of action. Thus, it is believed that there are three different recognition sites on the L channel, one each for verapamil, diltiazem and the dihydropyridines (such as nifedipine) (Scharwtz and Triggle, 1984; Katz, 1985). Nevertheless, this does not necessarily imply that Con A cannot be exerting its effect by activating ROC's. Thus, the possibility exists that these calcium channel blockers inhibit Ca<sup>2+</sup> entry via ROC or that ROC activation by Con A may indirectly lead to partial cellular depolarisation which automatically triggers the opening of VOC's (Cavero and Spedding, 1983). Unfortunately, radioisotopic  ${}^{45}Ca^{2+}$  studies were unable to monitor this putative lectin-induced Ca<sup>2+</sup> influx (see Results Section, Chapter 3). Although an increase in 45Ca<sup>2+</sup> uptake has been demonstrated in human peripheral blood lymphocytes stimulated with plant lectins (Whitney & Sutherland, 1972; Freedman et al, 1975; Greene et al, 1976) by a variety of methods, a more recent report indicates that Con A did not augment <sup>45</sup>Ca<sup>2+</sup> uptake into T cells and even the calcium ionophore A23187 (1 µg ml<sup>-1</sup>) only caused a slight increase in Ca<sup>2+</sup> influx (at a concentration which was optimum for stimulating DNA synthesis in human Tonsillar T cells) (Komada et al, 1985). These workers suggest that even though enhancement of  $Ca^{2+}$  uptake may not be an absolute requirement, intracellular Ca<sup>2+</sup> may act as an important messenger in T cell mitogenesis. Thus, it is possible that either the microcentrifugation technique used

in this study is inappropriate for estimating  ${}^{45}Ca^{2+}$  influxes or that the mitogenic dose of lectin, 5 µg ml<sup>-1</sup> and co-mitogenic concentration of A23187, 10<sup>-7</sup>M do not enhance  ${}^{45}Ca^{2+}$  uptake in murine splenocytes. However the data obtained from calcium antagonist studies make the first possibility more likely.

Rapid changes in the intracellular free calcium ion concentration  $[Ca^{2+}]i$  in murine splenocytes certainly were detected by the fluorescent probe fura-2 in response to Con A (5  $\mu$ g ml<sup>-1</sup>). The resting [Ca<sup>2+</sup>]i in these cells was approximately 185 nM. Following lectin stimulation the  $[Ca^{2+}]$  i rose rapidly to reach a maximal value by 1 1/2 minutes; the [Ca<sup>2+</sup>]i increased 1.45 fold over resting levels. Most other studies using fluorescent probes report a value of approximately 100-120 nM for the cytoplasmic free calcium concentration of resting lymphocytes (murine thymocytes) (Tsien et al, 1982; Metcalfe et al, 1985; Tsien & Poenie, 1986) which rose to between 150 nM and 200 nM following Con A treatment. The higher value for resting [Ca<sup>2+</sup>]i in splenocytes indicated in this study can probably be attributed to slight fura-2 leakage (see Section 2.3.1). The  $Ca^{2+}$  signal in response to Con A was partially diminished when  $Ca^{2+}$  was absent from the extracellular medium (Table 4). Subsequent  $Ca^{2+}$ addition to the media induced a further increase in  $[Ca^{2+}]$  i suggesting that the Con A provoked rise in  $[Ca^{2+}]i$  was due to both a mobilisation of intracellular  $Ca^{2+}ions$  and an influx of  $Ca^{2+}$  ions across the plasma membrane (or possibly an inhibition of  $Ca^{2+}$ extrusion). This Ca<sup>2+</sup> signal persisted for at least one hour and possibly longer; indeed Hesketh and colleagues (1983) report that the lectin induced Ca<sup>2+</sup> signal only returns to resting values after approximately 24 hours following the same time course as cap formation by Con A on the lymphocyte cell surface. This implies that the  $Ca^{2+}$ signal requires a continuous interaction of the mitogen with its receptor. The effects on the calcium signal of two calcium antagonists at concentrations known to significantly inhibit Con A induced DNA synthesis gave equivocal results (Tables 2 and 3). Thus, verapamil (5 x  $10^{-5}$ M) significantly diminished the lectin induced

 $[Ca^{2+}]$  i increase, suggesting that the  $Ca^{2+}$  influx component of the early  $Ca^{2+}$  signal may well be important in the molecular events following ligand-receptor interaction which ultimately lead to DNA synthesis. In contrast, diltiazem at a concentration (10<sup>-4</sup>M) known to suppress [<sup>3</sup>H]-thymidine incorporation into Con A stimulated splenocytes) had no effect on the early  $Ca^{2+}$  signal. However diltiazem might feasibly impair the persistent  $Ca^{2+}$  signal, since its effect on  $[Ca^{2+}]i$  was only monitored for approximately 5 minutes following lectin treatment. Fura-2 studies thus demonstrate that lectin mediated increases in [Ca<sup>2+</sup>]i were remarkedly reduced but not abolished in the absence of extracellular  $Ca^{2+}$  or by the addition of a calcium channel blocker, verapamil. This is in agreement with other recent studies (Mills et al, 1985; Hesketh et al, 1985; Gelfand et al, 1986) using murine thymocytes and human T lymphocytes. This implies that following binding of Con A murine splenocytes, both intra- and extracellular mediated changes in  $Ca^{2+}$  contribute to the early  $Ca^{2+}$  signal. Moreover, this Ca<sup>2+</sup> signal appears obligatory for subsequent DNA synthesis as witnessed by the abrogation of enhanced [<sup>3</sup>H]-thymidine incorporation in Con A stimulated splenocytes by the calcium channel blockers.

It is not unreasonable to assume that if it is indeed this early rise in intracellular calcium which initiates the proliferogenic response to Con A then the next step in the sequence of events leading to DNA synthesis and ultimately mitosis is the association of  $Ca^{2+}$  with its specific binding protein, calmodulin. Therefore, if calmodulin is activated during the mitotic response of T lymphocytes to Con A then calmodulin inhibitors such as trifluoperazine (TFP) and M&B 13753 should compromise this response. This was confirmed when the lectin-enhanced DNA synthesis was inhibited by treatment with these calmodulin antagonists ( $10^{-7}M$ ) (Figure 8). However, to attribute any effects of these drugs to the inhibition of calmodulin it is necessary to ascertain that the action results from inactivation of

metabolic effect (Cheung, 1984). Calmodulin antagonists have been employed to identify calmodulin regulation of specific biological processes and indeed many different physiological processes have been inhibited following their use. Rather than indicating calmodulin's ability to regulate diverse enzymes this may infer that calmodulin antagonists have multiple sites of action (Veigl et al, 1984a). Thus, TFP ( $\simeq$  50 µM) reduces lymphocyte ATP levels, so that any enzymic process that involves hydrolysis of ATP will be sensitive to these changes irrespective of any calmodulin regulation of the process (Corps et al, 1982). Additionally TFP (~ 50 µM) and other phenothiazines inhibit protein kinase C as well as inhibiting the function of Ca<sup>2+</sup> calmodulin, and therefore cannot be considered as specific inhibitors of either activity (Schwantke et al, 1985). However, there is evidence that at least some of the effects of TFP are specific to calmodulin. Thus, the growth of a macrophage like cell line (J774) is normally inhibited by TFP whilst a variant of this line lacking in a specific calmodulin-dependent binding protein is unaffected by TFP although it has a normal calmodulin content. The enzymatic activity associated with the calmodulin binding protein, if any, is unknown (Speaker et al, 1982). Since lower concentrations of TFP inhibited lectin induced DNA synthesis in murine splenocytes than those indicated in the literature it seems likely that calmodulin is implicated in the signal-transduction pathway. Thus, calcium via the formation of calci-calmodulin conformers appears to have an obligatory role in triggering the cascade of mitotic events linked to Con A stimulation. Regulation of cell division by calmodulin is thought to be related to the effective concentration of the protein in the cell at any given time. Thus, it is believed that in certain cell types (see Section 1.5.2) an elevation in calmodulin concentration occurs at the G1/S boundary of the cell cycle which is essential for progression of the cells into the S phase. Regulation of the cytoskeleton and protein phosphorylation, particularly histone phosphorylation by calmodulin might be crucial in the G1/S transition (Chafouleas and Means, 1982).

The [Ca<sup>2+</sup>]i response to Con A in thymocytes was antagonised by membrane-permeant cyclic AMP analogues or agents that elevate intracellular cyclic AMP (Hesketh et al, 1985). However, in the present studies using murine splenocytes, neither dibutyryl cAMP nor 8-bromo cAMP impaired the Con A induced Ca<sup>2+</sup> signal at any concentration tested. Moreover, only rather unphysiological concentrations of dibutyryl cAMP (10<sup>-4</sup>M) significantly inhibited Con A induced DNA synthesis (Figure 9). However, forskolin, an agent which directly activates adenylate cyclase in a variety of cells (Seamon et al, 1981), inhibited lectin stimulated [<sup>3</sup>H]-thymidine incorporation in a dose-dependent fashion (Figure 10). There are many reports suggesting that cAMP is indeed an inhibitory regulator of cell proliferation (see Section 1.5.1). Moreover, the inhibition of growth by cyclic AMP has been attributed to its ability to prevent excess calcium from entering cells at a critical growth restriction point in G1 phase of the cell cycle. Thus, Con A activation of murine thymocytes requires  $Ca^{2+}$  and the influx of  $Ca^{2+}$  into Con A treated cells is inhibited by dibutyryl cAMP, as is DNA synthesis (Tsien et al, 1982; Hesketh et al, 1985). Further support for this hypothesis is that cyclic AMP also reduces  $Ca^{2+}$ availability in rat thymocytes by stimulating  $Ca^{2+}$  extrusion whilst the calcium ionophore A23187 can bypass the effect of cAMP and induce thymocyte proliferation (Ralph, 1983). These observations imply that normally cAMP acts to decrease intracellular  $Ca^{2+}$ . In this study the membrane-permeant cAMP analogues apparently had no effect on the early  $Ca^{2+}$  signal induced by Con A as monitored by the fluorescent probe fura-2 although it is impossible to know what their effect on the persistent signal might have been. However, an objection to the studies in which cAMP is implicated as a  $Ca^{2+}$  antagonist are that the inhibitory effects on the  $Ca^{2+}$ signal were elicited by high concentrations of cAMP analogues (5-10 mM) (Hesketh et al, 1985) and could therefore be described as non-specific. Certainly it is unlikely that endogenously generated cAMP ever reaches such proportions.

An alternative hypothesis to explain the inhibitory effects of forskolin on Con A induced murine splenocyte proliferation is that rather than reducing the  $Ca^{2+}$  availability, cAMP formation might antagonise protein kinase C activity. Indeed, the inhibitory effect of forskolin on human platelet secretion and aggregation induced by 5-hydroxtryptamine is attributed to its ability to inhibit protein kinase C activity (De Chaffoy De Courcelles et al, 1987).

The inhibitory role of cAMP has recently been challenged on the basis that cAMP elevating agents act synergistically with other mitogens to stimulate DNA synthesis in 3T3 cells (Rozengurt, 1985) (see Section 1.5.1). Murine splenocytes incubated for 48 hours with forskolin alone showed no enhanced [<sup>3</sup>H]-thymidine incorporation, but these preliminary studies do not imply that cAMP is irrelevant in stimulus-mitosis coupling. Thus, it is possible that once cells have passed through the restriction point in G<sub>1</sub> (possibly Ca<sup>2+</sup> dependent) then a cAMP signal might activate certain metabolic processes, thereby enhancing cell cycle progression (Ralph, 1983). Furthermore, it is possible that after G<sub>1</sub> phase growth has commenced, subsequent cAMP-dependent events may be obligatory in certain cells (including lymphocytes) before DNA synthesis can eventually occur. In addition, a later cAMP surge in G<sub>2</sub> may also be required before the cell is able to progress into mitosis (Whitfield et al, 1987).

Whilst the role of cAMP remains unclear the early  $Ca^{2+}$  signal demonstrated by the fluorescent probe fura-2 is strongly implicated in the signal-transduction pathway ultimately leading to DNA synthesis in lectin stimulated murine splenocytes. The early  $Ca^{2+}$  signal appears to involve both mobilisation of intracellular  $Ca^{2+}$  ions and also depends on extracellular calcium. There is a large body of evidence which demonstrates that mobilisation of  $Ca^{2+}$  in the cytoplasm of cells stimulated by agents acting via cell surface receptors is nearly always associated

with the hydrolysis of phosphatidylinositol 4,5, bisphosphate (PIP2) to yield inositol trisphosphate (InsP<sub>3</sub>) and diacylglycerol (DG). InsP<sub>3</sub> can induce the release of  $Ca^{2+}$ ions from the endoplasmic reticulum (Berridge and Irvine, 1984; Spat, 1986; Taylor, 1988). However, the release of calcium from intracellular stores is only one component of the early  $Ca^{2+}$  signal in lectin-stimulated splenocytes. When cells were suspended in  $Ca^{2+}$  free medium, the rise in  $[Ca^{2+}]i$  was significantly reduced (Figure 7), suggesting that part of the intracellular  $[Ca^{2+}]i$  increment originates from the extracellular medium. The transfer of extracellular calcium to the cell cytosol upon activation does not necessarily imply the existence of a calcium channel. Thus, a possible explanation is that calcium constantly leaks into cells down the inwardly directed concentration gradient (1mM  $\rightarrow$  100nM) and that Ca<sup>2+</sup> is constantly pumped out. Any inhibition of the 'Ca<sup>2+</sup> pump' upon lectin stimulation would induce an increase in [Ca<sup>2+</sup>]i without any channel opening (Linch et al, 1987). If this hypothesis is indeed true for murine splenocytes, then fura-2 loaded splenocytes incubated in  $Ca^{2+}$  free medium should have demonstrated an increase in  $[Ca^{2+}]i$  upon  $Ca^{2+}$  restoration to match that invoked by restoration of  $Ca^{2+}$  ions to lectin-stimulated splenocytes in  $Ca^{2+}$  free medium (Figure 7). Since this wasn't the case it seems likely that Con A stimulation of these cells does involve a  $Ca^{2+}$  influx, rather than a 'leak', which serves as one component of the early  $Ca^{2+}$  signal. There are several explanations advanced to explain how hydrolysis of phospholipids may lead to Ca<sup>2+</sup> influx into cells (see Section 1.5.2). Thus, phosphatidic acid formed from DG might function as a  $Ca^{2+}$  ionophore, (Putney et al, 1980), or an increase in  $Na^+/Ca^{2+}$ exchange might be stimulated leading to Ca<sup>2+</sup> uptake dependent on the Na<sup>+</sup> gradient (Philipson & Nishimoto, 1984). A more recent hypothesis suggests that InsP4 (formed by phosphorylation of 1,4,5 InsP<sub>3</sub>) might first regulate transfer of Ca<sup>2+</sup> from the extracellular space to the intracellular ER pool. Subsequently InsP3 would then

induce mobilisation of this  $Ca^{2+}$  from the ER into the cytosol (Irvine & Moor, 1986; Taylor, 1987; Imboden, 1988) (see Section 1.5.2.).

The two limbs of the inositol lipid signalling system can be mimicked by calcium ionophore A23187 and the phorbol ester TPA. Studies undertaken to establish a possible role for this signal transduction pathway in stimulated splenocytes demonstrated that A23187 ( $10^{-7}$ M) and TPA ( $10^{-9}$ M) did indeed act synergistically to induce DNA synthesis in these cells (Figure 14). Since the synergy could not be

repeated with  $4\alpha$ -phorbol-12,13-didecanoate, a phorbol ester without protein kinase C (PKC) enhancing activity (Castagna et al, 1982) it appears that protein kinase C activation may also be implicated in T-lymphocyte proliferation. Since neither TPA nor A23187 alone were able to substantially enhance [<sup>3</sup>H]-thymidine incorporation (Figure 14) it is apparent that both PKC activation and an increase in [Ca<sup>2+</sup>]i are necessary signals in murine splenocyte proliferation. In this model, it is believed that the co-mitogens by-pass the physiological pathway acting through PIP<sub>2</sub> hydrolysis, by increasing [Ca<sup>2+</sup>]i and activating PKC directly. These are assumed to be obligatory secondary responses for lectin stimulation. Indeed, Con A stimulates the breakdown of PIP<sub>2</sub> in murine thymocytes (Moore et al, 1984) leading to these dual signals.

Protein kinase C and  $Ca^{2+}$  are now believed to be synergistic signals in the control of many biological responses in a variety of cell types (Downes and Michell, 1985) although such activation is not a universal phenomenon. For example,  $Ca^{2+}$  alone controls K<sup>+</sup> efflux from parotid acinar cells (Putney et al, 1984) and stimulation of oxygen radical production in neutrophils appears to relate exclusively to protein kinase C activation (Di Virgilio et al, 1984). However, in murine splenocytes the hypothesis that both signals are obligatory for mitosis to occur is substantiated by studies which demonstrated that a  $Ca^{2+}$  signal alone was probably insufficient to

induce DNA synthesis. Thus, A23187 ( $10^{-7}$ M) alone, produced an early Ca<sup>2+</sup> signal of greater magnitude than Con A but was not itself mitogenic (Figure 14).

Furthermore this  $Ca^{2+}$  signal was unaffected by either TPA (10<sup>-9</sup>M) or 4 $\alpha$ 12,13 phorbol didecanoate (10<sup>-9</sup>M) although only TPA which activates PKC was able to synergise with A23187 (Table 7, Figures 14 and 17). These results imply that in addition to the early Ca<sup>2+</sup> signal there must be another signal presumably PKC activation. Preliminary studies indicated that the co-mitogens A23187 and TPA did not enhance [<sup>3</sup>H]-thymidine incorporation to the same extent as Con A, implying that the lectin promotes a further unidentified signal in addition to the  $Ca^{2+}$  and PKC signals. Indeed, studies with macrophage-depleted human peripheral lymphocytes demonstrate that these two pathways, protein kinase C activation and Ca<sup>2+</sup> mobilisation are both essential and synergistic for promoting DNA synthesis. However, in addition to the appropriate concentrations of TPA and A23187 to mimick these signals, a small quantity of plant lectin, phytohaemagglutinin (PHA) is also required for maximal DNA synthesis to occur (Kaibuchi et al, 1985). Studies with murine splenocytes in contrast demonstrated that no concentration of Con A tested further enhanced DNA synthesis stimulated by the co-mitogens TPA and A23187 (Figure 18). Moreover, in these later studies, the stimulation achieved by A23187 and TPA matched that of Con A (5 µg ml<sup>-1</sup>) implying that PKC activation and an increase in  $[Ca^{2+}]$  are the only signals involved in lectin stimulation which mediate the cellular responses leading to DNA synthesis. However, earlier studies demonstrated that receptor occupancy of Con A was necessary for 4-6 hours whilst extracellular Ca<sup>2+</sup> was only required for the first 4 hours (Figures 13 and 11 respectively). This implies that receptor occupation by Con A performs another function in addition to the generation of InsP<sub>3</sub> and diacylglycerol which leads to Ca<sup>2+</sup> mobilisation and PKC activation. Indeed, it is believed that Con A-promoted cross-linking (achieved by

binding to and linking components of the T cell receptor through their carbohydrate moieties) is obligatory for the maximal mitotic response in T cells (Metcalfe et al, 1985).

Synergism between protein kinase C and Ca<sup>2+</sup> signals has been amply demonstrated in the proliferation of murine splenocytes by the appropriate combinations of TPA and ionophore. However fura-2 studies demonstrate that pre-treatment of these cells with a higher concentration of TPA ( $10^{-7}$ M) resulted in a diminution of the Ca<sup>2+</sup> signal induced by Con A (Tables 8 & 10, Figure 19). Since

 $4\alpha$  12,13 phorbol didecanoate, which lacks the ability to activate protein kinase C had no such antagonistic effect (Table 9) it appears that TPA via its protein kinase C enhancing activity can at high concentrations (10<sup>-7</sup>M) suppress the lectin-induced increase in [Ca<sup>2+</sup>]i whilst at lower concentrations (10<sup>-9</sup>M) can synergise with A23187 to induce proliferation of murine splenocytes. A similar situation exists in antigen-stimulated RBL basophilic leukaemia cells. Thus, A23187 and TPA synergise in producing a secretory response whilst TPA at higher concentrations suppresses the antigen-activated mobilisation of  $Ca^{2+}$ . In this manner, TPA can act either as a potentiator or inhibitor of secretion, depending on its concentration and when it is added (Di Virgilio et al, 1984). There are additional reports in the literature that TPA can antagonise the  $Ca^{2+}$  signal generated by various stimuli in different cells. Although TPA is thought to exert its affects on the  $[Ca^{2+}]i$  via protein kinase C activation in all cases, the way in which  $Ca^{2+}$  homeostasis is affected differs. Thus, TPA is believed to stimulate  $Ca^{2+}$  efflux from human platelets to restore  $[Ca^{2+}]i$  to resting levels following activation. PKC activation leads to Ca<sup>2+</sup> extrusion probably by Ca<sup>2+</sup>-ATPase and possibly by enhanced Na<sup>+</sup>-Ca<sup>2+</sup> exchange (Pollock et al, 1987). In contrast, TPA is thought to mediate its effects on  $Ca^{2+}$  homeostasis in PC12 and RINm5F cells by inhibition of the voltage-gated  $Ca^{2+}$  channel. Thus, in these two cell types, TPA (presumably via PKC activation) inhibits the cytosolic-free

 $Ca^{2+}$  concentration rise induced by depolarizing agents (Di Virgilio et al, 1986). Other reports suggest that TPA interferes in the stimulus-response sequence at a very early point which precedes the mobilisation of calcium. Thus, it is possible that the antagonistic effect of TPA on fMet-Leu-Phe-stimulated Ca<sup>2+</sup> mobilisation in rabbit neutrophils is due to phosphorylation of proteins, such as the receptors themselves or the G proteins of the receptor coupling systems which subsequently inhibits the generation of the signal,  $InsP_3$ , responsible for the mobilisation of  $Ca^{2+}$  (Naccache et al, 1985). This particular hypothesis is substantiated by other studies which suggest that phorbol esters disrupt coupling of the activated G protein to phospholipase C (Muldoon et al, 1987; Smith et al, 1987). Furthermore, it seems likely that this phorbol ester-mediated effect may mimic a negative feedback signal induced by protein kinase C activation stimulated by DG generated upon activation of phospholipase C. Thus it is suggested that PKC activation by  $Ca^{2+}$  and DG may be responsible for the normal decrease in InsP3 release, possibly by a modification resulting in the down-regulation or inhibition of the G protein responsible for relaying the mitotic signal from the cell surface receptor to the phospholipase C activity (Muldoon et al, 1987).

TPA pre-treatments have been demonstrated to depress Con A induced increases in  $[Ca^{2+}]i$  but rather than inhibiting the mitogenicity of Con A, TPA enhanced the lectin stimulated DNA synthesis in murine splenocytes (Figure 20).

Since the inactive phorbol ester,  $4\alpha$  12,13 phorbol didecanoate had no such effect it implies that PKC activation is responsible for the enhancement of DNA synthesis in lectin-stimulated cells. These results reaffirm the hypothesis that the Ca<sup>2+</sup> signal alone is insufficient to trigger splenocytes to divide following stimulation, since the magnitudes of the rises in  $[Ca^{2+}]i$  induced by Con A alone, or Con A and TPA in combination (Figure 19) did not correlate with their relative mitogenicity in the  $[^{3}H]$ -thymidine assay (Figure 20). Thus, rises in  $[Ca^{2+}]i$  cannot by themselves account for the stimulation of DNA synthesis in murine splenocytes. However, since the  $Ca^{2+}$  signal was not completely abolished by pretreatment with TPA the hypothesis that PKC activation and elevated  $[Ca^{2+}]i$  act synergistically to induce proliferation in murine splenocytes remains viable.

In summary therefore, Con A stimulation of murine splenocytes is believed to lead to both PKC activation and an early [Ca<sup>2+</sup>]i increment, subsequent cellular responses are set in motion ultimately leading to DNA synthesis. These two signals are assumed to be generated by hydrolysis of PIP<sub>2</sub> following lectin induced phospholipase C activity, under normal conditions this activity may be down-regulated by a negative feedback loop in which PKC disrupts the coupling of the activated G protein to phospholipase C. Generation of InsP3 and DG will therefore cease together with the subsequent  $Ca^{2+}$  mobilisation and PKC activation and the proliferative response will therefore wane. However, when cells are pretreated with TPA, the subsequent protein kinase C activation will down-regulate the generation of InsP3 and DG induced by Con A stimulation and therefore the  $Ca^{2+}$  signal will be impaired but whilst the physiological activator of PKC, DG may be suppressed, the continued presence of TPA which also activates PKC will bypass this negative feed-back loop and ultimately lead to proliferation. This super-stimulation of PKC by phorbol ester more than compensates for the reduced Ca<sup>2+</sup> signal since splenocytes pre-treated with TPA demonstrate a greater increase in DNA synthesis than cells stimulated by Con A alone (Figure 20). Nevertheless, since TPA alone does not match the mitogenicity of Con A it reiterates the hypothesis that both PKC activation and an increase in [Ca<sup>2+</sup>]i are obligatory for murine splenocyte proliferation.

One of the additional effects of protein kinase C activation appears to be the activation of the amiloride-sensitive  $Na^+/H^+$  antiport leading to cytoplasmic

205

alkalinization (Grinstein et al, 1985). In the present study, amiloride, an inhibitor of  $Na^+/H^+$  exchange (Besterman et al, 1985) significantly inhibited [<sup>3</sup>H]-thymidine incorporation into Con A stimulated murine splenocytes in a dose dependent manner (Figure 21).

An increase in pHi in response to Con A was detected using the fluorescent probe BCECF confirming other studies in which lectin stimulation of lymphocytes caused cellular alkalinzation (Gerson et al, 1982; Metcalfe et al, 1985). The pHi increase in response to Con A stimulation was slower than the Ca<sup>2+</sup> signal and was only fully developed after ten minutes (Figure 22). Since, the co-mitogenic concentration of TFP (10<sup>-9</sup>M) produced a comparable pHi increase it appears that protein kinase C activation does lead to cytoplasmic alkalinization probably due to stimulation of the Na<sup>+</sup>/H<sup>+</sup> antiport in splenic lymphocytes. The Con A induced increase in pHi might result from Na<sup>+</sup>/H<sup>+</sup> exchange initiated by protein kinase C activation or calci-calmodulin activation dependent on the earlier Ca<sup>2+</sup> signal (Villereal et al, 1985). Certainly in antigen-stimulated T cells of the leukaemia line, HPB-ALL the rise in pHi was dependent on the presence of extracellular  $Ca^{2+}$ , suggesting that an elevation of [Ca<sup>2+</sup>]i mediates the stimulation of the antiport (Rosoff and Cantley, 1985). Many other agents that activate Na<sup>+</sup>/H<sup>+</sup> exchange also induce an increase in cytoplasmic free  $Ca^{2+}$ . The change in  $[Ca^{2+}]i$  generally precedes the activation of the antiport (Hesketh et al, 1985) which suggests a causal relationship. The effects of A23187 on cytoplasmic pH in rat thymocytes, an agent chosen to define the role of  $Ca^{2+}$  in the regulation of the Na<sup>+</sup>/H<sup>+</sup> antiport, suggest that stimulation of Na<sup>+</sup>/H<sup>+</sup> exchange is due at least in part, to cell shrinking and does not require stimulation of PKC (Grinstein and Cohen, 1987). Whilst it is clear that Con A stimulation of murine splenocytes leads to cytoplasmic alkalinization it is as yet unclear whether it is an obligatory signal in the series of events triggered by ligand binding which result in proliferation. Thus, it may represent a passive accompaniment intended to increase

H<sup>+</sup> ejection in anticipation of an increased metabolic rate or it might increase the osmotic content of the cells leading to cell swelling which might be an early requirement for cellular growth (Gelfand et al, 1987). Amiloride inhibition of DNA synthesis in lectin stimulated splenocytes seems to imply that enhanced Na<sup>+</sup>/H<sup>+</sup> exchange leading to an increase in pHi might be an obligatory signal. However, amiloride is non-specific and has been demonstrated to block Ca<sup>2+</sup>/Na<sup>+</sup> exchange (Kaczorowski et al, 1985), to inhibit protein synthesis (L'Allemain et al, 1984) and growth factor receptor tyrosine kinases (Davis & Czech, 1985). Intriguingly, when more potent and specific structural analogues of amiloride were investigated for their effects on human peripheral lymphocytes stimulated with phytohaemagglutinin, it was found that concentrations of the analogues 10 times higher than those required to inhibit Na<sup>+</sup>/H<sup>+</sup> exchange had no effect on proliferation. This implies that neither activation of the antiport nor the associated intracellular alkalinization are essential for the proliferation of human T lymphocytes (Mills et al, 1986). Thus, whilst Con A does indeed initiate an increase in pHi in murine splenocytes it is as yet unclear whether it is an obligatory signal for proliferation to occur. Stimulation of Na<sup>+</sup>/H<sup>+</sup> antiport is only one of many possible targets of protein kinase C, thus protein kinase C has a potential role in the control or modulation of many metabolic processes (for review see Kikkawa and Nishizuka, 1986).

In summary, both protein kinase C activation and a  $Ca^{2+}$  signal are implicated in the Con A stimulation of splenocytes. The Na<sup>+</sup>/H<sup>+</sup> antiport may be a target protein for PKC, stimulation of the antiport leads to an increase in pHi which although monitored following Con A stimulation has not yet been irrevocably established as an essential signal for DNA synthesis to occur. The generation of pHi and Ca<sup>2+</sup> signals in response to lectin binding is consistent with the hypothesis that both signals are generated following PIP<sub>2</sub> hydrolysis; InsP<sub>3</sub> thus produced can release intracellular Ca<sup>2+</sup> whilst diacylglycerol may activate PKC which subsequently causes the pHi increase.

Another intracellular messenger, ornithine decarboxylase (ODC) has been implicated in both T lymphocyte proliferation and in the generation of  $Ca^{2+}$  signals in other cells (see Section 1.7). However, preliminary studies with the ODC inhibitor DFMO (Tables 11 and 12) failed to resolve whether or not ODC plays a role in murine splenocyte proliferation since the inhibitor appeared to be lacking in activity (see Chapter 3).

Having delineated the mechanism of Con A induced murine splenocyte proliferation as far as possible, the effects of known physiological modulators of immunological proliferation were investigated for their ability to interfere with the various steps of the outlined mechanism. Of the many hormones known to modulate the immune system, regulation by gonadal steroids is a well documented fact. This contention is supported by both clinical and experimental evidence (Grossman, 1985) (see Section 1.7). Despite the controversies which exist in the literature most agree that sex steroids depress the cell-mediated immune response (Grossman, 1985; Myers et al, 1986). With a view to determine the mode of action by which sex steroids modulate the cell-mediated immune system, their effects on T cells, the major functional units of the cellular immune system were investigated. Thus, the ability of sex steroids to inhibit Con A-induced murine splenocyte proliferation (an experimental model considered to represent the proliferative aspect of the immune response) was examined. The results of this study demonstrate clearly that sex steroids and certain of their metabolites are able to inhibit mitogen-induced blast transformation (Figures 23 and 24) and therefore confirm observations made in other T-lymphocytes (Schiff et al, 1975; Mendelsohn et al, 1977; Neifeld et al, 1977; Wyle and Kent, 1977; Homo et al, 1980). Since inhibition of lymphocyte stimulation is only observed at pharmacological concentrations of sex steroids together with the fact that most studies indicate that steroid receptors are present only within the epithelial reticulum of the lymphoid organ

(Grossman et al, 1979) it suggests that the inhibitory effect of sex steroids on splenocyte proliferation is the result of a nonspecific, nonreceptor mediated event.

This hypothesis is further substantiated by the finding that  $\alpha$ -oestradiol, a relatively rare and physiologically inert female sex steroid which does not bind to oestrogen receptors in target tissue (Noteboom and Gorski, 1965) is an equally effective inhibitor of lectin-induced proliferation.

There is a substantial amount of literature describing non-receptor-mediated actions of steroid molecules, many of these 'direct' effects of steroids relate to changes in membrane function, for a review see (Duval et al, 1983). Because of the analogy between the structure of the cholesterol molecule and that of steroids it was suggested that steroids could be inserted vertically in the phospholipid bilayer instead of the cholesterol moiety (Willmer, 1961). However it is now believed that steroids are not really incorporated in the membrane but lie flat or on their edge on the surface of the phospholipid polar heads (Khaiat et al, 1975). Accumulation of these hydrophobic steroid molecules around the membrane together with their interaction with membrane components is likely to produce alterations in membrane properties especially at concentrations above 10<sup>-5</sup>M (Read & McElhaney, 1976). The characteristics of these interactions between steroids and the membrane are different from those of usual steroid-receptor associations (Duval et al, 1983); moreover the affinity of these compounds for the membrane site appears relatively weak since most steroids are only active at pharmacological concentrations (Godeau et al, 1978). Steroid recognition sites may be formed by phospholipids or phospholipid clusters. This hypothesis is based on the similarities demonstrated between the effects of steroids and anaesthetics on the membranes. Administration of an anaesthetic molecule or a steroid will induce a change of this lipid core towards a liquid state and thus alter functions of certain membrane proteins (Lee, 1976). The anaesthetic properties of steroids have been recognised for a long time; (Selye, 1941). The structure-activity relationship of

steroids with anaesthetic activity is not clearly defined although it is believed that only hormonally active steroids have an anaesthetic effect (Selye, 1941) which might explain why testosterone glucuronide, a conjugated metabolite of testosterone and cholesterol had no inhibitory effect on lymphoblast transformation (Figures 23 and 25).

Interestingly, not only do Ca<sup>2+</sup> ions play an important role in controlling the physical property of the lipid and anaesthetic action but they also appear to be involved in steroid membrane interactions (Papahadjopoulos, 1972). Indeed, steroids (although not sex steroids) cause enhanced <sup>45</sup>Ca<sup>2+</sup> uptake into mouse thymocytes (Homo & Simon, 1981). Testosterone (10<sup>-4</sup>M) had no effect on the lectin-induced Ca<sup>2+</sup> signal (Table 13) implying that its inhibitory action on murine splenocytes must be mediated elsewhere. Since high concentrations of sex steroids (10<sup>-4</sup>M) have been shown to inhibit glucose transport within 1-2 minutes of administration in adipocytes (Livingston and Lockwood, 1975) it is possible that by interfering with glucose transport following lectin stimulation, testosterone inhibits murine splenocyte Another direct effect of steroids rather than by a classical proliferation. receptor-mediated event is their reported rapid inhibition of nucleoside transport in isolated mouse thymocytes. Sex steroids at high concentrations (10<sup>-5</sup>M and above) exerted a rapid inhibition of tritiated uridine uptake (Gagne et al, 1980). These rapid effects on precursor nucleoside pools could explain the inhibitory effect of sex steroids on lectin-induced stimulation, since testosterone (10<sup>-4</sup>M) caused a significant inhibition of DNA synthesis in murine splenocytes even when administered 44 hours following Con A treatment (Figure 26).

These *in vitro* studies demonstrate that the inhibitory effects of sex steroids are only observed at pharmacological concentrations. However there is a large volume of literature which suggests that gonadal steroids interact with the immune system under physiological conditions (see Section 1.7). The evidence is based on several

210

observations including, the sexual dimorphism which exists in the immune response, alterations in the immune response noticeable during pregnancy and the presence of specific receptors for gonadal steroids in the organs responsible for the immune response (Grossman, 1985). Results from this study demonstrate that gonadectomy does indeed have dramatic effects on the lymphoid organs of mice. Orchidectomy resulted in significant increases in both the spleen and thymic indices (Figure 28) whereas ovariectomy only caused thymic enlargement (Figure 27). These observations confirm earlier studies (Dougherty, 1952; Chiodi, 1976) and imply that the gonadal steroids, particularly testosterone are involved in involution of the lymphoid organs. Indeed, regeneration of the thymus after orchidectomy has been shown to be inhibited in a dose-related fashion by testosterone implants which release physiological concentrations of testosterone. These findings illustrate the important physiological link between the endocrine and immune systems (Greenstein et al, 1986; Fitzpatrick & Greenstein, 1987).

Orchidectomy did not appear to potentiate T cell function since splenocytes derived from orchidectomised mice responded to lectin stimulation in a comparable fashion as cells from control animals (Figure 29). In contrast, preliminary studies demonstrated that ovariectomy impaired the spleen cell mitogen response (Figure 30) implying that oestrogen potentiates T cell function in the intact animal. Interestingly, however, there are reports in the literature which suggest that females have decreased cellular immune responses as compared to males due in part to the presence of oestradiol (Huber et al, 1982) (see Section 1.7). Moreover, it is suggested that oestradiol rather than directly suppressing T cell function may depress the cell-mediated immune response indirectly through the generation of thymic serum factors (Grossman et al, 1982; Grossman & Roselle, 1983; Grossman, 1985; Myers et al, 1986). Furthermore, it is believed that removal of sex steroids in the male rat by castration results in an increase in thymus gland size and in the production of a serum

substance which enhances the response to the mitogen, Con A. Administration of physiological levels of oestradiol (it was suggested that testosterone is aromatized into oestradiol in the male rat) reversed these effects. Since thymectomy abolished the enhancing affects of castration on T cell transformation the thymus is likely to be the source of the active serum factor (Grossman et al, 1982). Indeed, the hypothesis is that oestradiol interacting with receptors in the thymic epithelial reticulum depresses the production and/or release of a thymic source of a serum factor that enhances proliferation induced by Con A. However, preliminary investigations in this study dramatically demonstrated that mouse serum fractions from any source (viz. control, gonadectomised and sham-operated animals) and at any concentration tested failed to support the enhanced DNA synthesis in murine splenocytes, normally achieved by lectin stimulation (Figure 31). Moreover, it is apparent that not only does mouse serum not support splenocyte proliferation but it actually inhibits in a dose-related fashion DNA synthesis in murine splenocytes stimulated with Con A in the presence of 10% foetal calf serum (FCS) (Figure 32). These results are substantiated by reports which suggest that mouse serum does not support a primary in vitro immune response to sheep red blood cells (Mishell & Dutton, 1967). In addition, there are indications that mouse serum exerts an inhibitory effect on the immune response in cells cultured in foetal calf serum supplemented media. A similar inhibition was obtained with gerbil, rat and guinea pig sera but not with human or rabbit serum (Veit & Michael, 1973; Schreier & Nordin, 1977). Indeed, many investigators using the immune response in vitro (the proliferative aspect of which is considered to be mimicked by lectin stimulation), agree that FCS has a pivotal role (Schreier & Nordin, 1977). Serum, as a medium supplement supplies macromolecules which are protective or promote growth in some ill-defined manner, small molecules, eg. nucleosides, vitamins, hormones and co-enzymes which supply essential trace nutrients not present in the medium, factors (including antibodies) which neutralise or combine with the

stimulant and foreign antigens (Ling and Kay, 1975). The heterologous sera (FCS) used in this study obviously provided a variety of potential antigens and factors which were critical in promoting/supporting murine splenocyte proliferation. Despite its widespread use it is difficult to reconcile in the light of these findings how this lectin-induced *in vitro* proliferative model, which depends on <u>heterologous</u> serum-supplemented media can in reality provide information relating to the *in vivo* murine immune response. Results of these and similar *in vitro* experiments should be interpreted cautiously and critically when delineating the immune response in the intact animal, since it is clear that no simple extrapolation from the *in vitro* data can be made.

These studies with mouse serum fractions were unable to shed any light on the presence or absence of thymic serum factors which might mediate the effects of gonadal steroids on the immune system. However of all the sera types tested, female 'control' serum alone significantly enhanced DNA synthesis (albeit at a severely reduced level when compared with FCS supplemented cultures) in response to lectin in murine splenocytes derived from female animals (Figure 31). These findings taken together with earlier experiments which demonstrated that ovariectomy impaired the spleen cell mitogen response, suggest that oestrogen potentiates rather than suppresses T cell function. This hypothesis is in stark contrast to reports in the literature which suggest that females have enhanced humoral reactivity but decreased cell-mediated immunity compared to males (Ahlqvist, 1976; Bell, 1981; Huber et al, 1982; Myers et al, 1986). Nevertheless, controversy exists in the literature and there are other reports, in agreement with these preliminary findings, which indicate that females are reported to have a greater capacity for cellular immunity than males, (Eidinger and Garrett, 1972). Thus, females reject tumours and homografts with greater efficiency than males (Graff et al, 1969) and cells derived from females perform better in mixed lymphocyte cultures than cells derived from males (Lieber et al, 1969). A more recent study suggests that oestradiol treatment decreases a subpopulation of T cells (Lyt-2+)

in the thymus and in the spleen whereas testosterone maintains this subpopulation of suppressor T cells (Ahmed et al, 1985). Oestradiol receptors have also been demonstrated on OKT-8<sup>+</sup> peripheral T cells (analogous to murine Lyt-2<sup>+</sup> cells), which have suppressor/cytotoxic function (Cohen et al, 1983). Thus, it is possible that the enhanced cellular immunity in females may relate in part to alterations induced by oestradiol in the number and function of suppressor cells (Ahmed et al, 1985). The tentative hypothesis based on preliminary investigations in this study that oestradiol potentiates T cell function could thus be due to the inhibition of the activity of suppressor T cells by oestradiol rather than to a direct mitogenic effect.

In contrast, the manner in which androgens modulate the immune system is difficult to discern, castration causes both thymus gland and spleen enlargment although the spleen cell mitogen response, on a cell for cell basis for lymphocytes derived from castrate animals was not significantly different from controls. This may be interpreted to mean that castration does not result in a net augmentation of individual T cell function, at least under the model system employed. However, the *in vivo* immune response of castrated animals may be enhanced due to an absolute increase in the number of lymphoid cells able to respond. Indeed, castration of males is believed to result in improved *in vivo* immune function (Graff et al, 1969; Castro & Hamilton, 1972).

The observation made in this study, that the inhibitory effect of sex steroids *in vitro* on murine splenocyte proliferation are only noticeable at high concentrations does not necessarily preclude a role of these hormones *in vivo* under some circumstances. Thus, progesterone concentrations close to  $10^{-5}$ M may occur within human trophoblastic tissue and may therefore play a role in the failure to reject the embryo (Siiteri et al, 1977). Whilst only pharmacological concentrations of sex steroids exert an inhibitory effect on isolated murine splenocytes stimulated to divide by lectin, it is evident that physiological levels of sex steroids do modulate the immune

214

response in vivo(Grossman, 1985). This difference may be accounted for when it is understood that the sex steroid receptors are most likely present in the epithelial reticulum of the lymphoid organ (Grossman et al, 1979a,b) and that these non-lymphoid cells are eliminated in the preparation of lymphocyte suspensions (Sasson & Meyer, 1981). Since lymphoid reticuloepithelial cells not only possess sex steroid receptors but also undergo hypertrophy following gonadectomy and marked alterations following treatment with gonadal steroids, it is conceivable that such changes are mediated by steroid-receptor interactions. It is also believed that these cells whilst undergoing hypertrophy also produce and release substances into the blood, thymic serum factors (Grossman & Roselle, 1983). Indeed it is believed that sex steroids function within the hypothalamic-pituitary-gonadal-thymic axis (HPG-thymic axis) (Grossman, 1985). Thus, gonadotropin releasing hormone (GnRH) from the hypothalamus stimulates luteinizing hormone (LH) release from the pituitary and LH stimulates sex steroid production. Through a negative feedback mechanism this causes GnRH release from the hypothalamus to be depressed and LH release from the pituitary to be inhibited. Increasing levels of sex steroids from the gonads depress thymosin release, whereas decreasing levels of sex steroid are thought to induce thymosin release. In turn increased levels of thymosin stimulate GnRH release. T cell function is modulated by various thymosins which stimulate maturation of the different T cell subpopulations (Grossman, 1985). These reproductive-immunological interactions are likely to be very complex. Indeed a recent paper suggests that a thymic factor inhibits the secretion of testosterone by rat Leydig cells in vitro (Pedernera et al, 1986). This report emphasises the bidirectional nature of the HPG-thymic axis.

Given the difficulties of interpretation associated with the Con A model referred to in the preceding section, the mechanism of action of a more physiological mitogen Interleukin-2 was investigated with a view to produce a more relevant *in vitro* 

model to examine proliferation. Thus, exposure of Con A blasts to human recombinant Human IL2 (100 U ml<sup>-1</sup>) stimulated cell proliferation (Figure 33). Nifedipine suppressed this response to IL2 (Figure 34) implying that a  $Ca^{2+}$  influx is necessary for cell division to occur. This is in agreement with Birx et al (1984) and Gearing et al (1985) who also observed a reduction in IL2-induced proliferation in T lymphocytes treated with calcium antagonists. However, measurement of  $[Ca^{2+}]i$  in Con A blasts following IL2 stimulation, although complicated by pronounced fura-2 leakage failed to detect a distinct Ca<sup>2+</sup> signal (Figure 35). Whilst calcium is relatively firmly established as a mediator in lectin stimulation of T lymphocytes (Tsien et al, 1982; Hesketh et al, 1983; Gelfand et al, 1984), there are conflicting reports as to whether IL2 interaction with its cell surface receptor generates a change in  $[Ca^{2+}]i$ . Thus, some workers are convinced that cytosolic  $Ca^{2+}$  does increase following IL2 administration (Gearing et al, 1985; Farrar et al, 1986; Rossio et al, 1986) whilst others have demonstrated that increases in [Ca<sup>2+</sup>]i are not associated with IL2-IL2 receptor interaction (Weiss et al, 1984; Mills et al, 1985b; O'Flynn et al, 1985; LeGrue et al, 1987). Since TFP the calmodulin antagonist also prevented proliferation of Con A blasts (Figure 36) it appears that IL2-provoked mitosis is  $Ca^{2+}$  dependent, with a subsequent calci-calmodulin mediated protein phosphorylation ultimately leading to mitosis. However, these results should be interpreted cautiously since both nifedipine and TFP only exerted their inhibitory action on IL2 induced proliferation at a concentration (10<sup>-5</sup>M), ten times greater than the concentration required to suppress Con A induced murine splenocyte proliferation (Figures 34 and 3, 36 and 8). Thus, the effects may relate to non-specific metabolic actions rather than to an indication of  $Ca^{2+}$  dependency.

Attempts to correlate IL2 ligand-receptor interaction with phosphatidylinositol turnover are just as equivocal (see Section 1.7). Amiloride, an agent used to inhibit the  $Na^+/H^+$  antiport (believed to be activated by PKC) inhibited
IL2-induced proliferation (Figure 37), suggesting that PKC activation mediated by DG one of the products of phosphatidylinositol hydrolysis may indeed be involved in stimulus-mitosis coupling. IL2 induced cytoplasmic alkalinization via the Na<sup>+</sup>/H<sup>+</sup> antiport has been monitored in lymphocytes (Mills et al, 1985a) although no alterations in labelled inositol phosphates were detected (Mills et al, 1986) or in the intracellular distribution of PKC (Gelfand et al, 1987). In contrast, other workers suggest that the IL2 signal does involve the activation of inositol phospholipid turnover (Bonvini et al, 1986; Farrar et al, 1986) and PKC translocation (Farrar and Anderson, 1985; Evans & Farrar, 1987). Thus, the precise biochemical and molecular events associated with IL2 regulation of lymphocyte proliferation await clarification. In peripheral T lymphocytes IL2 does not directly stimulate cell division, thus an antigenic or mitogenic stimulus is first required for cellular activation and expression of IL2 receptors. IL2 then provides a second signal which enables progression of activated cells from  $G_1$  to the replicative S phase (Stadler et al, 1981). Activation of lymphocytes by mitogenic lectins leads to proliferation, differentiation and secretion of lymphokines (Gillis, 1983). Mitogenic lectins or antigens in conjunction with monocytes initiate the sequential secretion of IL1 from activated monocytes and IL2 from activated T lymphocytes (Gillis, 1983). In a distinct activation pathway, mitogens or lectins promote IL2 receptor expression (Gillis, 1983; Robb, 1984) and binding of IL2 to this receptor then leads to cell proliferation. The lectin induced  $Ca^{2+}$ increase demonstrated in this study is believed to be required for IL2 production but not for IL2 receptor expression (Mills et al, 1985a). In the absence of a  $Ca^{2+}$  influx and subsequent IL2 production, lectins are unable to trigger DNA synthesis and cell proliferation.

Since the sex steroids inhibited IL2-induced proliferation at the same pharmacological concentrations necessary to suppress Con A-induced proliferation (Figure 38), it appears that Con A blasts are equally sensitive in terms of inhibition of DNA synthesis as murine splenocytes. Thus, signalling of cells after the binding of IL2 to its receptor, one of the final steps in the interleukin cascade leading to lymphocyte proliferation is as vulnerable to sex steroid as is activation of lymphocytes by lectin. It seems unlikely on the basis of this preliminary evidence that sex steroids mediate their inhibitory effects on lymphocyte proliferation by blocking the production of IL2 by a T cell subset in an analogous fashion to glucocorticoid inhibition of T lymphocyte activation (Smith et al, 1977). Indeed, since similar concentrations of sex steroid also inhibit B lymphocyte proliferation (Figure 42) it seems likely that their effect is the result of a nonspecific non-receptor-mediated event. Possibly, the incorporation/interaction of sex steroids with membrane lipids leads to a modification of membrane properties (eg. fluidity, permeability and membrane enzyme activities) which in turn could abolish critical membrane events, such as inhibition of nucleoside or glucose transport, necessary for DNA synthesis to occur.

The activation of quiescent T lymphocytes requires two stimuli: binding and cross-linking of the T cell antigen-receptor complex by antigen-bearing accessory cells or mitogenic lectins (Zanders, 1985; Isakov, 1987), and the accessory cell-derived

hormone, IL1 (Mizel, 1982). Of the two members of the IL1 family IL1 $\alpha$  and IL1 $\beta$ , only IL1 $\beta$  induced a significant increment in DNA synthesis when administered alone (Figured 39 and 40) and neither IL1 enhanced the proliferation induced by low concentrations of Con A (0.35 µg ml<sup>-1</sup>), presumably the dose of Con A chosen did not provide an adequate signal in the form of receptor cross-linking, although further investigations are required to substantiate this hypothesis. IL1 responsive lymphocytes bear a single class of high affinity receptors for IL1 (Mizel et al, 1987; Dower and Urdal, 1987). In the presence of IL1, the IL1 receptor undergoes extensive ligand-induced down regulation, the internalized IL1 receptor is not recycled but is either degraded or redirected to sites within the cell (Mizel et al, 1987). The

218

mechanism of IL1 action is unclear but does not appear to involve PIP<sub>2</sub> hydrolysis (Abraham et al, 1987), indeed the accumulation of relatively high levels of IL1 in the nucleus raises the possibility that IL1 may mediate its effects via interaction with nuclear receptors (Mizel et al, 1987).

The findings presented in this study emphasize the role of a rapid rise in [Ca<sup>2+</sup>]i and activation of protein kinase C as intracellular messengers in the signal transduction pathway following lectin stimulation of lymphocytes. Hydrolysis of PIP<sub>2</sub> is likely to be responsible for these two signals, since it yields IP<sub>3</sub> which mobilises intracellular Ca<sup>2+</sup> and DG, which activates PKC. PKC may subsequently enhance Na<sup>+</sup>/H<sup>+</sup> exchange leading to cellular alkalinization. The lectin seems to provide an additional signal, possibly receptor crosslinking which is obligatory for T cells to progress from G<sub>0</sub>, the resting stage of the cell cycle into the G<sub>1</sub> phase and subsequent completion of the mitotic cycle. Transition of activated lymphocytes from G1 into the S phase occurs as a result of IL2 production and binding to its high affinity receptors. The signals generated by IL2-IL2 receptor interaction remain elusive and await further clarification. Proliferation of T lymphocytes in vitro is vulnerable to pharmacological concentrations of sex steroids which mediate their effect probably by a non-specific, non-receptor-mediated event possibly leading to alterations in membrane properties. The mechanisms by which gonadal steroids modulate the immune system in the intact animal remain speculative and there are still many missing parts in the puzzle that must be resolved in the future.

## APPENDIX

## (A) Determination of Viability by Trypan Blue Exclusion

The number of viable splenocytes was determined by staining cell populations with trypan blue. Viable cells exclude the dye while nonviable cells take up the dye.

Equal volumes of spleen cell suspension and trypan blue saline solution (4 parts of 0.2% trypan blue with 1 part of 4.25% sodium chloride (w/v) were mixed and the number of unstained (viable) and stained (dead) cells were counted (at least 200 in total) within three minutes.

## **REFERENCES**

Abboud, C.N., Scully, S.P., Lichtman, A.H., Brennan, J.K. & Segel, G.B. The requirements for ionized calcium and magnesium in lymphocyte proliferation. J. Cell. Physiol., <u>122</u>, 64-72

Abraham, R.T., Ho, S.W., Barna, T.J. & McKean, D.J. Transmembrane signalling during interleukin 1-dependent T cell activation. J. Biol. Chem., <u>262</u>, 2719-2728

Aharonov, A., Pruss, R.M., Herschman, H.R., 1978. Epidermal growth factor, relationship between receptor regulation and mitogenesis in 3T3 cells. J. Biol. Chem. 253, 3970-3977

Ahlqvist, J. 1976 Endocrine influences on lymphatic organs, immune response, inflamation and autoimmunity. Acta. Endocrinol., (suppl) (Copenh) 206, 33-70

Ahmed, S.A. & Dauphinee, M.J. 1985. Effects of short-term administration of sex hormones on normal and auto-immune mice. J. Immunol., <u>134</u>, 204-211

Aickin, C.C., Brading, A.F. & Burdyga, T.V. 1984 Evidence for sodium-calcium exchange in the guinea pig ureter, J. Physiol., <u>347</u>, 411-430

Akerman, K.E.O. & Nicholls, D.G. 1983 Physiological and bioenergetic aspects of mitochondrial calcium transport. Rev. Physiol. Biochem. Pharmacol., <u>95</u>, 149-201

Alexander, P. & Currie, G, 1984 Concomitant synthesis of growth factors and their receptors, an aspect of malignant transformation. Biochem. Pharmacol., <u>33</u>, 941-943

Algranati, I.D., Goldemberg, S.H. 1977 Polyamines and their role in protein synthesis. Trends Biochem. Sci. 2, 272-279

Alitalo, K., Koskinen, P., Makela, J.P., Saksela, K., Sistonen, L. & Winqvist, R. 1987 myc oncogenes: activation and amplification. Biochem. Biophys. Acta. <u>907</u>, 1-32

Allan, D. & Michell, R.H. 1977 A comparison of the effects of phytohaemagglutinin and of calcium ionophore A23187 on the metabolism of glycerolipids in small lymphocytes. Biochem. J., <u>164</u>, 389-397

Allen, J.C. & Smith, C.J., 1979 'Chalones: A reappraisal'. Biochemical Soc. Transactions (Bioch. Reviews) 7, 584-92

Altura, B.M. & Altura, B.T., 1981 Magnesium ions and contraction of vascular smooth muscles: relationship to some vascular diseases. Fed. Proc. Fed. Am. Soc. Exp. Biol., <u>40</u>, 2672-2679

Anderson, J.N, 1984 The effect of steroid hormones on gene transcription. In 'Biological regulation and development', Vol 3B, Hormone Action; eds R.F. Goldberger, K.R. Yamamoto, (Plenum, New York), pgs 169-212

Ansar Ahmed, S., Dauphinée, M.J. & Talal, N. 1985 Effects of short-term administration of sex hormones on normal and autoimmune mice. J. Immunol. <u>134</u>, 204-210

Antoniades, H.N., Stathakos, D. & Scher, C.D. 1975 Isolation of a cationic polypeptide from human serum that stimulates proliferation of 3T3 cells. Proc. Natl. Acad. Sci. (USA), <u>72</u>, 2635-2639

Anzano, M.A., Roberts, A.B., De Larco, J.E., Wakefield, L.M., Assoian, R.K., Roche, N.S., Smith, J.M., Lazarus, J.E. & Sporn, M.B., 1985. Increased secretion of type beta transforming growth factor accompanies viral transformation of cells. Molecular & Cellular Biology, <u>5</u>, 242-247

Appleman, M.M., Thompson, W.J., Russell 1973. Cyclic nucleotide phosphodiesterases. Adv. Cyclic Nucleotide Res., 3, 65-98

Arai, K., Yokota, T., Miyajima, A., Arai, N. & Lee, F. 1986. Molecular biology of T cell derived lymphokines: a model system for proliferation and differentiation of haemopoietic cells. BioEssays, Vol 5, 166-171

Archer, J.A., Gordon, P. & Roth, J. 1975. Defect in insulin binding to receptors in obese man. Amerlioration with calorie restriction. J. Clin. Invest., 55, 166-174

Armato, U., Andreis, P.G., Draghi, E. 1981. Cyclic AMP and cyclic GMP as the respective mediators of the intracycle stimulation of DNA synthesis and mitosis induced by glucagon and insulin in primary neonatal rat hepatocytes. Life Sci., 29, 2763-2769

Assoian, R.K., Kamoriya, A., Meyers, C.A., Miller, D.M., & Sporn, M.B. 1983 Transforming growth factor ß in human platelets, J. Biol. Chem., <u>258</u>, 7155-7160

Assoian, R.K. & Sporn, M.B. 1986 Type ß transforming growth factor in human platelets: release during platelt degranulation and action on vascular smooth muscle cells. J. Cell Biol., <u>102</u>, 1217-1223

Averdunk, P. 1976 Early changes of 'leak flux' and the cation content of lymphocytes by Concanavalin A. Biochem. Biophys. Res. Commun., 70, 101-109

Babcock, S.K., Gill, R.G., Bellgrau, D. & Cafferty, K.J. 1987 Studies on the two-signal model for T cell activation in vivo. Transplantation Proceedings, 19, 303-306

Baldassare, J.J. & Fisher, G.J. 1986 Regulation of membrane-associated and cytosolic phospholipase C activities in human platelets by guanosine triphosphate. J. Biol. Chem., <u>261</u>, 11942-11944

Balk, S.D. 1971 Calcium as a regulator of the proliferation of normal but not of transformed chicken fibroblasts in a plasma-containing medium. Proc. Natl. Acad. Sci. (USA), <u>68</u>, 271-275

Balkwill, F.R., Taylor-Papadimitriou, J., 1978 Interferon affects both G1 and S+G2 in cells stimulated from quiescence to growth. Nature 274, 798-800

Bardin, C.W., Bullock, L.P., Mills, L.C., Lin, Y.C. & Jacob, S.T. 1978 Role of receptors in the anabolic action of androgens. In 'Receptors & Hormone Action Vol II', Eds O'Malley, B.W., Birnbaumer, L. (Academic Press, New York), pgs 83-103

Barnes, D. & Sato, G. 1980 Serum-free cell culture: a unifying approach. Cell 22, 649-655

Baserga, R. & Weiss, L. 1967 Inhibition of deoxyribonucleic acid synthesis by pentobarbital. Biochim. Biophys. Acta., <u>145</u>, 361-367

Baserga, R. 1981 The Cell Cycle. New England Journal of Medicine, 304, 453-459

Baserga, R. 1985 'The Cell Cycle' in "The Biology of Cell Reproduction", (Harvard University Press), pgs 3-21

Batchelor, J.R., 1968 Hormonal control of antibody formation in 'Regulation of the Antibody Response', ed by B. Cinader. Charles C. Thomas. (Springfield, USA), pg 276

Bauer, S., Gierschik, P., Grandt, R., Horn, F., Jakobs, K.H., 1987 Regulation of signal transduction by signal-controlled effector systems. Biochem. Soc. Transact., 15, 24-26

Baublieu, E.E., 1984 Pharmacology of steroid receptors: from molecular and cellular studies to efficient antihormonal drugs. Biochem. Pharmacol., <u>33</u>, 905-912

Baxter, J.D. & Harris, A.W., 1975 Mechanism of glucocorticol action: general features, with reference to steroid-mediated immunosuppression. Transpl. Proc. 7, 55-65

Becker, P.B., Gloss, B., Schmid, W., Strahle, U., Schutz, G. 1986 In vivo protein-DNA interactions in a glucocorticoid response element require the presence of the hormone. Nature <u>324</u>, 686-688

Beguinot, L., Hanover, J.A., Ho, S., Richert, N.D., Willingham, M.C. & Pastan, I., 1985 Phorbol esters induce transient internalization without degradation of unoccupied epidermal growth factor receptors. Proc. Natl. Acad. Sci., USA, <u>82</u>, 2774-2778

Beguinot, L., Lyall, R.M., Willingham, M.C. & Pastan, I. 1984 Down regulation of the epidermal growth factor receptor in KB cells is due to receptor internalization and subsequent degradation in lysosomes. Proc. Natl. Acad. Sci., USA, <u>81</u>, 2384-2388

Bell, P.A., 1981 Steroids and the cells of the immune system In 'Mechanisms of Steroid Action', eds: Lewis, G.P. & Ginsburg, M. (MacMillan Press), pgs 75-83

Benos, D.J. & Sapirstein, Y.S., 1983 Characteristics of an amiloride-sensitive sodium entry pathway in cultured rodent glial and neuroblastoma cells. J. Cell. Physiol., <u>116</u>, 213-220

Berlin, R.D., Oliver, J.H., Ukena, T.E. & Yin, H.H., 1974 Control of cell surface topography. Nature, 247, 45-46

Berridge, M.J., 1975 Control of cell division: a unifying hypothesis. J. Cyclic. Nucleot. Res., <u>1</u>, 305-321

Berridge, M.J., 1981 Phosphatidylinositol hydrolysis: a multifunctional transducing mechanism. Mol. Cell. Endocrinol., <u>24</u>, 115-140

Berridge, M.J., 1984a "Oncogenes: inositol lipids and cellular proliferation", Bio/Technology. <u>6</u>, 541-546

Berridge, M.J., 1984b Inositol trisphosphate and diacylglycerol as second messengers. Biochem. J., 220, 345-360

Berridge, M.J., 1986 Second messenger dualism in neuromodulation and memory. Nature, <u>323</u>, 294-295

Berridge, M.J., 1987 Inositol lipids and cell proliferation. Biochim. Biophys. Acta., 907, 33-45

Berridge, M.J., Brown, K.D., Irvine, R.F., Heslop, J.P., 1985 Phosphoinositides and cell proliferation. J. Cell. Sci. Suppl., 3, 187-198

Berridge, M.J., Irvine, R.F., 1984 Inositol trisphosphate: a novel second messenger in cellular transduction. Nature, <u>312</u>, 315-321

Besch, E.L. & Chou, B.J., 1971 Physiological responses to blood collection methods in rats. Proc. Soc. Exp. Biol. Med. <u>138</u>, 1019-1021

Besterman, J.M., May, W.S., LeVine, H., Cragoe, E.J. & Cuatrecasas, P. 1985. Amiloride inhibits phorbol ester-stimulated Na<sup>+</sup>/H<sup>+</sup> exchange and protein kinase C. An amiloride analog selectively inhibits Na<sup>+</sup>/H<sup>+</sup> exchange. J. Biol. Chem., <u>26</u>0, 1155-9

Besterman, J.M., Watson, S.P. & Cuatrecasas, P. 1986 Lack of association of epidermal growth factor -, insulin - and serum-induced mitogenesis with stimulation of phosphoinositide degradation in BALB/c3T3 fibroblasts. J. Biol. Chem., <u>261</u>, 723-727

Biden, T.J., Prentki, M., Irvine, R.F., Berridge, M.J. & Wollheim, C.B. 1984 'Inositol 1,4,5-triphosphate mobilizes intracellular Ca<sup>2+</sup> from permeabilized insulin-secreting cells'. Biochem. Journal, <u>223</u>, 467-473

Biden, T. & Wollheim, C.B. 1986 Ca<sup>2+</sup> regulates the inositol tris/tetrakisphosphate pathway in intact and broken preparations of insulin-secreting RINm5F cells. J. Biol. Chem., <u>261</u>, 11931-11934

Birnbaumer, L., Codina, J., Mattera, R., Cerione, R.A., Hildebrandt, J.O., Sunyer, T., Rojas, F.J., Caron, M.G., Lefkowitz, R.J., Ivengar R. 1985 Structural basis of adenylate cyclase stimulation and inhibition by distinct guanine nucleotide regulatory proteins. In 'Molecular mechanisms of transmembrane signalling', Eds: P. Cohen, M.D. Houslay, (Elsevier (Amsterdam)), pgs131-182

Birge, C.A., Peake, G.T., Manz, I.K. & Daugheday, W.H., 1967 Radioimmunoassayable growth hormone in the rat pituitary gland: effects of age, sex and hormonal state. Endocrinology, <u>81</u>, 195-204

Black, B., Jarrett, C. & McDonald, J. 1980 Relationship between ion transport and  $(Ca^{2+} + Mg^{2+})$  ATPase activity in adinpocyte endoplasmic reticulum. Biochim. Biophys. Acta. <u>596</u>, 359-371

Blat, C., Chatelain, G., Desauty, G. & Harel, L. 1986 Inhibitory diffusible factor IDF45, a G1 phase inhibitor. FEBS Letters 203, 175-180

Bodycote, J. & Wolfe, S 1986 Metabolic breakdown of (<sup>3</sup>H) thymidine and the inability to measure human lymphocyte proliferation by incorporation of radioactivity. Proc. Natl. Acad. Sci., <u>83</u>, 4749-4753

Bolton, T.B. 1979 Mechanism of action of transmitters and other substances on smooth muscle, Physiol. Rev. <u>59</u>, 606-671

Bolton, T.B. 1985 Calcium exchange in smooth muscle In 'Control and Manipulation of Calcium Movement', Ed: J.R. Parratt, Raven Press (N. York), pg147-168

Bombik, B.M. & Burger, M.M. 1973 cAMP and the cell cycle: inhibition of growth stimulation. Exp. Cell. Res. <u>80</u>, 88-94

Bonvini, E., Ruscetti, F.W., Ponzoni, M., Hoffman, T. & Farrar, W.L. 1987 Interleukin 2 stimulates turnover and breakdown of inositol phospholipids in an interleukin 2 dependent cell line. J. Biol. Chem., <u>262</u>, 4160-4164

Boonstra, J., Moolenaar, W.H., Harrison, P.H., Moed P., Van Der Saag, P.T. & De Laat, S.W. 1983 Ionic responses and growth stimulation induced by nerve growth factor and epidermal growth factor in rat phaeochromoytoma (PC12) cells. J. Cell. Biol., <u>97</u>, 92-98

Boorman, G.A., Luster, M.I., Dean. J.H. & Wilson, R.E., 1980 The effect of adult exposure to diethylstilbestrol in the mouse and macrophage function. J. Reticuloendothel. Soc., 28, 547-551

Borden, E.C., Hogan, T.F. & Voelkel, J.G., 1982 Comparative antiproliferative activity in vitro of natural interferons  $\alpha$  and  $\beta$  for diploid and transformed human cells. Cancer Res. <u>42</u>, 4948-4953

Bottenstein, J., Hayash, I., Hutchings, S., Masui, H. Mather, J. McClure, D.B., Ohasa, S., Rizzino, A., Sato, G., Serrero, G., Wolfe, R. & Wu, R. 1979, The growth of cells in serum-free hormone supplemented media, Methods enzymology. 58, 94-109

Bourgeade, M.F., Besancon, F. & Thang, M.N. 1983 Interferon affects intracelular calmodulin levels. Biochem. Biophys. Res. Commun. <u>111</u>, 430-437

Bourne, H.R., Masters, S.B. & Sullivan, K.A. 1987 Mammalian G proteins: structure and function, Biochem. Soc. Transact. <u>15</u>, 35-38

Bowen-Pope, D.F., Dicorleto, P.E. & Rass, R. 1983 Interactions between the receptors for platelet-derived growth factor and epidermal growth factor. J. Cell. Biol., <u>96</u>, 679-683

Boynton, A.L., Kleine, L.P. & Whitfield, J.F. 1985 Cyclic AMP elevators stimulate the initiation of DNA synthesis by calcium-deprived rat liver cells. In 'Control of annual cell proliferation', eds: A.L. Boynton; H.L. Leffert. (Academic Press (London)), pgs 121-149 Boynton, A.L., MacManus, J.P. & Whitfield, J.F. 1982 Stimulation of liver cell DNA synthesis by oncomodulin on MW 11500 calcium-binding protein from hepatoma. Exptl. Cell. Res., 138, 454-458

Boynton, A.L. & Whitfield, J.F. 1981 Cyclic AMP-dependent protein kinases and calmodulin stimulate the initiation of DNA synthesis by rat liver cells. Adv. Cyclic Nucleotide Res. <u>14</u>, 411-419

Boynton, A.L. & Whitfield, J.F., 1983 The role of cAMP in cell proliferation: a critical assessment of the evidence. Adv. Cyclic Nucelotide Res. 15, 193-294

Bradford, P.G. & Rubin, R.P. 1986 Guanine nucleotide regulation of phospholipase C activity in permeabilized rabbit neutrophils. Biochem. J. 239, 97-102

Bradshaw, R.A. 1983 Nerve growth factor and released hormones In 'Biochemical Actions of Hormones', Vol X, Ed: Litwack, G. (Academic Press Inc., New York), pgs 91-114

Brooks, R.F., Bennett, D.C. & Smith, J.A. 1980 Mammalian cell cycles need two random transitions. Cell, <u>19</u>, 493-504

Brown, K.D. & Blakeley, D.M. 1984 Transforming growth factors: sources, properties and possible roles in normal and malignant cell growth control. Biochemical Society Transactions, <u>12</u>, 168-73

Brown, K.D., Blay, J., Irvine, R.F., Heslop, J.P. & Berridge, M.J. 1984 Reduction of epidermal growth factor receptor affinity by heterologous ligands: evidence for a mechanism involving the breakdown of phosphoinositides and the activation of protein kinase C. Biochem. Biophys. Res. Commun. <u>123</u>, 377-384

Bullough, W. & Lawrence, E. 1968 Epidermal chalone and mitotic control in the Vx2 epidermal tumour. Nature, 220, 134-135

Bullough, W.S., 1975 Mitotic control in adult mammalian tissues. Biol. Rev. 50, 99-127

Burger, M.M., Bombik, B.M., Breckenridge, B.McL. & Seppard, J.R. 1972 Growth control and cyclic alteration of cyclic AMP in the cell cycle. Nature, <u>239</u>, 161-163

Burgess, A.W. 1985 Hematopoietic growth factors. In "Mediators in cell growth and differentiation", Eds: R.J. Ford & A.L. Maizel (Raven Press, New York), pgs159-169

Burgess, A.W., 1986 Growth factors, receptors and cancer. BioEssays, 5, 15-18

Burgess, A.W. & Metcalfe, D. 1980 The nature and action of granulocyte-macrophate colony stimulating factor. Blood, <u>56</u>, 1947-1958

Burgess, G.M., Godfrey, P.P., McKinney, J.S., Berridge, M.J., Irvine. R.F. & Putney, J.W. 1984 'The second messenger linking receptor activation to internal Ca release in liver'', Nature, <u>309</u>, 63-66

Burns, C.P. & Rozengurt, E. 1984 Extracellular Na<sup>+</sup> and initiation of DNA synthesis: role of intracellular pH and K<sup>+</sup>. J. Cell. Biol. <u>98</u>, 1082-1089

Burwen, S.J. & Jones, A.L. 1987 The association of polypeptide hormones and growth factors with the nuclei of target cells. Trends Biochem. Sci., <u>12</u>, 159-162

Calissano, P., Cattaneo, A., Biocca, S., Alor, L., Mercanti, D. & Levi-Montalani, R. 1984 The nerve growth factor. Exp. Cell. Research, <u>154</u>, 1-9

Callahan, M., Cochran, B.H. & Stiles, C.D. 1985 The PDGF inducible 'competence genes': intracellular mediators of the mitogenic response. In 'Growth factors in biology and medicine' (Ciba Foundation Symposium, 116), Eds: E. Evered, J. Nugent & J. Whelan, (Pitman Publishing Ltd., London), pgs87-91

Cameron, I.L., Smith, N.K.R., Pool, T.B. & Sparks, R.L 1980 Intracellular concentration of sodium and other elements as related to mitogenesis and oncogenesis in vivo. Cancer Res., <u>40</u>, 1493-1500

Campbell, A.K., 1986 What does Ca<sup>2+</sup> do in cells? Cell Calcium, 7, 285-286

Campbell, A.K., Dormer, A.L. & Hallett, M.B., 1985 Coelenterate photoproteins as indicator of cytoplasmic free  $Ca^{2+}$  in small cells. Cell Calcium, <u>6</u>, 69-82

Campisi, J., Gray, H.E., Pardee, A.B., Dean, M. & Sonenshein, G.E. 1984 Cell-cycle control of <u>c-myc</u> but not c-<u>ras</u>. Expression is lost following chemical transformation. Cell <u>36</u>, 241-247

Carney, S.A., 1982 The influence of cyclic nucleotides on DNA synthesis in skin. Br. J. Exp. Path., <u>63</u>, 351-353

Carpenter, G. 1983 The biochemistry and physiology of the receptor-kinase for epidermal growth factor. Mol. Cell. Endocrinol., <u>31</u>, 1-19

Carpenter, G, 1984 Properties of the receptor for epidermal growth factor. Cell <u>37</u>, 357-358

Carpenter, G. & Cohen, S. 1979 Epidermal growth factor. Annu. Rev. Biochem., 48, 193-216

Cassell, D., Rothenberg, P., Zhuang, Y-X., Deuel, T.F. & Glaser, L 1983 Platelet-derived growth factor stimulates Na<sup>+</sup>/H<sup>+</sup> exchange and induces cytoplasmic alkalinization in NR6 cells. Proc. Natl. Acad. Sci., USA, <u>80</u>, 6224-6228

Castagna, M., Takai, Y., Kanbuchi, K., Sano, K., Kikkawa, U & Nishzuka, Y. 1982 Direct activation of calcium activated phospholipid-dependent protein kinase by tumour-promoting phorbol esters. J. Biol. Chem., 257, 7847-7851

Castro, J.E., 1974 Orchidectomy and the immune response: Effect of orchidectomy on lymphoid tissues of mice. Proceedings of the Royal Society Series B. <u>185</u>, 425-436

Castro, J. & Hamilton, D. 1972 Adrenalectomy and orchidectomy as immunopotentiating procedures. Transplantation. <u>13</u>, 614-616

Caufriez, A. & Copinschi, G. 1986 Somatomedins & Steroids. Hormone Res., 24, 185-188

Cauvin, C., Loutzenhiser, R. & Van Breeman, 1983 Mechanisms of calcium antagonist-induced vasodilation. Annu. Rev. Pharmacol. Toxicol. 23, 373-396

Cavero, I. & Spedding, M. 1983 "Calcium antagonists", A class of drugs with a bright future. Part 1: Cellular calcium homeostasis and calcium as a coupling messenger", Life Sciences, <u>33</u>, 2571-2581

Cech, S.Y., Broaddus, W.C. & Maguire, M.E. 1980 Adenylate cyclase. The role of magnesium and other divalent cations. Molec. & Cell Biochem. 33, 67-92

Cech, S.Y. & Maguire, M.E. 1982 Magnesium regulation of the Beta-receptor-adenylate cyclase complex. Molecular Pharmacology, 22, 267-273

Chafouleas, J.G. & Means, A.R. 1982 "Calmodulin is an important regulatory molecle in cell proliferation", In Ions. Cell prolifertion and cancer', (Academic Press, New York), Eds: A.L. Boynton, W.L. McKeelan & J.F. Whitfield, pp440-465

Charest, R., Prpic, V., Exton, J.H. & Blackmore, P.F. 1985 Stimulation of inositol triphosphate formation in hepatocytes by vasopressin, adrenaline and angiotensin II and its relationship to changes in cytosolic free Ca<sup>2+</sup>. Biochem. J. 227, 79-90

Cheung, W.Y. 1982 "Calmodulin", Sci. American, 246, 48-56

Cheung, W.Y. 1984 Calmodulin: its potential role in cell proliferation and heavy metal toxicity, Federation Proceedings, <u>43</u>, 2995-2999

Chiarugi, V., Porciatti, F., Pasquali, F. & Bruni, P. 1985 Transformation of Balb/3T3 cells with EJ/T24/H-RAS oncogene inhibits adenylate cyclase response to ß-adrenergic agonist while increases muscarinic receptor dependent hydrolysis of inositol lipids. Biochem. Biophys. Res. Commun., 132, 900-907

Childs, C.B., Proper, J.A., Tucker, R.F. & Moses, H.L., 1982 Serum contains a platelet-derived transforming growth factor. Proc. Natl. Acad. Sci., USA, <u>79</u>, S312-S316

Chinkers, M. & Garbers, D.L. 1986 Suppression of protein tyrosine kinase activity of the epidermal growth factor receptor by epidermal growth factor. J. Biol. Chem., 261, 8295-8297

Chiodi, H. 1976 Thymus hypertrophy induced by castration in old male rats in mice. Federation Proceedings, <u>35</u>, 277

Clark, J.H. & Peck, E.J. 1977 Steroid hormone receptors: basic principles and measurements. In 'Receptors and Hormone Action', Eds: O'Malley, B.W. & Birnbaumer, L. (Academic Press, New York), pgs 383-410

Clarke, A.G. & MacLennan, K.A. 1986 The many facets of thymic involution. Immunol. Today 7, 204-205

Clemens, M.J. & McNurlan, M.A. 1985 Regulation of cell proliferation and differentiation by interferons. Biochem. J., <u>226</u>, 345-360

Clemmons, O.R., Underwood, L.E. & Van Wyck, J.J. 1981 Hormonal control of immunoreactive somatomedin production by cultured human fibroblasts, J. Clin. Invest., <u>67</u>, 10-17

Cochet, C., Gill, G.W., Meisenhelder, J., Cooper, J.A. & Hunter, T. 1984 C-kinase phosphorylates the epidermal growth factor receptor and reduces its epidermal growth factor stimulated tyrosine protein kinase activity, J. Biol. Chem., 259, 2553-2558

Cochran, B.H., Reffel, A.C. & Stiles, C.D. 1983 Molecular cloning of gene sequences regulated by platelet-derived growth factor. Cell, <u>33</u>, 939-947

Cochran, B.H., Zullo, J., Verma, I.M. & Stiles, C.D. 1984 Expression of the <u>c-fos</u> gene and of an fos-related gene is stimulated by platelet derived growth factor. Science, <u>226</u>, 1080-1082

Cockroft, S. 1984  $Ca^{2+}$  dependent conversion of phosphatidylinositol to phosphatidate in neutrophils stimulated with fMet-leu-phe or ionophore A23187, Biochem. Biophys. Acta., <u>795</u>, 37-46

Cockcroft, S., Barrowman, M.M. & Gomperts, B.D. 1985 Breakdown and synthesis of polyphosphoinositides in fMetLeuPhe-stimulated neutrophils. FEBS Lett., 181, 259-63

Cockroft, S. 1987 Phosphosphoinositide phosphodiesterase: regulation by a novel guanine nucleotide binding protein, Gp. Trends. Biochem. Sci., <u>12</u>, 75-78

Cockroft, S., Bennett, J.P. & Gomperts, B.D. 1981 The dependence on  $Ca^{2+}$  of phosphatidylinositol breakdown and enzyme secretion in rabbit neutrophils stimulated by formylmethonyl leucylphenylalanine or ionomycin. Biochem. J., 200, 501-508

Cockroft, S. & Gomperts, B.D. 1985 Role of guanine nucleotide binding protein in the activation of polyphosphoinositide phosphodiesterase. Nature, <u>314</u>, S34-S36

Codina, J., Hildebrandt, J.D., Sunyer, T., Sekura, R.D., Manclark, E.R., Ivengar, R. & Bimbaumer, L. 1983 Mechanisms in the vectorial receptor-adenylate cyclase signal transduction. In: 'Adv. Cyclic Nucleotide Research and Protein Phosphorylation', Vol 17, Ed: Greengard, P., Robison, G.A., Paoletti, R. and Nicosia, S., Raven Press (N. York), pgs111-125

Coffrey, R.G., Hadden, E.M. & Hadden, J.W. 1977 Evidence for cGMP and calcium mediation of lymphocyte activation by mitogens. J. Immunol., <u>119</u>, 1387-1394

Cohen, S. 1959 Purification and metabolic effects of nerve growth- promoting protein from snake venom. J. Biol. Chem., <u>234</u>, 1129-1137

Cohen, S. 1962 Isolation of a mouse submaxillary gland protein accelerating incisor eruption and eyelid opening in the newborn animal. J. Biol. Chem., 237, 1562-1568

Cohen, S. 1984 Epidermal growth factor (EGF): historical perspectives. In "Hormonal proteins and peptides", Vol 12. Ed: C.H. Li, (Academic Press, London), pgs 299-304

Cohen, S. & Taylor, J.M. 1974 Epidermal growth factor: chemical and biological characterization. Recent Progress in Hormone Res., <u>30</u>, 533-550

Cohn, D.A. 1979a High sensitivity to androgen as a contributing factor in sex differences in the immune response. Arthritis Rheum., 22, 1218-1221

Cohn, D.A. 1979b Sensitivity to androgen. A possible factor in sex differences in the immune response. Clin. Exp. Immunol., <u>38</u>, 218-222

Cohn, D.A. 1986 Low immune responses in high androgen responder CS7BR/cdj males. Clin. Exp. Immunol., <u>63</u>, 210-217

Collins, M.K.L. & Rozengurt, E. 1984 Homologous and heterologous mitogenic desensitization of Swiss 3T3 cells to phorbol esters and vasopressin: role of receptor and postreceptor steps. J. Cell Physiol., <u>118</u>, 133-142

Conner, C.G., Moore, P.B., Brady, R.C., Horn, J.P., Arlinghaus, B.B. & Dedman, J.R. 1983 The role of calmodulin in cell transformation. Biochem. Biophys. Res. Commun., <u>112</u>, 647-654

Coppock, D.L., Covey, L.R. & Straus, D.S. 1980 Growth response to insulin in mouse melanoma cells and fibroblast melanoma hybrids. J. Cell. Physiol., <u>105</u>, 81-92

Cooper, J.A., Sefton, B.M. & Hunter, T. 1984 Diverse mitogenic agents induce the phosphorylation of two related 42 000 dalton protein on tyrosine in quiescent chick cells. Mol. Cell. Biol., 4, 30-37

Corps, A.N., Hesketh, T.R. & Metcalfe, J.C. 1982 Limitations of the use of phenothiazines and local anaesthetics as indicators of calmodulin function in intact cells. FEBS Lett, <u>138</u>, 280-284

Corvera, S. & Czech, M.P. 1985 Mechanism of insulin action on membrane protein recycling: a selective decrease in the phosphorylation state of insulin-like growth factor II receptors in the cell surface membrane. Proc. Natl. Acad. Sci. USA, <u>82</u>, 7314-7318

Cotecchia, S., Leeb-Lundberg, L.M.F., Hagen, P.O., Lefkowitz, R.J. & Caron, M.G. 1985 Phorbol ester effects on  $\alpha$ -adrenoceptor binding and phosphatidylinositol metabolism in cultured vascular smooth muscle cells. Life Sciences, <u>37</u>, 2389-2398

Coussens, L., Parker, P.J., Rhee, L., Yang-Feng, T.L., Chen, E., Waterfield, M.D. & Ullrich, A. 1986 Multiple distinct forms of bovine and human protein kinase C suggest diversity in cellular signalling pathways. Science 233, 859-866

Cox, J.A. 1984 "Sequential events in calmodulin on binding with calcium and interaction with target enzymes", Fed.Proc., <u>43</u>, 3000-3004

Criss, W.E. & Kakiuchi, S. 1982 Calcium: calmodulin and cancer. Fed. Proc. 41, 2289-2291

Cuatecasas, P.A., 1974 Membrane receptors. Annu. Rev. Biochem., 43, 3000-3010

Curran, T., Bravo, R. & Miller, R. 1985 Transient induction of c-fos and c-myc. Cancer Surveys, <u>4</u>, 654-681

Currie, G., 1981 Platelet-derived growth factor requirements for in vitro proliferation of normal and malignant mesenchymal cells. Br. J. Cancer, <u>43</u>, 335-343

Cuthbertson, K.S.R., Whittingham, D.G. & Cobbold, P.H., 1981 Free  $Ca^{2+}$  increases in exponential phases during mouse oocyte activation. Nature, <u>294</u>, 754-757

Cutler, G.B., Cassorla, F.G., Ross, J.L., Pescovitz, O.H., Barnes, K.M., Comite, F., Feuillan, P.P., Laue, L., Foster, C.M., Kenigsberg, D., Carusco-Nicoletti, M., Garcia, H.B., Uriate, M., Hench, K.D., Skerda, M.C., Long, L.M. and Loriaux, D.L., 1986 'Pubertal growth: Physiology & Pathophysiology' Recent progress in hormone research. <u>42</u>, 443-450

Czech, M.P., 1985 The nature and regulation of the insulin receptor: structure and function. Ann. Rev. Physiol., <u>47</u>, 357-381

Czech, M.P., Oppeheimer C.L. & Massagué, J. 1983 Interrelationships among receptor structures for insulin and peptide growth factors. Fed. Proceedings, <u>42</u>, 2598-2601

Davis, R.J. & Czech, M.P., 1985 Tumour promoting phorbol diesters cause the phorylation of epidermal growth factor receptors in normal fibroblasts at threonine-654. Proc. Natl. Acad. Sci., USA, <u>82</u>, 1974-1978

Davis, R.J. & Czech, M.P., 1986 Inhibition of the apparent affinity of the epidermal growth factor receptor caused by phorbol diesters correlates with phosphorylation of threonine-654 but not other sites on the receptor. Biochem. J., 233, 435-441

Davis, R.J., Ganong, B.R., Bell, R.M. & Czech, M.P., 1985 Sn-1,2, Dioctanoylglycerol. A cell-permeable diacylglycerol that mimics phorbol diester action on the epidermal growth factor receptor and mitogenesis. J. Biol. Chem., 260, 1562-1566

Dawson, P. 1985 GTP enhances inositol triphosphate-stimulated Ca<sup>2+</sup> release from rat liver microsomes. FEBS Lett, <u>185</u>, 147-150

De Chaffoy, De Courcelles, D., Roevens, P., Van Belle, H. 1987 Prostaglandin E, and forskolin antagonize C-kinase activation in the human platelet. Biochem. J., 244, 93-99

De Coursey, T.E., Chandy, K.G., Gupta, S. & Cahalan, M.D. 1984 Voltage gated K<sup>+</sup> channels in human T lymphocytes: a role in mitogenesis. Nature, <u>307</u>, 465-468

Degliantoni, G., Murphy, M. Kobayashi, M. Francis, M.K., Perussia, B. & Trinchieri, G. 1985 Natural killer (NK) cell-derived haematopoietic colony-inhibiting activity and NK cytotoxic factor. J. Exp.Med. <u>162</u>, 1513-1530

DeLarco, J.E. & Todaro, G.J. 1978 Growth factors from murine sarcoma virus transformed cells. Proc. Natl. Acad. Sci., USA, 75, 4001-4005

Denton, R.M. & McCormack, J.G., 1980 On the role of the calcium transport cycle in heart and other mammalian mitochondria. FEBS Lett <u>119</u>, 1-8

Denton, R.M. & McCormack, J.G., 1985 Ca<sup>2+</sup> transport by mammalian mitochondria and its role in hormone action. Am. J. Physiol., 249, E543-E554

Deuel, T.F. & Huang, J.S. 1984 Platelet-derived growth factor. J. Clin. Investigation, <u>74</u>, 669-676

Deuel, T.F., Kimura, A., Maehama, S. & Tong, B.D. 1985 Platelet derived growth factor: roles in normal and V-sis transformed cells. Cancer Surveys, <u>4</u>, 634-653

Deutsh, C. & Price, M.A., 1982 Cell calcium in human perpheral blood lymphocytes and the effect of mitogen. Biochim. Biophys. Acta., <u>687</u>, 211-218

Dexter, T.M., Heyworth, C.M. & Whetton, A.D. 1985 The role of haemopoietic cell growth factor (Interleukin 3) in the development of haemopoietic cells. In 'Growth Factors in Biology and Medicine', (Ciba Foundation Symposium 116) (Pitman, London), p129-147

Diamantstein, T., Osawa, H., Mouzaki, A. & Josimovic-Alasevic, O. 1986 Regulation of interleukin-2 receptor expression and receptor release. Molecular Immunology, <u>23</u>, 1165-1172

Dipolo, R. & Beagué, L. 1983 The calcium pump and sodium-calcium exchange in squid axons. Annu. Rev. Physiol., 45, 313-24

Dipolo, R. & Beagué, L. 1986 Reverse Na-Ca exchange requires internal Ca and/or ATP in squid axons, Biochim. Biophys. Acta., 854, 298-306

Di Virgilio, F., Lew, D.P. & Pozzan, T. 1984 Protein kinase C activation of physiological processes in human neutrophils at vanishingly small cytosolic  $Ca^{2+}$  levels. Nature, <u>310</u>, 691-693

Di Virgilio, F., Pozzan, T. Wollheim, C.B., Vicentini, L.M. & Meldolesi, J. 1986 Tumour promoter phorbol myristate acetate inhibits  $Ca^{2+}$  influx through voltage-gated  $Ca^{2+}$  channels in two secretory cell lines, PC12 and RINm5F. J. Biol. Chem., <u>261</u>, 32-35

Dockray, G.J. & Hopkins, C.R. 1982 In "Functional Integration of Cells in Animal Tissues. British Society for Cell Biology Symposium 5", Eds: J.D. Pitts & M.E. Finbow, CU Press

Dohlman, H.G., Caron, M.G. & Lefkowitz, R.J., 1987 A family of receptors coupled to guanine nucleotide regulatory proteins. Biochemistry <u>26</u>, 2657-2664

Doolittle, R.F., Hunkapiller, M.W., Flood, L.E., Devare, S.G., Robbus, K.C., Aaronson, S.A. & Antoniades, H.N., simian sarcoma virus onc gene v-sis is derived from the gene (or genes) encoding a platelet derived growth factor. Science, 221, 275-277

Dormer, R.L., Hallett, M.B. & Campbell, A.K. 1985 Measurement of intracellular free Ca<sup>2+</sup>, importance during activation and injury of small cells. In "Control and Manipulation of Calcium Movement", ed: J.R. Parratt, (Raven Press, New York), pgs 1-27

Dougherty, R.W. & Niedel, J.E., 1986 Cytosolic calcium regulates phorbol diester binding affinity in intact phagocytes. J. Biol. Chem., 261, 4097-4100

Dougherty, T.P. 1952 Effect of hormones on lymphatic tissues. Physiol. Rev., <u>32</u>, 379-401

Dower, S.K. & Urdal, D.L. 1987 The interleukin-1 receptor. Immunol. Today, 8,46-51

Downes, C.P. 1985 Receptor-dependent generation of intracellular signals from inositol phospholipids in parotid gland and brain. Biochem. Soc. Transact. 13, 1107-1110

Downes, C.P. & Michell, R.H. 1985 Inositol phospholipid breakdown as a receptor-controlled generator of second messengers. In 'Molecular Mechanisms of Transmembrane Signalling', Eds: M.D. Houslay, P. Cohen, (Elsevier, Amsterdam), pgs 3-56

Downward, J., Waterfield, M.D. & Parker, P.J. 1985 Autophosphorylation and protein kinase C phosphorylation of the epidermal growth factor receptor. J. Biol. Chem., <u>260</u>, 14538-14546

Downward, J., Yarden, Y., Mayes, E., Scrace, G., Totty, N. Stockwell, P., Ullrich, A., Schlessinger, J. & Waterfield, M.D. 1984 Close similarity of epidermal growth factor receptor and <u>verb-B</u> oncogene protein sequences. Nature, <u>307</u>, 521-527

DuBois, E.L. & Tuffanelli, D.L. 1964 J. Am. Med. Assoc., 190, 104-111

Duff, G.W. & Durum, S.K. 1984 The pyrogenic and mitogenic actions of interleukin 1 are related, Nature, <u>304</u>, 449-451

Dulbecco, R. & Elkington, J. 1975 Induction of growth in resting fibroblastic cell cultures by  $Ca^{2+}$ , Proc. Natl. Acad. Sci., USA, <u>72</u>, 1584-1588

Duprez, V. & Dautry-Varsat, A. 1986 Receptor mediated endocytosis of interleukin 2 in a human tumour T cell line. J. Biol. Chem., <u>261</u>, 15450-15454

Dunn, W.A., Conolly, T.P. & Hubbard, A.L. 1986 Receptor mediated endocytosis of epidermal growth factor by rat hepatocytes: receptor pathway. J. Cell. Biol., <u>102</u>, 24-36

Durham, A.C.H. & Walton, J.M. 1982 Calcium ions and the control of proliferation in normal and cancer cells. Bioscience Reports 2, 15-30

Durkin, J.P., Whitfield, T.F. & MacManus, J.P. 1983 The role of calmodulin in the proliferation of transformed and phenotypically normal ts ASV-infected rat cells. J. Cell. Physiol., <u>115</u>, 313-319

Duval, D., Durant, S. & Homo-Delarche, F. 1983 Non-genomic effects of steroids: interactions of steroid molecules with membrane structures and functions. Biochim. Biophys. Acta., 737, 409-442

Edelman, I.S. 1975 Mechanism of action of steroid hormons. J. Steroid Biochem., <u>6</u>, 147-159

Eidinger, D. & Garrett, T.J., 1972 Studies of the regulatory effects of the sex hormones on antibody formation and stem cell differentiation. J. Exp. Med., <u>136</u>, 1098-1116

Ek, B., Westermark, B., Wasteson, A. & Heldin, C.H. 1982 Stimulation of tyrosine specific phosphorylation by platelet-derived growth factor. Nature, <u>295</u>, 419-420

Elgebaly, S., Kunkel, S., Lovett, E.J. & Virani, J. 1982 cAMP differences between clones of high and low malignant fibrosarcoma cells. Oncology, <u>39</u>, 163-166

Elliot, D.A. & Rizack, M.A. 1974 Adenylate cyclase stimulants seem to promote magnesium flux in rat cells. J. Biol. Chem., 249, 3985-3987

Erdos, J.J. & Maguire, M.E. 1983 Hormone sensitive magnesium transport in murine S49 lymphoma cells: characterization and specificity for magnesium. J. Physiol., <u>377</u>, 351-371

Evans, S.W. & Farrar, W.L. 1987 Identity of common phosphoprotein substrates stimulated by interleukin 2 and diacylglycerol suggests a role of protein kinase C for IL2 signal transduction. J. Cell Biochem., <u>34</u>, 47-59

Fantes, P. 1986, Growth factors, G<sub>0</sub> and cell cycle controls. BioEssays, <u>4</u>, 32-33

Farrar, W.L., Cleveland, J.L., Beckner, S.K., Bonvini, E. & Evans, S.W. 1986 Biochemical and molecular events associated with interleukin 2 regulation of lymphocyte proliferation. Immunological Reviews, <u>92</u>, 49-65

Farrar, W.L., Thomas, P.T. & Anderson, W.B. 1985 Altered cytosol/membrane enzyme redistribution on interleukin-3 activation of protein kinase C. Nature, <u>315</u>, 235-237

Fearn, J.C. & King, A.C. 1985 EGF receptor affinity is regulated by intracellular calcium and protein kinase C. Cell, <u>40</u>, 991-1000

Feher, J.J. & Briggs, F.N. 1982 The effect of calcium load on the calcium permeability of sarcoplasmic reticulum. J. Biol. Chem., 257, 10191-10199

Felber, S.M. & Brand, M.O. 1983 Concanavalin A causes an increase in sodium permeability and intracellular sodium content of pig lymphocytes. Biochem. J. 210, 893-897

Feldman, D., Funder, J.W. & Edelman, I.S. 1972 Subcellular mechanisms in the action of adrenal steroids. Am. J. Med., 53, 545-560

Fidelus, R.K., Laughter, A.H. & Twomey, J.J., 1984 The role of mitogens and lymphokines in the induction of ornithine decarboxylase (ODC) in T lymphocytes. J. Immunol., <u>132</u>, 1462-1465

Fisher, D.B. & Mueller, G.C. 1968 An early alteration in the phospholipid metabolism of lymphocytes by phytohemagglutin. In: Proc. Natl. Acad. Sci., USA, 60, 1396-1402

Fiskum, G. & Lehninger, A.L. 1980 The mechanisms and regulation of mitochondrial  $Ca^{2+}$  transport. Fed. Proceedings, <u>39</u>, 2432-2436

Fiskum, G. & Lehninger, A.L. 1982 Mitochondrial regulation of intracellular calcium. In 'Calcium and Cell Function' Vol II, Ed: Cheung, W.Y., (Academic Press (London)), pgs 38-80

Fitzpatrick, F.T.A. & Greenstein, B.D. 1987 Effects of various steroids on the thymus, spleen ventral prostate and seminal vesicles in old orchidectomized rats. J. Endocrinol., <u>113</u>, 51-55

Fitzpatrick, F.T.A., Kendall, M.D., Wheeler, M.J., Adcock, I.M. & Greenstein, B.D. 1985 Reappearance of thymus of ageing rats after orchidectomy. J. Endocrinol., <u>106</u>, R17-R19

Fleischman, L.F., Chahwala, S.B. & Cantley, L. 1985 <u>Ras-</u>transformed cells: altered levels of phosphatidylinositol-4,5-biphosphate and catabolites. Science, <u>231</u>, 407-410

Flier, J.S., Usher, P. & Moses, A.C. 1986 'Monoclonal antibody to the type 1 insulin-like growth factor (IGF-1) receptor blocks IFG-1 receptor mediated DNA synthesis: clarfication of the mitogenic mechanisms of IGF-1 and insulin in human skin fibroblasts'. Proc.Natl.Acad.Sci. (USA) <u>83</u>, 664-668

Ford, W.L. 1978 The preparation and labelling of lymphocytes. In 'Handbook of experimental immunology', Ed: D.M. Weir, (Blackwell Scientific Publications (Oxford)), pgs 23.1-23.22

Foukes, J.G. & Rosner, M.R. 1985 Tyrosine-specific protein kinases as mediators of growth control. In 'Molecular Mechanisms of Transmembrane Signalling', Eds: P. Cohen & M.D. Houslay, Elsevier, pgs217-252

Franks, D.J., Whitfield, J.F. & Durkin, J.P. 1987 Viral P21 KrRAS protein: a potent intracellular mitogen that stimulates adenylate cyclase activity in early G1 phase of cultured rat cells. J. Cell. Biochem., <u>33</u>, 87-94

Freedman, M.H., Raff, M.C. & Gomperts, B. 1975 "Induction of increased calcium uptake in mouse T lymphocytes by concanavalin A and its modulation by cyclic nucleotides", Nature, 255, 378-382

Frelin, C., Barbry, P., Green, R.D., Jean, T., Vigne, P. & Lazdunski, M. 1986 The Na<sup>+</sup>/H<sup>+</sup>antiport of eukaryotic cells: relationship between the kinetic properties of the system and its physiological function. Biochimie, <u>68</u>, 1279-1285

Frey-Wettstein, M. & Craddock, C.G. 1970 Testosterone induced depletion of thymus and marrow lymphocytes as related to lymphopoiesis and haematopoiesis. Blood, <u>35</u>, 257-261

Friedkin, M., Legg, A. & Rozengurt, E. 1979 Antitubilin agents enhance the stimulation of DNA synthesis by polypeptide growth factors in 3T3 mouse fibroblasts. Proc. Natl. Acad. Sci., USA, 76, 3909-3912

Friedkin, M. & Rozengurt, E. 1981 The role of cytoplasmic microtubules in the regulation of the activity of peptide growth factors. Adv. Enz. Regul., 19, 39-59

Friedman, D.L., Johnson, R.A. & Zeilig, C.E. 1976 The role of cyclic nucleotides in the cell cycle. Adv. Cyclic Nucl. Res., 7, 69-114

Froehlick, J.E. & Rachmeler, M. 1972 Effect of adenosine 3-5-cyclic monophosphate on cell proliferation. J. Cell. Biol., 55, 19-31

Froesch, E.R., Schmid, C., Schwander, J. & Zapf, J. 1985 Actions of insulin-like growth factors. Ann. Rev. Physiol., <u>47</u>, 443-467

Frykberg, L., Palmieri, S., Beug, H., Graf, T., Hayman, M.J. & Vennstrom, B. 1983 Transforming capacities of avian erythroblastosis virus mutants deleted in the erb A or erb B oncogenes. Cell, <u>32</u>, 227-238

Fujii, H., Nawa, Y., Tsuchiya, H., Matsuno, K., Fukumoto, T., Fukunda, S. & Kotani, M. 1975 Effect of a single administration of testosterone on the immune response and lymphoid tissue in mice. Cell. Immunol., 20, 315-320

Fujika, H., Tanaka, Y., Miyake, R., Kikkawa, U. & Nishizuka, V. 1984 Activation of calcium activated, phospholipid-dependent protein kinase (protein kinase C) by new classes of tumour promoters: Teleocidin and debromoaplysiatoxin. Biochem. Biophys. Res. Commun., <u>120</u>, 339-343

Gagne, D., Homo, F. & Duval, D. 1980 Steroid-induced inhibition of nucleoside uptkae in isolated mouse thymocytes. Biochim. Biophys. Acta., <u>603</u>, 27-35

Gaudin-Harding, F. 1981 Magnésium et système immunitaire: Données récontes. Mag.-Bull., <u>3</u>, 229-236

Gearing, A.J.H., Johnstone, A.P. & Thorpe, R. 1985 Production and assay of the interleukins. J. Immunol. Methods <u>83</u>, 1-27

Gearing, A.J.H., Wadhwa, M., Perris, A.D. 1985 Interleukin 2 stimulates T cell proliferation using a calcium flux. Immunol. Lett. 10, 297-302

Geisow, M.J. 1986 Cell surface receptors: puzzles & paradigms. BioEssays, 4, 149-151

Geisow, M.J. & Walker, J.H. 1986 New proteins involved in cell regulation by  $Ca^{2+}$  and phospholipids. Trends Biochem. Sci., <u>11</u>, 420-423

Gelfand, E.W., Cheung, R.K. & Grinstein, S. 1984 Role of membrane potential in the regulation of lectin-induced calcium uptake. J. Cell. Physiol., <u>121</u>, 533-539

Gelfand, E.W., Cheung, R.K., Grinstein, S. & Mills, G.B. 1986 Characterization of the role for calcium influx in mitogen-induced triggering of human T cells. Identification of calcium-dependent and calcium-independent signals. Eur. J. Immunol., <u>16</u>, 907-12

Gelfand, E.W., Mills, G.B., Cheung, R.K., Lee, J.W.W. & Grinstein, S. 1987 Transmembrane ion fluxes during activation of human T lymphocytes. Role of  $Ca^{2+}$ , Na<sup>+</sup>/H<sup>+</sup> exchange and phospholipid turnover. Immunological Reviews, <u>95</u>, 59-87

Gelfant, S. 1981 Cycling-non cycling cell transitions in tissue aging, immunological surveillance, transformation and tumour growth. Int. Rev. Cytol., 70, 1-25

Gerson, D.F., Kiefer, H. & Eufe, W. 1982 Intracellular pH of mitogen-stimulated lymphocytes. Science, <u>216</u>, 1009-1010

Gerzer, R., Hamet, P., Ross, A.H., Lawson, J.A. & Hardman, J.G. 1983 Calcium-induced release from platelet membranes of fatty acids that modulate soluble guanylate cyclase. J. Pharmacol. Exp. Ther., <u>226</u>, 180-186

Ghosh-Dastidar, P., Coty, W.A., Griest, R.E., Wood, D.D.L. & Fox, C.F. 1983 Progesterone receptor subunits are high affinity substrates for phosphorylation by epidermal growth factor receptor. Proc. Natl. Acad. Sci., USA, <u>81</u>, 1654-1658

Gibson, C.W., Rittling, S.R., Hirschorn, R.R., Kaczmarek, L., Calabretta, B., Stiles, C.D. & Baserga, R. 1986 Cell cycle dependent genes inducible by different mitogens in cells from different species. Mol. Cell Biochem., 71, 61-69

Gibson, K.D., Ben-porath, E., Doller, H.J. & Segen. B.J. 1977 Influence of thyroid hormones on growth and growth-related processes. In Growth Factors FEBS 1977, Vol 48, Eds: K.W. Kastrup, J.H. Nielsen (Pergamon Press), pgs101-110

Gillies, R.J. 1982 Calcium, calmodulin & cAMP and the control of cellular proliferation. Trends Biochem. Sci., 7, 233-235

Gillis, S. 1983 Interleukin 2: biology and biochemistry. J. Clin. Immunol., 3, 1-13

Gillis, S. & Mizel, S.B. 1981 T cell lymphoma model for the analysis of interleukin-1 mediated T-cell activation. Proc. Natl. Acad. Sci., 78, 1133-1137

Gilman, A.G. 1984 G proteins and dual control of adenylate cyclase. Cell, <u>36</u>, 577-579

Glucksman, A. & Cherry, C.P. 1968 The effect of castration, oestrogens, testosterone and the oestrous cycle on the cortical epithelium of the thymus in male and female rats. J. Anat. 103, 113-133

Godeau, J.F., Schorderet-Slatkine, S., Hubert, P. & Baulieu, E.E. 1978 Induction of maturation in Xenopus laevis oocytes by a steroid linked to a polymer. Proc. Natl. Acad. Sci. USA, 75, 2353-2357

Godeau, F., Persson, H., Gray, H.E. & Pardee, A.B. 1986 C-myc expression is dissociated from DNA synthesis and cell division in <u>Xenopus</u> oocyte and early embryonic development. Embo. J., <u>5</u>, 3571-3577

Goldberg, N.D., Haddox, M.K., NIcol, S.E., Glass, D.B., Sanford, C.H., Kuehl, F.A. & Estersen, R. 1975 Biologic regulation through opposing influences of cyclic GMP and cyclic AMP: the yin yang hypothesis. Adv. Cyclic Nucleotide Res., <u>5</u>, 307-330

Goldspink, D.F. & Goldberg, A.F. 1975 "Influence of pituitary growth hormone on DNA synthesis in rat tissues", Americal Journal of Physiology, <u>228</u>, 302-317

Goldstein, J.L., Brown, M.S., Anderson, R.G.W., Russell, D.W. & Schneider, W.J. 1985 Receptor-mediated endocytosis: concepts emerging from the LDL receptor system. Annu. Rev. Cell. Biol., 1; 1-39

Goldwasser, E. 1975 Erythropoietin and the differentiation of red blood cells. Fed. Proc., <u>34</u>, 2285-2291

Goldwasser, E., Krantz, S.B., Wang, F.F. 1985 Erythropoieti and erythroid differentiation. In 'Mediators in Cell Growth and Differentiation'', Eds: R.J. Ford & A.L. Maizel, (Raven Press, New York), p103-107

Gonzales, R.C., Whitfield, J.F., Boynton, A.L., MacManus, J.P. & Rixon, R.H. 1984 Prereplicative changes in the soluble calmodulin of isoproterenol-activated rat parotid glands. J. Cell. Physiol., <u>118</u>, 257-261

Gospodarowicz, D. 1984 Fibroblast growth factor. In 'Hormonal Proteins and Peptides', Ed: C.H. Li, Vol 12, Academic Press, New York, p205-230

Gorski, J., Toft, D., Shyamala, G., Smith, D. & Notides, A. 1968 Hormone receptors: studies on the interaction of oestrogen with the uterus. Recent Prog. Horm. Res., 24, 45-80

Goren, H.J., White, M.F. & Kahn, C.R. 1987 Separate domains of the insulin receptor contain sites of autophosphorylation and tyrosine kinase activity. Biochemistry, <u>26</u>, 2374-2382

Gospodarowicz, D. 1985 'Epidermal and Fibroblast Growth Factor', p61-90. In 'Control of Animal Cell Proliferation', Vol 1. Ed: A.L. Boynton, H.L. Leffert, Academic Press Inc.

Gospodarowicz, D., Bialecki, H. & Thakral, T.K., 1979 The angiogenic activity of the fibroblast and epidermal growth factor. Exp. Eye. Res., 28, 501-514

Gospodarowicz, D. & Mescher, A.L. 1981 Fibroblast growth factor and vertebrate regeneration. Adv. Neurol., 29, 149-171

Goustin, A.S., Leof, E.B., Shipley, G.D. & Moses, H.L. 1986 Growth factors and cancer. Cancer Research, <u>46</u>, 1015-1029

Goyette, M., Petropoulos, C.J., Shank, P.R. & Fauston, N. 1984 Regulated transcription of c-Ki-ras and c-myc during compensatory growth of rat liver. Mol. Cell Biol., <u>4</u>, 1493-1498

Graft, R., Lappe, M. & Snell, G. 1969 The influence of the gonads and adrenal glands on the immune response to skin grafts. Transplantation, 7, 105-110

Greaves, M. & Janossy, G. 1972 Elicitation of selective T and B lymphocyte responses by cell surface binding ligands. Transplant. Rev., <u>11</u>, 87-92

Green, S. & Chambon, P. 1986 A superfamily of potentially oncogenichormone receptors. Nature, <u>324</u>, 615-617

Green, S., Walter, P., Kumar, V., Krust, A., Bornert, J.M., Argos, P. & Chambon, P. 1986 Human oestrogen receptor cDNA: sequence, expression and homology to v-erb-A. Nature, <u>320</u>, 134-139

Greene, W.C., Parker, C.M. & Parker, C.W. 1976 Opposing effects of mitogenic and non-mitogenic lectins on lymphocyte activation. Evidence that wheat germ agglutinin produces a negative signal. J. Biol. Chem., <u>251</u>, 4017-25

Greenberg, M.E. & Ziff, E.B. 1984 Stimulation of 3T3 cells induces transcription of the c-fos proto-oncogene. Nature, <u>311</u>, 433-438

Greenberg, P.L. & Rozengurt, E. 1982 Role of intracellular degradation of epidermal growth factor in mitogenesis. Exp. Cell. Res., <u>142</u>, 111-117

Greenstein, B.D., Fitzpatrick, F.T.A., & Adcock, I.M 1986 Reappearance of the thymus in old rats after orchidectomy: inhibition of regeneration by testosterone. J. Endocrinol., <u>110</u>, 417-422

Greenstein, B.D., Fitzpatrick, F.T.A., Adcock, I.M., Kendall, M.D. & Wheeler, M.J. 1986 Reappearance of the thymus in old rats after orchidectomy, inhibition of regeneration by testosterone. J. Endocrinol., <u>110</u>, 417-422

Grey, G.O. & Thalhimer, W. 1924 Observations on the effects of insulin introduced into the medium of tissue culture. <u>82</u>, 1609

Grinstein, S. & Cohen, S. 1987 Cytoplasmic  $[Ca^{2+}]$  and intracellular pH in lymphocytes. J. Gen. Physiol. <u>89</u>, 185-213

Grinstein, S., Cohen, S., Goetz, J., Rothstein, A. & Gelfand, E. 1985 Characterization of the activation of Na<sup>+</sup>/H<sup>+</sup> exchange in lymphocytes by phorbol esters: change in cytoplasmic pH dependence of the antiport. Proc. Natl. Acad. Sci., USA, <u>82</u>, 1429-1433

Grinstein, S. & Goetz, J.D. 1985 Control of free cytoplasmic calcium by intracellular pH in rat lymphocytes. Biochim. Biophys. Acta., <u>819</u>, 267-270

Grody, W.W., Schrader, W.T. & O'Malley, B.W. 1982 Activation, transformation and subunit structure of steroid hormone receptors. Endocr. Rev., <u>3</u>, 141-63

Grossman, C.J. 1985 Interactions between the gonadal steroids and the immune system. Science, <u>227</u>, 257-261

Grossman, C.J., Nathan, P., Taylor, B.B. & Sholiton, L.J. 1979a Rat thymic dihydrotestosterone receptor: preparation, location and physiochemical properties. Steroids, <u>34</u>, 539-553

Grossman, C.J. & Roselle, G.A. 1983 The interrelationship of the HPG-thymic axis and immune system regulation. J. Steroid Biochem. <u>19</u>, 461-467

Grossman, C.J., Sholiton, C.J. & Nathan, P. 1979b Rat thymic estrogen receptor-1. Preparation, location and physiochemical properties. J. Steroid Biochem., <u>11</u>, 1233-1240

Grossman, C.J., Sholiton, L.J. & Roselle, G.A. 1982 Estradiol regulation of thymic lymphocyte function in the rat: mediation by serum thymic factors. J. Steroid. Biochem., <u>16</u>, 683-690

Grubbs, R.D. & Maguire, M.E. 1987 Magnesium as a regulatory cation: criteria and evaluation. Magnesium, <u>6</u>, 113-127

Grynkiewicz, G., Poenie, M. & Tsien, R.Y. 1985 A new generation of  $Ca^{2+}$  indicators with greatly improved fluorescent properties. J. Biol. Chem., <u>260</u>, 3440-3450

Gunther, T. & Averdunk, R. 1979 Reduced lectin stimulation of lymphocytes from magnesium-deficient rats. J. Clin. Chem. Clin. Biochem., <u>17</u>, 51-55

Gunther, T. & Vormann, J. 1986 Probable role of protein phosphorylation in the regulation of  $Mg^{2+}$  efflux via  $Na^+/H^+$  antiport. Mag. Bull., <u>8</u>, 307-309

Gunther, T., Vormann, J. & Forster, R. 1984 Regulation of intracellular magnesium by Mg<sup>2+</sup> efflux. Biochem. Biophys. Res. Commun., <u>119</u>, 124-131

Gupta, D. 1977 Androgens and anhandrogens in relation to human sexual maturity. In 'Androgens and Antiandrogens', Ed: Martini, L. & Motta, M., (Raven Press, New York), p295-308

Gurari-Rotman, D., Revel, M., Tartakovsky, B., Segal, S., Harn, T., Handzel, Z. & Levin, S. 1978 Lymphoblastogenesis in Down's Syndrome and its inhibition by human interferon. FEBS Lett <u>94</u>, 187-190

Gutowski, J.K. & Cohen, S. 1983 Induction of DNA synthesis in isolated nuclei by cytoplasmic factors from spontaneously proliferating and mitogen activated lymphoid cells. Cell Immunol., <u>75</u>, 300-311

Guy, G.R., Gordon, J., Michell, R.H. & Brown, G. 1985 Synergism between diacylglycerols and calcium ionophore in the induction of human B cell proliferation mimics the inositol lipid phosphate brakdown signals induced by crosslinking surface immunoglobulin. Biochem. Biophys. Res. Commun., 131, 484-491

Hadden, J.W., Coffey, R.G., Ananthatrishnan, R. & Hadden, E.M. 1979 Cyclic ucleotides and calcium in lymphocyte regulation and activation. Ann. NY. Acad. Sci., <u>332</u>, 241-254

Haddox, M.K. & Greenfield, A.R.L. 1982 "Cyclic AMP and ornithine decarboxylase in cell proliferation", In Ions, Cell Proliferation and Cancer, (Academic Press), pgs433-447

Halenda, S.P. & Feinstein, M.B. 1984 Phorbol myi state acetate stimulates formation of phosphatidyl inositol 4-phosphate and phosphatidyl inositol 4,5-biphosphate in human platelets. Biochem. Biophys. Res. Comm., <u>124</u>, 507-513

Hall, K. & Sara, V.R. 1983 Growth and somatomedins. Vitamins and hormones, 40, 175-233,

Halsey, D.L., Girard, P.R., Kuo, J.F. & Blackshear, P.J. 1987 Protein kinase C in fibroblasts. J. Biol. Chem., <u>262</u>, 2234-2241

Hamilton, M.S. & Hellstrom, I. 1977 Altered immune responses in pregnant mice. Transplantation, 23, 423

Haupe, A., Gober, M., Sherr, G.J. & Galibert, F. 1984 Nucleotide sequence of the feline retroviral oncogen u-fms shows unexpected homology with oncogenes encoding tyrosine-specific protein kinases, Proc. Natl. Acad. Sci., <u>81</u>, 85-89

Hann, S.R., Thompson, C.B. & Eisenman, R.N., 1985 C-myc oncogene protein synthesis is independent of the cell cycle in human and avian cells. Nature, <u>314</u>, 366-369

Haring, H., Kirsch, D., Obermaier, B., Ermel, B. & Machicao, F. 1986 Tumour-promoting phorbol esters increase the Km of the ATP binding site of the insulin receptor kinase from rat adipocytes. J. Biol. Chem., <u>261</u>, 3869-3875

Harrison, R.W., Fairfield, S. & Orth, D. 1976 Multiple glucocorticoid binding components of intact AT-20 D-1 mouse pituitary tumour cells. Biochim. Biophys. Acta. <u>444</u>, 487-496

Hashimoto, Y. & Soderling, T.R. 1987 Calcium, calmodulin-dependent protein kinase 11 and calcium. Phospholipid-dependent protein kinase activities in rat tissues assayed with a synthetic peptide. Arch. Biochem. Biophys., 252, 418-425

Kaukkamaa, M. 1987 Membrane associated steroid hormone receptors. In 'Steroid Hormone Receptors: their intracellular localization", Ed: C.R. Clark, (Ellis Horwood Ltd, (Chichester, England)), pgs155-169

Hayles, J. & Nurse, P. 1986 Cell Cycle Regulation in Yeast. J. Cell. Sci. Suppl., 4, 155-170

Healy, J.W. & Neeman, S. 1967 haemolysin response of rats treated neonatally with steroids. Endocrinology, <u>80</u>, 762-765

Heby, O. & Anderson, G. 1980 Polyamines and the cell cycle. In 'Polyamines and Biomedical Research', Ed: J.M. Gaugas, (Wiley (New York)), pgs17-34

Heizmann, C.W. & Berchfold, M.W. 1987 Expression of parvalbumin and other  $Ca^{2+}$ -binding proteins in normal and tumour cells. A topical review. Cell calcium,  $\underline{8}$ , 1-41

Heldin, C.H., Johnsson, A., Wennergren, S., Wernstedt, C., Betsholtz, C. & Westermark, B. 1986 A human osteosarcoma cell line secretes a growth factor structurally related to a homodimer of PDGF-A-chains. Nature, <u>319</u>, 511-514

Heldin, C.H. & Westermark, B. 1984 Growth factors: mechanismof action and relation to oncogenes. Cell, <u>37</u>, 9-20

Helmreich, E.J.M. & Elson, E.L. 1984 Mobility of proteins and lipids in membranes. In 'Advances in Cyclic Nucleotide and Protein Phosphorylation Research', Vol 18, Eds: P. Greengard, G.A. Robison, (Raven Press, New York), pgs1-61

Hemler, M.E., Brenner, M.B., McLean, J.H. & Strominger, J.L. 1984 Antigenic stimulation regulates the expression of interleukin 2 receptor on human T cells. Proc. Natl. Acad. Sci. (USA), <u>81</u>, 2172-2175

Hendry, I.A. 1977 Cell division in the developing sympathetic nervous system. Neurocytology, <u>6</u>, 299-309

Henquin, J.C., Nenquin, M., Awouters, P. & Cogneau, M. 1986 Magnesium uptake by pancreatic islet cells is modulated by stimulators and inhibitors of the B-cell function. Eur. J. Biochem. <u>155</u>, 653-657

Henquin, J.C., Tanagawa, T., Nenquin, M. & Cogneau, M. 1983 Glucose modulates Mg<sup>2+</sup> fluxes in pancreatic islet cells. Nature, <u>301</u>, 73-74

Heskleth, T.R., Bavetta, S., Smith, G.A. & Metcalfe, J.C. 1983 Duration of the calcium signal in the mitogenic stimulation of thymocytes. Biochem. J., <u>214</u>, 575-579

Hesketh, T.R., Moore, J.P., Morris, J.D.H., Taylor, M.V., Rogers, J., Smith, G.A. & Metcalfe, J.C. 1985 A common sequence of calcium and pH signals in the mitogenic stimulation of eukaryotic cells, Nature, <u>313</u>, 481-484

Hesketh, T.R., Smith, G.A. & Metcalfe, J.C 1982 "Calcium and lymphocyte activation", In 'Ions, Cell Proliferation and Cancer', Eds: A.L. Boynton, W.L. McKeehan, J.F. & Whitfield, (Academic Press (New York)), pgs397-415

Hesketh, T.R., Smith, G.A., Moore, J.P., Taylor, M.V. & Metcalfe, J.C. 1983 "Free cytoplasmic calcium concentration and the mitogenic stimulation of lymphocytes", J. Biol. Chem., <u>258</u>, 4876-4882

Higashijima, T., Ferguson, K.M., Sternweis, P.C., Smigel, M.D. & Gilman, A.G. 1987 Effects of  $Mg^{2+}$  and the  $\beta\nu$ -subunit complex on the interactions of guanine nucleotides with G proteins. J. Biol. Chem., <u>262</u>, 762-766

Hirata, M., Sasaguri, T., Hamachi, T., Hashimoto, T., Kukita, M. & Koga, T. 1985 Irreversible inhibition of  $Ca^{2+}$  release in saponin-treated macrophages by the photoaffinity derivative of inositol-1,4,5-trisphosphate. Nature, <u>317</u>, 723-725

Hizuka, N., Takano, K., Shizume, K., Asakawa, K., Miyakawa, M., Tanaka, I. & Horikawa, R. 1986 Insulin-like growth factor 1 stimulates growth in normal growing rats. Eur. Jou. of Pharmacology <u>125</u>, 143-46

Hollenberg, M.D. 1982 "Membrane receptors and hormone action 1: New trends related to receptor structure and receptor regulation", In 'More About Receptors', Ed: J.W. Lambe, (Elsevier Press), pgs 1-6

Hollenberg, S.M., Weinberger, C., Ong, E.S., Cerelli, G., Oro, A., Lebo, R., Thompson, E.B., Rosenfield, M.G. & Evans, R.M. 1985 Primary structure and expression of a functional human glucocorticoid receptor cDNA, Nature, <u>318</u>, 635-641

Holley, R.W. 1975 Control of growth of mammalian cells in cell culture. Nature, 258, 487-90

Holley, R.W., Baldwin, J.H., Greenfield, S. & Armour, R. 1985 A growth regulatory factor that can both inhibit and stimulate growth. In 'Growth factors in biology and medicine', Pitman (Ciba Foundation Symposium, 116, London), pgs241-252

Holley, R.W., Bohlen, P., Fava, R. & Baldwin, J.H. 1980 Purification of kidney epithelial cell growth inhibitors, Proc. Natl. Acad. Sci. (USA), 77, 5989-5992

Holley, R.W. & Kiernan, J.A. 1968 "Contact inhibition" of cell division in 3T3 cells. Proc. Natl. Acad. Sci., <u>60</u>, 300-304

Holley, R.W. & Kiernan, J.A. 1974 Control of the initiation of DNA synthesis in 3T3 cells: serum factors. Proc. Natl. Acad. Sci., USA, 71, 2908-2911

Hollingsworth, E.B., Ukena, D. & Daly, J.W. 1986 The protein kinase C activator phorbol-12-myristate-13-acetate enhances cyclic AMP accumulation in phenochromocytoma cells. FEBS Lett., <u>196</u>, 131-4

Homo, F., Dardenne, M. & Duval, D. 1980 "Effects of steroids on concanavalin A-induced blast transformation of mouse lymphoid cells", Cell. Immunol., <u>56</u>, 381-391

Homo, F. & Simon, J. 1981 Early effect of steroids on 45 calcium uptake by mouse thymocytes. Biochem. Biophys. Res. Commun. <u>102</u>, 458-465

Hopkins, C.R. 1985 Gated pits and their role in membrane receptor internalization. In "Molecular mechanisms of transmembrane signalling", Ed: P. Cohen & M.D. Houslay, pgs337-357, Elsevier (Amsterdam)

Hopkins, C.R. 1986a Uptake and intracellular processing of cell surface receptors: current concepts and prospects. In 'New Insights into Cell and Membrane Transport Processes', Eds: G. Poste & S.T. Crooke, (Plenum Press, New York), pgs347-360

Hopkins, C.R. 1986b Membrane boundaries involved in the uptake and intracellular processing of cell surface receptors. Trends. Biochem. Sci., <u>11</u>, 473-477

Horowitz, J.B., Kaye, J., Conrad, P.J., Katz, M.E. & Janeway, C.A. 1986 Autocrine growth inhibition of a cloned line of helper T cells. Proc. Natl. Acad. Sci., (USA), <u>83</u>, 1886-1890

Hosang, M. & Shooter, E.M. 1987 The internalization of nerve growth factor by high affinity receptors on pheoctromocytoma PC12 cells. EMBO J. <u>6</u>, 1197-1202

Hosseini, J.M. & Elin, R.J. 1985 Magnesium variability of lymphocytes from cell cuure. J. Am. Coll. Nutr., <u>4</u>, 613-7

Houslay, M.D. 1985 Renaissance for cyclic GMP? Trends Biochem. Sci., 10, 465-466

Houslay, M.D. 1987 Egg activation unscrambles a potential role for  $IP_4$ . Trends Biochem. Sci., <u>12</u>, 1-2

Hsu, Y.M. & Wang, J.L. 1986 Growth control in cultured 3T3 fibroblasts V Purification of an M. 13,000 polypeptide responsible for growth inhibitory activity. J. of Cell Biology, <u>102</u>, 362-369

Huber, S.A., Job, L.P. & Auld, K.R. 1982 Influence of sex hormones on coxsackie B-3 virus infections in Balb/c mice. Cell Immunol., <u>67</u>, 173-179

Huckel, C., Brock, J., Rychly, J., Schulz, U. & Schulze, H.A. 1985 Qualitative and quantitative Con A binding to mouse lymphocytes, T cell enriched spleen cells and allogeneically stimulated lymphocytes. Archium Imminologiae Therapur. Experimentalis, <u>33</u>, 645-57

Huggins, J.P. & England, P.J., 1985 The control of calcium pumps and channels and their role in regulating intracellular calcium. In 'Molecular mechanisms of transmembrane signalling", Eds: P. Cohen & M.D. Housley, (Elsevier (Amsterdam)), pgs57-87

Hui, D.Y. & Harmony, J.A.K. 1980 Phosphatidylinositol turnover in mitogen-activated lymphocytes. Biochem. J., <u>192</u>, 91-98

Hunt, N.H. & Martin, T.J. 1979 Cyclic nucleotide metabolism in tumours. Aust. N.Z. Med. <u>9</u>, 584-599

Hunt, N.H. & Martin, T.J. 1980 Hormone receptors and cyclic nucleotides, significance for growth and function of tumours. Molec. Aspects Med. <u>3</u>, 59-118

Hunter, T. 1982 Phosphotyrosine - a new protein modification. Trends. Biochem. Sci., <u>8</u>, 246-249

Hunter, T. 1984 The proteins of oncogenes. Sci. American. 251, 60-69

Hunter, T. & Cooper, J.A. 1985 Protein-tyrosine kinases. Annu. Rev. Biochem., 54, 897-930

Hunter, T., Ling, N. & Cooper, J.A. 1984 Protein kinase C phosphorylation of the EGF receptor at a threonine residue close to the cytoplasmic face of the plasma membrane, Nature, <u>311</u>, 480-483

Ihle, J.N. 1985 Biochemical and biological properties of interleukin-3, a lymphokine mediating the differentiation of a lineage of cells that includes prothymocytes and mastlike cells. Contemp. Top. Mol. Immunol., 10, 93-119

Imboden, J.B. 1988 The regulation of intracellular signals during lymphocyte activation. Immunology Today, <u>9</u>, 17-18

Imboden, J.B. & Stobo, J.D. 1985 Transmembrane signalling by the T cell antigen receptor. J. Exp. Med. <u>161</u>, 446-456

Imboden, J.B., Weiss, A. & Stobo, J.D. 1985 Transmembrane signalling by the T3-antigen receptor complex. Immunol. Today, <u>6</u>, 328-331

Iqbal, Z. & Koenig, H. 1985 Polyamines appear to be second messengers in mediating  $Ca^{2+}$  fluxes and neurotransmitter release in potassium depolarised synaptosomes. Biochem. Biophys. Res. Commun., <u>133</u>, 563-573

Irvine, R.F. & Moor, R.M. 1986 Micro-injection of inositol 1,3,4,5-tetratrisphosphate activates sea urchin eggs by a mechanism dependent on external  $Ca^{2+}$ . Biochem. J. <u>240</u>, 917-920

Isakov, N., Mally, M.I., Scholz, W. & Altman, A. 1987 T-lymphocyte activation: the role of protein kinase C and the bifurcating inositol phospholipid signal transduction pathway. Immunological Reviews, <u>95</u>, 89-111

Isakou, N., Scholz, W. & Altman, A. 1986 Signal transduction and intracellular events in T-lymphocyte activation. Immunology Today, 7, 271-277

Isaksson, O.G.P., Jansson, J.O. & Gause, I.A.M. 1982 Growth hormone stimulates longitudinal bone growth directly. Science, <u>216</u>, 1237-39

Isaksson, O.G.P., Eden, S. & Jansson, J.O. 1985 Mode of action of pituitary growth hormone on target cells, Ann. Rev. Physiol., <u>47</u>, 483-499

Ives, H.E. & Daniel, T.O. 1987 Interrelationship between growth factor-induced pH changes and intracellular Ca<sup>2+</sup>. Proc. Natl. Acad. Sci. USA, <u>84</u>, 1950-1954

Iyengar, R. & Birnbaumer, L. 1982 Hormone receptor modulates the regulatory component of adenylyl cyclase by reducing its requirement for  $Mg^{2+}$  and enhancing its extent of activation by guanine nucleotides. Proc. Natl. Acad. Sci., USA, <u>79</u>, 5179-5183

Iype, P.T. & McMahon, J.B. 1984 Hepatic proliferation inhibitor. Mol. Cell. Biochem., <u>59</u>, 57-80

Jacobs, S. & Cuatrecasas, P. 1976 The mobile receptor hypothesis and co-operativity of hormone binding. Biochim. Biophys. Acta. 433, 482-495

Jacobs, S., Kull, F.C., Earp, H.S., Svoboda, M.E., Van Wyck, J.J. & Cuatrecasas, P. 1983 Somatomedin-C stimulates the phosphorylation of the B-subunit of its own receptor. J. Biol. Chem., 258, 9581-9584

Jakobs, K.H., Atories, K. & Schultz, G. 1984 Mechanisms and components involved in adenylate cyclase inhibition by hormones. In 'Advances in Cyclic Nucleotide and Protein Phosphorylation Research', Vol 17, Eds: Greengard, P., Robison, G.A., Paoletti, R. & Nicosia, S., Raven Press (N. York), pgs135-143

James, R. & Bradshaw, R.A. 1984 Polypeptide growth factors. Annu. Rev. Biochem. 53, 259-92

Janne, J., Poso, H. & Raina, A. 1978 Polyamines in rapid growth and cancer. Biochim. Biophys. Acta. <u>473</u>, 241-293

Jansson, J., Ekberg, S., Isaksson, O. & Eden, S. 1984 Influence of gonadal steroids on age- and sex-related secretory patterns of growth hormone in the rat. Endocrinology, <u>114</u>, 1287-1047

Jansson, J., Ekberg, S., Isaksson, O., Mode, A. & Gustafsson, J. 1985 Imprinting of growth hormone secretion, body growth and hepatic steroid metabolism by neonatal testosterone. Endocrinology 117, 1881-1889

Jensen, E.V. & DeSombre, E.R., 1972 Mechanism of action of the female sex hormones. Annu. Rev. Biochem., <u>41</u>, 203-230

Jensen, E.V. 1987 Cytoplasmic versus nuclear localization of steroid hormone receptors: a historical perspective. In 'Steroid hormone receptors: their intracellular localization', Ed: C.R. Clark, (Ellis Horwood Ltd (Chichester, England)), pgs 13-25

Jensen, E.V., Greene, G.L. Closs, L.E., DeSombre, E.R. & Nadhji, M. 1982 Receptors reconsidered: A 20 year perspective. Rec. Prog. Horm. Res., <u>38</u>, 1-34

Jensen, E.V., Suzuki, T., Kawashima, T., Stumpf, W.E., Jungblut, P.W. & DeSombre, E.R. 1968 A two step mechanism for the interaction of estradiol with rat uterus. Proc. Natl. Acad. Sci., USA, <u>59</u>, 632-638

Jentzsch, K.D., Wellmitz, G., Heder, G., Petzold, E., Buntrock, P. & Gehme, P. 1980 A bovine brain fraction with fibroblast growth factor activity inducing articular cartilage regeneration in vivo. Acta. Biol. Med. Ger., 39, 967-971

Jimenez de Asua, L., O'Farrell, M.K., Clinigan, D. & Rudlannd, P.S. 1977 Temporal sequence of hormonal interactions during the prereplicative phase of quiescent cultured 3T3 fibroblasts. Proc. Natl. Acad. Sci., (USA), 74, 3845-3849

Jimenez de Asua, L. 1980 An order sequence of temporal steps regulate the rate of initiation of DNA synthesis in cultured mouse cells. In 'Control Mechanisms in Animal Cells', Eds: L. Jimenez de Asua, R., Levi-Montalcim, R., Shields, & S. Iacobelli. (Raven Press, New York), pgs173-197

Johnson, J., Epel, D. & Paul, M. 1976 Intracellular pH and activation of sea urchin eggs after fertilization. Nature, <u>262</u>, 661-664

Johnson, L.F., Rao, L.G. & Muench, A.J. 1982 Regulation of thymidine kinase enzyme level in serum stimulated mouse 3T6 fibroblasts. Exptl. Cell. Res., <u>138</u>, 79-85

Joseph, S.K., Coll, K.E., Thomas, A.P., Rubin, R. & Williamson, J.R. 1985 The role of extracellular  $Ca^{2+}$  in the response of the hepatocyte to  $Ca^{2+}$  dependent hormones. J. Biol. Chem. <u>260</u>, 12508-12515

Joseph, S.K., Thomas, A.P., Williams, R.J., Irvine, R.F. & Williamson, J.R. 1984 myo inositol 1,4,5-trisphosphate. A second messenger for the hormonal mobilisation of intracellular Ca<sup>2+</sup>in liver. J. Biol. Chem., <u>259</u>, 3077-3081

Kaczorowski, G.F., Barros, F., Dathmers, J.K., Trimble, M.J. & Cragoe, E.J. 1985 Inhibition of Na<sup>+</sup>/Ca<sup>2+</sup> exchange in pituitary plasma membrane vesicles by analogues of amiloride. Biochemistry, <u>24</u>, 1394-1403

Kaever, V. & Resch, K. 1985 Are cyclic nucleotides involved in the initiation of mitogenic activation of human lymphocytes. Biochim. Biophys. Acta., <u>846</u>, 216-225

Kahn, C.R. 1985 The molecular mechanism of insulin action. Annu. Rev. Med., 36, 429-451

Kaibuchi, K., Takai, Y. & Nishizuka, Y. 1985 Protein kinase C and calcium ion in mitogen response of macrophage depleted human peripheral lymphocytes. J. Biol. Chem., <u>260</u>, 1366-1369

Kaibuchi, K., Tsuda, T., Kikuchi, A., Tanimoto, T., Yamashita, T. & Takai, Y. 1986 Possible involvement of protein kinase C and calcium ion in growth factor-induced expression of c-myc oncogene in Swiss 3T3 fibroblasts. J. Biol. Chem., <u>261</u>, 1187-1192

Kalland, T. 1980 Reduced natural killer activity in female mice after neonatal exposure to diethylstilbesterol. J. Immunol., <u>124</u>, 1297-1301

Kalland, T. & Fosberg, J.G. 1978 Delayed hypersensitivity response to oxazolone in enonatally estrogenized mice. Cancer Lett., <u>4</u>, 141-143

Kamine, J. & Rubin, H. 1976 Magnesium required for serum-stimulation of growth in cultures of chick embryo fibroblasts. Nature, <u>263</u>, 143-145

Kamisaka, Y., Toyoshima, S. & Osawa, T. 1986 Phosphatidylinositol-specific phospholipase C of murine lymphocytes. Arch. Biochem. Biophys., 249, 569-578

Kaplan, P.L. & Ozanne, B. 1983 Cellular responsiveness to growth factors correlates with a cell's ability to express the transformed phenotype. Cell., <u>33</u>, 931-938

Kappas, A., Jones, H.E.J. & Roitt, I.M. 1963 Effects of steroid sex hormones on immunological phenomena. Nature, 198, 902-905

Karaki, H., Hatano, K. & Weiss, G.B. 1983 Effects of magnesium on <sup>45</sup>Ca uptake and release at different sites in rabbit aortic smooth muscle. Pflugers Arch., <u>398</u>, 27-32

Kasuga, M., Fujita-Yamaguchi, Y., Blithe, O.L. & Kahn, C.R. 1983 Tyrosine-specific protein kinase activity is associated with the purified insulin receptor. Proc. Natl. Acad. Sci., USA, <u>80</u>, 2137-41

Kasuga, M., Zick, Y., Blithe, D.L., Crettaz, M. & Kahn, C.R. 1982 Insulin stimulates tyrosine phosphorylation of the insulin receptor in a cell-free system. Nature, <u>298</u>, 667-669

Katz, A.M., 1985 Basic cellular mechanisms of action of the calcium-channel blockers. Am. J. Cardiol., 55, 2B-9B

Kawashima, K., Iwata, S. & Endo, H. 1980 Growth stimulative effect of parathyroid hormone, calcitonin and N6, 02'-dibutyryl adenosine 3,'5'-cyclic monophosphoric acid on chick embryonic cartilage cultivated in a chemically defined medium. Endocrinol. Japon., <u>27</u>, 349-56

Keenan, B.S., Meyer, W.T., Hadjian, A.J. & Migeon, C.J. 1975 Androgen receptor in skin fibroblasts: characterization of a specific 17B-hydroxy- $5\alpha$ -androstane-3-one-protein complex in cell sonicates and nuclei. Steroids, 25, 535-52

Kelleher, D.J., Pessin, J.E., Ruoho, A.E. & Johnson, G.L. 1984 Phorbol ester induces desensitization of adenylate cyclase and phosphorylation of the ß-adrenergic receptor in turkey erythrocytes. Proc. Natl. Acad. Sci., USA, <u>81</u>, 4316-4320

Kelly, K., Cochran, B.H., Stiles, C.D. & Leder, P. 1983 Cell specific regulation of the c-myc gene by lymphocyte mitogens and platelet-derived growth factor. Cell, <u>35</u>, 603-610

Kelly, K., Cochran, B., Stiles, C. & Leder, P. 1984 Regulation of c-myc by growth signals. Curr. Top. Microbiol. Immunol., <u>113</u>, 117-126

Keski-Oja, J., Leof, E.B., Lyons, R.M., Coffey, R.J. & Moses, H.L. 1987 Transforming growth factors and control of neoplastic cell growth. Journal of Cellular Biochemistry, <u>33</u>, 95-107

Khajat, A., Keteui, P., Ter-Minassian-Saraga, L., Citanova, N. & Jayle, M.F. 1975 Surface behaviour of oestradiol-17 beta. Biochem. Biophys. Acta., <u>401</u>, 1-5

Kiess, W. & Belohradsky, B.H. 1986 Endocrine regulation of the immune system. Klin. Wochenschr. <u>64</u>, 1-7

Kikkawa, U. & Nishizuka, Y. 1986 The role of protein kinase C in transmembrane signalling. Annu. Rev. Cell. Biol., 2, 149-178

Killander, D. & Zetterberg, A. 1965 A quantitative cytochemical investigation of the relationship between cell mass and initiation of DNA synthesis in mouse fibroblasts in vitro. Exp. Cell Research <u>40</u>, 12-20

King, A.C. & Cuatrecasas, P. 1982 Resolution of high and low affinity epidermal growth factor receptors. J. Biol. Chem., <u>257</u>, 3053-3060

King, C.S. & Cooper, J.A. 1986 Effects of protein kinase C activation after epidermal growth factor binding on epidermal growth factor receptor phosphorylation. J. Biol. Chem., <u>261</u>, 10073-10078

King, G.L. & Kahn, C.R. 1985 Effect of insulin on growth in vivo and cells in culture. In 'Control of Animal Cell Proliferation', Vol 1, Ed: A.L. Boynton & H.L. Leffert, (Academic Press Inc)., pgs201-249

King, R.J.B. 1986 Receptor structure: a personal assessment of the current status. J. Steroid Biochem., 25, 451-454

King, W.J. & Greene, G.C. 1984 Monoclonal antibodfies localize oestrogen receptor in the nuclei of target cells. Nature, <u>307</u>, 745-747

Kinoshita, Y., Kumura, S. & Fukanizu, M. 1974 Cytolytic effects of glucorticoids on thymic medullary lymphocytes in vitro. Exp. Cell. Res., 87, 387-397

Kishimoto, A., Takai, Y., Mori, T., Kikkawa, U. & Nishizuka, Y. 1980 Activation of calcium and phospholipid-dependent protein kinase by diacylglycerol, its possible relation to phosphatidylinositol turnover. J. Biol. Chem., <u>255</u>, 2273-2276

Klaus, G.C.B. 1986 Unravelling the control of B cells. Nature, 324, 16-17

Klaus, G.C.B. & Hawrylowicz, C.M. 1984 Cell cycle control in lymphocyte stimulation. Immunology Today, 5, 15-19

Klaus, G.C.B., O'Garra, A., Bijsterbosch, M.K. & Holman, M. 1986 Activation and proliferation signals in mouse B cells. VIII. Induction of DNA synthesis in B cells by a combination of calcium ionophores and phorbol myristate acetate. Eur. J. Immunol., <u>16</u>, 92-97

Klausner, R.D., Harford, J. & Van Reswoude, J. 1984 Rapid internalization of the transferrin receptor in KS62 cells is triggered by ligand-binding or treatment with a phorbol ester. Proc. Natl. Acad. Sci., USA, <u>81</u>, 3005-3009

Klee, C.B., Crouch, T.H. & Richman, P.G. 1980 Calmodulin. Annu. Rev. Biochem., 49, 489-515

Klee, C.B. & Newton, D.L., 1985 Calmodulin: an overview. In 'Control and manipulation of calcium movement', Ed: J.R. Parratt, Raven Press (New York), pgs131-146'

Klein, H.H., Freidenberg, G.R., Kladde, M. & Olefsky, J.M. 1986 Insulin activation of insulin receptor tyrosine kinase in intact rat adipocytes. J. Biol. Chem., 261, 4691-4697

Klein, J. & Nagy, Z.A. 1982 Mhc restriction and Ir genes. Adv. Cancer Res., 37, 233-317

Knopf, J.L., Lee, M.H., Sultzman, L.A., Kriz, R.W., Loomis, C.R., Hewick, R.M. & Bell, R.M. 1986 Cloning and expression of multiple protein kinase C cDNA's. Cell, <u>46</u>, 491-502

Koch, B., Lukz-Bucher, B., Briaud, B. & Mialhe, C. 1977 Glucocorticoid binding to plasma membrane of the adenohypophysis. J. Endocrinology, <u>73</u>, 399-400

Koch, K.S. & Leffert, H.L. 1979 Increased sodium ion influx is necessary to initiate rat hepatocyte proliferation. Cell, <u>18</u>, 153-163

Koenig, H., Goldstone, A. & Lu, C.Y. 1983 "Polyamines regulate calcium fluxes in a rapid plasma membrane response", Nature, <u>305</u>, 530-536

Koenig, H., Goldstone, A.D. & Lu, C.Y. 1983b "ß adrenergic stimulation of  $Ca^{2+}$  fluxes, endocytosis, hexose transport and amino acid transport in mouse kidney cortex is mediated by polyamine synthesis", Proc. Natl. Acad. Sci., USA, <u>80</u>, 7210-7214

Koga, T., Nakano, S., Nakayama, M., Kounoue, E., Nagafuchi, S., Nitro, Y. & Yamada, H. 1986 Identification and partial purification of a low-molecular weight growth inhibitor formed by density-inhibited, tumorigenic V79 Chinese hamster cells. Cancer Research, <u>46</u>, 4431-4437

Kohn, L.D. & Shifrin, S. 1982 Receptor structure and function: an exploratory approach using the throtropin receptor as a vehicle. Horiz. Biochem. Biophys.,  $\underline{6}$ , 1-42

Koike, T. 1983 Nerve growth factor-induced neurite outgrowth of rat pheochromocytoma PC12 cells: dependence on extracellular  $Mg^{2+}$  and  $Ca^{2+}$ . Brain Research, 289, 293-303

Komada, H., Nakabayashi, H., Hara, M. Sawada, M., Takaliashi, T., Shiomi, T. & Izutsu, K. 1985 Mechanisms of interleukin-2 mediated signalling and role of calcium in T cell mitogenesis. Immunol. Invest., <u>14</u>, 435-442

Koontz, J.W. & Iwahashi, M. 1981 Insulin as a potent specific growth factor in a rat hepatoma cell line. Science, 211, 947-949

Kram, R., Mamont, P. & Tomkins, G.M. 1973 Pleiotypic control of adenosine 3':5'-cyclic monophosphate: a model for growth control in animal cells. Proc. Natl. Acad. Sci., USA, <u>70</u>, 1432-1436

Kruijer, W., Cooper, J., Hunter, T. & Verma, I.M. 1984 Platelet-derived growth factor induces rapid but transient expression of the c-<u>fos</u> gene and protein. Nature, <u>312</u>, 711-716

Krupp, M.N. & Livingston, J.N. 1978 Insulin binding to solubilized material from fat cell membranes: evidence for two binding species. Proc. Natl. Acad. Sci., USA, 75, 2593-2597

Krust, A., Green, S., Argos, P., Kumar, V., Walter, P. Bornert, J.M. & Chambon, P. 1986 The chicken oestrogen receptor sequence: homology with v-erb A and the human oestrogen and glucocorticoid receptors. Embo. J., <u>5</u>, 891-897

Kwok-Keung Fung, B. 1985 The light-activated cyclic GMP phosphodiesterase system in retinal rods. In 'Molecular mechanisms of transmembrane signalling', Eds: P. Cohen & M.D. Houslay, (Elsevier (Amsterdam)), pgs183-214

Lacal, J.C., Srivastava, S.K., Anderson, P.S. & Aaronson, S.A. 1986 <u>Ras</u> p21 proteins with high or low GTPase activity can efficiently transform NIH/3T3 cells. Cell, <u>44</u>, 609-617

Lachman, L.B. 1985 The purification and biological properties of human interleukin-1, In 'Mediators in Cell Growth and Differentiation', Ed: R.J. Ford & A.L. Maizel, (Raven Press, New York), pgs171-183

Laduron, P.M. 1984 Criteria for receptor sites in binding studies. Biochem. Pharmacol., <u>33</u>, 833-839

Lahita, R.G. 1985 Sex steroids and the rheumatic diseases. Arthritis Rheum., 28, 121-126

L'Allemain, G., Franchi, A., Cragoe, E. & Pouysségur, J. 1984 Blockade of the Na/H antiport abolishes growth factor-induced DNA synthesis in fibroblasts. Structure-activity relationships in the amiloride series. J. Biol. Chem., 259, 4313-4319

Landreth, G.E. & Shooter, E.M. 1980 Nerve growth factor receptors on PC12 cells: ligand-induced conversion from low to high-affinity states. Proc. Natl. Acad. Sci., USA, <u>77</u>, 4651-55

Larsson, E.L. & Coutinho, A. 1979 The role of mitogenic lectins in T-cell triggering. Nature, 280, 239-241

Larsson, O., Zetterberg, A. & Engstrom, W. 1985 Cell-cycle specific induction of quiescence achieved by limited inhibition of protein synthesis: counteractive effect of addition of purified growth factors. J. Cell Sci., <u>73</u>, 375-387

Laychock, S.G. 1983 Identification and metabolism of polyphosphoinositides in isolated islets of Langerhans. Biochem. J., <u>216</u>, 101-106

Ledbetter, J.A., Gentry, L.E., June, C.H., Rabinovitch, P.S. & Purchio, A.F. 1987 Stimulation of T cells through the CD3/T-cell receptor complex: role of cytoplasmic calcium, protein kinase C translocation, and phosphorylation of pp60<sup>c-src</sup> in the activation pathway. Molec. Cell Biol., <u>7</u>, 650-656

Lee, L.S. & Weinstein, I.B. 1978 Tumour-promoting phorbol esters inhibit binding of epidermal growth factor to cellular receptors, Science, <u>202</u>, 313-15

Lee, L.S. & Weinstein, I.B. 1979 Mechanism of tumour promoter inhibition of cellular binding of epidermal growth factor. Proc. Natl. Acad. Sci. USA, <u>76</u>, 5168-72

Lee, M. & Nurse, P. 1987 Complementation used to clone a human homologue of the fusion yeast cell cycle control gene cdc2. Nature, <u>327</u>, 31-35

Leeb-Lundberg, L.M.F., Cotecchia, S., Comasney, J., DeBernardis, J.F., Lefkowitz, R.J. & Caron, M.G 1985 Phorbol esters promote α, adrenergic receptor

phosphorylation and receptor uncoupling from inositol phospholipid metabolism. Proc. Natl. Acad. Sci. USA, <u>82</u>, 5651-5655

Leffert, H.L. & Koch, K.S. 1986 Growth regulation by sodium ion influxes. In 'Control of animal cell proliferation', Ed: A.L. Boynton & H.L. Leffert, (Academic Press (London)), pgs367-413

LeGrue, S.J. 1987 Interleukin-2 stimulus-response coupling is calcium independent. Lynphokine Research, <u>6</u>, 1-11

Leof, E.B., Wharton, W., Van Wyk, J.J., Pledger, W.J. 1982 Epidermal growth factor (EGF) and somatomedin C regulate  $G_1$  progression in competent BALB/C-3T3 cells. Exp. Cell Res. <u>141</u>, 107-115

Le Peuch, C.J., Haiech, J. & Demaille, J.B. 1979 Concerted regulation of cardiac sarcoplasmic reticulum calcium transport by cAMP dependent and Ca<sup>2+</sup> calmodulin dependent phosphorylatoins. Biochemistry, <u>18</u>, 5150-5157

Levi-Montalcini, R. 1952 Effects of mouse tumour transplantation on the nervous system. Ann. NY Acad. Sci., 55, 330-344

Levi-Montalcini, R. 1987 The nerve growth factor: thirty-five years later. EMBO J. <u>6</u>, 1145-1154

Levi Montalcini, R. & Angeletti, P.U., 1968 Nerve growth factor. Physiol. Rev., 48, 534-569

Levi-Montalcini, R. & Booker, B. 1960 Destruction of the sympathetic ganglia in mammals by an antiserum to a nerve growth protein. Proc. Natl. Acad. Sci., USA, 46, 384-391

Levine, B.S. & Coburn, J.W. 1984 Magnesium, the mimic/antagonist of calcium. New Engl. J. Med., <u>310</u>, 1253-1255

Ling, N.R. & Kay, J.E. 1975 DNA synthesis and the cell cycle. In 'Lymphocyte stimulation', (North Holland Publishing Company (Oxford)), pgs357-378

Lichstein, D. & Rodbard, D. 1987 A second look at the second messenger hypothesis. Life Sci., <u>40</u>, 2041-2051

Lichtman, A.H., Segel, G.G. & Lichtman, M.A. 1983 The role of calcium in lymphocyte proliferation (An interpretive review). Blood, <u>61</u>, 413-421

Lieber, E., Hirschhorn, K. & Fundenberg, H.H. 1969 Response of agammaglobulinaemic lymphocytes in mixed lymphocyte culture. Clin. Exp. Immunol., <u>4</u>, 83-87

Lillien, L.E. & Claude, P. 1985 Nerve growth factor is a mitogen for cultured chromaffin cells. Nature, <u>317</u>, 632-634

Linch, D.C., Wallace, D.L. & O'Flynn, K. 1987 Signal transduction in human T lymphocytes. Immunological Reviews, <u>95</u>, 137-159

Lipton, A. 1985 Platelet growth factors: presence and biological significance. In 'Control of Animal Cell Proliferation', Vol 1, Ed: A.L. Boynton & H.L. Leffert, (Academic Press Inc.), pgs151-166

Liskay, R.M. & Prescott, D.M. 1978 Genetic analysis of the  $G_1$  period: isolation of mutants (or variants) with a  $G_1$  period from a Chinese hamster cell line lacking  $G_1$ , Proc. Natl. Acad. Sci., <u>75</u>, 2873-2877

Litosch, I. & Fain, J.N. 1986 Regulation of phosphoinositide breakdown by guanine nucleotides. Life Sci., <u>39</u>, 187-194

Livingston, J.N. & Lockwood, D.H. 1975 Direct measurements of sugar uptake in small and large adipocytes from young and adult rats. J. Biol. Chem., 250, 8353-8360

Lopez-Rivas, A., Adelberg, E.A. & Rozengurt, E. 1982 Intracellular K<sup>+</sup> and the mitogenic response of 3T3 cells to peptide factors. Proc. Natl. Acad. Sci., USA, <u>79</u>, 6275-6279

Lopez-Rivas, A. & Rozengurt, E. 1983 Serum rapidly mobilizes calcium from an intracellular pool in quiescent fibroblastic cells. Biochem. Biophys. Res. Commun. 114, 240-247

Lubin, M. 1982 The potassium sensitivity shift and other matters. In 'Ions, Cell Proliferation and Cancer', Eds: A.L. Boynton, W.L. McKeehan & J.F. Whitfield, (Academic Press (New York)), pgs131-150

Luckasen, J.R., White, J.G. & Kersey, J.H. 1974 Mitogenic properties of a calcium ionophore, A23187, Proc. Natl. Acad. Sci., <u>71</u>, 5088-5090

McCaffrey, P. Ran, W., Campisi, J. & Rosner, M.R. 1987 Two independent growth factor-generated signals regulate c-fos and c-myc mRNA levels in Swiss 3T3 cells. J. Biol. Chem., <u>262</u>, 1442-1445

McCann, P.P., 1980 Regulation of ornithine decarboxylase in eukaryotes. In 'Polyamines in Biomedical Research', Ed: J.M. Gaugas, (Wiley (New York)), pgs109-124
McClain, D.A. & Edelman, G.M. 1976 Analysis of the stimulation-inhibition by lymphocytes exposed to convanavalin. American Jou. Exp. Med., 144, 1494-1508

McCormack, J.G. & Denton, R.M. 1986  $Ca^{2+}$  as a second messenger within mitochondria. Trends Biochem. Sci., <u>11</u>, 258-262

McKeehan, W.L. & Ham, R.G. 1978 Calcium and magnesium ions and the regulation of multiplication in normal and transformed cells. Nature, <u>275</u>, 756-758

McMahon, J.B., Farelly, J.G. & Iype, P.T. 1982 Purification and properties of a rat liver protein that specifically inhibits the proliferation of non-malignant epithelial cells from rat liver. Proc. Natl. Acad. Sci., USA, <u>79</u>, 456-460

McNeil, P.L., McKenna, M.P. & Taylor, D.L. 1985 A transient rise in cytosolic calcium follows stimulation of quiescent cells with growth factors and is inhibitable with phorbol myristate acetate. J. Cell Biol., <u>101</u>, 372-379

Macara, I.G. 1985 Oncogenes, ions and phospholipids. Am. J. Physiol., 248; C3-C11

Macara, I.G., Marinetti, G.V. & Bladuzzi, P.C. 1984 Transforming protein of avian sarcoma virus UR2 is associated with phoshatidylinositol kinase activity: possible role in tumorigenesis. Proc. Natl. Acad. Sci., USA, <u>81</u>, 2728-2732

Macara, I.G., Marinetti, G.V., Livingston, J.N. & Balduzzi, P.C. 1985 Lipid phosphorylating activities and tyrosine kinases: a possible role for phosphatidylinositol turnover in transformation. In 'Growth Factors and Transformation', Ed: J. Feramisco, (Cold Spring Harbor, New York), pgs365-368

MacDonald, H.R. & Nabholz, M. 1986 T cell activation. Annu. Rev. Cell Biol., 2, 231-253

Macleod, R.M., Abad, A. & Eidson, L.L. 1969 In vivo effect of sex hormones on the in vitro synthesis of prolactin and growth hormone in normal and pituitary tumour bearing rats. Endocrinology, 84, 1475-1483

MacManus, J.P., Boynton, A.L. & Whitfield, J.F. 1978 cAMP and calcium as intracycle regulators in the control of cell proliferation. Adv. Cyclic Nucleotide Res., 2, 485-491

MacManus, J.P., Brewer, L.M. & Whitfield, J.F. 1985 The widely-distributed tumour protein oncomodulin, is a normal constituent of human and rodenta placentas. Cancer Lett., <u>27</u>, 145-151

MacManus, J.P. & Whitfield, J.F. 1983 Oncomodulin: a calcium binding protein from hepatoma. In 'Calcium and Cell Function', Ed: W.Y. Cheung, (Academic Press (London)), pgs411-440

Madsen, K., Friberg, U., Roos, P., Eden, S. & Isaksson, O. 1983 Growth hormone stimulates the proliferation of cultured chrondrocytes from rabbit ear and rat rib growth cartilage. Nature, <u>304</u>, 545-47

Maguire, M.E. 1984 Hormone sensitive magnesium transport and magnesium regulation of adenylate cyclase. Trends Pharmacol. Sci., 5, 73-77

Maguire, M.E. & Erdos, J.J. 1978 Magnesium but not calcium accumulation is inhibited by  $\beta$  adrenergic stimulation in S49 lymphoma cells. J. Biol. Chem., <u>253</u>, 6633-6636

Maguire, M.E. & Grubbs, R.D. 1983 Magnesium transport: pharmacology, compartmentation and distinction from calcium transport. Fed. Proc. Ged. Am. Soc. Exp. Biol. <u>42</u>, 300

Mainwarig, V.I.P. 1977 'The mechanism of action of androgens', (Springer-Verlag). Monographs on Endocrinology, Vol 10

Manalan, A.S. & Klee, C.B. 1984 Calmodulin. Adv. Cyclic Nucleotide Protein Phoshoryl. Res., <u>18</u>, 227-2784

March, C.J., Mosley, B., Larsen, A., Cerretti, D.P., Braedt, G., Price, V., Gillis, S., Henney, C.S., Kronheim, S.R., Grabslein, K., Conlon, P.J., Hopp, T.P. & Cosman, D. Cloning, sequence and expression of two distinct human interleukin-1 complementary DNAs. Nature, <u>315</u>, 641-646

Marcu, K.B., 1987 Regulation of expression of the c-myc proto-oncogene. BioEssays, <u>6</u>, 28-32

Marti, U., Burwen, S.J., Barker, M.E., Ferer, A.M., Huling, S. & Jones, A.L. 1987 Hepatocyte nuclear binding of <sup>125</sup>I-epidermal growth factor (EGF). J. Cell Biochem. (Suppl.), <u>11A</u>, 35

Marquardt, H., Huntapiller M.W., Hood, L.E., Twardsk, D.R., De Larco, J.E., Stephenson, J.R. & Todaro, G.J. 1983 Transforming growth factors produced by retrouirus transformed rodent fibroblasts and human melanoma cells: Amino acid sequence homology with epidermal growth factor. Proc. Natl. Acad. Sci., USA, <u>80</u>, 4684-4688

Marquardt, H., Todaro, G.J., Henderson, L.E. & Oroszlan, S. 1981 Purification and primary structure of a polypeptide with multiplication-stimulation activity from rat liver cell cultures. J. Biol. Chem., <u>256</u>, 6895-65

Massagué, J., Blinderman, L.A. & Czech, M.P. 1982 The high affinity insulin receptors mediates growth stimulation in rat hepatoma cells. J. Biol. Chem., <u>257</u>, 13958-63

Mastro, A.M. & Smith, M.C. 1983 Calcium dependent activation of lymphocytes by ionophore, A23187 and a phorbol ester tumor promoter. J. Cell. Physiol., <u>116</u>, 51-56

Mathur, S., Mathur, R.S., Goust, J.M., Williamson, H.O. & Fundenburg, H.H., 1979 Cyclic variations in white cell sub-populations in the human menstrual cycle: correlations with progesterone and oestradiol. Clin. Immunol. Immunopathol., 13, 246-249

Matteson, D.R. & Deutsch, C. 1984 K channels in T lymphocytes: a patch clamp study using monoclonal antibody adhesion. Nature, <u>307</u>, 468-471

Mazia, D. 1974 The cell cycle. Sci. American 230, 55-64

Means, A.R. & Dedman, J.R. 1980 Calmodulin - an intracellular calcium receptor. Nature, <u>285</u>, 73-77

Mendelsohn, J., Muller, M.M. & Bernheim, J.L. 1977 Inhibition of human lymphocyte stimulation by steroid hormones: cytokinetic mechanisms. Clin. Exp. Immunol., <u>27</u>, 127-34

Merritt, J.E., Taylor, C.W., Rubin, R.P. & Putney, J.W. 1986 Evidence suggesting that a novel guanine nucleotide regulatory protein couples receptors to phospholipase C in exocrine pancreas. Biochem. J., 236, 337-343

Metcalf, D. 1966 The thymus, In 'Recent results in cancer research', Vol 5, (Springer-Verlag)

Metcalf, D. 1971 Acute antigen-induced elevation of serum colony stimulating factor (CSF) levels. Immunology, <u>21</u>, 427-436

Metcalf, D. 1981 Haemopoietic colony stimulating factors, p343-384 In 'Tissue Growth Factors', (Springer-Verlag, Berlin), pgs343-384

Metcalf, D. 1985 The granulocyte-macrophage colony stimulating factor, Cell. <u>43</u>, 5-6

Metcalf, J.C., Hesketh, T.R. & Smith, G.A. 1985 Free cytosolic  $Ca^{2+}$  measurements with fluorine labelled indicators using <sup>19</sup>FNMR. Cell Calcium, <u>6</u>, 183-195

Metcalfe, J.C., Hesketh, T.R., Smith, G.A., Morris, J.D.H., Corps, A.N. & Moore, J.P. 1985 Early response pattern analysis of the mitogenic pathway in lymphocytes and fibroblasts', J. Cell. Suppl., <u>3</u>, 199-228

Metcalfe, J.C., Pozzan, T., Smith, G.A. & Hesketh, T.R. 1980 "A calcium hypothesis for the control of cell growth", Biochem. Soc. Symp. <u>45</u>, 1-26

Meuer, S.C., Butteroth, T.H. & Meyer zum Buschenfelde, K.H. 1986 Monoclonal antibodies as probes to define critical surface structures involved in T cell activation. In 'Leukocyte Typing II', Vol 1, Eds: E.L. Reinherz, B.F. Haynes, L.M. Nadler & I.D. Bernstein, (Springer-Verlag (New York)), pgs67-78

Michell, R.H. 1975 Inositol phospholipids and cell surface receptor function. Biochim. Biophys. Acta., <u>415</u>, 81-147

Michell, R.H. 1979 Inositol phospholipids in membrane function. Trends Biochem. Sci., <u>4</u>, 128-131

Michell, R.H. 1982a 'Phosphatidylinositol turnover in signal transduction', Neurosciences Res. Prog. Bull., 20, 338-350

Michell, R.H. 1982b Inositol lipid metabolism in dividing and differentiating cells, Cell Calcium, <u>3</u>, 429-440

Michell, R.H. 1986a Inositol phosphates. Profusion and confusion. Nature, 319, 176-177

Michell, R.H. 1986b A second messenger function for inositol tetrakisphosphate. Nature, <u>324</u>, 613

Michell, R.H., Jafferji, S.S. & Jones, L.M. 1977 The possible involvement of phosphotidylinasitol breakdown in the mechanism of stimulation response coupling at receptors which control cell-surface calcium gates', Adv. Exp. Med. Biol., <u>83</u>, 447-464

Migliaccio, A., Rotondi, A. & Auricchio, F. 1984 Calmodulin-stimulated phosphorylation of 17ß-estradiol receptor on tyrosine. Proc. Natl. Acad. Sci., USA, 81, 5921-5925

Migliacco, A., Rotondi, A. & Auricchio, F. 1986 Estradiol receptor: phosphorylation on tyrosine in uterus and interaction with anti-phosphotyrosine antibody. Embo. J. <u>5</u>, 2867-2872

Mills, G.B., Cheung, R.K., Grinstein, S. & Gelfand, E.W. 1985a Increase in cytosolic free calcium concentration is an intracellular messenger for the production of Interleukin 2 but not for the expression of the Interleukin 2 receptor. J. Immunol., 134, 1640-1643

Mills, G.B., Cheung, R.K., Grinstein, S. & Gelfand, E.W. 1985b Interleukin 2-induced lymphocyte proliferation is independent of increases in cytosolic-free calcium concentrations. J. Immunol., 134, 2431-2435

Mills, G.B., Stewart, D.J., Mellors, A. & Gelfand, E.W. 1986 Interleukin 2 does not induce phosphatidylinositide hydrolysis in T cells: evidence against signalling through phosphatidylinositide hydrolysis. J. Immunol., <u>136</u>, 3019-24

Milner, J. & Milner, S. 1981 SV40-53K antigen: a possible role for 53K in normal cells. Virology, <u>112</u>, 785-788

Mishell, R.I. & Dutton, R.W. 1967 Immunization of dissociated spleen cell cultures from normal mice. J. Exp. Med., <u>126</u>, 423-442

Mix, L.L., Dinerstein, R.J. & Villereal, M.L. 1984 Mitogens and melattin stimulate an increase in intracellular free calcium concentration in human fibroblasts. Biochem. Biophys. Res. Commun., <u>119</u>, 69-75

Miyake, T., King, C.K.H. & Goldwasser, E. 1977 Purification of human erythropoietin. J. Biol. Chem., 252, 5558-5564

Mizel, S.B., Kilian, P.L., Lewis, J.C., Paganelli, K.A. & Chizzonite, R.A. 1987 The Interleukin 1 receptor dynamics of interleukin 1 binding and internalization in T cells and fibroblasts. J. Immunol., <u>138</u>, 2906-2912

Mizel, S.B. 1982 Interleukin 1 and T cell activation. Immunol. Rev., 63, 51-72

Moolenaar, W.H. 1986 Regulation of cytoplasmic pH by Na<sup>+</sup>/H<sup>+</sup> exchange. Trends Biochem. Sci., <u>11</u>, 141-143

Moolenaar, W.H., Mummery, C.L., van der Saag, P.T. & De Laat, S.W. 1981 Rapid ionic events in the interaction of growth in serum-stimulated neuroblastoma cells. Cell <u>23</u>, 789-798 Moolenaar, W.H., Tertoolen, L.G.J. & De Laat, S.W. 1984 Phorbol ester and diacylglycerol mimic growth factors in raising pH. Nature, <u>312</u>, 371-374

Moolenaar, W.H., Tsien, R.Y., Van der Saag, P.T. & De Laat, S.W. 1983  $Na^+/H^+$  exchange and cytoplasmic pH in the action of growth factors in human fibroblasts. Nature, <u>304</u>, 645-648

Murayama, T. & Ui, M. 1985 Receptor-mediated inhibition of adenylate cyclase and stimulation of arachidonic acid release in 3T3 fibroblasts. J. Biol. Chem., <u>260</u>, 7226-7233

Myers, M.J., Butler, L.D. & Petersen, B.H. 1986 Estradiol-induced alteration in the immune system. II, Suppression of cellular immunity in the rat is not the result of direct estrogenic action. Immunopharmacology, <u>11</u>, 47-55

Mooradian, A.D., Morley, J.E. & Korenman, S.G. 1987 Biological actions of androgens. Endocrine Reviews, 8, 1-28

Moore, J.P., Johannsson, A., Hesketh, T.R., Smith, G.A. & Metcalfe, J.C. 1984 Calcium signals and phospholipid methylation in eukaryotic cells. Biochem. J., <u>221</u>, 675-684

Moore, J.P., Todd, J.A., Hesketh, R. & Metcalfe, J.C. 1986 c-fos and c-myc gene activation, ionic signals and DNA synthesis in thymocytes. J. Biol. Chem., 261, 8158-8162

Moore, R.D., 1983 Effects of insulin upon ion transport. Biochim. Biophys. Acta., 737, 1-49

Morgan, J.I. & Curran, T. 1986 Role of ion flux in the control of c-fos expression. Nature, <u>322</u>, 552-555

Morgan, J.I., Wigham, C.G. & Perris, A.D. 1984 The promotion of mitosis in cultured thymic lymphocytes by acetylcholine and catecholamines. J. Pharm. Pharmacol., <u>36</u>, 511-515

Morris, J.D.H., Metcalfe, J.C., Smith, G.A., Hesketh, T.R. & Taylor, M.V. 1984 Some mitogens cause rapid increases in free calcium in fibroblasts. FEBS Lett., <u>169</u>, 189-193

Moscat, J., Aracil, M., Diez, E., Balsinde, J., Barreno, P.G. & Municio, A.M. 1986 Intracellular Ca<sup>2+</sup> requirements for zymoson-stimulated phosphoinositide hydrolysis in mouse peritoneal macrophages. Biochem. Biophys. Res. Comm., 134, 367-371

Moses, A.C. & Pilistine, S.J. 1985 Insulin-like growth factors. In 'Control of animal cell proliferation', Vol 1, Eds: A.L. Boynton & H.L. Leffert, (Academic Press Inc., New York), pgs91-120

Mukku, V.R. & Stancel, G.M. 1985 Regulation of Epidermal growth factor receptor by estrogen. J. Biol. Chem., <u>260</u>, 9820-9824

Muller, R., Bravo, R., Burckhardt, J. & Curran, T. 1984 Induction of c-fos gene and protein by growth factors precedes activation of c-myc. Nature, <u>312</u>, 716-720 Mulcahy, L.S., Smith, M.R. & Stacey, D.W. 1985 Requirement for <u>ras</u> proto-oncogene function during serum-stimulated growth of NIH3T3 cells. Nature, <u>313</u>, 241-246

Muldoon, L.L., Jamieson, G.A. & Villereal, M.L. 1987 Calcium mobilization in permeabilized fibroblasts: effects of inositol trisphosphate, orthovanadate, mitogens, phorbol ester and guanosine triphosphate. J. Cell. Phys., <u>130</u>, 29-36

Murray, A.W. 1987a A cycle is a cycle is a cycle. Nature, 327, 14-15

Mustelin, T., Poso, H. & Anderson, L.L., 1986 Role of G proteins in T cell activation: non hydrolysable GTP analogues induce early ornithine decarboxylase activity in human T lymphocytes. EMBO J., 5, 3287-3290

Mustelin, T., Poso, H., Lapinjoki, S.P., Gunther, J., Andersson, L.C. 1987 Growth signal transduction: rapid activation of covalently bound ornithine decarboxylase during phosphatidylinositol breakdown. Cell, <u>49</u>, 171-176

Myers, M.J., Butler, L.D. & Petersen, B.H. 1986 Estradiol-induced alteration in the immune system. II. Suppression of cellular immunity in the rat is not the result of direct estrogenic action. Immunopharmacology, <u>11</u>, 47-55

Myers, M.J. & Petersen, B.H. 1985 Estradiol induced alterations of the immune system. 1. Enhancement of IgM production. Int. J. Immunopharmacol., 7, 207-210

Naccache, P.H., Molski, T.F.P., Boregeat, P., White, J.R. & Sha'afi, R.I., 1985 Phorbol esters inhibit the fMet-Leu-Phe- and leukotriene  $B_4$ -stimulated calcium mobilization and enzyme secretion in rabbit neutrophils. J. Biol. Chem., <u>260</u>, 2125-2131

Nahorski, S.R. & Batty, I. 1986 Inositol tetrakisphosphate: recent developments in phosphoinositide metabolism and receptor function. Trends Pharamcol. Sci., 7, 83-85

Nakamura, T., Teramoto, H., Tomito, Y. & Ichihara, A. 1986 Two types of growth inhibitor in rat platelets for primary cultured rat hepatocytes. Biochem. & Biophys. Res. Commun., <u>134</u>, 755-763

Nakamura, T., Tomita, Y., Hirai, R., Yamaoka, K., Kaji, K. & Ichihara, A. 1985 Inhibitory effect of transforming growth factor ß on DNA synthesis of adult rat hepatocytes in primary culture. Biochem. Biophys. Res. Comm., <u>133</u>, 1042-1050

Nambi, P., Peters, J.R., Sibley, D.R. & Lefkowitz, R.J., 1985 Desensitization of the turkey erythrocyte ß-adrenergic receptor in a cell-free system. J. Biol. Chem., 260, 2165-2171

Nandi, S., Yang, J., Richards, J., Guzman, R., Rodrigues, R. & Imagawa, W. 1980 Role of growth factor and classical hormone in the control of growth and function of mammary tissues. In 'Hormones and Adaptation and Evolution', Ed: S. Ishii, (Springer Verlag Berlin), pgs145-155

Neifeld, J.P., Lippman, M.E. & Tormey, D.C. 1977 Steroid hormone receptors in normal human lymphocytes. J. Biol. Chem., 252, 2972-2977

Nicholls, D. 1978 The regulation of extra mitochondrial free calcium ion concentration by rat liver mitochondria. Proc. Natl. Acad. Sci., (USA), <u>71</u>, 1286-1290

Nicol, T., Bilbey, D.L.J., Charles, L.M., Cordingley, J.L. & Vernon-Roberts, B. 1964 Oestrogen: the natural stimulant of body defence. J. Endocrinol., <u>30</u>, 277-91

Nicola, N.A. & Metcalf, D. 1985 The colony-stimulating factors and myeloid leukaemia, Cancer Surveys 4, 789-815

Nisbet-Brown, E., Lee, J.W.W., Cheung, R.K. & Gelfand, E.W. 1985 Antigen-dependent increases in cytosolic free calcium in specific human T lymphocyte clones. Nature, <u>316</u>, 545-547

Nishimura, J., Huang, J.S. & Deuel, T.F. 1982 Platelet-derived growth factor stimulates tyrosine-specific protein kinase activity in Swiss mouse 3T3 cell membranes. Proc. Natl. Acad. Sci., USA, 79, 4303-4307

Nishimura, S. & Sekiya, T. 1987 Human cancer and cellular oncogenes. Biochem. J., <u>243</u>, 313-327

Nishizuka, Y. 1984a Turnover of phospholipids and signal transduction. Science, 225, 1365-1370

Nishizuka, Y. 1984b The role of protein kinase C in cell surface signal transduction and tumour promotion. Nature, <u>308</u>, 693-698

Nishizuka, Y. 1986 Studies and perspectives of protein kinase C. Science, 233, 305-12

Norton, J.M. & Wira, C.R. 1977 Dose related effects of the sex hormones and cortisol on the growth of the bursa of Fabricius in chick embryos. J. Steroid. Biochem., <u>8</u>, 985-987

Noteboom, W. & Gorski, J. 1965 Stereoscopic binding of estrogens in the rat uterus. Arch. Biochem. Biophys., <u>111</u>, 559-568

Nowell, P.C., Koretzy, G.A., Reed, J.C. & Hannan-Harris, A.C. 1985 Interleukins and inhibitors in human lymphocyte regulation, In 'Mediators in Cell Growth and Differentiation', R.J. Ford & A.L. Maizel, (Raven Press, New York), p199-212

Oettgen, H.F., Binaghi, R.A. & Benacerraf, B. 1966 The effect of various hormones on the serum level of anaphylactic antibody in the rat. Int. Arch. Allergy Appl. Immunol., 29, 131-135

Oettgen, H.C., Terhorst, C., Cantley, L.C. & Rosoff, P.M. 1985 Stimulation of the T3-T cell receptor complex induces a membrane-potential-sensitive calcium influx. Cell, <u>40</u>, 583-590

O'Flynn, K., Linch, D.C. & Tatham, P.E.R. 1984 The effect of mitogenic lectins and monoclonal antibodies on intracellular free calcium concentration in human T-lymphocytes. Biochem. J., 219, 661-666

O'Flynn, K., Zanders, E.D., Lamb, J.R., Beverley, P., Wallace, D.L., Tatham, P.E.R., Tax, W.J.M. & Linch, D.C. 1985 Investigation of early T-cell activation: analysis of the effect of specific antigen, IL-2 and monoclonal antibodies on intracellular free calcium concentration. Eur. J. Immunol., 15, 7-11

Oka, Y. & Orth, D.N. 1983 Human plasma Epidermal growth factor/ß Urogastrone is associated with blood platelets. J. Clin. Invest., <u>72</u>, 249-259

O'Keefe, E.J. & Pledger, W.J. 1983 'A model of cell cycle control: sequential events regulated by growth factors', Molecular and Cellular Endocrinology, <u>31</u>, 167-186

Olashaw, N.E., O'Keefe, E.J. & Pledger, W.J. 1986 Platelet-derived growth factor modulates epidermal growth factor receptors by a mechanism distinct from that of phorbol esters. Proc. Natl. Acad. Sci., USA, <u>83</u>, 3834-3838

Olashaw, N.E. & Pledger, W.J. 1985 Mechanisms initiating cellular proliferation, In 'Mediators in Cell Growth and Differentiation', Ed: R.J. Ford & A.L. Maizel, (Raven Press, New York), pgs31-44

Oppenheim, J.J., 1969 Relationship of <u>in vitro</u> lymphocyte transformation to delayed hypersensitivity in guinea pigs and man. Fed. Proc. Fed. Am. Soc. Exp. Biol., <u>27</u>, 21-28

Oppenheim, J.J. & Gery, I. 1982 Interleukin 1 is more than an Interleukin. Immunology Today, <u>3</u>, 113-119

Oppenheimer, C.L., Pessin, J.E., Massagué, J., Gitomer, W. & Czech, M.P. 1983 Insulin action rapidly modulates the apparent affinity of the Insulin-like growth factor II receptor. J. Biol. Chem., <u>258</u>, 4824-30

Otten, J., Johnson, G.S. & Pastan, I. 1971 Cyclic AMP levels in fibroblasts: relationship to growth rate and contact inhibition of growth. Biochem. Biophys. Res. Commun., <u>44</u>, 1192-1198

Owen, N.E. & Villereal, M.E., 1982 Effect of the intracellular  $Ca^{2+}$  antagonist TMB8 on the serum stimulate Na<sup>+</sup> influx in human fibroblasts. Biochem. Biophys. Res. Commun., <u>109</u>, 762-768

Owen, N.E. & Villereal, M.L. 1983 Lys-Bradykinin stimulates Na<sup>+</sup> influx and DNA synthesis in cultured human fibroblasts. Cell, <u>32</u>, 979-985

Owens, R.J., Gallagher, C.J. & Crumpton, M.J. 1984 Cellular distribution of p68 a new calcium binding protein from lymphocytes. Embo. J., 3, 945-952

Paavonen, T., Andersson, L. & Adlercreutz, H. 1981 Sex hormone regulation of in <u>vitro</u> immune response. Estradiol enhances Human B cell maturation via inhibition of suppressor T cells in pokeweed mitogen-stimulated cultures. J. Exp. Med., <u>154</u>, 1935-1945

Packard, B. 1986 Signal transduction through the cytoplasm. Trends Biochem. Sci., <u>11</u>, 154

Paetkau, V., Mills, G., Gerhart, S. & Monticone, V. 1976 Proliferation of murine thymic lymphocytes in vitro is mediated by a concanavlin A-induced release of a lymphokine (costimulator), J. Immunol., <u>117</u>, 1320-1324

Palmer, S., Hawkins, P.T., Michell, R.H. & Kirk, C.J., 1986 The labelling of polyphosphoinositides with [<sup>32</sup>P] Pi and the accumulation of inositol phosphates in vasopressin-stimulated hepatocytes. Biochem. J., <u>238</u>, 491-499

Papahadjopoulos, D. 1972 Studies on the mechanism of action of local anaesthetics with phospholipid model membranes. Biochim. Biophys. Acta., <u>265</u>, 169-86

Pappritz, G., Meezor, H. & Herberg, H. 1977 Estrous cycle controlled cell proliferation in the adrenal cortex of female rats. Cell Tissue Res., <u>184</u>, 269-279

Pardee, A.B., 1974 A restriction point for control of normal animal cell proliferation. Proc. Natl. Acad. Sci., USA, <u>71</u>, 1286-1290

Pardee, A.B., Dubrow, R., Hamlin, J.L. & Kletzien, R.F. 1978 Animal cell cycle. Annu. Rev. of Biochem., <u>47</u>, 715-750

Pardee, A.B., Campisi, J. & Croy, R.G. 1982 Differences in growth regulation of normal and tumour cells. Ann. N.Y. Acad. Sci., <u>397</u>, 121-129

Paridiso, A.M., Tsien, R.Y. & Machen, T.E. 1984 Na<sup>+</sup>-H<sup>+</sup> exchange in gastric glands as measured with a cytoplasmic-trapped fluorescent pH indicator. Proc. Natl. Acad. Sci., <u>81</u>, 7436-7440

Paris, S. & Pousségur, J. 1983 Biochemical characterisation of the amiloride-sensitive Na<sup>+</sup>/H<sup>+</sup> antiport in Chinese hamster lung fibroblasts. J. Biol. Chem., <u>258</u>, 3503-3508

Paris, S. & Pousségur, J. 1984 Growth factors activate the  $Na^+/H^+$  antiporter in quiescent fibroblasts by increasing its affinity for intracellular H<sup>+</sup>. J. Biol. Chem., 259, 10989-10994

Parker, C.J. 1976 Control of lymphocyte function. New Engl. J. Med., 295, 1180-1186

Parker, M.G. 1987 Oncogenes from hormone receptors, J. Endocr., 112, 1-2

Parker, P.J., Coussens, L., Totty, N., Rhee, L., Young, S., Chen, E., Stable, S., Waterfield, M.D. & Ullrick, A. 1986 The complete primary structure of protein kinase C - the major phorbol ester receptor. Science, 233, 853-859

Parker, P.J., Stabel, S. & Waterfield, M.D. 1984 Purification to homogeneity of protein kinase C from bovine brain: identity with the phorbol ester receptor. Embo. J., <u>3</u>, 953-959

Pastan, I.H., Johnson, G.S. & Anderson, W.B. 1975 Role of cyclic nucleotides in growth control. Annu. Rev. Biochem., <u>44</u>, 491-522

Patt, L. & Houck, J.C. 1980 The incredible shrinking chalone. FEBS Lett., 120, 163-170

Paul, D. 1985 Growth factors, oncogenes and transformation part 1. Growth factors and cell cycle control. Arzneim-Forsch/Drug Res., <u>35</u>, 773-779

Paul, D. & Ristow, H.J. 1979 Cell cycle control by  $Ca^{2+}$  ions in mouse 3T3 cells and in transformed 3T3 cells. J. Cell Physiol., <u>98</u>, 31-40

Peach, M.J. 1981 Molecular actions of angiotensin. Biochem. Pharmacol., <u>30</u>, 2745-2751

Pearce, P., Khalid, A.K. & Funder, J.W. 1981 Androgens and the thymus. Endocrinology, <u>109</u>, 1073-1077

Pearlmutter, A.F. & Soloff, M.S. 1979 Characterisation of the metal ion requirement for oxytocin receptor in rat mammary gland membranes. J. Biol.Chem., <u>254</u>, 3899-3906

Pedernera, E., Diaz-Osuna, J. & Claagno, M. 1986 A thymus factor influences the in vitro testosterone secretion of Leydig cells in the rat. Life Sci., <u>38</u>, 779-787

Pedersen, P.L. & Carafoli, E. 1987 Ion motive ATPase 1. Ubiquity, properties, and significance to cell function. Trends. Biochem. Sci., 12, 146-150

Pegg, A.E. 1986 Recent advances in the biochemistry of polyamines in eukaryotes. Biochem. J., 234, 249-262

Pegg, A.E. & McCann, P.P., 1982 Polyamine metabolism and function. Am. J. Physiol., <u>243</u>, C212-C221

Pegg, A.E. & Williams-Ashman, H.G. 1981 Biosynthesis of putrescine. In 'Polyamines in Biology and Medicine', Ed: D.R. Morris & L.J. Marton, (Dekker (New York), pgs3-42

Peleg, S. & Lindrer, H.R. 1980 Role of steroid hormones and prostaglandin in the regulation of DNA synthesis by decidual cells in culture. Mol. Cellular Endocrin., 20, 209-218

Perris, A.D., MacManus, J.P., Whitfield, J.F. & Weiss, L.A. 1971 The parathyroid glands and mitotic stimulation in rat bone marrow after haemorrhage. Am. J. Physiol., <u>220</u>, 773-778

Perris, A.D. & Morgan, J.I. 1976 The interaction of divalent cations, hormones and cyclic nucleotides in control of mitosis. Calcified Tissue Res., (supplement), <u>21</u>, 15-20

Perris, A.D., Weiss, L.A. & Whitfield, J.F. 1970 Parathyroidectomy and the induction of thymic atrophy in normal, adrenalectomised and orchidectomised rats. J. Cell Physiol., <u>76</u>, 141-150

Perrot-Applanat, M., Logeat, F., Groyer-Picard, M.J. & Milgrom, E. 1985 Immunocytochemical study of mammalian progesterone receptor using monoclonal antibodies. Endocrinology, <u>116</u>, 1473-1485 Pershadsingh, H.A., McDaniel, M.L., Landt, M., Bry, C.G., Lacy, P.E. & McDonald, J.M. 1980  $Ca^{2+}$ -activated ATPase and ATP-dependent calmodulin-stimulated  $Ca^{2+}$  transport in islet cell plasma membranes. Nature, 288, 492-495

Pfizenmaier, K., Scheurich, P., Daiibener, W., Kronke, M., Rollinghoff, M. & Wagner, H. 1984 Quantitative representation of all T cells committed to develop into cytotoxic effector cells and/or Interleukin 2 activity-producing helper cells within murine T lymphocyte subsets. Eur. J. Immunol., <u>14</u>, 33-39

Philipson, K.D. & Nishimoto, A.N. 1984 Stimulation of Na<sup>+</sup>-Ca<sup>2+</sup> exchange in cardiac sacrolemmal vesicles by phospholipase D. J. Biol. Chem., <u>259</u>, 16-19

Phillips, L.S. & Vassilopoulon-Sellin, R. 1980 Somatomedins. The New England Journal of Medicine, 302, 371-380

Pietras, R.J. & Szego, C.M. 1979 Metabolic and proliferative response to oestrogen by hepatocytes selected for plasma membrane binding sites specific for estradiol 17B. J. Cell. Physiol., <u>98</u>, 145-160

Pietras, R.J. & Szego, C.M. 1984 Specific internalization of estrogen and binding to nuclear matrix in isolated uterine cells. Biochem. Biophys. Res. Comm., <u>123</u>, 84-91

Pike, L.J., Bowen-Pope, D.F., Ross, R. & Krebs, E.G. 1983 Characterization of platelet-derived growth factor stimulated phosphorylation in cell membranes. J. Biol. Chem., <u>258</u>, 9383-9390

Pike, L.J. & Krebs, E.G. 1986 Protein tyrosine kinase activity of hormone and growth factor receptors. In 'The Receptors', Vol 3. Ed: P.M. Conn (Academic Press, New York), pgs93-134

Plaut, M. 1987 Lymphocyte hormone receptors. Ann. Rev. Immunol., 5, 621-669

Pledger, L.J., Hart, C.A., Locetell, K.L. & Scher, C.D. 1981 Platelet-derived growth factor-modulated proteins: constructive synthesis by a transformed cell line. Proc. Natl. Acad. Sci., USA, <u>78</u>, 4358-4362

Pledger, W.J., Stiles, C.D., Antoniades, H.N. & Scher, C.D. 1977 Induction of DNA synthesis in BAIB/c3T3 cells by serum components: Re-evaluation of the commitment process. Proc. Natl. Acad. Sci., USA, 74, 4481-4485

Pledger, W.J., Stiles, C.D., Antoniades, H.N. & Scher, C.D. 1978 An ordered sequence of events is required before Balb/c3T3 cells become committed to DNA synthesis. Proc. Natl. Acad. Sci., USA, 75, 2839-2843

Pollock, W.K., Sage, S.O. & Rink, T.J. 1987 Stimulation of  $Ca^{2+}$  efflux from fura-2-loaded platelets activated by thrombin or phorbol myristate acetate. FEBS Lett, <u>210</u>, 132-136

Pousségur, J. 1985 The growth factor-activable  $Na^+/H^+$  exchange system: a genetic approach. Trends Biochem. Sci., <u>10</u>, 453-455

Pousségur, J., Chambard, J.C., Franchi, A., Paris, S. & Van Obberghen-Schilling, E. 1982 Growth factor activation of an amiloride-sensitive  $Na^+/H^+$  exchange system in quiescent fibroblasts: coupling to ribosomal protein S6 phosphorylation. Proc. Natl. Acad. Sci., USA, <u>79</u>, 3935-3939

Powell, A.E. & Leon, M.A. 1970 Reversible interaction of human lymphocytes with the mitogen Con A. Exptl. Cell Res., <u>62</u>, 315-325

Powers, M.L. & Florini, J.R. 1975 A direct effect of testosterone on muscle cells in tissue culture. Endocrinology, <u>97</u>, 1043-1047

Pozzan, T., Corps, A.N., Hesketh, J.R. & Metcalfe, J.C. 1981 Mitogenic stimulation and the redistribution of Concanavalin A receptors on lymphocytes. Expl. Cell Res., <u>134</u>, 399-408

Prasad, K.V.S., Severini, A. & Kaplan, J.G. 1987 Sodium ion influx in proliferating lymphocytes: an early component of the mitogenic signal. Arch. Biochem. Biophys., 252, 515-525

Prentki, M., Biden, T.J., Janjic, D., Irvine, R.F., Berridge, M.J. & Wollheim, C.B. 1984 Rapid mobilisation of  $Ca^{2+}$  from rat insulinoma microsomes by inositol-1,4,5-triphosphate, Nature, 309, 562-564

Putney, J.W., 1978 Stimulus-permeability coupling: role of calcium in the receptor regulation of membrane permeability. Pharmacol. Rev., <u>30</u>, 209-45

Putney, J.W., 1986 A model for receptor-regulated calcium entry. Cell Calcium, 7, 1-12

Putney, J.W., McKinney, J.S., Aub, D.L. & Leslie, B.A. 1984 Phorbol ester-induced protein secretion in rat partoid glands. Relationship to the role of inositol lipid breakdown and protein kinase C activation in stimulus-secretion coupling. Mol. Pharmacol., <u>26</u>, 261-266

Putney, J.W., Weiss, S.J., Van de Walle, C.M. & Haddas, R.A. 1980 Is phosphatidic acid a calcium ionophore under neurohumoral control? Nature, <u>284</u>, 345-347

Quastel, M.R. & Kaplan, J.G. 1970 Early stimulation of potassium uptake in lymphocytes treated with PHA. Exp. Cell Res., <u>63</u>, 230-233

Rakowicz-Szulazynska, E.M. & Kopowski, H. 1986 Identification of NGF receptor in chromatins of melanoma cells using monoclonal antibody to cell surface NGF receptor. Biochem. Biophys. Res. Commun., <u>140</u>, 174-180

Rakowicz-Szulczynska, E.M., Rodeck, U., Herlyn, M. & Koprowski, H. 1986 Chromatin binding of epidermal growth factor, nerve growth factor and platelet derived growth factor in cells bearing the appropriate surface receptors. Proc. Natl. Acad. Sci., USA, <u>83</u>, 3728-3732

Rall, T.W. & Sutherland, E.W. 1962 The enzymatically catalysed formation of adenosine 3', S' monophosphate and inorganic pyrophosphate for adenosine triphosphate. J. Biol. Chem., 237, 1228-32

Ralph, R.K. 1983 Cyclic AMP, calcium and control of cell growth. FEBS Lett., 161, 1-8

Rampini, C. & Dubois, C. 1981 Changes in phospholipid metabolism of SV3T3 transformed fibroblasts restricted by plating and cultivation at low serum concentration and subsequently re-stimulated by serum. Biochem. Biophys. Res. Commun., <u>100</u>, 1222-1228

Rao, G.S. 1981 Mode of entry of steroid and thyroid hormones into cells. Molecular & Cellular Endocrinology, <u>21</u>, 97-108

Rao, M.L., Rao, G.S. & Brueuer, H. 1977 Uptake of estrone, estradiol-17ß and testosterone by isolated rat liver cells. Biochem. Biophys. Res. Commun., <u>77</u>, 566-573

Rasmussen, H. 1970 Cell communication, calcium ion and cyclic adenosine monophosphate. Science, <u>170</u>, 404-412

Rasmussen, H. & Barrett, P.Q. 1984 Calcium messenger system: an integrated view. Physiol. Rev., <u>64</u>, 938-984

Rasmussen, H. & Goodman, B.P. 1977 Calcium and cyclic nucleotides. Physiol. Rev., <u>57</u>, 421-509

Rasmussen, H., Kojima, I. & Barrett, P. 1986 Information flow in the calcium messenger system. In 'New Insights into Cell and Membrane Transport Processes', Eds: G. Poste & S.T. Crooke. (Plenum Press (New York)), pgs 145-174

Rebhun, L.I. 1977 Cyclc nucleotides, calcium and cell division. Int. Rev. Cytol., 49, 1-54

Read, B.D. & McElhaney, R.N. 1976 Influence of membrane lipid fluidity on glucose and uridine facilitated diffusion in human erythrocytes. Biochim. Biophys. Acta., <u>419</u>, 331-341

Reich, N.C. & Levine, A.J. 1984 Growth regulation of a cellular tumour antigen, p53, in nontransformed cells. Nature, <u>308</u>, 199-201

Reinherz, E.L. 1987 T cell receptors, who needs more? Nature, 325, 660-663

Reinherz, E.L. & Scholssman, S.F. 1983 The characterization and function of human immunoregulatory T lymphocyte subsets. In 'T lymphocytes today', Ed: J.R. Inglis, Elsevier (Amsterdam), pgs104-109

Renard, D., Poggioli, B. & Claret, M. 1987 How far does phospholipase C activity depend on the cell calcium concentration? Biochem. J., 243, 391-398

Rink, T.J., Tsien, R.Y. & Pozzan, T. 1982 "Cytoplasmic pH and free  $Mg^{2+}$  in lymphocytes. J. Cell Biol., <u>95</u>, 189-196

Rittenhouse, S.E. 1982 Preparation of selectively labelled phosphatidylinositol and assay of phosphatidylinositol-specific phospholipase. Methods Enzymol., <u>86</u>, 3-11

Rixon, R.H. & Whitfield, J.F. 1972 Hypoplasia of the bone marrow in rats following removal of the parathyroid glands. J. Cell. Physiol., <u>79</u>, 343-352

Rixon, R.H. & Whitfield, J.F. 1976 The control of liver regeneration by parathyroid hormone and calcium. J. Cell Physiol., <u>87</u>, 147-155

Robb, R.J. 1984 Interleukin 2 - the molecule and its function. Immunology Today, 5, 203-209

Robbins, K.C., Antoniades, H.N., Devare, S.G., Hunkapiller, M.W. & Aaronson, S.A. 1983 Structural and immunological similarities between simian sarcoma virus gene product(s) and human platelet-derived growth factor. Nature, <u>305</u>, 605-608

Roberts, A.B., Azano, M.A., Lamb, L.C. & Smith, J.M. 1981 'New class of transforming growth factors potentiated by epidermal growth factor: isolation from non-neoplastic tissues', Proc. Natl. Acad. Sci., USA, 78, 5339-5343

Roberts, A.B., Azano, M.A., Wakefield, L.M., Roche, N.S., Stern, D.F. & Sporn, M.B. 1985 Type ß transforming growth factor: A bifunctional regulator of cellular growth. Proc. Natl. Acad. Sci., USA, <u>82</u>, 119-123

Roberts, A.B. & Sporn, M.B. 1985 Transforming growth factors. Cancer Surveys, 4, 683-689

Roberts, A.B., Frolik, C.A., Anzano, M.A. & Sporn, M.B. 1983 "Transforming growth factors from neoplastic and non neoplastic tissues", Fed. Proceedings, <u>42</u>, 2621-2626

Rogers, J., Hesketh, T.R., Smith, G.A. & Metcalfe, J.C. 1983 Intracellular pH of stimulated thymocytes measured with a new fluorescent indicator. J. Biol. Chem., 258, 5994-5997

Ronot, X. & Adolphe, M. 1986 Update on the concept of the cell cycle: the contribution of flow cytometry. Biology of the Cell, <u>58</u>, 113-120

Rosenthal, A.S. & Shevach, E.M. 1973 Function of macrophages in antigen recognition by guinea pig T lymphocytes 1. Requirement for histocompatible macrophages and lymphocytes. J. Exp. Med., <u>138</u>, 1194-1212

Rosoff, P.M. & Cantley, L.C. 1985 Stimulation of the T3-T cell receptor associated  $Ca^{2+}$  influx enhances the activity of the Na<sup>+</sup>/H<sup>+</sup> exchanger in a leukaemic human T cell line. J. Biol. Chem., <u>260</u>, 14053-9

Ross, R., Glomset, B., Kariya, B. & Harker, L. 1974 A platelet dependent serum factor that stimulates the proliferation of arterial smooth muscle cells in vitro. Proc. Natl. Acad. Sci. USA, <u>71</u>, 1207-1210

Rossio, J.L., Ruscetti, F.W. & Farrar, W.L. 1986 Ligand-specific calcium mobilization in IL2 and IL3 dependent cell lines. Lymphokine Res., 5, 163-172

Roth, R.A. & Cassell, D.J. 1983 Insulin receptor: evidence that it is a protein kinase. Science, <u>219</u>, 299-301

Rothenberg, P., Reuss, L. & Claser, L. 1982 Serum and epidermal growth factor transiently depolarize quiescent BSC-1 epithelial cells. Proc. Natl. Acad. Sci., USA, <u>79</u>, 7783-7787

Roubinian, J.R., Papoian, R. & Talal, N. 1977 Androgenic hormones modulate autoantibody responses and improve survival in murine lupus. J. Clin. Invest., <u>59</u>, 1066-1699

Roubinian, J.R., Talal, N., Greenspan, J.S., Goodman, J.R. & Sitteri, P.K. 1978 Effect of castration and sex hormone treatment on survival, anti-nucleic acid antibodies, and glomerulonephritis in NZB.NZW F mice. J. Exp. Med., <u>147</u>, 1568-1572

Rozengurt, E. 1976 Co-ordination of early membrane changes in growth stimulation. J. Cell Physiol., <u>89</u>, 627-632

Rozengurt, E. 1980 Stimulation of DNA synthesis in quiescent cultured cells. Exogenous agents, internal signals and early events. Cur. Top. Cell Reg., <u>17</u>, 59-88

Rozengurt, E. 1982a Synergistic stimulation of DNA synthesis by cAMP derivatives and growth factors in mouse 3T3 cells. J. Cell Physiol., <u>112</u>, 243-249

Rozengurt, E. 1982b Monovalent ion fluxes, cyclic nucleotides and the stimulation of DNA synthesis in quiescent cells. In 'Ions, Cell Proliferation and Cancer', Eds: A.L. Boynton, W.L. McKeehan & J.F. Whitfield, Academic Press (New York), pgs259-277

Rozengurt, E. 1985 The mitogenic response of cultured 3T3 cells: integration of early signals and synergistic effects in a unified framework. In 'Molecular Mechanisms of transmembrane Signalling', Eds: P. Cohen & M.D. Houslay, (Elsevier (Amsterdam)), pgs429-452

Rozengurt, E., Brown, K.O. & Pettican, P. 1981 Vasopressin inhibition of epidermal growth factor binding to cultured mouse cells. J. Biol. Chem., <u>256</u>, 716-722

Rozengurt, E. & Burns, C.P. 1984 Extracellular Na<sup>+</sup> and initiation of DNA synthesis: role of intracellular pH and K<sup>+</sup>. J. Cell Biol., <u>98</u>, 1082-1089

Rozengurt, E. & Collins, M. 1983 Molecular aspects of growth factor action: receptors and intracellular signals. Journal of Pathology, <u>141</u>, 309-331

Rozengurt, E., Collins, M., Brown, K.D. & Pettican, P. 1982 Inhibition of epidermal growth factor binding to mouse cultured cells by fibroblast-derived growth factor. J. Biol. Chem., 257, 3680-3686

Rozengurt, E., Legg, A. & Pettican, P. 1979 Vasopressin stimulation of mouse 3T3 cell growth. Proc. Natl. Acad. Sci., USA, <u>76</u>, 1284-1287

Rozengurt, E., Legg, A., Strang, G. & Courtenay-Luck, A. 1981 Cyclic AMP: a mitogenic signal for Swiss 3T3 cells. Proc. Natl. Acad. Sci., USA, 78, 4392-4396

Rosengurt, E. & Mendoza, S.A. 1980 Monovalent ion fluxes and the control of cell proliferation in cultured fibroblasts. Ann. NY. Acad. Sci. USA, <u>339</u>, 175-190

Rozengurt, E. & Mendoza, S.A. 1985 Synergistic signals in mitogenesis: role of ion fluxes, cyclic nucleotides and protein kinase C in Swiss 3T3 cells. J. Cell Sci. Suppl. <u>3</u>, 229-242

Rozengurt, E., Mendoza, S.A., Wigglesworth, N.W. & Pohjanpelto, P. 1980 Na<sup>+</sup> energy and Na-K pump activity in murine, hamster and human cells - Effects of monensin, serum platelet extract, and viral transformation. J. Cell Physiol., <u>103</u>, 17-27

Rozengurt, E., Stroobant, P., Waterfield, M.D., Deuel, T.D. & Keehan, M. 1983 Platelet-derived growth factor elicits cyclic AMP accumulation in Swiss 3T3 cells: Role of prostaglandin production. Cell <u>34</u>, 265-272

Rubin, H. 1975 Central role for magnesium in co-ordinate control of metabolism and growth in animal cells, Proc. Natl. Acad. Sci., USA, <u>72</u>, 3551-3555

Rubin, H. 1977 Magnesium deprivation reproduces the co-ordinate effects of serum removal or corticol addition on transport and metabolism in chick embryo fibroblasts. J. Cell Physiol., <u>89</u>, 613-626

Rubin, H. 1981 Growth regulation reverse transformation and adaptability of 3T3 cells in decreased  $Mg^{2+}$  concentration. Proc. Natl. Acad. Sci., <u>78</u>, 328-332

Rubin, H. 1982 Effect of magnesium content on density-dependent regulation of the onset of DNA synthesis in transformed 3T3 cells. Cancer Res., <u>42</u>, 1761-1768

Rubin, H. & Chu, B. 1978 Reversible regulation by magnesium of chick embryo fibroblasts proliferation. J. Cell Physiol., <u>94</u>, 13-20

Rubin, H. & Koide, T. 1976 Mutual potentiation by magnesium and calcium of growth in animal cells. Proc. Natl. Acad. Sci., USA, 73, 168-172

Rubin, H., Terasaki, M. & Samui, H. 1979 Major intracellular cations and growth control: correspondence among magnesium content, protein synthesis, and the onset of DNA synthesis in Balb/c3T3 cells. Proc. Natl. Acad. Sci., <u>76</u>, 3917-3924

Rubin, J.B., Shia, M.A. & Pilch, P.F. 1983 Stimulation of tyrosine-specific phosphorylation in vitro by insulin like growth factor 1. Nature, <u>305</u>, 438-440

Ruddon, R.W. 1981 Phenotypic characteristics of transformed malignant cells. In 'Cancer Biology', (Oxford University Press), pgs120-210

Rudland, P.S., Gospodarowicz, D. & Seifert, W. 1974 Activation of guanyl cyclase and intracellular cyclic GMP by fibroblast growth factor. Nature, <u>250</u>, 741-742

Russell, D.H., Byus, C.V. & Manen, C.A. 1976 Proposed model of major sequential biochemical events of a trophic respose. Life Sciences, 19, 1297-1306

Russell, D.H. 1981 Ornithine decarboxylase: a key regulatory protein. Med. Biol., 59, 286-95

Sadler, S.E., Bower, M.A. & Meller, J.L. 1985 Studies of a plasma membrane steroid receptor in <u>Xenopus</u> oocytes using the synthetic progrestin Ru 486. J. Steroid Biochem., <u>22</u>, 419-426

Sadler, S.E. & Maller, J.L. 1982 Identification of a steroid receptor on the surface of <u>Xenopus</u> oocytes by photoaffinity labelling. J. Biol. Chem., <u>257</u>, 355-361

Sadler, S.E. & Maller, J.L. 1984 'Plasma membrane steroid hormone receptors', The Receptors Vol 1, Ed: P.M. Conn (Academic Press, London), pgs431-463

Sadler, S.E., Schechter, A.L., Tabin, C.J. & Maller, J.L. 1986 Antibodies to the ras gene product inhibit adenylate cyclase and accelerate progesterone-induced cell division in <u>xenopus laevis</u> oocytes. Mol. Cell. Biol., <u>6</u>, 719-722

Salmon, D.M. & Honeyman, T.W. 1980 Proposed mechanism of cholinergic action in smooth muscle. Nature, <u>284</u>, 344-345

Salmon, W.D. & DuVall, M.R. 1970 A serum fraction with "sulphation factor activity" stimulates <u>in vitro</u> incorporation of leucine and sulphate into protein-polysaccharide complexes, uridine into RNA, and thymidine into DNA of costal cartilage from hypophysectomized rats. Endocrinology, <u>86</u>, 721-727

Samui, H. & Rubin, H. 1982a Changes of intracellular and externally bound cations accompanying serum stimulation of mouse BalB/C3T3 cells. Exp. Cell Research, 139, 15-25

Sanui, H. & Rubin, H. 1982b The role of magnesium in cell proliferation and transformation. In 'Ions, Cell Proliferation and Cancer', Eds: A.L. Boynton, W.L. McKeehan & J.F Whitfield, (Academic Press (New York)), pgs517-537

Sap, J., Muroz, A., Damm, K., Goldberg, Y., Ghysdael, J., Leutz, A., Beug, H. & Vennstrom, B. 1986 The c-erb-A protein is a high affinity receptor for thyroid hormone. Nature, <u>324</u>, 635-640

Sasaki, Y. & Hidaka, H. 1982 Calmodulin and Cell Proliferation. Biochem. Biophys. Res. Commun., <u>104</u>, 451-456

Sasson, S. & Meyer, M. 1981 Effect of androgenic steroids on rat thymus and thymocytes in suspension. J. Steroid Biochem., 14, 509-517

Sassone-Corsi, P. & Verma, I.M. 1987 Modulation of c-fos gene transcription by negative and positive cellular factors. Nature, <u>326</u>, 597-510

Sawyer, S.T. & Cohen, S. 1981 Enhancement of calcium uptake in Ptd Ins turnover by epidermal growth factor in A 431 cells. Biochemistry, <u>20</u>, 6280-6286

Seamon, K.B., Padgett, W. & Daly, J.W. 1981 Forskolin: unique diterpene activator of adenylate cyclase in membranes and in intact cells. Proc. Natl. Acad. Sci., USA, <u>78</u>, 3363-3367

Schatzman, H.J. 1985 Calcium extrusion across the plasma membrane by the Ca<sup>2+</sup> pump and the Ca<sup>2+</sup>-Na<sup>+</sup> exchange system. In 'Calcium and Cell Physiology', Ed: D. Marme, (Springer-Verlag), pgs18-52

Schatzman, H.J. & Vincenzi, F.F. 1969 Calcium movement across the membrane of human red cells. J. Physiol., 201, 369-395

Schechter, Y., Hernawez, L., Schlessinger, J. & Cuatrecasas, P. 1979 Local aggregation of hormone-receptor complexes is required for activation by epidermal growth factor, Nature <u>278</u>, 835-838

Scher, C.D., Hendrickson, S.L., Whippe, A.P., Gottesman, M.M. & Pledger, W.P. 1982 Constitutive synthesis of platelet derived growth factor modulated proteins by a tumorigenic cell line. In 'Cold Spring Harbor Conference', on Cell Proliferation, (Vol. 9), (Cold Spring Harbor Press, New York), pgs289-303

Scher, C.D., Shepard, R.C., Antoniades, H.N. & Stiles, C.D. 1979 Platelet-derived growth factor and the regulation of the mammalian fibroblast cell cycle. Biochim. Biophys. Acta., <u>560</u>, 217-241

Scher, C.D., Stathakos, D. & Antoniades, H.N. 1974 Dissociation of cell division stimulating capacity for Balb/C3T3 from the insulin-like activity in human serum. Nature, <u>247</u>, 279-281

Schiff, R.I., Mercier, D. & Buckley, R.H. 1975 Inability of gestational hormones to account for the inhibitory effects of pregnancy plasma on lymphocyte responses in vitro. Cell. Immunol., <u>20</u>, 69-80

Schlessinger, J., Schreiber, A.B., Levi, A., Lax, I., Liberman, T. & Yarden, Y. 1983 Regulation of cell proliferation by epidermal growth factor. CRC Critical Reviews in Biochemistry, <u>14</u>, 93-111

Schrader, W.T. 1984 New model for steroid hormone receptors? Nature, <u>308</u>, 17-18

Schreier, M.H. & Nordin, A.A. 1977 An evaluation of the immune response in vitro. In 'B and T cells in immune recognition', Eds: Loor, F. & Roelants, G.E., (J. Wiley & Sons), pgs127-152

Schreiber, A.B., Libermann, T.A., Lax, I., Yarden, Y. & Schlessinger, J. 1983 Biological role of epidermal growth factor-receptor clustering. Investigation with monoclonal anti-receptor antibodies. J. Biol. Chem. <u>258</u>, 846-853

Schwantke, N., Le Bouffant, F., Dorée, M. & Le Peuch, C.J. 1985 Protein kinase C: properties and possible role in cellular division and differentiation. Biochimie, <u>67</u>, 1103-1110

Schwartz, A. & Triggle, P.J. 1984 Cellular action of calcium channel blocking drugs. Annu. Rev. Med., 35, 325-339

Scott, I.G., Poso, H., Akerman, K.E.O. & Andersoon, L.C. 1985 Rapid activation of ornithine decarboxylase by mitogenic (but not nonmitogenic) ligands in human T lymphocytes. Eur. J. Immunol., <u>15</u>, 783-787

Scow, R.O. & Hagan, S.N. 1965 Effects of testosterone propionate and growth hormone on growth and chemical composition of muscle and other tissues in hypophysectomized male rats. Endocrinology, 77, 852-858

Sefton, B.M. 1985 The role of tyrosine protein kinases in the action of growth factors. In 'Control of Animal Cell Proliferation (Vol 1)', Ed: A.L. Boynton & H.L Leffert (Academic Press Inc, London Ltd), pgs315-330

Sefton, B.M. & Hunter, T. 1984 Tyrosine protein kinases. Advances in cyclic nucleotide and protein phosphorylation research, Vol 18. Eds: P. Greengard & G.A. Robison (Raven Press, New York), pgs195-226

Seifert, W.E. & Rudland, P.S. 1974 Cyclic nucleotides and growth control in cultured mouse cells: correlation of changes in intracellular 3': 5'cGMP concentration with a specific phase of the cell cycle. Proc. Natl. Acad. Sci., 71, 4920-4924

Selye, H. 1941 Anaesthetic effect of steroid hormones. Proc. Soc. Exp. Biol. Med., <u>46</u>, 116-121

Shear, H.L., Roubinian, J.R., Gil, P. & Talal, N. 1981 Clearance of sensitized erythrocytes in NZB and NZW mice. Effects of castration and sex hormone treatment. Eur. J. Immunol., 11, 776-780

Sherman, M.R. & Stevens, J. 1984 Structure of mammalian steroid receptors: evolving concepts and methodological developments. Annu. Rev. Physio., <u>46</u>, 83-105

Sherwood, J.B. & Goldwasser, E. 1979 A sensitive radioimmunoassay for erythropoietin. Blood, 54, 885-893

Shields, R. 1978 Growth factors for tumours. Nature, 272, 670-671

Shields, R., Brooks, R.F., Riddle, P.N., Capellaro, D.F. & Delia, D. 1978 Cell size, cell cycle and transition probability in mouse fibroblasts. Cell, <u>15</u>, 469-474

Shipley, G.D., Childs, C.B., Volkenant, H.E. & Moses, H.L. 1984 Differential effects of epidermal growth factor, transforming growth factor and insulin on DNA and protein synthesis and morphology in serum-free cultures of AKR-2B cells. Cancer Res., <u>44</u>, 710-716

Shipley, G.D. & Ham, R.G. 1981 Improved medium and culture conditions for clonal growth with minimal serum protein and for enhanced serum-free survivial of Swiss 3T3 cells. In vitro, <u>17</u>, 656-70

Schuldiner, S. & Rozengurt, E. 1982 Na<sup>+</sup>/H<sup>+</sup> antiport in Swiss 3T3 cells: mitogenic stimulation leads to cytoplasmic alkalinization. Proc. Natl. Acad. Sci., USA, <u>79</u>, 7778-7782

Sibley, D.R. & Lefkowitz, R.J. 1985 Molecular mechanisms of receptor desensitization using the B adrenergic receptor-coupled adenylate cyclase system as a model. Nature, <u>317</u>, 124-129

Sibley, D.R., Nambi, P., Peters, J.R. & Lefkowitz, R.J. 1984b Phorbol diesters promote ß-adrenergic receptor phosphorylation and adenylate cyclase desensitization in duck erythrocytes. Biochem. Biophys. Res. Commun., <u>121</u>, 973-979

Sibley, D.R., Peters, J.R., Nambi, P., Caron, M.G. & Lefkowitz, R.J. 1984a Desensitization of turkey erythrocyte adenylate cyclase. J. Biol. Chem. 259, 9742-9749

Simmons, V.P. 1987 Thymic 'Atropy' at puberty is a myth. Medical Hypothesis, 22, 299-301

Simpson, J.G., Gray, E.S. & Beck, J.S. 1975 Age involution in the normal adult thymus. Clin. Exp. Immunol., 19, 261-264

Sirbasku, D.A. 1978 Oestrogen induction of growth factors specific for hormone responsive mammary pituitary and kidney tumour cells. Proc. Natl. Acad. Sci., USA, <u>75</u>, 3786-3790

Sirbasku, D.A., Ikeda, T. & Danielpour, D. 1985 'Endocrine and autocrine estromedins for mammary and pituitary tumor cells', In 'Mediators in cell growth and differentiation', Vol 37, M.D. Anderson, Eds: R.J. Ford & A.L. Maizel, Symposium on Fundamental Cancer Research, (Raven Press, New York), pp213-232

Sirbasku, D.A. & Moo, J.M. 1982 Characterisation of hypothalamic origin growth factor(s) for rat pituitary tumour cells in culture (abstract), Fed. Proc. <u>41</u>: 489

Sirbasku, D.A., Officer, J.M., Leland, F.E. & Iio, M. 1982 Evidence of a new role for pituitary derived hormones and growth factors in mammary tumour cell growth <u>in</u> <u>vivo</u> and <u>in vitro</u>. In 'Growth of Cells in Hormonally Defined Media', Vol 9, Cold Spring Harbor, Conference on Cell Proliferation, Cold Spring Harbor Laboratory, Cold Spring Harbor, New York, Eds: G.H. Sato, A.B. Pardee & D.A. Sirbasku, pp763-778

Sittman, D.B., Graves, R.A. & Marzluff, W.F. 1983 Histone mRNA concentrations are regulated at the level of transcription and mRNA degradation. Proc. Natl. Acad. Sci., USA, 80, 1849-1853

Slijivic, V.S. & Warr, G.W. 1973 Effect of oestrogens on lymphoid tissue. Period. Biol., 75, 231-244

Smith, C.D., Cox, C. & Snyderman, R. 1986 Receptor-coupled activation of phosphoinositide-specific phospholipase C by an N protein. Science, 232, 97-100

Smith, C.A. & Rennick, D.M. 1986 Characterization of a murine lymphokine distinct from IL-2 and IL-3 possessing a TCGF activity and an MCGF activity that synergises with IL-3. Proc. Natl. Acad. Sci., (USA), <u>83</u>, 1857-1861

Smith, C.D., Lane, B.C., Kusaka, I., Vergheses, M.W. & Snyderman, R. 1985 Chemoattractant receptor-induced hydrolysis of phosphaldyinositol 4,5-bisphosphate in human polymorphonuclear leukocyte membranes. J. Biol. Chem., <u>260</u>, 5875-5878

Smith, C.D., Uhing, R.J. & Snyderman, R. 1987 Nucleotide regulatory protein-mediated activation of phospholipase C in human polymorphonuclear leukocytes is disrupted by phorbol esters. J. Biol. Chem., <u>262</u>, 6121-6127

Smith, J.A. & Martin, L. 1973 Do cells cycle? Proc. Natl. Acad. Sci., USA, 70, 1263-1267

Smith, K.A. 1985 The determinants of T-cell growth. In 'Mediators in Cell Growth and Differentiation', Ed: R.J. Ford & A.L. Maizel, (Raven Press, New York), pgs185-192

Smith, K.A., Crabtree, G.R., Kennedy, S.J. & Munck, A.U. 1977 Glucocorticoid rptors and gucocortcoid sensitivity of mitogen stimulated and unstimulated human lymphocytes. Nature, <u>267</u>, 523-6

Smith, K.A., Gurson, H.L., Ridell, A.J. & Perris, A.D. 1975 Inhibitory action of oestrogen on calcium induced mitosis in rat bone marrow and thymus. J. Endocr., 65, 45-53

Smith, V.L. & Dedmar, J.R. 1986 An immunological comparison of several novel calcium-binding proteins. J. Biol. Chem., <u>261</u>, 15815-15818

Snowdone, K.W. & Borte, A.B. 1984 Changes in cytosolic ionized calcium induced by activators of secretion in  $GH_3$  cells. American Journal of Physiol., <u>246</u>, E198-E201

Snowdowne, K.W. & Borle, A.B. 1984 Measurement of cytosolic free calcium in mammalian cells with aequorin. Am. J. Physiol., 247, C396-C408

Sobhon, P. & Jirasattham, C. 1974 Effect of sex hormones on the thymus and lymphoid tissue of ovarectomized rats. Acta. Anat. <u>89</u>, 211-225

Soll, A.H., Goldfine, I.D., Roth, J., Kahn, C.R. & Neville, D.M. 1974 Thymic lymphocytes in obese (Ob/Ob) mice. J. Biol. Chem., 249, 4127-31

Soll, A.H., Kahn, C.R. & Neville, D.M. 1975 Insulin binding to liver plasma membranes in the (ob/ob) mouse. Demonstration of a decreased number of functionally normal receptors. J. Biol. Chem., 250, 4702-4707

Spandidos, D.A. 1986 A unified theory for the development of cancer. Bioscience Reports, <u>6</u>, 691-708

Spat, A., Bradford, P.G., McKinney, J.S. Rubin, R.P. & Putney, J.W. 1986 A saturable receptor for <sup>32</sup>P-inositol-1,4,5-trisphosphate in hepatocytes and neutrophils. Nature, <u>319</u>, 514-516

Speaker, M.G., Orlow, S.J., Sturgill, T.W. & Rosen, D.M. 1983 Characterization of a calmodulin-binding protein that is deficient in trifluoperazine-resistant variants of the macrophage-like cell line J774. Proc. Natl. Acad. Sci., USA, <u>80</u>, 329-333

Spedding, M. 1987 Three types of  $Ca^{2+}$  channel explain discrepancies. Trends in Pharmacol. Sci., <u>8</u>, 115-117

Spencer, E.M., Ross, M. & Smith, B. 1983 The identity of human insulin like growth factors I and II with somatomedins C and A and homology with rat IGCI and II. In 'Insulin-like growth factors/somatomedins', (Walter de Gruyter & Co), p81-96.

Sporn, M.B. & Roberts, A.B. 1985 Autocrine, paracrine and endocrine mechanisms of growth control. Cancer Surveys, <u>4</u>, 627-632

Sporn, M.B. & Roberts, A.B. 1985 Autocrine growth factors and cancer. Nature, 313, 745-747

Sporn, M.B., Roberts, A.B., Shull, J.H., Smith, J.M., Ward, J.M. & Sodek, J. 1983 Polypeptide transforming growth factors isolated from bovine sources and used for wound healing in vivo. Science, 219, 1329-1331

Sprent, J. 1977 Migration and lifespan of lymphocytes. In 'B and T cells in Immune Recognition', Eds: F. Loor & G.E. Roelants, (J. Wiley & Sons (London)), pgs59-82

Straddon, J.M. & Hansford, R.G. 1986 4  $\beta$ -phorbol 12-myristate 13-acetate attenuates the glucagon-induced increase in cytoplasmic free Ca<sup>2+</sup> concentration in isolated rat hepatocytes. Biochem. J., <u>238</u>, 737-743

Stadel, J.M., Nambi, P., Short, R.G.L., Sawyer, D.F., Caron, M.G. & Lefkowitz, R.J. 1983 Catecholamine-induced desensitization of turkey erythrocyte adenylate cyclase is associated with phosphorylation of the ß-adrenergic receptor. Proc. Natl. Acad. Sci., USA, <u>80</u>, 3173-3177

Stadel, J.M., Rebar, R., Shorr, R.G.L., Nambi, P. & Crooke, S.T. 1986 Biochemical characterization of phosphorylated ß-adrenergic receptors from catecholamine-desensitized turkey erythrocytes. Biochemistry, <u>25</u>, 3719-3724

Stadler, B.M., Dougherty, S.F., Farrar, J.J. & Oppenheim, J.J. 1981 Relationship of cell cycle to recovery of IL2 activity from human mononuclear cells, human and mouse T-cell lines. J. Immunol., <u>127</u>, 1936-1940

Steinberg, A.D., Huston, D.P., Taurog, J.D., Cowdery, J.S. & Raveche, E.S. 1981 The cellular and genetic basis of murine lupus. Immunol. Rev., 55, 121-

Steinman, G.G., Klaus, B. & Muller-Hermelink, H-K 1986 The involution of the ageing human thymic epithelium is independent of puberty. Scand. J. Immunol., <u>22</u>, 563-573

Stephenson, E.W. 1981 Magnesium effects on activation of skinned fibres from striated muscle. Fed. Proceed. <u>40</u>, 2661-2666

Stewart, W.E. 1979 The interferon system. (Springer-Verlag, New York)

Stiles, C.D., Capone, G.T., Scher, C.D., Antoniades, H.N., Van Wyck, J.J. & Pledger, W.J. 1979 Dual control of cell growth by somatomedins & platelet derived growth factor. PNAS (USA), <u>76</u>, 1279-1283

Stobo, J.O. 1972 PHA and Con A: Probes for murine T cell activation and differentiation. Transplant. Rev., 11, 60-72

Stoscheck, C.M. & Carpenter, G. 1984 Down regulation of epidermal growth factor receptors: direct demonstration of receptor degradation in human fibroblasts. J. Cell. Biol., <u>98</u>, 1048-1053

Strauss, D.S. 1981 Effects of insulin on cellular growth and proliferation. Life Science, 29, 2131-2139

Streb, H., Irvine, R.F., Bendge, M.J. & Schulz, I. 1983 Release of  $Ca^{2+}$  from a nonmitochondrial intracellular store in pancreatic acinar cells by inositol-1,4,5,triphosphate. Nature, <u>306</u>, 67-68

Sugden, D., Vanecek, J., Klein, D.C., Thomas, T.P. & Anderson, W.B. 1985 Activation of protein kinase C potentiates isoprenaline-induced cyclic AMP accumulation in rat pineal ocytes. Nature, <u>314</u>, 359-361 Sugimoto, Y., Whitman, M., Cantley, L.C. & Erikson, R.L. 1984 Evidence that Rous Sarcoma virus transforming gene product phosphorylates phosphatidylinositol and diacylglycerol. Proc. Natl. Acad. Sci., USA, <u>81</u>, 2117-2121

Sumner, J.B. 1919 The globulins of the jack bean. Canavalia ensiformis. J. Biol. Chem. <u>37</u>, 137-141

Sumner, J.B. & Howell, S.F. 1936 The identification of the hemagglutin of the jackbean with Con A. J. Bacteriol., <u>32</u>, 227-231

Swain, S.L. & Dutton, R.W. 1983 Mouse T-lymphocyte sub-populations: relationships between function and Lyt antigen phenotype. T Lymphocyte Today, Ed: J.R. Inglis, (Elsevier (Amsterdam)), pgs100-103

Swann, K. & Whitaker, M. 1985 Stimulation of the Na/H exchanger of sea urchin eggs by phorbol ester. Nature, <u>314</u>, 274-277

Swierenga, S.H.H., Whitfield, J.F., Boynton, A.L., MacManus, J.P. Rixon, R.H., Sikorska, M., Tsang, B.K. & Walker, P.R. 1981 Regulation of proliferation of normal and neoplastic rat liver cells by calcium and cAMP. Ann. NY. Acad. Sci., 349, 295-311

Takagi, K., Okabe, Y., Yoshimura, K. & Ichikawa, Y. 1986 Changes in intracellular K<sup>+</sup> and Na<sup>+</sup> ion concentrations during cell growth and differentiation. Cell Structure & Function, <u>11</u>, 235-243

Takai, Y., Kaibuchi, K., Tsuda, T. & Hoshijima, M. 1985 Possible modes of action of growth factors and tumour promoters in the activation of the c-myc gene in Swiss 3T3 fibroblasts. J. Cell. Biochem., 29, 143-155

Takai, Y., Kishimoto, A., Iwasa, Y., Kawahara, Y. Hori, T. & Nishizuka, Y. 1979 Calcium dependent activation of a multifunctional protein kinase by membrane phospholipids. J. Biol. Chem., 254, 3692-3695

Takai, Y., Minakuchi, R., Kikkawa, U., Sano, K., Kaibuchi, K., Yu, B., Matsubara, T. & Nishizuka, Y. 1982 Membrane phospholipid turnover, receptor function and protein phosphorylation. Progr. Brain Res., <u>56</u>, 287-301

Takayama, S., White, M.F., Lauris, V. & Kahn, C.R. 1984 Phorbol esters modulate insulin receptor phosphorylation and insulin action in cultured hepatoma cells. Proc. Natl. Acad. Sci., USA, <u>81</u>, 7797-7801

Takuwa, N., Takuwa, Y. & Rasmussen, H. 1987 A tumour promoter, 12-0-tetra decanoylphorbol 13-acetate, increases cellular 1,2-diacylglycerol content through a mechanism other than phosphoinositide hydrolysis in Swiss mouse 3T3 fibroblasts. Biochem. J., 243, 647-653

Tata, J.R. 1976 Growth promoting actions of peptide hormones. Ciba Found. Symp, <u>41</u>, 297-312

Taub, M., Chuman, L., Saier, M.H. & Sato, G. 1979 Growth of Madin-Darby canine kidney epithelial cell (MDCK) line in hormone-supplemented, serum-free medium. Proc. Natl. Acad. Sci., USA, <u>76</u>, 3338-3342

Taylor, C.W. 1986 Growth factors control a network of interacting messengers. Trends Pharmacol. Sci., 7, 467-471

Taylor, C.W. 1987 Receptor regulation of calcium entry. Trends Pharmacol. Sci., 8, 79-80

Taylor, C.W. 1988 Time and space - novel aspects of hormone action. Trends. Pharmacol. Sci., 2, 43-45

Taylor-Papadimitriou, J., Ebsworth, N. & Rozengurt, E. 1985 Possible mechanisms of interferon-induced growth inhibition, In "Mediators in Cell Growth and Differentiation", Eds: R.J. Ford & A.L. Maizel, (Raven Press, New York), p283-298

Taylor-Papadimitriou, J., Shearer, M., Balkwill, F.R. & Fantes, K.H. 1982 Effects of Hu IFN- $\alpha$ 2 and Hu IFN- $\alpha$  (Namalwa) on breast cancer cells grown in culture and as xenografts in the nude mouse. J. Inferferon Res., 2, 479-491

Terrin, H.M. 1967 Studies on carcinogenesis by avian sarcoma viruses VI. Differential multiplication of infected and of converted cells in response to insulin. J. Cell Physiol., <u>69</u>, 377-384

Teng, M.H., Bartholomew, J.C. & Bissell, M.J. 1976 Insulin effect on the cell cycle: analysis of the kinetics of growth parameters in confluent chick cells. Proc. Natl. Acad. Sci., USA, <u>73</u>, 3173-3177

Terris, S., Hofmann, C. & Steiner, D.F. 1979 Mode of uptake and degradation 125 I-labelled insulin by isolated hepatocytes and H4 hepatoma cells. Can. J. Biochem., 57, 459-68

Thoenen, H. & Barde, Y.A. 1980 Physiology of nerve growth factor. Physiol. Rev., <u>60</u>, 1284-1335

Thoenen, H., Korsching, S., Heumann, R. & Acheson, A. 1985 Nerve growth factor. In 'Growth Factors in Biology and Medicine', Eds: D. Evered, J. Nugent & J. Whelan, Ciba Foundation Symposium 116, (Ciba Foundation), pgs113-123

Thomas, J.A., Buchsbaum, R.N., Zimniak, A. & Racker, E. 1979 The intracellular pH measurements in Erhlich ascities tumour cells inhibiting spectroscopic probes generated in situ, Biochemistry, <u>18</u>, 2210-2218

Thompson, C.B., Challoner, P.B., Neiman, P.E. & Groudine, M. 1985 Levels of c-myc oncogene mRNA are invariant throughout the cell cycle. Nature, <u>314</u>, 363-366

Thrall, C.L. & Spelsberg, T.C. 1980 Factors affecting the binding of chick oviduct progesterone receptor to deoxyribonucleic acid - evidence the deoxyribonucleic acid is not the nuclear acceptor site. Biochemistry, <u>19</u>, 4130-38

Todano, G.J., Fryling, C. & Delarco, J.E. 1980 Transforming growth factors produced by certain human tumour cells: polypeptides that interact with epidermal growth factor receptors. Proc. Natl. Acad. Sci., USA, <u>77</u>, 5258-5262

Tonna, E.A. 1973 Hormonal influence on skeletal growth and regeneration. In Humoral Control of Growth and Differentiation, Vol 1", Eds: J. Lobue & A.S. Gordon (Academic Press Inc., New York), pgs275-359

Tosi, P., Kraft, R., Luzi, P., Cintorino, M., Frankhauser, G., Hess, M.W. & Cottier, H. 1982 Involution patterns of the human thymus 1. Size of the corticol area as a function of age. Clin. Exp. Immunol., <u>47</u>, 497-501

Towart, R. 1981 The selective inhibition of serotonin-induced contractions of rabbit cerebral vascular smooth muscle by calcium antagonists dihydropyridines. An investigation of the mechanism of action of nimodipine. Circ. Res., <u>48</u>, 650-657

Towle, A.C. & Sze, P.V. 1983 Steroid binding to synaptic plasma membrane: differential binding of glucocorticoids and gonadal steroids. J. Steroid Biochem., 18, 135-143

Truneh, A., Albert, F., Goldstein, P. & Schmitt-Verhirst, A.M. 1985 Early steps of lymphocyte activation bypassed by synergy between calcium ionophores and phorbol ester. Nature, <u>313</u>, 318-320

Tsang, B.K., Whitfield, J.F. & Rixon, R.H. 1981 The reversible inability of isoproterenol-activated parotid gland cells to stimulate DNA synthesis in the hypocalcaemic thyroparathyroidectomized rat. J. Cell Physiol., 107, 41-46

Tsien, R.Y. 1980 New calcium indicators and buffers with high selectivity against magnesium and protons. Design synthesis and properties of prototype structures. Biochemistry, 19, 2396-2404

Tsien, R.Y. & Poenie, M. 1986 Fluorescence ratio imaging: a new window into intracellular ionic signalling. Trends Biochem. Sci., <u>11</u>, 450-455

Tsien, R.Y., Pozzan, T. & Rink, T.J. 1982 Calcium homeostasis in intact lymphocytes: cytoplasmic free calcium monitored with a new, intracellularly trapped fluorescent indicator. J. Cell Biol., <u>94</u>, 325-335

Tsien, R.Y., Pozzan, T. & Rink, T.J. 1984 Measuring and manipulating cytosolic  $Ca^{2+}$  with trapped indicators. Trends Biochem. Sci., 9, 263-266

Tsien, R.Y. & Rink, T.J. 1980 Neutral carrier ion selective microelectrodes for measurement of intracellular free calcium. Biochim. Biophys. Acta., <u>599</u>, 623-638

Tsien, R.Y., Rink, T.J. & Poenie, M. 1985 Measurement of cytosolic free  $Ca^{2+}$  in individual small cells using fluorescence microscopy with dual excitation wavelengths. Cell Calcium <u>6</u>, 145-157

Turlapaty, D.D.M.V. & Altura, B.M. 1978 Extracellular magnesium ions control  $Ca^{2+}$  exchange and content of vascular smooth muscle. Eur. Journal Pharmacol., 52, 421-423

Uhing, R.J., Prpic, V., Jiang, H. & Exton, J.H. 1986 Hormone-stimulated polyphosphoinositide breakdown in rat liver plasma membranes. Roles of guanine nucleotides and calcium. J. Biol. Chem., <u>261</u>, 2140-2146

Ullrich, A., Bell, J.R., Chen, E.Y., Petruzelli, L.M., Dull, T.J., Gray, A., Coussens, L., Liao, Y-C., Tsubokawa, M., Mason, A., Seeburg, P.H., Grunfeld, C., Rosen, O.M. & Ramachandran, J. 1985 Human insulin receptor and its relationship to the tyrosine kinase family of oncogenes. Nature, <u>313</u>, 756-760

Ushiro, H. & Cohen, S. 1980 Identification of phosphotyrosine as a product of Epidermal growth factor - activated protein kinase in A-431 cell membranes. J. Biol. Chem., <u>255</u>, 8363-8365

Vara, F., Schneider, J.A., Rozengurt, E. 1985 Ionic responses rapidly elicited by activation of protein kinase C in quiescent Swiss 3T3 cells. Proc. Natl. Acad. Sci., USA, 82, 2384-2388

Varon, S. & Shaper, S.D. 1983 The Na<sup>+</sup> K<sup>+</sup>-pump may mediate the control of nerve cells by nerve growth factor. Trends Biochem. Sci., <u>8</u>, 22-25

Van Breeman, C. & Siegel, B. 1980 The mechanism of  $\alpha$  adrenergic activation of the dog coronary artery. Circ. Res., <u>46</u>, 426-29

Van Wyck, J.J. 1984 The somatomedins: biological actions and physiologic control mechanisms. Hormonal proteins and peptides, Vol XII, Ed: C.H. Li, (Academic Press Inc.), pgs81-125,

Veigl, M.L., Yanaman, T.C., Branch, M.E. & Sedwick, L.D. 1984b Differences in calmodulin levels of normal and transformed cells as determined by culture conditions. Cancer Research, <u>44</u>, 3184-3189

Veigl, M.L., Vanaman, T.C. & Sedwick, W.D. 1984a Calcium and calmodulin in cell growth and transformation. Biochim. Biophys. Acta. 738, 21-48

Veit, B. & Michael, J.G. 1973 Characterization of an immunosuppressive factor present in mouse serum. J. Immunol., <u>111</u>, 341-351

Verghese, M.W., Smith, C.D. & Snyderman, R. 1985 Potential role for a guanine nucleotide regulatory protein in chemottractant receptor mediated polyphosphoionsitide metabolism,  $Ca^{2+}$  mobilisation and cellular responses by leukocytes. Biochem. Biophys. Res. Comm., 127, 450-457

Verheul, H.A.M., Tittes, E.V., Kelder, J. & Schuurs, A.H.W.M. 1986 Effects of steroids with different endocrine profiles on the development, morphology and function of the bursa of Fabricius in chickens. J. Steroid. Biochem., <u>25</u>, 665-675

Vicentini, L.M. & Villereal, M.L. 1986 Inositol phosphates turnover, cytosolic  $Ca^{2+}$  and pH: putative signals for the control of cell growth. Life Sci., <u>38</u>, 2269-2276

Villereal, M., Owen, N., Vincentini, L., Mix-Muldoon, L. & Jamieson, G. 1985 In 'Cancer Cells', Chapter 3. Eds: J. Farasmisco, B. Ozanne & C. Stiles, (Cold Spring Harbor), pgs417-424

Vogel, A., Raines, E., Kariya, B., Rivest, M.J. & Ross, R. 1978 Co-ordinate control of 3T3 cell proliferation by platelet-derived growth factor and plasma components. Proc. Natl. Acad. Sci., USA, 75, 2810-2814

Vogel, S. & Hoppe, J. 1986 Polyamines stimulate the phosphorylation of phosphatidylinositol in membranes from A431 cells. Eur. J. Biochem., <u>154</u>, 253-257

Vormann, J. & Gunther, T. 1987 Effect of concanavalin A and extracellular magnesium on the concentration of intracellular free  $Ca^{2+}$  in thymocytes from normal and Mg-deficient rats. Mag. Bull., 9, 33-35

Wagner, H., Hardt, C., Heeg, K., Pfizenmaier, K., Solbach, W., Bartlett, R., Stockinger, H. & Rollinghoff, M. 1980 T-T cell interactions during cytotoxic T lymphocyte CTL responses: T cell derived helper factor (Interleukin 2) as a probe to analyse CTL responsiveness and thymic maturation of CTL progenitors. Immunol., Rev., <u>51</u>, 215-256

Waite, W.I. & Hirschorn, K. 1978 Lymphocyte response to activators. In 'Handbook of Experimental Immunology', Ed: D.M. Weir, (Blackwell Scientific Publications), pgs26.1-26.10

Wakelam, M.J.O., Davies, S.A., Mousley, M.D., McKay, I., Marshall, C.J. & Hall, A. 1986 Normal p21<sup>N-ras</sup> couples bombesin and other growth factor receptors to inositol phosphate production. Nature, <u>323</u>, 173-176

Walker, G.M. 1986 Magnesium and cell cycle control: an update. Magnesium, <u>5</u>, 9-23

Walker, G.M. & Duffus, J.H. 1983 Magnesium as the fundamental regulator of the cell cycle. Magnesium, 2, 1-16

Wallace, M.A. & Fain, J.N. 1985 Guanosine 5'-O-Thiotriphosphate stimulates phospholipase C activity in plasma membranes of rat hepatocytes. J. Biol. Chem., 260, 9527-9530

Walter, P., Green, S., Greene, G.L., Krust, A., Bomert, J.M. & Jeltsch, J.M. 1985 Cloning of the human estrogen receptor cDNA. Proc. Natl. Acad. Sci., USA, <u>82</u>, 7889-7893

Walters, C.S. & Claman, H.N. 1975 Age related changes in cell mediated immunity in BALB/c mice. J. Immunol., 115, 1438-1443

Walters, M.R. 1985 Steroid hormone receptors and the nucleus, Endocrine Reviews, <u>6</u>, 5120-543

Walthall, B.J. & Ham, R.G. 1981 Multiplication of human diploid fibroblasts in a synthetic medium supplemented with EGF, insulin and dexamethasone. Exp. Cell Res., <u>134</u>, 301-309

Waltman, S.R., Burde, R.M. & Berricos, J. 1971 Prevention of corneal homograft rejection by oestrogens. Transplantation, <u>11</u>, 195-198

Wang, J.L. & Hsu, Y.M. 1986 Negative regulators of cell growth. Trends. Biochem. Sci., <u>11</u>, 24-26

Wang, J.H. & Waisman, D.M. 1979 Calmodulin and its role in the second messenger system. Curr. Top. Cell. Regul., 15, 47-107

Wang, Z.W. & Rozengurt, E. 1983 Interplay of cAMP and microtubules in modulating the initiation of DNA synthesis in 3T3 cells. J. Cell Biol., <u>96</u>, 1743-1750

Waterfield, M.D., Scrace, T., Whittle, N., Stroobant, P., Johnsson, A., Wasterson, A., Westermark, I.J., Heldin, C.H., Huang, J.S. & Deuel, T.F. 1983 Platelet derived growth factor is structurally related to the putative transforming protein p28 of simian sarcoma virus. Nature, <u>304</u>, 35-39

Watson, J. 1976 The involvement of cyclic nucleotide metabolism in the initiation of lymphocyte proliferation induced by mitogens. J. Immunol., <u>117</u>, 1656-1663

Wedner, H.J., Dankner, R. & Parker 1975 Cyclic GMP and lectin-induced lymphocyte activation. J. Immunol., <u>115</u>, 1682-1687

Weinberg, R.A. 1985 The action of oncogenes in the cytoplasm and nucleus, Science, 230, 770-776

Weinberger, C., Hollenberg, S.M., Ong, E.S., Harmon, J.M., Brower, S.T., Gollowski, J., Thompson, E.B., Rosenfield, M.G. & Evans, R.M. 1985 Identification of human glucocorticoid receptor complementary DNA clones by eprtope selection. Science, 228, 740-742

Weinberger, C., Hollenberg, S.M., Rosenfield, M.G. & Evans, R.M. 1985 Domain structure of human glucocorticoid receptor and its relationship to the v-erb-A oncogene product. Nature, <u>318</u>, 670-672

Weinberger, C., Thompson, C.C., Ong, E.S., Lebo, R., Gruol, D.J. & Evans, R.H. 1986 The c-erb-A gene encodes a thyroid hormone receptor. Nature, <u>324</u>, 641-646

Weinberg, R.A. 1983 A molecular basis of cancer. Sci. American, 249, 126-142

Weinstein, I.B. 1987 Growth factors, oncogenes and multistage carcinogenesis. J. Cell Biochem., 33, 213-224

Weiss, A., Imboden, J., Shoback, D. & Shobo, J. 1984 Role of  $T_3$  surface molecules in human T cell activation:  $T_3$  dependent activation results in a rise in cytoplasmic free calcium. Proc. Natl. Acad. Sci., <u>81</u>, 4169-4173

Wells, J.N. & Hardman, J.G. 1977 Cyclic nucleotide phosphodiesterases. Adv. Cyclic Nucleotide Res., <u>8</u>, 119-144

Welshons, W.V., Lieberman, M.E. & Gorski, J. 1984 Nuclear localization of unoccupied oestrogen receptors. Nature, <u>307</u>, 747-749

Werth, D.K. & Pastan, I. 1984 Vinculin phosphorylation in response to calcium and phorbol esters in intact cells. J. Biol. Chem., <u>259</u>, 5264-5270

Westermark, B., Heldin, C.H., Ek, B., Johnsson, A., Mellstrom, K., Nister, M., Wasteson, 1983 Biochemistry and biology of platelet-derived growth factor, In 'Growth and Maturation Factors', Ed: G. Guroff (Wiley, New York), pgs73-114 Westermark, B. & Wasteson, A. 1976 A platelet factor stimulating human normal glial cells. Exp. Cell Res., <u>98</u>, 170-174

Wharton, W., Gillespie, G.Y., Russell, S.W. & Pledger, W.J. 1982 Mitogenic activity elaborated by macrophage-like cell lines acts as competence factor(s) for BALB/c-3T3 cells. J. Cell Physiol., <u>110</u>, 93-100

Whitfield, J.F., Boynton, A.L., Macmanus, J.P., Sikorska, M. & Tsang, B.K. 1979 The regulation of cell proliferation by calcium and cyclic AMP. Mol. Cell. Biochem., <u>27</u>, 155-179

Whitfield, J.F. 1982 The roles of calcium and magnesium in cell proliferation: an overview. In 'Ions, Cell Proliferation and Cancer', Eds: A.L. Boynton, W.L. McKeenan and J.F. Whitfield, (Academic Press, (New York)),pgs283-293

Whitfield, J.F., Boynton, A.L., MacManus, J.P., Rixon, R.H., Sikorska, H., Tsang, B., Walker, P.R. & Swienenga, S.H.H. 1981 The roles of calcium and cAMP in cell proliferation. Ann. NY. Acad. Sci., 339, 216-240

Whitfield, J.F., Boynton, A.L., Rixon, R.H. & Youdale, T. 1985 The control of cell proliferation by calcium,  $Ca^{2+}$ -calmodulin and cyclic AMP. In 'Control of Animal Cell Proliferation', Eds: A.L. Boynton & H.L Leffert, (Academic Press (London)), pgs332-365

Whitfield, J.F., Durkin, J.P., Franks, D.J., Kleine, L.P., Raptis, L., Rixon, R.H., Sikorska, M. & Walker, P.R. 1987 Calcium, cyclic AMP and protein kinase C - partners in mitogenesis. Cancer and Metastasis Reviews, 5, 205-250

Whitfield, J.F., Perris, A.D. & Youdale, T. 1969 The calcium mediated promotion of mitotic activity in rat thymocyte populations by growth hormone, neurohormones, parathyroid hormone and prolactin. J. Cell. Physiol., <u>73</u>, 203-212

Whitman, M., Kaplan, O., Cantley, L., Roberts, T.M. & Schaffhausen, B. 1986 Phophoinositide kinase activity and transformation. Fed. Proceed., <u>45</u>, 2647-2652

Whitney, R.B. & Sutherland, R.M. 1972 Requirement for calcium ions in lymphocyte transformation stimulated by phytohemagglutinin. J. Cell Physiol., <u>80</u>, 329-333

Whitney, R.B. & Sutherland, R.M. 1983 Characteristics of calcium accumulation by lymphocytes and alterations in the process induced by phytohemagglutinin. J. Cell Physiol., <u>82</u>, 9-20

Williams, K.A., Murphy, W. & Haslam, R. 1987 Effects of activation of protein kinase C on the agonist-induced stimulation and inhibition of cyclic AMP formation in intact human platelets. Biochem. J., 243, 667-678

Williams, T.C., Corson, D.C. & Sykes, B.D. 1987 Oncomodulin. J. Biol. Chem., 262, 6248-6256

Willmer, E.N. 1961. Steroids and cell surfaces. Biol. Rev., 36, 368-398

Willingham, M.C., Johnson, G.S. & Pastan, I. 1972 Control of DNA synthesis and mitosis in 3T3 cells by cyclic AMP. Biochem. Biophys. Res. Commun., <u>48</u>, 743-748

Wolberg, W.H. 1971 Alternative pathways for DNA-thymic synthesis in stimulated leucocytes. Clin. Exp. Immunol., <u>8</u>, 177-186

Wolf, M., LeVine, H., May, W.S.M., Cuatrecasas, P. & Sahyoun, N. 1985 A model for intracellular translocation of protein kinase C involving synergism between  $Ca^{2+}$  and phorbol esters. Nature, <u>317</u>, 546-549

Wolff, B., Dickson, R.B. & Hanover, J.A. 1987 A nuclear localization signal in steroid hormone receptors. Trends Pharmacol. Sci., <u>8</u>, 119-121

Woo, D.D.L., Fay, S.P., Griest, R., Coty, W., Goldfine, I. & Fox, C.F. 1986 Differential phosphorylation of the progesterone receptor by insulin, epidermal growth factor, and platelet-derived growth factor receptor tyrosine protein kinases. J. Biol. Chemistry, <u>261</u>, 460-67

Woodgett, J.R. & Hunter, T. 1987 Immunological evidence for two physiological forms of protein kinase C. Molecular Cell Biol., 7, 85-96

Wrann, M., Fox, C.F. & Ross, R. 1980 Modulation of epidermal growth factor receptors on 3T3 cells by platelet derived growth factors. Science, <u>210</u>: 1363-65

Wright, N.A. & Morley, A.R. 1971 The effect of testosterone on the growth fraction of the mouse small intestine. J. Endocrinol., <u>50</u>, 351-352

Wyatt, H.V. 1961 Cation requirements of HeLa cells in tissue culture. Expl. Cell Res., 23, 97-107

Wyle, F.A. & Kent, J.R. 1977 Immunosuppression by sex steroid hormones. The effect upon PHA- and PDP-stimulated lymphocytes. Clin. Exp. Immunol., <u>27</u>, 407-15

Yamamoto, K.R. & Alberts, B.M. 1976 Steroid receptors: elements for modulation of eukaryotic transcription. Annu. Rev. Biochem., <u>45</u>, 721-746

Youdale, T. & MacManus, J.P. 1975 Failure of tritiated thymidine incorporation into DNA to reflect the autoradiographically demonstrable calcium-induced increase in thymic lymphoblast DNA synthesis. J. Cell Physiol., <u>86</u>, 495-502

Yunis, A.A., Wu, M-C. & Muller, A.M. 1983 Colony-stimulating factor In "Growth and Maturation Factors, Vol. 1", Ed: G. Guroff (John Wiley & Sons), pgs211-247

Yu, K.T. & Czech, M.P. 1986 Tyrosine phosphorylation of insulin receptor  $\beta$  subunit activates the receptor tyrosine kinase in intact H-35 hepatoma cells. J. Biol. Chem., <u>261</u>, 4715-4722

Zapf, J. & Froesch, E.R. 1986 Insulin like growth factors/somatomedins, structure, secretion, biological actions and physiological role. Hormone Research, <u>24</u>, 121-130

Zapf, J., Schoenel, E. & Froesch, E.R. 1978 Insulin-like growth factors I and II: Some biological actions and receptor binding characteristics of two purified constituents of nonsuppressible insulin-like activity of human serum. Eur. J. Biochem., <u>87</u>, 285-296 Zapf, J., Schoenle, E. & Froesch, E.R. 1985 <u>In vivo</u> effects of the insulin-like growth factors (IGFs) in the hypophysectomized rat: comparison with human growth hormone and the possible role of the specific IGF carrier proteins, In 'Growth Factors in Biology and Medicine', Eds: D. Evered, J. Nugent & J. Whelan, Ciba Foundation Symposium 116, (Ciba Foundation), p169-187

Zanders, E.D. 1985 Lymphocyte receptors and transmembrane signalling. In 'Molecular mechanisms of transmembrane signalling', Eds: P. Cohen & M.D. Houslay, Elsevier Science (Amsterdam), pgs412-427

Zapf, J., Walter, H. & Froesch, E.R. 1981 Radioimmunological determination of insulin-like growth factors I and II in normal subjects and in patients with growth disorders and extrapancreatic tumour hypoglycaemia. J. Clin. Invest., <u>68</u>, 1321-1330

Zetterberg, A. & Larsson, O. 1985 Kinetic analysis of regulatory events in  $G_1$  leading to proliferation or quiescence of Swiss 3T3 cells. Proc. Natl. Acad. Sci., USA, <u>82</u>, 5365-5369

Zilberman, Y., Howe, L.R., Moore, J.P., Hesketh, T.R. & Metcalfe, J.C. 1987 Calcium regulates inositol 1,3,4,5-tetrakisphosphate production in lysed thymocytes and in intact cells stimulated with concanavalin A. Embo. J., <u>6</u>, 957-962

Zinkernagel, R.M. & Doherty, P.C. 1977 Major transplantation antigens, viruses and specificity of surveillance T cells. Contemp. Top. Immunobiol., 7, 179-220

Zwiller, J., Revel, M-O. & Malviya, A.N. 1985 Protein kinase C catalyses phosphorylation of guanylate cyclase in vitro. J. Biol. Chem., <u>260</u>, 1350-1353